22

November 2022

Prospectus

Morgan Stanley Investment Funds Société d'Investissement à Capital Variable Luxembourg ("SICAV")

Contents

		Page
Defini	itions	3
Sectio	on 1	7
1.1	The Company and the Funds	7
1.2	Investment Objectives and Policies	11
1.3	Methodology for Calculating the Global Exposure	157
1.4	Financial Indices	160
1.5	Risk Factors	160
Sectio	on 2	192
2.1	Share Class Description	192
2.2	Issue of Shares, Subscription and Payment Procedure	197
2.3	Redemption of Shares	200
2.4	Conversion of Shares	203
2.5	Charges and Expenses	206
2.6	Publication of Net Asset Value	211
2.7	Net Asset Value Determination	212
2.8	Best Execution	215
2.9	Conflicts of Interests	215
2.10	Dividend Policy	216
2.11	Taxation	219
2.12	Pooling	227
Sectio	on 3	228
3.1	General Information	228
Apper	ndix A – Investment Powers and Restrictions	241
Apper	ndix B – Collateral Policy	255
Apper	ndix C - Additional Information for U.K. Investors	257
Apper	ndix D - Additional Information for Irish Investors	261
Apper	ndix E - Additional Information for Chilean Investors	264
Apper	ndix F – Additional Information for Taiwanese Investors	266
Apper	ndix G – Additional Information for Peruvian Investors	267
Apper	ndix H – Additional Information for Asian-Pacific and Middle Eastern Investors	268
Apper	ndix I – Data Protection Notification	275
	ndix J – List of sub-custodians used by the Depositary and sub-delegates that may arise fro Depositary's duties	
Apper	ndix K – Facilities Agents and Services	293

Prospectus of the Morgan Stanley Investment Funds

Société d'Investissement à Capital Variable Luxembourg ("SICAV")

Morgan Stanley Investment Funds (the "Company") is registered in the Grand Duchy of Luxembourg as an undertaking for collective investment pursuant to Part I of the Law of 17 December 2010 on undertakings for collective investment (the "2010 Law"). Such registration however does not imply a positive assessment by the supervisory authority of the quality of the shares of the Company (the "Shares") offered for sale. Any representation to the contrary is unauthorised and unlawful. The Company is an Undertaking for Collective Investment in Transferable Securities ("UCITS") for the purpose of the Directive 2009/65/EC of the European Parliament and of the Council of 13 July 2009 on the coordination of laws, regulations and administrative provisions relating to undertakings for collective investment in transferable securities, as amended (the "UCITS Directive"). The Company has appointed MSIM Fund Management (Ireland) Limited as management company (the "Management Company").

The Luxembourg law of 4 June 2009 transposing the Oslo Convention on Cluster Munitions introduced in article 3 a prohibition on the financing, with full knowledge, of cluster munitions and explosive sub-munitions. As such both the Management Company and the Company have adopted a policy designed to comply with the abovementioned Luxembourg law.

Subscriptions can be accepted only on the basis of the current prospectus (the "Prospectus"), which is valid only if accompanied by a copy of the latest Annual Report containing the audited accounts, and of the semi-annual report if such report is published after the latest Annual Report. These reports form an integral part of the Prospectus.

The Company draws the investors' attention to the fact that any investor will only be able to fully exercise his investor rights directly against the Company, notably the right to participate in general shareholders' meetings, if the investor is registered himself and in his own name in the shareholders' register of the Company. In cases where an investor invests in the Company through an intermediary investing into the Company in his own name but on behalf of the investor, it may not always be possible for the investor to exercise certain Shareholder rights directly against the Company. Shareholders are advised to take advice on their rights.

No person is authorised to make any representation other than as contained in the Prospectus or in the documents referred to in the Prospectus. Such documents are available to the public at the registered office of Morgan Stanley Investment Funds, Luxembourg.

This Prospectus was prepared in English and may be translated into other languages. Any such translation shall only contain the same information and have the same meanings as the English language document. Where there is any inconsistency between the English language document and the document in another language, the English language document shall prevail except to the extent (but only to the extent) required by the laws of any jurisdiction where the Shares are sold, so that in an action based upon disclosure in a document of a language other than English, the language of the document on which such action is based shall prevail.

Important: If you are in any doubt about the contents of this document, you should consult your stockbroker, bank manager, solicitor, accountant or other financial adviser.

The Company may, if, in the opinion of the Directors of the Company (the "Directors"), it is likely to be fiscally beneficial for the Company, invest via one or more wholly-owned subsidiaries located in any jurisdiction in the world.

The distribution of this Prospectus and the offering of the shares may be restricted in certain jurisdictions. It is the responsibility of any persons in possession of this Prospectus and any persons wishing to make application for Shares pursuant to this Prospectus to inform themselves of and to observe all applicable laws and regulations of any relevant jurisdictions.

This Prospectus has been prepared solely for, and is being made available to investors for the purposes of evaluating an investment in Shares in the Funds. Investors should only consider investing in the Funds if they understand the risks involved including the risk of losing all capital invested. Distributors and other intermediaries which offer, recommend or sell Shares in the Funds must comply with all laws, regulations and regulatory requirements as may be applicable to them. Also, such distributors and other intermediaries must consider such information about the Funds as is made available by the Management Company for the purposes of the EU's Product Governance regime, including, without limitation, target market information. Distributors and

intermediaries may obtain such information on request from MSIM Fund Management (Ireland) Limited, Luxembourg Branch at *cslux@morganstanley.com*.

In particular, the Shares have not been and will not be registered under the United States Securities Act of 1933, as amended (nor has the Company been registered under the United States Investment Company Act of 1940, as amended) and may not be offered or sold, directly or indirectly, in the United States of America or its territories or possessions or areas subject to its jurisdiction, or to citizens or residents thereof other than in accordance with the laws of the United States, provided however that no provisions of this Prospectus shall prevent a Morgan Stanley entity or any of its affiliates or subsidiaries from owning Shares.

Similarly, the Shares of the Morgan Stanley Investment Funds Indian Equity Fund may not be offered or sold, directly or indirectly, to persons resident in India.

Following the implementation of the Directive (EU) 2019/1160 of the European Parliament and of the Council of 20 June 2019, amending the Directive 2009/65/EC and 2011/61/EU with regard to cross-border distribution of collective investment undertakings (the "CBDF Directive"), Member States shall ensure that a UCITS makes available, in each Member State where it intends to market its units, facilities to perform the tasks listed under article 92 of the UCITS Directive. In order to comply with such new requirements, the Company has appointed different entities in the various Member States of the EU to render these tasks. The exhaustive list of (i) the tasks to be rendered and (ii) the entities in charge of rendering such tasks is set out in "Appendix K - Facilities Agents and Services" and is also available on the following website: www.morganstanleyinvestmentfunds.com.

The date of this Prospectus is November 2022.

The Management Company acts as the global distributor for marketing the Shares and the Management Company may appoint sub-distributors (each a "distributor"). The duties of the distributors may include passing the subscription, redemption and conversion orders to the Company's central administration in Luxembourg. The distributors may not offset the orders received or carry out any duties connected to the individual processing of the subscription, redemption and conversion orders. In addition, any investor may deal directly with the Management Company in order to subscribe for, redeem or convert Shares.

The Directors have taken all reasonable care to ensure that at the date of this Prospectus the information contained herein is accurate and complete in all material respects. The Directors accept responsibility accordingly. However, the Directors do not accept responsibility with regard to the content of the Prospectus or any information relating to the Shares other than to the Shareholders of the Company.

Any information given by any person not mentioned in the Prospectus should be regarded as unauthorised. The information contained in the Prospectus is considered to be accurate at the date of its publication. To reflect material changes, this document may be updated from time to time and potential subscribers should enquire of the Company as to the issue of any later Prospectus.

It should be remembered that the price of the Shares can go down as well as up. An investor may not get back the amount he has invested, particularly if Shares are redeemed soon after they are issued and the Shares have been subject to a Sales Charge or transaction charge. Changes in exchange rates may also cause the value of Shares in the investor's base currency to go up or down.

The Company determines the principles of the calculation of the price or net asset value of its Shares, which are implemented by the Management Company on a forward basis. This means that it is not possible to know in advance the Net Asset Value per Share at which Shares will be bought or sold (exclusive of any Sales Charge). The Net Asset Value per Share is calculated at the valuation point following the Cut-Off Point (as defined below).

The Company's Funds are not designed for investors with short term investment horizons. Activities which may adversely affect the interests of the Company's Shareholders (for example that disrupt investment strategies or impact expenses) are not permitted. Specifically, market timing is not permitted.

Whilst recognising that Shareholders may have legitimate needs to adjust their investments from time to time, the Management Company in its discretion may, if it deems such activities adversely affect the interests of the Company's Shareholders, take action as appropriate to deter such activities.

Accordingly if the Management Company determines or suspects that a Shareholder has engaged in such activities, it may suspend, cancel, reject or otherwise deal with that Shareholder's subscription or conversion applications and take any action or measures as appropriate or necessary to protect the Company and its Shareholders.

Such measures may include the imposition of a redemption fee on the redemption proceeds of Shareholders whom the Management Company determines to have engaged in such activities, or the imposition of limits on the number of conversions of Shares between Funds which are permitted, as described under "Redemption of Shares" and "Conversion of Shares".

Potential subscribers or purchasers of Shares should inform themselves as to (a) the possible tax consequences, (b) the legal requirements, and (c) any foreign exchange restrictions or exchange control requirements which they might encounter under the laws of the countries of their citizenship, residence or domicile and which might be relevant to the subscription, purchase, holding, conversion or sale of Shares.

Definitions

All references in the Prospectus to:

"Business Day" refers to any day on which banks are open for business in London,

Luxembourg, New York and Tokyo as the case may be;

"CET" refers to Central European Time;

"Cash Equivalents" means bank deposits (excluding bank deposits at sight), money market

instruments, money market funds, or other eligible assets listed under

article 41(1) of the 2010 Law;

"China A-Shares" refers to shares denominated and traded in Chinese Yuan on the

Shanghai Stock Exchange or the Shenzhen Stock Exchange and issued

by Chinese companies;

"Contingent Convertible Instrument" means a debt security which may be converted into equity securities

or suffer capital losses through decreasing its face value if prespecified events occur, depending in particular on the capital ratio

levels of the issuer of the security;

"Controlling Person" means the natural persons who exercise control over an entity. In the

case of a trust, the settlor(s), the trustee(s), the protector(s) (if any), the beneficiary(ies) or class(es) of beneficiaries, and any other natural person(s) exercising ultimate effective control over the trust, and in the case of a legal arrangement other than a trust, such term means persons in equivalent or similar positions. The term "Controlling Persons" must be interpreted in a manner consistent with the Financial Action

Task Force Recommendations;

"Cut-Off Point" refers to the time by which the Transfer Agent must receive applications for subscription, conversion or redemption in respect of a

Dealing Day for such application to be processed on such Dealing Day.

This shall mean 1 pm CET on a Dealing Day for all Funds;

"Dealing Day" refers to any full Luxembourg Business Day for all Funds, save where a particular Fund has a different definition included in the investment policy section relating to that Fund. A Dealing Day shall not fall within a period of suspension of calculation of the Net Asset Value of a

> particular Fund. The Directors may also take into account periods during which principal stock exchanges or other markets on which a substantial portion of the investments of the Company attributable to a Fund from time to time is quoted or dealt in are closed otherwise than for ordinary holidays, or dealings therein are restricted or suspended, provided that such restriction or suspension affects the valuation of the investments of the Company attributable to such Fund quoted on such

> principal stock exchanges or other markets. The Directors may elect to treat such closures as "non-Dealing Days" for Funds which trade a substantial amount of their portfolio on these closed principal stock exchanges or other markets. A list of expected "non-Dealing Days" for

> the Funds is available from the Company's website (www.morganstanleyinvestmentfunds.com) and will be updated in

advance, at least semi-annually;

"EEA" refers to the European Economic Area;

"ESG" means environmental, social and governance factors, which are a subset of non-financial performance indicators which include

sustainable, ethical and corporate governance issues such as, without

limitation, the impact of a company on the environment, the conduct of social and business relationships and governance ethics. These three factors, as determined by the Investment Adviser or Investment Sub-Adviser for any particular Fund, may be considered in addition to traditional financial analysis, securities selection and portfolio construction processes;

"ETF"

refers to exchange traded fund;

"EU"

refers to the European Union;

"Euro"

refers to the currency of the member states of the EU that adopt the single currency in accordance with the Treaty establishing the European Economic Community (signed in Rome on 25 March 1957), as amended by the Treaty on EU (signed in Maastricht on 7 February 1992);

"Eurozone"

refers to those member states of the EU which have adopted the Euro as their national currency;

Fixed Income Securities"

shall mean:

- (i) transferable securities other than equity securities; and
- (ii) money market instruments.

For the avoidance of doubt, this includes:

- (i) both fixed and floating rate instruments;
- (ii) debt securities of any type, including all types of bonds and debentures, and including securitised debt of any type, including asset-backed securities, whether backed by mortgages including uniform mortgage-backed securities ("mortgage-backed securities") or other receivables such as credit card debt or other loans that have been securitised; and
- (iii) all instruments that can be considered money market instruments, including without limitation commercial paper.

For these purposes subordinated and/or hybrid securities, convertible bonds and Contingent Convertible Instruments are considered to be "transferable securities other than equity securities".

For the avoidance of doubt, this definition shall only include instruments of the above types that are also Eligible Assets for investments of UCITS, subject to the provisions of Directive 2007/16/EC of 19 March 2007 implementing Council Directive 85/611/EEC on the coordination of laws, regulations and administrative provisions relating to undertakings for collective investment in transferable securities (UCITS) as regards the clarification of certain definitions (the "Directive 2007/16/EC");

"JPY" or "Yen"

refer to the currency of Japan;

the "Law"

refers to the laws of the Grand Duchy of Luxembourg;

"Located"

means the Investment Adviser's opinion on where an issuer is located, for the purposes of a specific Fund. This may differ from Fund to Fund.

The Investment Adviser's opinion will be based on one or more of the following factors:

- (i) The primary market on which the issuer's equity securities are listed, quoted or traded;
- (ii) The issuer's country of incorporation;
- (iii) The issuer's country of domicile;
- (iv) The countries from which the issuer derives its revenue;
- (v) Any other factors which the Investment Adviser, acting reasonably, considers denote an economic exposure to a particular location;

"Moody's"

means Moody's Investors Service, Inc.;

"Net Asset Value" or "NAV"

refer to the net asset value of the Funds or the Classes within a Fund depending on the context;

"OECD"

refers to the Organisation for Economic Co-operation and Development;

"Reference Currency"

refers to the reference currency as defined for each fund under Section 1.1 "The Company and the Funds" of this Prospectus;

"S&P"

means Standard & Poor's Corporation;

"Settlement Period"

means the contractual settlement period for subscriptions, redemptions and conversions of Shares, and unless otherwise indicated in Section 1.2 "Investment Objectives and Policies", means up to five (5) Business Days after the relevant Dealing Day for Saudi Equity Fund and up to three (3) Business Days after the relevant Dealing Day for all other Funds. If banks or interbank settlement systems in the country of the settlement currency or the Share Class Currency are closed or not operational on the settlement date, or if the settlement date falls on a holiday as set out in the Company's list of expected "non-Dealing Days" (which can be www.morganstanleyinvestmentfunds.com www.morganstanley.com/im), settlement will be delayed until such banks or interbank settlement systems are open and operating. Any day within the Settlement Period that is a holiday or a non-Dealing Day as set out in the Company's list of expected "non-Dealing Days" or as determined by the Directors for a Fund will be excluded when determining the settlement date;

"SFTs"

means securities financing transactions within the meaning of Regulation (EU) 2015/2365 of the European Parliament and of the Council of 25 November 2015 on transparency of securities financing transactions and of reuse;

"Shanghai Stock Connect"

means the Shanghai-Hong Kong Stock Connect program;

"Shenzhen Stock Connect"

means the Shenzhen-Hong Kong Stock Connect program;

"Sterling" or "£"

refer to the currency of the United Kingdom;

"Stock Connect"

means the Shanghai Stock Connect and the Shenzhen Stock Connect which allow non-Chinese investors to purchase certain China A-Shares via brokers in Hong Kong and/or any other similar stock connect program between any other city of the People's Republic of

China and Hong Kong when it becomes available to, and can be utilised by, the Company;

"Taxonomy Regulation"

refers to Regulation (EU) 2020/852 of the European Parliament and of the Council of 18 June 2020 on the establishment of a framework to facilitate sustainable investment;

"USD", "US\$", "US Dollars" and "\$"

refer to the currency of the United States of America; and

"VaR"

refer to Value at Risk.

Section 1

1.1 The Company and the Funds

The Company

The Company is an open-ended investment company with variable capital ("société d'investissement à capital variable") incorporated in Luxembourg with limited liability under Part I of the 2010 Law. The Company is presently structured to provide both institutional and individual investors with a variety of funds (hereinafter referred to collectively as the "Funds" or singularly as a "Fund"). The Company offers a range of equity, bond, asset allocation and alternative investment funds denominated in the currencies specified below (the "Reference Currencies").

Some Funds are authorised to use derivatives and efficient portfolio management techniques as an important part of their investment strategies as described in the Funds' investment objectives. For the purpose of the UCITS Directive and the European and Luxembourg regulation following therefrom (the "UCITS Regulations") as applicable to the Company, these Funds are classified according to the methodology adopted by the Management Company in order to calculate the global risk exposure of each Fund. Such a classification can be found under Section 1.3 "Methodology for Calculating the Global Exposure".

Equity Funds

Morgan Stanley Investment Funds American Resilience Fund, (Reference Currency: USD) (the "American Resilience Fund")¹

Morgan Stanley Investment Funds Asia Opportunity Fund, (Reference Currency: USD) (the "Asia Opportunity Fund")

Morgan Stanley Investment Funds Asian Property Fund, (Reference Currency: USD) (the "Asian Property Fund")

Morgan Stanley Investment Funds Calvert Sustainable Climate Aligned Fund, (Reference Currency: USD) (the "Calvert Sustainable Climate Aligned Fund")

Morgan Stanley Investment Funds Calvert Sustainable Climate Transition Fund, (Reference Currency: USD) (the "Calvert Sustainable Climate Transition Fund")

Morgan Stanley Investment Funds Calvert Sustainable Developed Europe Equity Select Fund, (Reference Currency: Euro) (the "Calvert Sustainable Developed Europe Equity Select Fund")

Morgan Stanley Investment Funds Calvert Sustainable Developed Markets Equity Select Fund, (Reference Currency: USD) (the "Calvert Sustainable Developed Markets Equity Select Fund")

Morgan Stanley Investment Funds Calvert Sustainable Diversity, Equity and Inclusion Fund, (Reference Currency: USD) (the "Calvert Sustainable Diversity, Equity and Inclusion Fund")

Morgan Stanley Investment Funds Calvert Sustainable Emerging Markets Equity Select Fund, (Reference Currency: USD) (the "Calvert Sustainable Emerging Markets Equity Select Fund")

Morgan Stanley Investment Funds Calvert Sustainable Global Equity Engagement Fund, (Reference Currency: USD) (the "Calvert Sustainable Global Equity Engagement Fund")¹

Morgan Stanley Investment Funds Calvert Sustainable US Equity Select Fund, (Reference Currency: USD) (the "Calvert Sustainable US Equity Select Fund")

Morgan Stanley Investment Funds China A-shares Fund, (Reference Currency: USD) (the "China A-shares Fund")

Morgan Stanley Investment Funds China Equity Fund, (Reference Currency: USD) (the "China Equity Fund")

Morgan Stanley Investment Funds Counterpoint Global Fund, (Reference Currency: USD) (the "Counterpoint Global Fund")

¹ The Fund is not available for subscription at the date of this Prospectus. The Fund may be launched at the Directors' discretion, at which time, confirmation of the launch of the Fund will be made available at the registered office of the Company.

Morgan Stanley Investment Funds Developing Opportunity Fund, (Reference Currency: USD) (the "Developing Opportunity Fund")

Morgan Stanley Investment Funds Emerging Leaders Equity Fund, (Reference Currency: USD) (the "Emerging Leaders Equity Fund")

Morgan Stanley Investment Funds Europe Opportunity Fund, (Reference Currency: Euro) (the "Europe Opportunity Fund")

Morgan Stanley Investment Funds European Property Fund, (Reference Currency: Euro) (the "European Property Fund")

Morgan Stanley Investment Funds Global Brands Fund, (Reference Currency: USD) (the "Global Brands Fund")

Morgan Stanley Investment Funds Global Brands Equity Income Fund, (Reference Currency: USD) (the "Global Brands Equity Income Fund")

Morgan Stanley Investment Funds Global Core Equity Fund, (Reference Currency: USD) (the "Global Core Equity Fund")

Morgan Stanley Investment Funds Global Endurance Fund, (Reference Currency: USD) (the "Global Endurance Fund")

Morgan Stanley Investment Funds Global Focus Property Fund, (Reference Currency: USD) (the "Global Focus Property Fund")

Morgan Stanley Investment Funds Global Infrastructure Fund, (Reference Currency: USD) (the "Global Infrastructure Fund")

Morgan Stanley Investment Funds Global Insight Fund, (Reference Currency: USD) (the "Global Insight Fund")

Morgan Stanley Investment Funds Global Opportunity Fund, (Reference Currency: USD) (the "Global Opportunity Fund")

Morgan Stanley Investment Funds Global Permanence Fund, (Reference Currency: USD) (the "Global Permanence Fund")

Morgan Stanley Investment Funds Global Property Fund, (Reference Currency: USD) (the "Global Property Fund")

Morgan Stanley Investment Funds Global Quality Fund, (Reference Currency: USD) (the "Global Quality Fund")

Morgan Stanley Investment Funds Global Sustain Fund, (Reference Currency: USD) (the "Global Sustain Fund")

Morgan Stanley Investment Funds Indian Equity Fund, (Reference Currency: USD) (the "Indian Equity Fund")

Morgan Stanley Investment Funds International Equity (ex US) Fund, (Reference Currency: USD) (the "International Equity (ex US) Fund")

Morgan Stanley Investment Funds Japanese Equity Fund, (Reference Currency: Yen) (the "Japanese Equity Fund")

Morgan Stanley Investment Funds Japanese Small Cap Equity Fund, (Reference Currency: Yen) (the "Japanese Small Cap Equity Fund")

Morgan Stanley Investment Funds Latin American Equity Fund, (Reference Currency: USD) (the "Latin American Equity Fund")

Morgan Stanley Investment Funds NextGen Emerging Markets Fund, (Reference Currency: Euro) (the "NextGen Emerging Markets Fund")

Morgan Stanley Investment Funds Parametric Emerging Markets Fund, (Reference Currency: USD) (the "Parametric Emerging Markets Fund")¹

Morgan Stanley Investment Funds Parametric Global Defensive Equity Fund, (Reference Currency: USD) (the "Parametric Global Defensive Equity Fund") 1

Morgan Stanley Investment Funds Saudi Equity Fund, (Reference Currency: USD) (the "Saudi Equity Fund")

Morgan Stanley Investment Funds Sustainable Asia Equity Fund, (Reference Currency: USD) (the "Sustainable Asia Equity Fund")

Morgan Stanley Investment Funds Sustainable Emerging Markets Equity Fund, (Reference Currency: USD) (the "Sustainable Emerging Markets Equity Fund")

Morgan Stanley Investment Funds US Advantage Fund, (Reference Currency: USD) (the "US Advantage Fund")

Morgan Stanley Investment Funds US Core Equity Fund, (Reference Currency: USD) (the "US Core Equity Fund")

Morgan Stanley Investment Funds US Focus Property Fund, (Reference Currency: USD) (the "US Focus Property Fund")

Morgan Stanley Investment Funds US Growth Fund, (Reference Currency: USD) (the "US Growth Fund")

Morgan Stanley Investment Funds US Insight Fund, (Reference Currency: USD) (the "US Insight Fund")

Morgan Stanley Investment Funds US Permanence Fund, (Reference Currency: USD) (the "US Permanence Fund")

Morgan Stanley Investment Funds US Property Fund, (Reference Currency: USD) (the "US Property Fund")

Morgan Stanley Investment Funds US Value Fund, (Reference Currency: USD) (the "US Value Fund")1

Morgan Stanley Investment Funds Vitality Fund, (Reference Currency: USD) (the "Vitality Fund")

Bond Funds

Morgan Stanley Investment Funds Calvert Sustainable Global Green Bond Fund, (Reference Currency: USD) (the "Calvert Sustainable Global Green Bond Fund")

Morgan Stanley Investment Funds Emerging Markets Corporate Debt Fund, (Reference Currency: USD) (the "Emerging Markets Corporate Debt Fund")

Morgan Stanley Investment Funds Emerging Markets Debt Fund, (Reference Currency: USD) (the "Emerging Markets Debt Fund")

Morgan Stanley Investment Funds Emerging Markets Debt Opportunities Fund, (Reference Currency: USD) (the "Emerging Markets Debt Opportunities Fund")¹

Morgan Stanley Investment Funds Emerging Markets Domestic Debt Fund, (Reference Currency: USD) (the "Emerging Markets Domestic Debt Fund")

Morgan Stanley Investment Funds Emerging Markets Fixed Income Opportunities Fund, (Reference Currency: USD) (the "Emerging Markets Fixed Income Opportunities Fund")

Morgan Stanley Investment Funds Emerging Markets Local Income Fund, (Reference Currency: USD) (the "Emerging Markets Local Income Fund")¹

Morgan Stanley Investment Funds Euro Bond Fund, (Reference Currency: Euro) (the "Euro Bond Fund")

Morgan Stanley Investment Funds Euro Corporate Bond Fund, (Reference Currency: Euro) (the "Euro Corporate Bond Fund")

Morgan Stanley Investment Funds Euro Corporate Bond – Duration Hedged Fund, (Reference Currency: Euro) (the "Euro Corporate Bond – Duration Hedged Fund")

Morgan Stanley Investment Funds Euro Strategic Bond Fund, (Reference Currency: Euro) (the "Euro Strategic Bond Fund")

Morgan Stanley Investment Funds European Fixed Income Opportunities Fund, (Reference Currency: Euro) (the "European Fixed Income Opportunities Fund")

Morgan Stanley Investment Funds European High Yield Bond Fund, (Reference Currency: Euro) (the "European High Yield Bond Fund")

Morgan Stanley Investment Funds Floating Rate ABS Fund, (Reference Currency: USD) (the "Floating Rate ABS Fund)¹

Morgan Stanley Investment Funds Frontier Markets Local Debt Fund, (Reference Currency: USD) (the "Frontier Markets Local Debt Fund")¹

Morgan Stanley Investment Funds Global Asset Backed Securities Fund, (Reference Currency: USD) (the "Global Asset Backed Securities Fund")

Morgan Stanley Investment Funds Global Bond Fund, (Reference Currency: USD) (the "Global Bond Fund")

Morgan Stanley Investment Funds Global Convertible Bond Fund, (Reference Currency: USD) (the "Global Convertible Bond Fund")

Morgan Stanley Investment Funds Global Credit Fund, (Reference Currency: USD) (the "Global Credit Fund")

Morgan Stanley Investment Funds Global Credit Opportunities Fund, (Reference Currency: USD) (the "Global Credit Opportunities Fund")¹

Morgan Stanley Investment Funds Global Fixed Income Opportunities Fund, (Reference Currency: USD) (the "Global Fixed Income Opportunities Fund")

Morgan Stanley Investment Funds Global High Yield Bond Fund, (Reference Currency: USD) (the "Global High Yield Bond Fund")

Morgan Stanley Investment Funds Global Macro Fund, (Reference Currency: USD) (the "Global Macro Fund")1

Morgan Stanley Investment Funds Short Duration US Government Income Fund, (Reference Currency: USD) (the "Short Duration US Government Income Fund")¹

Morgan Stanley Investment Funds Short Maturity Euro Bond Fund, (Reference Currency: Euro) (the "Short Maturity Euro Bond Fund")

Morgan Stanley Investment Funds Short Maturity Euro Corporate Bond Fund, (Reference Currency: Euro) (the "Short Maturity Euro Corporate Bond Fund)¹

Morgan Stanley Investment Funds Sustainable Euro Corporate Bond Fund, (Reference Currency: Euro) (the "Sustainable Euro Corporate Bond Fund")

Morgan Stanley Investment Funds Sustainable Euro Strategic Bond Fund, (Reference Currency: Euro) (the "Sustainable Euro Strategic Bond Fund")

Morgan Stanley Investment Funds Sustainable European High Yield Bond Fund, (Reference Currency: Euro) (the "Sustainable European High Yield Bond Fund") 1

Morgan Stanley Investment Funds Sustainable Global Fixed Income Opportunities Fund, (Reference Currency: USD) (the "Sustainable Global Fixed Income Opportunities Fund")¹

Morgan Stanley Investment Funds Sustainable Global High Yield Bond Fund, (Reference Currency: USD) (the "Sustainable Global High Yield Bond Fund")

Morgan Stanley Investment Funds Sustainable US High Yield Bond Fund, (Reference Currency: USD) (the "Sustainable US High Yield Bond Fund") 1

Morgan Stanley Investment Funds US Dollar Corporate Bond Fund, (Reference Currency: USD) (the "US Dollar Corporate Bond Fund")

Morgan Stanley Investment Funds US Dollar High Yield Bond Fund, (Reference Currency: USD) (the "US Dollar High Yield Bond Fund")

Morgan Stanley Investment Funds US Dollar Short Duration Bond Fund, (Reference Currency: USD) (the "US Dollar Short Duration Bond Fund")

Morgan Stanley Investment Funds US Dollar Short Duration High Yield Bond Fund, (Reference Currency: USD) (the "US Dollar Short Duration High Yield Bond Fund")

Asset Allocation Funds

Morgan Stanley Investment Funds Global Balanced Defensive Fund, (Reference Currency: Euro) (the "Global Balanced Defensive Fund")

Morgan Stanley Investment Funds Global Balanced Fund, (Reference Currency: Euro) (the "Global Balanced Fund")

Morgan Stanley Investment Funds Global Balanced Income Fund (Reference Currency: Euro) (the "Global Balanced Income Fund")

Morgan Stanley Investment Funds Global Balanced Risk Control Fund of Funds, (Reference Currency: Euro) (the "Global Balanced Risk Control Fund of Funds")

Morgan Stanley Investment Funds Global Balanced Sustainable Fund, (Reference Currency: Euro) (the "Global Balanced Sustainable Fund")

Morgan Stanley Investment Funds Global Multi-Asset Opportunities Fund, (Reference Currency: Euro) (the "Global Multi-Asset Opportunities Fund")

Morgan Stanley Investment Funds Multi-Asset Risk Control Fund, (Reference Currency: USD) (the "Multi-Asset Risk Control Fund")

1.2 Investment Objectives and Policies

The purpose of the Company is to provide investors with an opportunity for investment in a professionally managed fund range which seeks to spread investment risks in order to achieve an optimum return from the capital invested.

For this purpose the Company offers a range of Funds which allow investors to make their own strategic allocation by combining holdings in the various Funds in proportions of their own choosing.

In accordance with the UCITS regime, the information below includes a description of the profile of the typical investor for whom each Fund has been designed. Please note however that this information does not constitute investment advice and potential investors should consult their own professional advisers concerning the acquisition, holding or disposal of any Shares in any of the Funds. Neither the Company nor the Management Company makes any representation in relation to the suitability, appropriateness or otherwise of an investment in Shares in any of the Funds.

Sub-distributors or other intermediaries who require the manufacturer's target market assessment for the purposes of MiFID II should contact the Management Company at: cslux@morganstanley.com. Funds will not be offered for sale to the public in a given jurisdiction until all proper authorisations in the relevant jurisdiction are obtained.

Each of the Funds is managed in accordance with the investment and borrowing restrictions specified in Appendix A.

The Funds are authorised to use derivatives either for hedging or efficient portfolio management purposes or as part of their investment strategies as described in the Funds' investment objectives. Unless stated otherwise in the investment objectives of a Fund, a Fund which uses derivatives will do so for hedging and/or efficient portfolio management purposes only. Funds using derivatives will do so within the limits specified in Section 3 of Appendix A. Investors should refer to Section 1.5 "Risk Factors" for special risk considerations applicable to derivatives. The Funds may gain exposure to eligible financial indices or reference assets which are in line with their investment objectives through one or several total return swaps ("TRS") or similar financial derivative instruments, as further described under Appendix A.

The Funds listed below may engage in the trading of futures, options on futures or on commodities and/or certain swaps. Although Morgan Stanley Investment Management Inc., a Sub-Adviser, is registered with the U.S. Commodity Futures Trading Commission ("CFTC") as a commodity trading advisor ("CTA") and is a member of the U.S. National Futures Association in such capacity, in respect of the Funds below for which it acts as sub-adviser, Morgan Stanley Investment Management Inc. relies on an exemption from registration as a CTA under CFTC Rule 4.14(a)(8) and, accordingly, will provide commodity interest trading advice to the Funds as if it were exempt from registration as a CTA. Morgan Stanley Investment Management Limited and the remaining Sub-Advisers are exempt from registration with the CFTC as a CTA under Rule 3.10(c)(3) in respect of the Funds below for which each entity acts as Investment Adviser or Sub-Adviser, as applicable.

American Resilience Fund Asia Opportunity Fund Calvert Sustainable Climate Aligned Fund Calvert Sustainable Climate Transition Fund Calvert Sustainable Developed Europe Equity Select Fund

Calvert Sustainable Developed Markets Equity Select Fund

Calvert Sustainable Diversity, Equity and Inclusion Fund

Calvert Sustainable Emerging Markets Equity Select Fund

Calvert Sustainable Global Equity Engagement Fund

Calvert Sustainable Global Green Bond Fund

Calvert Sustainable US Equity Select Fund

Counterpoint Global Fund

Developing Opportunity Fund

Emerging Markets Corporate Debt Fund

Emerging Markets Debt Fund

Emerging Markets Debt Opportunities Fund

Emerging Markets Domestic Debt Fund

Emerging Markets Fixed Income Opportunities Fund

Emerging Markets Local Income Fund

Europe Opportunity Fund

European Fixed Income Opportunities Fund

Floating Rate ABS Fund

Frontier Markets Local Debt Fund

Global Asset Backed Securities Fund

Global Balanced Sustainable Fund

Global Bond Fund

Global Convertible Bond Fund

Global Core Equity Fund

Global Credit Fund

Global Credit Opportunities Fund

Global Endurance Fund

Global Fixed Income Opportunities Fund

Global Focus Property Fund

Global High Yield Bond Fund

Global Infrastructure Fund

Global Insight Fund

Global Macro Fund

Global Multi-Asset Opportunities Fund

Global Opportunity Fund

Global Permanence Fund

Global Property Fund

Latin American Equity Fund

NextGen Emerging Markets Fund

Parametric Emerging Markets Fund

Parametric Global Defensive Equity Fund

Short Duration US Government Income Fund Short Maturity Euro Corporate Bond Fund

Short Watarity Early Corporate Bond Fund

Sustainable Emerging Markets Equity Fund

Sustainable European High Yield Bond Fund Sustainable Global Fixed Income Opportunities Fund

Sustainable Global High Yield Bond Fund

Sustainable US High Yield Bond Fund

US Advantage Fund

US Core Equity Fund

US Dollar Corporate Bond Fund

US Dollar High Yield Bond Fund

US Dollar Short Duration Bond Fund

US Dollar Short Duration High Yield Bond Fund

US Focus Property Fund

US Growth Fund

US Insight Fund

US Permanence Fund

US Value Fund

Vitality Fund

The Funds are authorised to employ efficient portfolio management techniques within the limits specified in Section 3 of Appendix A. The attention of investors is drawn to the fact that all Funds may from time to time use efficient portfolio management techniques such as securities lending and repurchase and reverse repurchase transactions. Investors should refer to Section 1.5 "Risk Factors" for special risk considerations applicable to efficient portfolio management techniques and derivatives.

An investment in any of the Company's Funds is not a deposit in a bank or other insured depository institution. Investment may not be appropriate for all investors. Each Fund is not intended to be a complete investment programme and investors should consider their investment goals and financial needs when making an investment decision about a particular Fund. An investment in the Company's Funds is intended to be a medium or long term investment. Funds should not be used as trading vehicle.

General information relating to the Funds

The term "limited extent" will refer to a level equal to that of 10% or less of the actually invested assets (net assets after deducting Cash Equivalents) of each Fund. Individual Funds may impose different levels and these will be outlined in the objectives for that Fund. The investment objectives and strategies of the Funds may not exhaustively disclose all ancillary eligible investments under the 2010 Law that the Funds may hold, provided such eligible investments do not exceed 5% of the net assets of the Funds.

Each Fund may hold ancillary liquid assets (*i.e.*, bank deposits at sight, such as cash held in current accounts with a bank accessible at any time) up to 20% of their net assets in order to cover current or exceptional payments, or for the time necessary to reinvest in eligible assets provided under article 41(1) of the 2010 Law or for a period of time strictly necessary in case of unfavourable market conditions. Such restriction may, under exceptionally unfavourable market conditions (for instance in highly serious circumstances such as the September 11 attacks or the bankruptcy of Lehman Brothers in 2008), temporarily be exceeded for a period of time strictly necessary up to 100% of their net assets, in order to take measures to mitigate risks relative to such exceptional unfavourable market conditions, in the best interest of the Shareholders.

Unless otherwise specified in their investment objectives and strategies, all Funds may hold Cash Equivalents up to 100% of their net assets in order to (i) achieve their investment goals, and/or for (ii) treasury purposes, and/or (iii) in case of unfavourable market conditions. Money market instruments include without limitation commercial papers, certificates of deposit and short term government bonds. All Funds may also invest in short-term money market funds and money market funds as defined by CESR's Guidelines on a common definition of European money market funds dated 19 May 2010 (CESR/10-049) (Short-Term Money Market Funds and Money Market Funds). This may include investment in any sub-funds of the Morgan Stanley Liquidity Funds or any other Short-Term Money Market Funds and Money Market Funds managed by the Investment Advisers or any of the Sub-Advisers.²

The Russian Trading Stock Exchange and the Moscow Interbank Currency Exchange are the only exchanges in the Russian Federation that qualify as Recognised Exchanges within the meaning of Article 41 (1) of the 2010 Law.

Any investment in SPACs will be limited to 10% of the net assets of the Funds.

The Funds may invest in China A-Shares via Stock Connect to the extent authorised under their investment objective. Investors of the relevant Funds should refer to Section 1.5 "Risk Factors" for special risk considerations applicable to investment in China A-Shares via Stock Connect.

If a currency is mentioned in brackets in the name of a Fund, such currency is the reference currency of such Fund and is used for performance measurement and accounting purposes. It may differ from the investment currency of the Fund.

There can be no guarantee that the investment objectives of the Funds will be met.

Investors should consult the key investor information document of the relevant Class of Share, for information relating to historic performance.

General information relating to Sustainability Risks integration

In keeping with Morgan Stanley Investment Management's commitment to sustainable investing, the Company has adopted a Sustainable Investing Policy which can be found on www.morganstanley.com/im/sustainable-investing and on www.morganstanleyinvestmentfunds.com. The Company's Sustainable Investing Policy outlines how the Company's governance structure, ESG integration, ESG Funds and stewardship, engagement and voting activities support the Company to fulfil its fiduciary duty and respond to investor preferences and needs. Stewardship activities in particular, are integral to our role as active long-term investors as they enable us to encourage companies and issuers to improve their performance on a range of sustainability issues, risks and opportunities. Approaches to engagement and stewardship vary across our investment teams based on the specific security/company, asset class, investment strategy and materiality of sustainability issues, risks and opportunities. Further details can be found in

² The Morgan Stanley Liquidity Funds (including all of its sub-funds) are not approved for distribution to non-qualified investors in Switzerland.

our <u>Sustainable Investing Policy</u> which can be found on <u>www.morganstanley.com/im/sustainable-investing</u> and on www.morganstanleyinvestmentfunds.com.

Pursuant to EU Regulation (EU) 2019/2088 on sustainability-related disclosures in the financial services sector (the "SFDR"), the Funds are required to disclose the manner in which Sustainability Risks (as defined in section 1.5.1. General Risk Factors hereafter) are integrated into the investment decision and the results of the assessment of the likely impacts of Sustainability Risks on the returns of the Funds.

The Company recognizes that various Sustainability Risks can threaten the investments at individual asset level and portfolio level. These Sustainability Risks may include climate change transition and physical risks, natural resources depletion, waste intensity, labor retention, turnover and unrest, supply chain disruption, corruption and fraud and reputational concerns associated with human rights violations.

The Company recognizes that the universe of relevant Sustainability Risks will grow and evolve over time. The materiality of such risks and financial impacts on an individual asset and on a portfolio as a whole depends on industry, country, asset class, and investment style. The Investment Advisers, the Sub-Advisers, or the Management Company, as the case may be, are ultimately responsible for the incorporation of materially relevant Sustainability Risks into due diligence and research, valuation, asset selection, portfolio construction, and ongoing investment monitoring alongside with other material risk factors.

The assessment of the likely impact of the relevant Sustainability Risks must therefore be conducted at each Fund level. The Funds, having a diversified portfolio, may be exposed to different Sustainability Risks varying from issuers, markets, sectors, financial instruments, geographical regions, etc. Unless specific information is given in relation to a Fund under Section 1.2 "Investment Objectives and Policies", it is not anticipated that any single Sustainability Risk will drive a material negative financial impact on the value of a Fund. Additional information about SFDR as applicable to the Company, as the case may be, can be found on www.morganstanley.com/im.

The Funds do not promote environmental or social characteristics nor have a sustainable investment as their objective and are considered as falling within the scope of Article 6 of the SFDR except the Funds listed below³:

- American Resilience Fund (Article 8 of the SFDR);
- Asia Opportunity Fund (Article 8 of the SFDR);
- Calvert Sustainable Climate Aligned Fund (Article 9 of the SFDR);
- Calvert Sustainable Climate Transition Fund (Article 9 of the SFDR);
- Calvert Sustainable Developed Europe Equity Select Fund (Article 9 of the SFDR);
- Calvert Sustainable Developed Markets Equity Select Fund (Article 9 of the SFDR);
- Calvert Sustainable Diversity, Equity and Inclusion Fund (Article 9 of the SFDR);
- Calvert Sustainable Emerging Markets Equity Select Fund (Article 9 of the SFDR);
- Calvert Sustainable Global Equity Engagement Fund (Article 9 of the SFDR);
- Calvert Sustainable Global Green Bond Fund (Article 9 of the SFDR);
- Calvert Sustainable US Equity Select Fund (Article 9 of the SFDR);
- Counterpoint Global Fund (Article 8 of the SFDR);
- Developing Opportunity Fund (Article 8 of the SFDR);
- Emerging Leaders Equity Fund (Article 8 of the SFDR);
- Emerging Markets Corporate Debt Fund (Article 8 of the SFDR);
- Emerging Markets Debt Fund (Article 8 of the SFDR);
- Emerging Markets Debt Opportunities Fund (Article 8 of the SFDR);
- Emerging Markets Domestic Debt Fund (Article 8 of the SFDR);
- Emerging Markets Fixed Income Opportunities Fund (Article 8 of the SFDR);
- Emerging Markets Local Income Fund (Article 8 of the SFDR);
- Euro Bond Fund (Article 8 of the SFDR);

³ As from the 12 December 2022, the following Funds will fall within the scope of Article 8 of the SFDR (please refer to the relevant footnotes under their respective investment objectives and policies below):

o European Property Fund;

14

Asian Property Fund;

Global Focus Property Fund;

Global Infrastructure Fund;

o Global Property Fund;

o US Focus Property Fund; and

US Property Fund,

- Euro Corporate Bond Fund (Article 8 of the SFDR);
- Euro Corporate Bond Duration Hedged Fund (Article 8 of the SFDR);
- Euro Strategic Bond Fund (Article 8 of the SFDR);
- Europe Opportunity Fund (Article 8 of the SFDR);
- European Fixed Income Opportunities Fund (Article 8 of the SFDR);
- European High Yield Bond Fund (Article 8 of the SFDR);
- Floating Rate ABS Fund (Article 8 of the SFDR);
- Frontier Markets Local Debt Fund (Article 8 of the SFDR);
- Global Asset Backed Securities Fund (Article 8 of the SFDR);
- Global Balanced Fund (Article 8 of the SFDR);
- Global Balanced Sustainable Fund (Article 8 of the SFDR);
- Global Bond Fund (Article 8 of the SFDR);
- Global Brands Equity Income Fund (Article 8 of the SFDR);
- Global Brands Fund (Article 8 of the SFDR);
- Global Convertible Bond Fund (Article 8 of the SFDR);
- Global Credit Fund (Article 8 of the SFDR);
- Global Credit Opportunities Fund (Article 8 of the SFDR);
- Global Endurance Fund (Article 8 of the SFDR);
- Global Fixed Income Opportunities Fund (Article 8 of the SFDR);
- Global High Yield Bond Fund (Article 8 of the SFDR);
- Global Insight Fund (Article 8 of the SFDR);
- Global Opportunity Fund (Article 8 of the SFDR);
- Global Permanence Fund (Article 8 of the SFDR);
- Global Quality Fund (Article 8 of the SFDR);
- Global Sustain Fund (Article 8 of the SFDR);
- Japanese Equity Fund (Article 8 of the SFDR);
- NextGen Emerging Markets Fund (Article 8 of the SFDR);
- Short Maturity Euro Bond Fund (Article 8 of the SFDR);
- Short Maturity Euro Corporate Bond Fund (Article 8 of the SFDR);
- Sustainable Asia Equity Fund (Article 8 of the SFDR);
- Sustainable Emerging Markets Equity Fund (Article 8 of the SFDR);
- Sustainable Euro Corporate Bond Fund (Article 8 of the SFDR);
- Sustainable Euro Strategic Bond Fund (Article 8 of the SFDR);
- Sustainable European High Yield Bond Fund (Article 8 of the SFDR);
- Sustainable Global Fixed Income Opportunities Fund (Article 8 of the SFDR);
- Sustainable Global High Yield Bond Fund (Article 8 of the SFDR);
- Sustainable US High Yield Bond Fund (Article 8 of the SFDR);
- US Advantage Fund (Article 8 of the SFDR);
- US Dollar Corporate Bond Fund (Article 8 of the SFDR);
- US Dollar High Yield Bond Fund (Article 8 of the SFDR);
- US Dollar Short Duration Bond Fund (Article 8 of the SFDR);
- US Dollar Short Duration High Yield Bond Fund (Article 8 of the SFDR);
- US Growth Fund (Article 8 of the SFDR);
- US Insight Fund (Article 8 of the SFDR);
- US Permanence Fund (Article 8 of the SFDR); and
- Vitality Fund (Article 8 of the SFDR).

EQUITY FUNDS

The various Equity Funds have the following objectives:

American Resilience Fund

The American Resilience Fund's investment objective is to seek long term capital appreciation, measured in US Dollars by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)) of companies located in the U.S. An issuer is considered to be within the US if it meets the following test: (i) its securities are traded on a recognised stock exchange in the U.S.; (ii) alone or on a consolidated basis it derives 50% or more of its annual revenues or profits from either goods produced, sales made or services performed in the U.S.; (iii) it is organised or has a principal office in the U.S.; (iv) it has at least 50% of its assets, core business operations and/or employees in the U.S., or (v) any other factor which the Investment Adviser, acting reasonably, considers denote an economic exposure to the U.S.

The Investment Adviser seeks to invest in high quality companies that can generate sustainable and high returns on operating capital by identifying companies with dominant market positions typically underpinned by powerful, hard-

to-replicate intangible assets (including brands, networks, licenses and patents) and pricing power, resulting in high gross margins. The Investment Adviser also seeks to identify capable management teams able to allocate capital effectively to grow the franchise, maintain the intangible assets and sustain and/or improve returns on operating capital.

The Fund may also invest, on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

Investment restrictions:

- The Fund will apply climate-related restrictions to exclude investments in any company that the Investment Adviser determines:
 - o has any tie to fossil fuels (such as oil, gas and coal); or
 - o whose core business activity involves energy, construction materials, utilities (excluding renewable electricity and water utilities), metals and mining.
- In addition, the Fund shall not knowingly include any company:
 - o whose core business activity involves weapons or civilian firearms; or
 - that is defined by the MSCI ESG Business Involvement Screening Research ("MSCI ESG BISR") database to have any tie to controversial weapons.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted under either the first or second bullet point above after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term sustainably high returns on operating capital including ESG factors and seeks to engage with company management teams as part of this (Subject to the Fund's investment objective the Fund retains discretion over which investments are selected. Whilst ESG considerations are an integrated and fundamental part of the investment process, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance high returns on operating capital of a company.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening that the Investment Adviser sources from third party providers, including UN Global Compact violations, as well as its own engagement with company management and research. The Investment Adviser reviews securities of issuers where it believes a significant breach of the above standards and principles has occurred and typically excludes such issuers where, after conducting our research and/or engagement, the Investment Adviser believes the breach is material to the sustainability of returns on operating capital, poses significant financial and reputational risk and the issuer has not committed to appropriate remedial action. Such exclusions are determined by the Investment Adviser in its discretion rather than by reliance on third party analysis. The analysis may be supported by third party ESG controversies analysis and business involvement metrics.

The Fund relies on its research capabilities, analytical resources, and judgment to identify and monitor high quality businesses meeting their investment criteria. The Investment Adviser believes that the number of issuers with high quality businesses meeting their criteria may be limited, and accordingly, the Fund may concentrate its holdings in a relatively small number of companies.

The Investment Adviser generally considers selling a portfolio holding when it determines that the holding no longer satisfies the investment criteria or that replacing the holding with another investment should improve the Fund's valuation and/or quality.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the American Resilience Fund's investment objective, it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Asia Opportunity Fund

The Asia Opportunity Fund's investment objective is to seek long term appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs)), of issuers Located in Asia, excluding Japan, and China A-Shares via Stock Connect.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser emphasises a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser seeks high quality established and emerging companies that the Investment Adviser believes are undervalued at the time of purchase. The Investment Adviser typically favours companies it believes have sustainable competitive advantages that can be monetised through growth. The investment process integrates analysis of sustainability with respect to disruptive change, financial strength, environmental and social externalities and governance (also referred to as ESG). The Investment Adviser generally considers selling a portfolio holding when it determines that the holding no longer satisfies its investment criteria.

The Investment Adviser employs a holistic approach to ESG within its company quality assessment by analyzing potential impacts to humanity's health, environment, liberty and productivity and corporate governance practices to ensure agency, culture and trust. The Investment Adviser views incorporating ESG-related potential risks and opportunities within the investment process as important to ensure long-term stewardship of capital. Over extended time horizons, the Investment Adviser believes that ESG risks are more likely to materialize and externalities not borne by the investee company are more likely to be priced into the value of securities. Since ESG risks could potentially impact the risk and reward profile of investment opportunities, the Investment Adviser typically engages company management in constructive discussions on a range of ESG issues the Investment Adviser deems materially important.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including **civilian firearms** (securities of issuers that generate 10% or more of revenue from the manufacture or production of civilian firearms or ammunition) and **cluster munitions and anti-personnel mines** (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented

on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Asia Opportunity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Asian Property Fund⁴

⁴ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the first paragraph will be amended, as follows:

"The Asian Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in equity securities of companies in the real estate industry Located throughout Asia and Oceania. Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment."

In addition, the third paragraph will be replaced by the following paragraphs:

"The investment process utilizes internal proprietary research to invest in public real estate companies that may offer the best relative value relative to their underlying assets and earnings. The Investment Adviser utilizes a bottom-up approach, valuing each security within the investment universe to arrive at an estimate of net asset value and forward cash flows. Real estate specific factors, broader equity factors, and ESG factors are assessed in the fundamental analysis to calculate appropriate valuation metrics.

The Investment Adviser also incorporates a top-down approach in the portfolio construction process by integrating several factors which may include forecasted fundamental inflections, macroeconomic considerations, and geopolitical and country risk assessments, to achieve diversified exposure across regions, countries and/or sectors.

The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related

The Asian Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in equity securities of companies in the real estate industry Located throughout Asia and Oceania. Companies in the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property and collective investment vehicles with exposure to property such as publicly quoted property unit trusts, closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment.

The Fund may also invest, on an ancillary basis in preference shares, debt securities convertible into common shares, China A-Shares via Stock Connect, warrants and other equity linked instruments. The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Sub-Adviser may engage with company management regarding corporate governance practices as well as what the Sub-Adviser deems to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy	Regul	lation	disclosure
----------	-------	--------	------------

strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser:
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Asian Property Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Climate Aligned Fund

The Calvert Sustainable Climate Aligned Fund's investment objective is to provide long-term capital appreciation, measured in US Dollars, primarily investing in equity securities (as listed below) of companies Located in developed markets that are involved in economic activities that address climate transition and/or are aligned to the long term de-carbonisation objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, of at least 50% less than that of the underlying market benchmark (MSCI World Index), and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million enterprise value including cash.

Although the focus is intended to be on climate-related objectives, the Fund may invest across all of the environmental objectives described below.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser utilizes an ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing the climate transition and/or identifies companies that demonstrate commitment to the goal of net zero greenhouse gas emissions by 2050 or sooner. The Investment Adviser evaluates carbon reduction commitments, carbon emission trends and progress toward meeting net zero commitments in its assessment. Once a universe of climate transition and/or aligned companies is established, the Investment Adviser then constructs the portfolio through an optimization process that tilts the portfolio towards companies with lower carbon risk, while minimizing factor risks relative to the MSCI World Index.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI World Index.

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

The Investment Adviser utilizes a quantitative and qualitative ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing climate transition and/or alignment.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the MSCI World Index for the purposes of comparing the Fund's carbon footprint.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Climate Aligned Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Climate Transition Fund

The Calvert Sustainable Climate Transition Fund's investment objective is to seek to provide an attractive level of total return, measured in US Dollars, by investing primarily in a concentrated portfolio of equity securities, including preference shares and depositary receipts (including American Depositary Receipts (ADRs), European Depositary Receipts (EDRs), and Global Depositary Receipts (GDRs)), of companies located throughout the world that are engaged in business models that help to mitigate or adapt to climate change and that qualify as sustainable investments.

The Fund may also invest on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments, in Fixed Income Securities, debt securities convertible into common shares, warrants and other equity linked instruments issued by any corporations engaged in climate change businesses, as well as in China A-Shares via Stock Connect provided such investments qualify as sustainable investments. The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect.

Companies engaged in climate change-focused businesses may be involved directly or indirectly in, among other areas, renewable power, water management, waste management, electrification, battery storage, mobility, hydrogen, biofuels, and agriculture, as well as other operations that help to facilitate a lower carbon environment.

The Fund aims to achieve its climate change mitigation and adaptation objective by investing in companies in the aforementioned areas, and whose business models either have a current, direct impact on lowering carbon emissions or which are investing significant capital in Property, Plan, and Equipment (PP&E), technology, and processes that will help facilitate lower carbon emissions in the future and which qualify as sustainable investments. Companies are further delineated between those that are providers of an end-product or service ("**Providers**") or part of the supply chain/process enabling an end-product or service ("**Enablers**"). Broad definitions of each category are articulated below.

- **Providers and Enablers:** Companies whose products and services directly help to provide climate change mitigation support or those that enable mitigation through the supply chain of Providers;
- **Transitioning Companies:** Companies that are not currently Providers or Enablers, but are in the process of introducing new business lines and/or changing old ones such that they will become Providers or Enablers in the future:
- Adapters: Companies which currently provide or in the future will provide products and services that help other companies or human stakeholders to adapt to the risks posed by climate change.

The Investment Adviser actively manages the Fund utilizing internal proprietary research to implement a fundamentally driven investment process that combines bottom-up and top-down research. The Fund will be comprised of those securities which the Investment Adviser considers offer higher forecasted total return potential relative to other opportunities in the investment universe based upon absolute valuation within the described universe.

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe, and which then seeks to differentiate companies based on the extent to which their business models help to mitigate or adapt to climate change, while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanley.com/im. The Investment Adviser will strive to engage company management around ESG issues where it is determined it is possible to generate positive impact on the sustainable investment objective, on a company's environmental or social performance factors or its corporate governance practices.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as a selection of companies across various industry groups that fit into GICS (Global Industry Classification Standard) categories including but not restricted to Industrials, Utilities, Materials, Energy, Financials, Information Technology, Consumer Discretionary and Consumer Staples and which are engaged in the climate-focused business areas described above. This selection is based on the quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment as mentioned in the paragraph above.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

In addition to the above ESG objectives, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

In relation to this Fund, the use of "Calvert" in the name refers to Calvert Research and Management ("Calvert"), whose role in relation to this Fund is limited to the provision of non-discretionary investment advice to the Investment Adviser to assist the Investment Adviser in its management of the Fund. Calvert has no discretion to make

discretionary investment decisions on behalf of the Fund, such investment discretion being vested solely in the Investment Adviser.

The Fund is actively managed, and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Climate Transition Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Developed Europe Equity Select Fund

The Calvert Sustainable Developed Europe Equity Select Fund's investment objective is to provide long-term capital appreciation, measured in Euro, primarily investing in equity securities (as listed below) of companies Located in Europe's developed markets, based on classification in the MSCI Europe Index, that are involved in economic activities that address global environmental or societal challenges such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations, or are leaders in managing financially material environmental or social risks and opportunities, such as carbon emission management and diversity, equity and inclusion. This strategy further takes into account the long-term carbon reduction objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, in the range of 50% than that of the underlying market benchmark (MSCI Europe Index) and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million revenue.

Independent from the carbon net zero profile pursued by the Fund, the Fund will seek to maintain higher levels of diversity at a portfolio level than the MSCI Europe Index as measured by weighted average number of women at the board level. The Investment Adviser obtains board gender diversity data from third party vendors, which provide data on the number of women on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the weighted average for the Fund and the MSCI Europe Index.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser seeks to identify companies it considers leaders in areas of environmental, social, and governance performance that are material to the long-term performance of the company. The Investment Adviser's investment strategy has three main elements: 1) company identification, 2) portfolio optimization, and 3) corporate engagement.

The Investment Adviser utilizes a quantitative and qualitative ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing global environmental or societal challenges, or their management of material environmental and social Sustainability Risk exposures and opportunities. Through the quantitative process, the Investment Adviser seeks to identify companies that are performing in the top 20-40% of their peer group on environmental or social factors determined to be financially material to the company, while not falling in the bottom 20-40% in any other material environmental, social, or governance issue based on the Investment Adviser's proprietary research. The ESG scores are determined by the Investment Adviser, using a combination of third party and customized ESG data as a base, having regard to ESG themes such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations. Companies identified also undergo a qualitative review.

Once a universe of ESG leaders is established, the Investment Adviser then constructs the portfolio through an optimization process that tilts the portfolio towards ESG-leading companies that have more gender diversity and lower carbon risk, while minimizing factor risks relative to the MSCI Europe Index.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI Europe Index.

Investment restrictions

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. The Fund references the MSCI Europe Index for the purpose of defining a geographical allocation of countries the Fund will invest into. In the context of the Fund's ESG integration, the Investment Adviser also uses the MSCI Europe Index for the purposes of comparing the Fund's carbon footprint and gender diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some

cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Developed Europe Equity Select Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Developed Markets Equity Select Fund

The Calvert Sustainable Developed Markets Equity Select Fund's investment objective is to provide long-term capital appreciation, measured in US Dollars, primarily investing in equity securities (as listed below) of companies Located in developed markets that are involved in economic activities that address global environmental or societal challenges, such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations, or are leaders in managing financially material environmental or social risks and opportunities, such as carbon emission management and diversity, equity and inclusion. This strategy further takes into account the long-term carbon reduction objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, in the range of 50% than that of the underlying market benchmark (MSCI World Index) and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million revenue.

Independent from the carbon net zero profile pursued by the Fund, the Fund will seek to maintain higher levels of diversity at a portfolio level than the MSCI World Index as measured by weighted average number of women at the board level. The Investment Adviser obtains board gender diversity data from third party vendors, which provide data on the number of women on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the weighted average for the Fund and the MSCI World Index.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (*e.g.*, an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser seeks to identify companies it considers leaders in areas of environmental, social, and governance performance that are material to the long-term performance of the company. The Investment Adviser's investment strategy has three main elements: 1) company identification, 2) portfolio optimization, and 3) corporate engagement.

The Investment Adviser utilizes a quantitative and qualitative ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing global environmental or societal challenges, or their management of material environmental and social Sustainability Risk exposures and opportunities. Through the quantitative process, the Investment Adviser seeks to identify companies that are performing in the top 20-40% of their peer group on environmental or social factors determined to be financially material to the company, while not falling in the bottom 20-40% in any other material environmental, social, or governance issue based on the Investment Adviser's proprietary research. The ESG scores are determined by the Investment Adviser, using a combination of third party and customized ESG data as a base, having regard to ESG themes such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations. Companies identified also undergo a qualitative review.

Once a universe of ESG leaders is established, the Investment Adviser then constructs the portfolio through an optimization process that tilts the portfolio towards ESG-leading companies that have more gender diversity and lower carbon risk, while minimizing factor risks relative to the MSCI World Index.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI World Index.

Investment restrictions

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the MSCI World Index for the purposes of comparing the Fund's carbon footprint and gender diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Developed Markets Equity Select Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Diversity, Equity and Inclusion Fund

The Calvert Sustainable Diversity, Equity and Inclusion Fund's investment objective is to provide long-term capital appreciation and advance diversity, equity and inclusion, measured in US Dollars, primarily investing in equity securities (as listed below) of companies Located globally, that demonstrate leadership or meaningful improvement in having a diverse workforce and an equal and inclusive work culture. The Investment Adviser will use an optimisation process to pick the positions that the Fund takes, based on a proprietary strategy.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs)).

The Fund will seek to maintain higher levels of workforce diversity at a portfolio level than the underlying market benchmark (MSCI World Index) as measured by data available in relation to the average number of women at the board level and average number of board members representing underrepresented ethnicities. The Investment Adviser obtains board gender and ethnic diversity data from third party vendors, which provide data on the number of woman and members representing underrepresented ethnicities on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the average of women and members representing underrepresented ethnicities on the board for the Fund and the MSCI World Index.

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser utilizes an ESG research process that seeks to identify companies that exhibit leadership in having a gender balanced workforce across different levels and/or demonstrates leadership in ethnic diversity members relative to the country's demographic in certain applicable countries. In addition, a company's leadership in other dimensions of diversity among board members, specifically age, cultural background, and skill sets, are also

considered. To identify companies that reflect an equal and inclusive work culture, the Investment Adviser seeks to invest in companies that have policies and procedures that adequately support equal opportunity in the hiring process, equal pay and fair promotion among all diversity groups.

The Investment Adviser also includes companies that are demonstrating meaningful improvement in their diversity practices, as evidenced by: 1) Increased diversity in workforce, specifically more gender balanced or more ethnically diverse; or 2) demonstrated progress after a severe controversy related to diversity and inclusion issues; or 3) resolved shareholder proposals (only companies based in US and Canada) on diversity and inclusion issues; or 4) identified as an engagement target by the Investment Adviser on diversity and inclusion issues.

Companies are excluded from the portfolio if the Investment Adviser determines they do not show improvement in diversity practices and they meet one of the following criteria: 1) lack diversity among their board members, specifically not gender balanced or not ethnically diverse; or 2) show significant risks associated with human capital practices; or 3) have severe controversies related to diversity and inclusion issues.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI World Index.

The Investment Adviser may engage company management around financially material ESG issues, including diversity and inclusion, that it deems will have a positive impact on society and the environment.

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the MSCI World Index for the purposes of comparing the Fund's gender and ethnic diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Diversity, Equity and Inclusion Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";

• accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Emerging Markets Equity Select Fund

The Calvert Sustainable Emerging Markets Equity Select Fund's investment objective is to provide long-term capital appreciation, measured in US Dollars, primarily investing in equity securities (as listed below) of companies Located in the world's emerging markets, that are involved in economic activities that address global environmental or societal challenges, such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations, or are leaders in managing financially material environmental or social risks and opportunities, such as carbon emission management and diversity, equity and inclusion. This strategy further takes into account the long-term carbon reduction objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, in the range of 50% than that of the underlying market benchmark (MSCI Emerging Markets Index) and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million revenue.

Independent from the carbon net zero profile pursued by the Fund, the Fund will seek to maintain higher levels of diversity at a portfolio level than the MSCI Emerging Markets Index as measured by weighted average number of women at the board level. The Investment Adviser obtains board gender diversity data from third party vendors, which provide data on the number of women on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the weighted average for the Fund and the MSCI Emerging Markets Index.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser seeks to identify companies it considers leaders in areas of environmental, social, and governance performance that are material to the long-term performance of the company. The Investment Adviser's investment strategy has three main elements: 1) company identification, 2) portfolio optimization, and 3) corporate engagement.

The Investment Adviser utilizes a quantitative and qualitative ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing global environmental or societal challenges, or their management of material environmental and social Sustainability Risk exposures and opportunities. Through the quantitative process, the Investment Adviser seeks to identify companies that are

performing in the top 20-40% of their peer group on environmental or social factors determined to be financially material to the company, while not falling in the bottom 20-40% in any other material environmental, social, or governance issue based on the Investment Adviser's proprietary research. The ESG scores are determined by the Investment Adviser, using a combination of third party and customized ESG data as a base, having regard to ESG themes such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations. Companies identified also undergo a qualitative review.

Once a universe of ESG leaders is established, the Investment Adviser then constructs the portfolio through an optimization process that tilts the portfolio towards ESG-leading companies that have more gender diversity and lower carbon risk, while minimizing factor risks relative to the MSCI Emerging Markets Index.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI Emerging Markets Index.

Investment restrictions

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the MSCI Emerging Markets Index for the purposes of comparing the Fund's carbon footprint and gender diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Emerging Markets Equity Select Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable Global Equity Engagement Fund

The Calvert Sustainable Global Equity Engagement Fund's investment objective is to provide high total return, consistent with reasonable risk, measured in US Dollars, by primarily investing in sustainable business models that can deliver value over time. The Fund seeks companies that are involved in economic activities that address global environmental or societal challenges, such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations, or are leaders in managing financially material environmental or social risks and opportunities, such as carbon emission management and diversity, equity and inclusion. This strategy further takes into account the long-term carbon reduction objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, in the range of 50% than that of the underlying market benchmark (MSCI All Country World Index) and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million revenue.

Independent from the carbon net zero profile pursued by the Fund, the Fund will seek to maintain higher levels of diversity at a portfolio level than the MSCI All Country World Index as measured by weighted average number of women at the board level. The Investment Adviser obtains board gender diversity data from third party vendors, which provide data on the number of women on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the weighted average for the Fund and the MSCI All Country World Index.

The Fund will seek to achieve its investment objective by investing primarily in equity securities of companies located in the world's developed markets, including depositary receipts (including American Depositary Receipts (ADRs), European Depositary Receipts (EDRs), and Global Depositary Receipts (GDRs)).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that the relevant companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (*e.g.*, an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments including those located in emerging markets, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Investment Adviser believes that an investment approach focused on sustainability, ESG integration and structured engagement is best positioned to minimize risk and capitalize on financial success, which it believes compounds over time. The Investment Adviser believes the real value in a stock lies in its ability to compound cash flow over time. For this reason, the Investment Adviser focuses on the long-term ownership of companies with sustainable businesses.

The Investment Adviser seeks to identify high or improving quality companies with secular growth characteristics, high or improving returns on invested capital, sustainable competitive advantages, durable balance sheets, and a strong capital allocation record that are trading at a discount to the Investment Adviser's estimate of the security's intrinsic value. Strong attention is also paid to the positive impact that each company is making on the environment and society.

The broad spectrum of ESG work completed by the Investment Adviser is integrated throughout the investment process employed by the Fund. The Investment Adviser takes a multifaceted approach to integrating ESG into its process and firmly believes that strong ESG characteristics are critical to the sustainability of a company's business model over time. When assessing the sustainability of a business model, the Investment Adviser places ESG related issues on the same level as inputs such as returns on invested capital or free-cash-flow generation. Each issuer is evaluated relative to an appropriate peer group based on financially material ESG factors as determined by the

Investment Adviser. The evaluation of a particular security's responsible investing characteristics generally involves both quantitative and qualitative analysis. In assessing investments, the Investment Adviser generally focuses on the ESG factors most relevant to the issuer's operations, and an issuer may be acceptable for investment based primarily on such assessment.

The Investment Adviser also seeks to engage company management on financially material ESG issues identified through its fundamental and ESG research processes. Engagement seeks to drive positive change, to improve the sustainability of each company, and to enhance long-term value creation. The engagement approach may include active proxy voting, dialogue with corporate management teams and boards, working in coalitions with other organizations, filing shareholder resolutions, and other initiatives.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the MSCI All Country World Index.

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, assessing data on ESG controversies and utilizing information sourced from third party providers. The Investment Adviser will consider controversies that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im, together with the Calvert Principle for Responsible Investment.

In relation to this Fund, the use of "Calvert" in the name refers to Calvert Research and Management ("Calvert"), whose role in relation to this Fund is limited to the provision of non-discretionary investment advice to the Investment Adviser to assist the Investment Adviser in its management of the Fund. Calvert has no discretion to make discretionary investment decisions on behalf of the Fund, such investment discretion being vested solely in the Investment Adviser.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange trades and over-the-counter options, futures and other derivatives for efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the MSCI All Country World Index for the purposes of comparing the Fund's carbon footprint and gender diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Global Equity Engagement Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;

- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Calvert Sustainable US Equity Select Fund

The Calvert Sustainable US Equity Select Fund's investment objective is to provide long-term capital appreciation, measured in US Dollars, primarily investing in equity securities (as listed below) of large-capitalisation US companies that are involved in economic activities that address global environmental or societal challenges that relate to areas such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations, or are leaders in managing financially material environmental or social risks and opportunities, such as carbon emission management and diversity, equity and inclusion. This strategy further takes into account the long-term carbon reduction objectives of the Paris Agreement.

The Fund will maintain a carbon profile that follows the net zero objectives of the Paris Agreement, as reflected in the appropriate Paris-Aligned benchmark, or in the absence of an approved Paris-Aligned benchmark it will maintain a substantially lower carbon footprint, in the range of 50% than that of the underlying market benchmark (Russell 1000 Index) and taking into account the long-term carbon reduction objectives of the Paris Agreement which may warrant a revision of the targeted range of reduction over time. Carbon footprint shall be measured as weighted average carbon intensity, defined as tonnes of CO2 per \$1 million revenue.

Independent from the carbon net zero profile pursued by the Fund, the Fund will seek to maintain higher levels of diversity at a portfolio level than the Russell 1000 Index as measured by weighted average number of women at the board level. The Investment Adviser obtains board gender diversity data from third party vendors, which provide data on the number of women on boards at the issuer level. The Investment Adviser collects this data for all names in its investment universe for which data is available and calculates the weighted average for the Fund and the Russell 1000 Index.

The Fund will seek to achieve its investment objective by investing primarily in equity securities including depositary receipts (such as American Depository Receipts (ADRs), Global Depositary Receipts (GDRs), and European Depository Receipts (EDRs).

The Fund utilizes a quantitative and qualitative ESG research process that applies the Calvert Principles for Responsible Investment to define the investment universe while making sure that such companies do not significantly harm any environmental or social objective. The Calvert Principles for Responsible Investment can be found on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The Investment Adviser may engage company management around financially material ESG issues that it deems to have a positive impact on society or the environment.

The Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/ engagement with company management (e.g., an engagement with a company on its material ESG risks or opportunities such as climate change, diversity, labor and human rights and ESG disclosure, either solely or as part of a shareholder coalition, see "General information relating to Sustainability Risks integration" for further information), or make adjustments to allocations.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities, provided such securities meet the Investment Adviser's ESG criteria.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser seeks to identify companies it considers leaders in areas of environmental, social, and governance performance that are material to the long-term performance of the company. The Investment Adviser's investment strategy has three main elements: 1) company identification, 2) portfolio optimization, and 3) corporate engagement.

The Investment Adviser utilizes a quantitative and qualitative ESG research process that seeks to differentiate companies based on the contribution of their products and services in addressing global environmental or societal challenges, or their management of material environmental and social Sustainability Risk exposures and

opportunities. Through the quantitative process, the Investment Adviser seeks to identify companies that are performing in the top 20-40% of their peer group on environmental or social factors determined to be financially material to the company, while not falling in the bottom 20-40% in any other material environmental, social, or governance issue based on the Investment Adviser's proprietary research. The ESG scores are determined by the Investment Adviser, using a combination of third party and customized ESG data as a base, having regard to ESG themes such as environmental sustainability and resource efficiency, diversity, equity and inclusion, respect for human rights, product responsibility, human capital management and accountable governance and transparent operations. Companies identified also undergo a qualitative review.

Once a universe of ESG leaders is established, the Investment Adviser then constructs the portfolio through an optimization process that tilts the portfolio towards ESG-leading companies that have more gender diversity and lower carbon risk, while minimizing factor risks relative to the Russel 1000 Index.

The aforementioned ESG criteria will result in a 20% or more reduction of the investment universe in a significantly engaging manner, and the aim of the Investment Adviser will be to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations. The investment universe for these purposes is defined as the constituents of the Russel 1000 Index.

Investment restrictions

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- involved with severe ESG controversies without material remediation and improvement;
- manufacturing or production of controversial weapons and civilian firearms;
- manufacturing or production of tobacco; and
- gambling.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the Russel 1000 Index for the purposes of comparing the Fund's carbon footprint and gender diversity at the level of the boards of the portfolio components.

The Fund references third party ESG data and its own proprietary research data during the security research process. The Investment Adviser may also utilize third party data in its portfolio construction process. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable US Equity Select Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

China A-shares Fund

The China A-shares Fund's investment objective is to provide capital appreciation over the long term measured in U.S. Dollars.

The Fund will seek to achieve its investment objective by investing primarily in China A-Shares of companies listed on the Shanghai Stock Exchange and/or the Shenzhen Stock Exchange via Stock Connect. The Fund may also invest in issuers of other countries including, but not limited to, those that are listed on the Hong Kong Exchange (including China H shares and red chips) and depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)) of issuers located in any country and referring to China companies.

For the purposes of the investment policy of the China A-shares Fund, the term "China A-Shares" means shares of companies listed on the Shanghai Stock Exchange and/or the Shenzhen Stock Exchange that are quoted in Chinese Renminbi, which may include: (i) stocks, (ii) participatory notes, (iii) preferred shares, and (iv) equity warrants.

The Fund may invest, on an ancillary basis, in equity securities that do not meet the definition of China A-Shares, which may include: (i) preference shares, (ii) debt securities convertible into common shares or preferences shares, and (iii) other equity linked securities. The Fund may also invest, to a limited extent, in ETFs, warrants on securities and other instruments.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the China A-shares Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days for the China A-shares Fund

The China A-shares Fund invests primarily through Stock Connect. Dealing Days for the China A-shares Fund therefore take into account dates on which Stock Connect is closed for trading and settlement. In this Prospectus references to Dealing Day in relation to the China A-shares Fund mean any day except: (a) any other day on which commercial banks in Luxembourg, China or Hong Kong are authorized by law or executive order to close (including Saturday and Sunday), and (b) any day on which Stock Connect is closed for trading.

China Equity Fund

The China Equity Fund's investment objective is to seek to provide an attractive risk-adjusted annualized return, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities of companies located in China, for the avoidance of doubt, including China A-shares of companies listed on the Shanghai Stock Exchange and/or the Shenzhen Stock Exchange via Stock Connect, and in issuers of other countries including, but not limited to, those that are listed on the Hong Kong Exchange (including China H shares and red chips) and depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), where the underlying companies are located in China. The Fund may also invest, as part of its primary investment universe, in equity securities of companies domiciled in or exercising the predominant part of their economic activity in China and which are listed on any exchanges or over the counter markets (subject to Appendix A – Investment Powers and Restrictions).

For the purposes of the investment policy of the China Equity Fund, the term "China A-Shares" means shares of companies listed on the Shanghai Stock Exchange and/or the Shenzhen Stock Exchange that are quoted in Chinese Renminbi, which may include: (i) stocks, (ii) participatory notes, (iii) preferred shares, and (iv) equity warrants.

The Fund may invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, which may include: (i) preference shares, (ii) debt securities convertible into common shares or preferences shares, and (iii) other equity linked securities. The Fund may also invest, to a limited extent, warrants on securities and other instruments including open-ended ETFs which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The investment process takes into account information about ESG issues when making investment decisions. The Sub-Adviser focus on engaging company management around corporate governance practices as well as what we deem to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the China Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days for the China Equity Fund

The China Equity Fund invests through Stock Connect. Dealing Days for the China Equity Fund therefore take into account dates on which Stock Connect is closed for trading and settlement. In this Prospectus references to Dealing Day in relation to the China Equity Fund mean any day except: (a) any other day on which commercial banks in Luxembourg, China or Hong Kong are authorized by law or executive order to close (including Saturday and Sunday), and (b) any day on which Stock Connect is closed for trading.

Counterpoint Global Fund

The Counterpoint Global Fund's investment objective is to seek long term appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in the equity securities, including depositary receipts (including ADRs and GDRs), of established and emerging companies Located throughout the world, with capitalisations within the range of companies include in the MSCI All Country World Index.

Investment process

The investment team has a long track record in the management of different investment strategies. The portfolio managers within the team will collaborate on an ongoing basis and will allocate and rebalance the Fund's assets across each manager's underlying and independently managed investment strategies on an opportunistic basis given current market conditions. The Fund may invest outside of these underlying and independently managed investment strategies to the extent such investments fit within the permissible investment universe of the Fund.

The investment team emphasises a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in companies it believes have strong name recognition and sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis, in debt securities convertible into common shares, preference shares, China A-Shares via Stock Connect, warrants on securities, Cash Equivalents and other equity linked securities. The Fund will not invest more than 15% of its net asset value in China A-Shares via Stock Connect. The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and openended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- coal mining (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of the MSCI All Country World Index. The Fund references the MSCI All Country World Index for the purpose of setting the capitalization range of issuers the Fund will invest in.

The Fund's performance is measured against the MSCI All Country World Index as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Counterpoint Global Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Developing Opportunity Fund

The Developing Opportunity Fund's investment objective is to seek long term appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities of issuers Located in developing markets including depositary receipts (including American Depositary Receipts (ADRs), Global

Depositary Receipts (GDRs)) and China A-Shares via Stock Connect.

Under normal market conditions, the Fund's assets will be invested primarily in equity securities of issuers Located in developing, emerging or frontier emerging market countries. A country may be considered a developing, emerging or frontier emerging market based on classification in the MSCI Emerging Markets Net Index or similar classification as a developing economy by an organization such as the International Monetary Fund, the United Nations, or the World Bank. For the purpose of this Fund, "developing markets," "developing market countries," "emerging market countries" and similar terms are used interchangeably but refer to the same underlying markets and countries.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser emphasises a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser seeks high quality companies that the Investment Adviser believes are undervalued at the time of purchase. The Investment Adviser typically favours companies it believes have sustainable competitive advantages that can be monetised through growth. The investment process integrates analysis of sustainability with respect to disruptive change, financial strength, environmental and social externalities and governance (also referred to as ESG). The Investment Adviser generally considers selling a portfolio holding when it determines that the holding no longer satisfies its investment criteria.

The Investment Adviser employs a holistic approach to ESG within its company quality assessment by analyzing potential impacts to humanity's health, environment, liberty and productivity and corporate governance practices to ensure agency, culture and trust. The Investment Adviser views incorporating ESG-related potential risks and opportunities within the investment process as important to ensure long-term stewardship of capital. Over extended time horizons, the Investment Adviser believes that ESG risks are more likely to materialize and externalities not borne by the investee company are more likely to be priced into the value of securities. Since ESG risks could potentially impact the risk and reward profile of investment opportunities, the Investment Adviser typically engages company management in constructive discussions on a range of ESG issues the Investment Adviser deems materially important.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of the MSCI Emerging Markets Net Index. The Fund references the MSCI Emerging Markets Net Index for the purpose of defining a geographical allocation of countries the Fund will invest into. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Developing Opportunity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Leaders Equity Fund

The Emerging Leaders Equity Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, through investment primarily in a concentrated portfolio of equity securities, including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs) and China A-Shares via Stock Connect, in emerging and frontier countries. In targeting its investment objective, the Investment Adviser will invest in companies with strong performance and best-in-class practices relative to their peers on one or more ESG metrics (as detailed below).

A country may be considered emerging or frontier based on classification in the MSCI Emerging Markets Net Index or similar classification by an organisation such as the International Monetary Fund, the United Nations or the World Bank, provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law.

On an ancillary basis the Fund may invest in Cash Equivalents, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments and, for the purpose of efficient portfolio management (including hedging), exchange traded and over-the-counter options, futures and other derivatives.

The Fund may invest in the equity securities (including ADRs and GDRs) of companies organised and located in countries other than an emerging or frontier market where the value of the company's securities will reflect principally conditions in an emerging or frontier country, or where the principal securities trading market for the company is in an emerging or frontier country, or where 35% of the company's revenue, sales, assets, EBITDA or profit before tax is derived from either goods produced, sales made or services performed in emerging or frontier countries. The Fund may invest in participatory notes that may be used to gain exposure to securities and markets which may not be efficiently accessed through direct investment. Exposure to participatory notes will not exceed 45% of the actually invested assets (gross assets after deducting Cash Equivalents). The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser integrates the consideration of ESG issues in its investment decision-making.

The Investment Adviser integrates Sustainability Risks into its investment decision-making process, including in the conduct of due diligence and research, valuation, asset selection, portfolio construction, and ongoing investment monitoring and portfolio management. In doing so, the Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/engagement with company management, or make adjustments to the top-down allocations to geographies, sectors, or asset classes. In implementing its integration of Sustainability Risks, the Investment Adviser may utilize a combination of information sources, including company-disclosed information, non-company disclosed information, and third-party research and data.

The Investment Adviser integrates the consideration of ESG issues in its investment decision-making. The Fund will also strive to align with the objectives of the Paris Agreement on climate change. The Fund seeks to achieve a lower carbon footprint than the MSCI Emerging Markets Net Index.

The Investment Adviser believes companies with forward-looking management teams that establish proactive strategies on these sustainability and ESG issues will be better positioned from a business and financial perspective over the long term than companies that do not consider them.

In addition to investing in companies that follow good governance practices and that exhibit strong management of sustainability factors relative to peers, the Fund may invest in companies in Emerging Markets that offer scalable and profitable solutions to address pressing sustainability issues including climate change and environmental/resource management. The Fund will seek to exclude certain companies that are highly exposed to

Sustainability Risks, applying both a best-in-class and best effort approach to its investment selection in this regard. The aim of the Investment Adviser will be to ensure that at least 90% of the Fund's portfolio is subject to the sustainability analysis described above.

ESG criteria are considered by the Investment Adviser during both the investment and research process to limit exposure to Sustainability Risks. These criteria may include, but are not limited to carbon emissions, water scarcity, waste management, biodiversity, labour management, gender diversity, health & safety, product safety, data privacy & security, executive remuneration, board independence and shareholder rights. The Investment Adviser focuses on engaging company management on what it deems to be materially important governance, environmental and/or social issues facing a company. The application of the above ESG criteria should result in a 20% or more reduction of the investible universe, as defined in the first four paragraphs above.

Investments shall not knowingly include any company involved in the manufacturing or production of:

- tobacco⁵;
- adult entertainment;
- civilian firearms;
- controversial weapons;
- coal⁶;
- oil sands⁷;
- Arctic oil⁸; and
- gambling⁹.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will $\underline{www.morgan stanley investment funds.com}$ disclosed as they implemented are on www.morganstanley.com/im. The exclusions are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusion criteria are applied to all equity investments within the Fund. The exclusion criteria will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusion criteria are subject to periodic review and any changes will be reflected in the exclusion policy document. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

The Fund will limit the use of derivatives to efficient portfolio management and for hedging purposes only.

The Fund is actively managed and uses the MSCI Emerging Markets Net Index as a comparator benchmark, which is representative of the Emerging Markets equities universe. The benchmark is used for performance comparison purposes only and does not integrate environmental or social characteristics.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

⁵ The Fund does not invest in securities of issuers that generate 5% or more of revenue from the manufacture of tobacco products or from the supply of key products necessary for the production of tobacco products, such as filters.

⁶ The Fund does not invest in securities of issuers that generate 10% or more of their revenue from the mining and extraction of thermal coal, or from coal-fired power.

⁷ The Fund does not invest in securities that derive 5% or more of their revenues from the extraction or production of oil sands.

⁸ The Fund does not invest in securities of issuers that generate 5% or more of their revenue from oil extraction or production in the Arctic region, including in the Arctic National Wildlife Refuge (ANWR).

⁹ The Fund does not invest in securities of issuers that generate 5% or more of revenue from gambling activities.

Profile of the typical investor

In light of the Emerging Leaders Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Europe Opportunity Fund

The Europe Opportunity Fund's investment objective is to seek long term capital appreciation, measured in Euro.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depository receipts (American Depository Receipts (ADRs), Global Depository Receipts (GDRs) and European Depository Receipts (EDRs)), of companies Located in Europe.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments. The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser emphasises a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser seeks high quality established and emerging companies that the Investment Adviser believes are undervalued at the time of purchase. The Investment Adviser typically favours companies it believes have sustainable competitive advantages that can be monetised through growth. The investment process integrates analysis of sustainability with respect to disruptive change, financial strength, environmental and social externalities and governance (also referred to as ESG). The Investment Adviser generally considers selling a portfolio holding when it determines that the holding no longer satisfies its investment criteria.

The Investment Adviser employs a holistic approach to ESG within its company quality assessment by analyzing potential impacts to humanity's health, environment, liberty and productivity and corporate governance practices to ensure agency, culture and trust. The Investment Adviser views incorporating ESG-related potential risks and opportunities within the investment process as important to ensure long-term stewardship of capital. Over extended time horizons, the Investment Adviser believes that ESG risks are more likely to materialize and externalities not borne by the company are more likely to be priced into the value of securities. Since ESG risks could potentially impact the risk and reward profile of investment opportunities, the Investment Adviser typically engages company management in constructive discussions on a range of ESG issues the Investment Adviser deems materially important.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Europe Opportunity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

European Property Fund¹⁰

¹⁰ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the first paragraph will be amended, as follows:

"The European Property Fund's investment objective is to seek long term capital appreciation, measured in Euro, by investing primarily in equity securities of companies in the European real estate industry Located throughout Europe. Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment."

In addition, the third paragraph will be replaced by the following paragraphs:

"The investment process utilizes internal proprietary research to invest in public real estate companies that may offer the best relative value relative to their underlying assets and earnings. The Investment Adviser utilizes a bottom-up approach, valuing each security within the investment universe to arrive at an estimate of net asset value and forward cash flows. Real estate specific factors, broader equity factors, and ESG factors are assessed in the fundamental analysis to calculate appropriate valuation metrics.

The Investment Adviser also incorporates a top-down approach in the portfolio construction process by integrating several factors which may include forecasted fundamental inflections, macroeconomic considerations, and geopolitical and country risk assessments, to achieve diversified exposure across regions, countries and/or sectors.

The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and

The European Property Fund's investment objective is to seek long term capital appreciation, measured in Euro, by investing primarily in equity securities of companies in the European real estate industry Located throughout Europe. Companies in the real estate industry may include property development companies, companies principally engaged in the development and/or ownership of income-producing property and collective investment vehicles with exposure to property such as publicly quoted property unit trusts, closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment

The Fund may also invest on an ancillary basis in preference shares, debt securities convertible into common shares, warrants and other equity linked instruments.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Investment Adviser may engage with company management regarding corporate governance practices as well as what the Investment Adviser deems to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of the benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the European Property Fund's investment objective it may be appropriate for investors who:

• seek to invest in equity securities;

fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser:
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Brands Equity Income Fund

The Global Brands Equity Income Fund's investment objective is to provide a regular income stream and long-term capital growth.

The Fund seeks to achieve its investment objective by investing primarily in high quality steady dividend yield distributing equity securities of issuers located in the world's developed markets.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in equity securities of issuers Located in emerging markets, including China A-Shares via Stock Connect, and in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities to gain exposure to issuers Located in developed and emerging markets. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund will invest in a concentrated portfolio of companies whose success the Investment Adviser believes depends on intangible assets (for example, but not limited to, brand names, copyrights, or methods of distribution) underpinning a strong business franchise.

The Fund will enter into financial derivative instruments with Morgan Stanley International Plc to enhance the level of income earned. This strategy is expected to include taking exposure via one or more swaps (the "Swaps") to options on equity indices. This is expected to increase the amount of income earned, through receipt of the premiums received for writing the underlying options. However, in a rising market, the risk of capital losses on the Swaps resulting from the exercise of the options may be mitigated by the growth in the value of the underlying assets.

Investment restrictions:

- The Fund will apply climate-related restrictions to exclude investments in any company that the Investment Adviser determines:
 - o have any tie to fossil fuels (such as oil, gas and coal); or
 - o whose core business activity involves energy, construction materials, utilities (excluding renewable electricity and water utilities), metals and mining.
- In addition, the Fund shall not knowingly include any company:
 - o whose core business activity involves weapons or civilian firearms; or
 - o that is defined by the MSCI ESG Business Involvement Screening Research ("MSCI ESG BISR") database to have any tie to controversial weapons.

The above restrictions are applied by the Investment Adviser to 100% of the equity securities in the Fund but are not able to be applied to the Swaps. The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted under either the first or second bullet point above after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

Investment process:

The Fund's investment process focuses on high quality companies with sustainably high returns on operating capital. As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term sustainably high returns on operating capital including ESG factors and seeks to engage with company management teams as part of this. Subject to the Fund's investment objective the Investment Adviser retains discretion over which investments are selected. Whilst ESG considerations are an integrated and fundamental part of the investment process, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance the high returns on operating capital of a company.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening that the Investment Adviser sources from third party providers, including UN Global Compact violations, as well as its own engagement with company management and research. The Investment Adviser reviews securities of issuers where it believes a significant breach of the above standards and principles has occurred and typically excludes such issuers where, after conducting our research and/or engagement, the Investment Adviser believes the breach is material to the sustainability of returns on operating capital, poses significant financial and reputational risk and the issuer has not committed to appropriate remedial action. Such exclusions are determined by the Investment Adviser in its discretion rather than by reliance on third party analysis. The analysis may be supported by third party ESG controversies analysis and business involvement metrics.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Brands Equity Income Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Brands Fund

The Global Brands Fund's investment objective is to seek an attractive long term rate of return, measured in US Dollars, through investment primarily in equity securities of companies in the world's developed countries. The Fund will invest in a concentrated portfolio of companies whose success the Company believes depends on intangible assets (for example, but not limited to, brand names, copyrights or methods of distribution) underpinning a strong business franchise.

The Fund may also invest, on an ancillary basis in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities to gain exposure to companies in developed and emerging markets, as well as equity securities of emerging market companies and China A-Shares via Stock Connect. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund will limit the use of derivatives to hedging purposes only.

Investment restrictions:

- The Fund will apply climate-related restrictions to exclude investments in any company that the Investment Adviser determines:
 - o have any tie to fossil fuels (such as oil, gas and coal); or
 - o whose core business activity involves energy, construction materials, utilities (excluding renewable electricity and water utilities), metals and mining.
- In addition, the Fund shall not knowingly include any company:
 - whose core business activity involves weapons or civilian firearms; or
 - that is defined by the MSCI ESG Business Involvement Screening Research ("MSCI ESG BISR") database to have any tie to controversial weapons.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted under either the first or second bullet point above after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

Investment process:

The Fund's investment process focuses on high quality companies with sustainably high returns on operating capital. As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term sustainably high returns on operating capital including ESG factors and seeks to engage with company management teams as part of this. Subject to the Fund's investment objective the Investment Adviser retains discretion over which investments are selected. Whilst ESG considerations are an integrated and fundamental part of the investment process, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance the high returns on operating capital of a company.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening that the Investment Adviser sources from third party providers, including UN Global Compact violations, as well as its own engagement with company management and research. The Investment Adviser reviews securities of issuers where it believes a significant breach of the above standards and principles has occurred and typically excludes such issuers where, after conducting our research and/or engagement, the Investment Adviser believes the breach is material to the sustainability of returns on operating capital, poses significant financial and reputational risk and the issuer has not committed to appropriate remedial action. Such exclusions are determined by the Investment Adviser in its discretion rather than by reliance on third party analysis. The analysis may be supported by third party ESG controversies analysis and business involvement metrics.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Brands Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Core Equity Fund

The Global Core Equity Fund's investment objective is to seek long-term capital appreciation, measured in USD. The Fund invests primarily in equity securities issued by mid to large capitalisation companies, including depositary receipts (such as American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)) and closed-end Real Estate Investment Trusts (REITS), issued by companies worldwide.

The investment process uses a combination of quantitative models, to assess market drivers, and fundamental research of individual stocks to identify companies with attractive valuations, above-average appreciation potential and competitive dividend yields.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the above requirements, debt securities convertible into common shares, preference shares, Cash Equivalents, warrants and other equity linked instruments.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of any benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Core Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Endurance Fund

The Global Endurance Fund's investment objective is to seek long-term capital appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of established and emerging companies located throughout the world, with capitalizations within the range of companies included in the MSCI All Country World Index.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, China A-Shares via Stock Connect, warrants and other equity linked instruments. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The investment process will emphasize a bottom up stock selection process seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser seeks to invest in companies with sustainable competitive advantages. The Investment Adviser typically favours companies with rising returns on invested capital, above-average business visibility, strong free cash flow generation and an attractive risk/reward.

The Fund will make long-term investments in companies globally that the Investment Adviser believes have the most durable long-term competitive advantages.

The Fund may also invest in more moderate growth companies, companies with lower earnings volatility and/or ones with some cyclicality in their end markets.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the

Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of the MSCI All Country World Index. The Fund references the MSCI All Country World Index for the purpose of setting the capitalization range of issuers the Fund will invest in. The Fund's performance is measured against the MSCI All Country World Index as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Endurance Fund's investment objective, it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Focus Property Fund¹¹

¹¹ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the second and fourth paragraphs will be amended, as follows:

"Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment.

[...]

The Fund may also invest on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments including but not limited to: preference shares and convertible preferred securities, debt securities convertible into common shares, rights and warrants to purchase equity securities, shares of investment companies, limited partnership interests, and other equity linked instruments."

In addition, the seventh paragraph will be replaced by the following paragraphs:

"The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies

The Global Focus Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in a focused, concentrated portfolio of equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of companies in the real estate industry or closely related to the real estate industry Located throughout the world that the Investment Adviser considers offer higher forecasted total return potential relative to other opportunities within the above mentioned asset classes.

Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property, companies that operate, construct, finance or sell real estate, companies with substantial real estate related holdings and/or services related to the real estate industry, including, but not limited to, real estate management, brokers, and building products, and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment.

The Fund will invest primarily in companies located in the developed countries of North America, Europe and Asia, but may also invest in companies located in emerging markets.

The Fund may invest on an ancillary basis in in equity securities not meeting the criteria of the Fund's primary investments, preference shares and convertible preferred securities, debt securities convertible into common shares, rights and warrants to purchase equity securities, shares of investment companies, China A-Shares via Stock Connect, limited partnership interests, and other equity linked instruments. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law. In accordance with Appendix A "Investment Powers and Restrictions" non-US Dollar currency exposure may be

to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser;
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

partially or fully hedged back to US Dollars. With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Investment Adviser actively manages the Fund using a high-conviction, disciplined, bottom-up, fundamentally driven investment methodology. The Fund will be comprised of those securities which the Investment Adviser considers offer higher forecasted total return potential based upon valuation relative to other opportunities in the investment universe. The Investment Adviser will assess real estate specific factors, broader equity factors, as well as ESG factors in their fundamental analysis in order to calculate appropriate valuation metrics. Top-down considerations are incorporated into the portfolio construction process and seek to achieve exposure across regions, countries and/or sectors and integrate forecasted fundamental inflections, macroeconomic considerations, geopolitical and country risk assessments, among other factors. The Investment Adviser actively selects positions in a limited number of equity securities.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Investment Adviser and Sub-Advisers may engage with company management regarding corporate governance practices as well as what the Investment Adviser and Sub-Advisers deem to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Focus Property Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Infrastructure Fund¹²

_

¹² As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the third paragraph will be replaced by the following paragraphs:

"The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their infrastructure operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

The Global Infrastructure Fund's investment objective is to seek to provide long term capital appreciation, measured in US Dollars, by investing primarily in equity securities issued by companies, for the avoidance of doubt including closed-end Real Estate Investment Trusts (REITS), Located throughout the world that are engaged in the infrastructure business. Companies in the infrastructure business may be involved in, among other areas, the transmission and distribution of electric energy; the storage, transportation and distribution of natural resources, such as natural gas, used to produce energy; the building, operation and maintenance of highways, toll roads, tunnels, bridges and parking lots; the building, operation and maintenance of airports and ports, railroads and mass transit systems; telecommunications; water treatment and distribution; waste operations; renewable power generation; and other emerging infrastructure sectors.

The Fund may also invest on an ancillary basis in preference shares, debt securities convertible into common shares, warrants and other equity linked instruments issued by any corporations engaged in infrastructure related business.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Sub-Advisers may engage with company management regarding corporate governance practices as well as what the Sub-Advisers deem to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Infrastructure Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
 - owning, manufacturing or production of tobacco;
 - owning, manufacturing or production of controversial weapons and civilian firearms;
 - owning or operating gambling activity; and
 - owning or operating adult entertainment.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser:
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Additionally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

• accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Insight Fund

The Global Insight Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in securities issued by companies on a global basis. Under normal market conditions, the Fund's investment objective will be pursued by investing primarily in equity securities of established and emerging companies. The investment process will emphasize a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in unique companies it believes have and sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also be invested, on an ancillary basis, in equities of companies not meeting the above requirements, debt securities convertible into common shares, China A-Shares via Stock Connect, depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), preference shares, warrants on securities, Cash Equivalents and other equity linked securities. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including **civilian firearms** (securities of issuers that generate 10% or more of revenue from the manufacture or production of civilian firearms or ammunition) and **cluster munitions and anti-personnel mines** (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Insight Fund's investment objective, it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;

- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Opportunity Fund

The Global Opportunity Fund's investment objective is to seek long term appreciation, measured in US Dollars. The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of issuers Located in any country.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis, in debt securities convertible into common shares, preference shares, China A-Shares via Stock Connect, warrants on securities, Cash Equivalents and other equity linked securities. The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser emphasises a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser seeks high quality established and emerging companies that the Investment Adviser believes are undervalued at the time of purchase. The Investment Adviser typically favours companies it believes have sustainable competitive advantages that can be monetised through growth. The investment process integrates analysis of sustainability with respect to disruptive change, financial strength, environmental and social externalities and governance (also referred to as ESG). The Investment Adviser generally considers selling a portfolio holding when it determines that the holding no longer satisfies its investment criteria.

The Investment Adviser employs a holistic approach to ESG within its company quality assessment by analyzing potential impacts to humanity's health, environment, liberty and productivity and corporate governance practices to ensure agency, culture and trust. The Investment Adviser views incorporating ESG-related potential risks and opportunities within the investment process as important to ensure long-term stewardship of capital. Over extended time horizons, the Investment Adviser believes that ESG risks are more likely to materialize and externalities not borne by the investee company are more likely to be priced into the value of securities. Since ESG risks could potentially impact the risk and reward profile of investment opportunities, the Investment Adviser typically engages company management in constructive discussions on a range of ESG issues the Investment Adviser deems materially important.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including **civilian firearms** (securities of issuers that generate 10% or more of revenue from the manufacture or production of civilian firearms or ammunition) and **cluster munitions and anti-personnel mines** (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take

into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Opportunity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Permanence Fund

The Global Permanence Fund's investment objective is to seek long-term capital appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of established companies located throughout the world, with capitalizations within the range of companies included in the MSCI All Country World Index.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, China A-Shares via Stock Connect, warrants and other equity linked instruments. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The investment process will emphasize a bottom up stock selection process seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in companies it believes have strong name recognition and sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward.

The Fund will make long-term investments in companies globally that the Investment Adviser believes have the most durable long-term competitive advantages. The Fund may also invest in more moderate growth companies, companies with lower earnings volatility and/or ones with some cyclicality in their end markets.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including **civilian firearms** (securities of issuers that generate 10% or more of revenue from the manufacture or production of civilian firearms or ammunition) and **cluster munitions and anti-personnel mines** (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented

on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of the MSCI All Country World Index. The Fund references the MSCI All Country World Index for the purpose of setting the capitalization range of issuers the Fund will invest in. The Fund's performance is measured against the MSCI All Country World Index as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Permanence Funds' investment objective, it may be appropriate for investors who;

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Property Fund¹³

¹³ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the first paragraph will be amended, as follows:

"The Global Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, primarily through investment in the equity securities of companies in the real estate industry Located throughout the world. Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment. The Fund may also invest on an ancillary basis in preference shares, debt securities convertible into common shares, warrants and other equity linked instruments."

In addition, the second paragraph will be replaced by the following paragraphs:

"The investment process utilizes internal proprietary research to invest in public real estate companies that may offer the best relative value relative to their underlying assets and earnings. The Investment Adviser utilizes a bottom-up approach, valuing each security within the investment universe to arrive at an estimate of net asset value and forward cash flows. Real estate specific factors, broader equity factors, and ESG factors are assessed in the fundamental analysis to calculate appropriate valuation metrics.

The Investment Adviser also incorporates a top-down approach in the portfolio construction process by integrating several factors which may include forecasted fundamental inflections, macroeconomic considerations, and geopolitical and country risk assessments, to achieve diversified exposure across regions, countries and/or sectors.

The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with

The Global Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, primarily through investment in the equity securities of companies in the real estate industry Located throughout the world. Companies in the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property and collective investment vehicles with exposure to property such as publicly quoted property unit trusts, closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment. The Fund may also invest on an ancillary basis in preference shares, debt securities convertible into common shares, warrants and other equity linked instruments.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Investment Adviser may engage with company management regarding corporate governance practices as well as what the Investment Adviser deems to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser;
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Property Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Quality Fund

The Global Quality Fund's investment objective is to seek an attractive long term rate of return, measured in US Dollars, through investment principally in equity securities of companies in the world's developed countries. The strategy will invest in a concentrated portfolio of high quality businesses with strong managements that can sustain high cross-cycle returns on capital. The Investment Adviser believes that long term returns can best be achieved through compounding and downside protection and that high quality companies can generate superior returns for their owners over the long term. The search for high quality companies will focus on identifying companies with dominant market positions, typically underpinned by hard to replicate intangible assets. Characteristics include resilient revenue streams, pricing power, typically low capital intensity and the opportunity for organic growth. The Investment Adviser aims to buy these stocks at attractive valuations relative to their cashflow based fundamental values.

The Fund may also invest, on an ancillary basis, in equity securities of companies Located in emerging markets, including China A-Shares via Stock Connect, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities. The Fund will limit the use of derivatives to hedging purposes only. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

Investment restrictions:

- The Fund will apply climate-related restrictions to exclude investments in any company that the Investment Adviser determines:
 - o have any tie to fossil fuels (such as oil, gas and coal); or
 - whose core business activity involves energy, construction materials, utilities (excluding renewable electricity and water utilities), metals and mining.
- In addition, the Fund shall not knowingly include any company:
 - o whose core business activity involves weapons or civilian firearms; or
 - o that is defined by the MSCI ESG Business Involvement Screening Research ("MSCI ESG BISR") database to have any tie to controversial weapons.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanley.com/im. Investments that are held by the Fund but become restricted under either the first or second bullet point above after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

Investment process:

The Fund's investment process focuses on high quality companies with sustainably high returns on operating capital. As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term sustainably high returns on operating capital including ESG factors and seeks to engage with company management teams as part of this. Subject to the Fund's investment objective the Investment Adviser retains discretion over which investments are selected. Whilst ESG considerations are an integrated and fundamental part of the investment process, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance the high returns on operating capital of a company.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening that the Investment Adviser sources from third party providers, including UN Global Compact violations, as well as its own engagement with company management and research. The Investment Adviser reviews securities of issuers where it believes a significant breach of the above standards and principles has occurred and typically excludes such issuers where, after conducting our research and/or engagement, the Investment Adviser

believes the breach is material to the sustainability of returns on operating capital, poses significant financial and reputational risk and the issuer has not committed to appropriate remedial action. Such exclusions are determined by the Investment Adviser in its discretion rather than by reliance on third party analysis. The analysis may be supported by third party ESG controversies analysis and business involvement metrics.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Quality Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Sustain Fund

The Global Sustain Fund's investment objective is to seek an attractive long-term rate of return, measured in US Dollars. The Investment Adviser will also apply ESG criteria that seek to achieve a greenhouse gas ("GHG") emissions intensity for the Fund that is significantly lower than that of the reference universe (which is defined, only for the purposes of comparing GHG emissions intensity, as the MSCI AC World Index). For the purpose of this Fund, the term GHG shall be as defined by the GHG protocol and include carbon dioxide (CO2), methane (CH4), nitrous oxide (N2O), hydrofluorocarbons (HFCs), perfluorocarbons (PFCs) and sulphur hexafluoride (SF6) in metric tonnes and in tonnes of CO2 equivalent.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts, of issuers Located in any jurisdiction. The Fund may, on an ancillary basis, invest in equity securities of companies Located in emerging markets, including China A-Shares via Stock Connect, as well as in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

In order to meet the low GHG emissions intensity criteria:

- the Fund will firstly exclude investments in any company that the Investment Adviser determines:
 - o have any tie to fossil fuels (such as oil, gas and coal),
 - o any company whose core business activity involves energy, construction materials, utilities (excluding renewable electricity and water utilities), metals and mining, or
 - o for which GHG emissions intensity estimates are not available and/or cannot be estimated (in the Investment Adviser's discretion); and
- the remaining issuers will then be ranked according to their GHG emissions intensity estimates, and those with the highest intensity will be excluded from the reference universe.

The GHG emissions based investment criteria should collectively result in a reduction of the reference universe by at least 20% in a significantly engaging manner and the Investment Adviser will ensure that at least 90% of the issuers in the portfolio are assessed against these criteria.

Investment restrictions

- In addition to the above GHG-related exclusions, investments shall not knowingly include any company whose core business activity involves the following.
 - a) tobacco;
 - b) alcohol:
 - c) adult entertainment;
 - d) gambling;

- e) civilian Firearms; or
- f) weapons.
- The Fund will not invest in any company that is defined by the MSCI ESG Business Involvement Screening Research ("MSCI ESG BISR") database to have any tie to controversial weapons.
- The Fund may invest no more than 10% of its net assets in China A-Shares via Stock Connect.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted under either the first or second bullet point above after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

Investment process

The Investment Adviser seeks to invest in a concentrated portfolio of high quality, dominant franchises characterised by sustainable and high returns on operating capital, powerful intangible assets including brands, networks, licences and patents, and pricing power. The Investment Adviser seeks to identify capable management teams able to allocate capital effectively to grow the franchise and sustain and/or improve the return on operating capital.

The Fund's investment process focuses on high quality companies with sustainably high returns on operating capital. As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term sustainably high returns on operating capital including ESG factors and seeks to engage with company management teams as part of this. Subject to the Fund's investment objective and the investment restrictions outlined above, the Investment Adviser retains discretion over which investments are selected. In exercising this discretion, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance the high returns on operating capital of a company.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening that the Investment Adviser sources from third party providers, including UN Global Compact violations, as well as its own engagement with company management and research. The Investment Adviser reviews securities of issuers where it believes a significant breach of the above standards and principles has occurred and typically excludes such issuers where, after conducting our research and/or engagement, the Investment Adviser believes the breach is material to the sustainability of returns on operating capital, poses significant financial and reputational risk and the issuer has not committed to appropriate remedial action. Such exclusions are determined by the Investment Adviser in its discretion rather than by reliance on third party analysis. The analysis may be supported by third party ESG controversies analysis and business involvement metrics.

As noted above, the Fund aims to achieve a GHG emissions intensity that is significantly lower than that of its reference universe by excluding issuers on the basis of their relative GHG emissions intensity estimates. The Investment Adviser uses third-party data providers to ascertain the GHG emissions intensity of specific issuers and in some cases data on specific issuers may not be available or may be estimated by an alternative third-party data provider or the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may vary and if the preferred data provider changes its methodology, the GHG emissions intensity estimates for one or more issuers may also change significantly. The Investment Adviser will also use its best efforts to exclude any company that the Investment Adviser determines have a tie to fossil fuels (e.g. reserve ownership, related revenues and power generation) but in some cases the data may not be available or fully representative of the activities of the issuer.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against the MSCI World Index as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Sustain Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Indian Equity Fund

The Indian Equity Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investment, directly or indirectly, primarily in the equity securities of companies domiciled or exercising the predominant part of their economic activity in India. The Fund, either directly or indirectly, may also invest, on an ancillary basis, in companies listed on Indian stock exchanges which may not be domiciled or exercising the predominant part of their economic activity in India, in depositary receipts (including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs) and European Depositary Receipts (EDRs)), debt securities convertible into common shares, preference shares, debentures, warrants on securities and other equity linked securities, including participatory notes. The Fund, either directly or indirectly, will invest in a concentrated portfolio of companies.

The investment process takes into account information about ESG issues when making investment decisions. The Investment Adviser focuses on engaging company management around corporate governance practices as well as what it deems to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Indian Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days for the Indian Equity Fund

Dealing days for the Indian Equity Fund take into account dates on which the Bombay Stock Exchange (BSE) and the National Stock Exchange of India (NSE) are closed for trading and settlement. In this Prospectus references to Dealing Day in relation to the Indian Equity Fund mean any day except: (a) any other day on which commercial banks in Luxembourg or India are authorized by law or executive order to close (including Saturday and Sunday), and (b) any day on which the BSE and NSE are closed for trading.

International Equity (ex US) Fund

The International Equity (ex US) Fund's investment objective is to seek an attractive long term rate of return, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing principally in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs), of companies located in the world's developed countries excluding the United States of America. The Fund will invest in a diversified portfolio of companies using in-depth bottom-up fundamental company research. The investment process relies on a disciplined, value-oriented approach to identify companies with capable management, dominant market positions, have the potential to generate stable, consistent returns. The Investment Adviser aims to buy these companies at attractive valuations relative to their cash-flow based fundamental values.

As an essential and integrated part of the investment process, the Investment Adviser assesses relevant factors material to long-term returns on operating capital including ESG factors and seeks to engage with company management teams as part of this. Subject to the Fund's investment objective, the Investment Adviser retains discretion over which investments are selected. Whilst ESG considerations are an integrated and fundamental part of the investment process, ESG factors are not the sole determinant of whether an investment can be made or a holding can remain in the Fund's portfolio, but instead the Investment Adviser considers material risks or opportunities in any of the ESG areas which could threaten or enhance the long-term sustainability or direction of a company's returns.

The Fund may also invest, on an ancillary basis, in equity securities of companies located in emerging markets, including in China A-Shares via Stock Connect; or, as a result of a corporate action, located in the United States of America, in preference shares, debt securities convertible into common shares or preference shares, warrants on securities and other equity linked securities. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund will limit the use of derivatives to currency forwards for hedging purposes only.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is compared against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the International Equity (ex US) Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Japanese Equity Fund

The Japanese Equity Fund's investment objective is to seek long term capital appreciation, measured in Yen, through investment primarily in the equity securities of companies that meet the Investment Adviser's ESG criteria while integrating ESG characteristics by maintaining higher sustainability scores than the MSCI Japan Index. The Fund will primarily invest in companies domiciled in or exercising the predominant part of their economic activity in Japan and which are listed on any exchanges or over the counter markets (subject to Appendix A – Investment Powers and Restrictions). The Fund may also invest, on an ancillary basis, in debt securities convertible into common shares, warrants and other equity linked instruments.

The Investment Adviser integrates the consideration of ESG issues in its investment decision-making.

The Investment Adviser applies proprietary assessment and scoring methodologies on the equity securities the Fund may invest in. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with companies, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on the value of a security or portfolio. These criteria may include, but are not limited to ESG themes such as climate change, human rights and diversity, health and safety, governance, and disclosure, which the Investment Adviser considers as "material ESG issues". The Investment Adviser focuses on engaging company management on health and safety, human resources, misconduct and other factors, subject to specific engagement from each company, based on material ESG issues. Companies are scored based on the expectation for improvement and how the company's ESG activities may improve corporate value.

The Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will exclude controversy cases that it views as being very severe based on ratings by relevant ESG data providers.

The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers in order to measure and evaluate the contribution of the securities to the ESG themes described above. The indicators will be measured and evaluated at on an annual basis.

Investments shall not knowingly include any company involved in the manufacturing or production of:

- tobacco¹⁴;
- gambling¹⁵;

¹⁴ The Fund does not invest in securities of issuers that generate 5% or more of their revenues from tobacco related

products or supplies such as filters.

15 The Fund does not invest in securities of issuers that generate 5% or more of their revenue from gambling activities.

- adult entertainment¹⁶;
- civilian firearms¹⁷;
- controversial weapons¹⁸; and
- thermal coal¹⁹.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The exclusions are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusion criteria are applied to all equity investments within the Fund. The exclusion criteria are subject to periodic review and any changes will be reflected in the exclusion policy document. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Japanese Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days for the Japanese Equity Fund

Dealing Days for the Japanese Equity Fund take into account dates (i) on which Japan Exchange Group markets are closed for trading and settlement or (ii) the Directors, in consultation with the Investment Adviser, have determined in their absolute discretion. In this Prospectus references to Dealing Day in relation to the Japanese Equity Fund mean any day except: (a) any other day the Directors, in consultation with the Investment Adviser, have determined in their absolute discretion, and (b) any day on which Japan Exchange Group markets and commercial banks in Luxembourg or Japan are authorized by law or executive order to close (including Saturday and Sunday) are closed for trading. The list of expected "non-Dealing Days" will be available on the Company's website (www.morganstanleyinvestmentfunds.com) and will be updated in advance, at least semi-annually. However, the list may be further updated from time to time in the presence of exceptional circumstances where the Directors believe that it is in the best interests of the Shareholders of the Fund.

Japanese Small Cap Equity Fund

_

¹⁶ The Fund does not invest in securities of issuers that generate 5% or more of their revenue from adult entertainment.

¹⁷ The Fund does not invest in securities of issuers that generate 5% or more of their revenues from producing civilian firearms.

¹⁸ The Fund does not invest in securities of issuers that manufacture whole controversial weapons systems including intended use components.

¹⁹ The Fund does not invest in securities of issuers that generate 10% or more of their revenue from thermal coal mining, and/or 10% or more of their revenue from thermal coal power generation.

The Japanese Small Cap Equity Fund's investment objective is to seek long term capital appreciation, measured in Yen, through investment primarily in equity securities of small-cap companies Located in Japan and which are listed on any exchanges or over the counter markets (subject to Appendix A – Investment Powers and Restrictions). The Fund may also invest, on an ancillary basis, in securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, warrants, and other equity linked instruments.

For the purpose of this Fund, small-cap companies are companies with capitalizations within the range of companies included in the MSCI Japan Small Cap Index. The MSCI Japan Small Cap Index is designed to measure the performance of the small cap segment of the Japanese market. With over 900 constituents the index represents approximately 14% of the free float adjusted market capitalization of the Japanese equity universe. The average market cap of a company in the MSCI Japan Small Cap Index is approximately JPY 100 billion with a median of approximately JPY 60-70 billion (these numbers can fluctuate). Please refer to www.msci.com for the latest information on this index.

The Investment Adviser employs a bottom-up investment approach. The Investment Adviser seeks to exploit market inefficiencies applicable to small cap securities which often have limited sell-side coverage while managing the associated volatility. The Fund will include those securities which the Investment Adviser considers offer higher forecasted total growth potential based upon valuation relative to other opportunities.

The Fund is actively managed and is not designed to track a benchmark, and therefore not constrained by the composition of the MSCI Japan Small Cap Index. The Fund references the MSCI Japan Small Cap Index for the purpose of setting the capitalization range of issuers the Fund will invest in. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Japanese Small Cap Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days for the Japanese Small Cap Equity Fund

Dealing Days for the Japanese Small Cap Equity Fund take into account dates on which Japan Exchange Group markets are closed for trading and settlement. In this Prospectus references to Dealing Day in relation to the Japanese Small Cap Equity Fund mean any day except any day on which Japan Exchange Group markets and commercial banks in Luxembourg or Japan are authorized by law or executive order to close (including Saturday and Sunday) are closed for trading. The list of expected "non-Dealing Days" will be available on the Company's website (www.morganstanleyinvestmentfunds.com) and will be updated in advance, at least semi-annually. However, the Directors may decide that the day before a day on which Japan Exchange Group markets and commercial banks in Luxembourg or Japan are authorized by law or executive order to close and are closed for trading would also be a "non-Dealing Day", when it is in the best interest of Shareholders.

Latin American Equity Fund

The Latin American Equity Fund's investment objective is to seek to maximise total return, measured in US Dollars, through investment primarily in the equity and equity related securities of companies incorporated or exercising the predominant part of their economic activity in Latin American countries.

Such Latin American countries include Argentina, Bolivia, Brazil, Chile, Colombia, Costa Rica, Dominican Republic, Ecuador, El Salvador, Guatemala, Honduras, Mexico, Nicaragua, Panama, Paraguay, Peru, Uruguay and Venezuela provided that the markets in these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law. As markets in other countries develop, the Fund expects to expand and further diversify the Latin American markets in which it invests. From time to time, the Fund's investments may be concentrated in a limited number of countries. Investments in securities listed on exchanges which are not Recognised Exchanges shall be treated as investments in non-listed securities (see Appendix A – Investment Powers and Restrictions) until such time as such exchanges are deemed to be Recognised Exchanges.

As part of its primary investment universe, the Fund may also invest in preference shares, warrants and other equity linked instruments, including depositary receipts (such as American Depositary Receipts (ADRs) and Global

Depositary Receipts (GDRs)), and debt securities convertible into common shares of companies incorporated or exercising the predominant part of their economic activity in Latin American countries.

Financial derivative instruments may be used for efficient portfolio management (in accordance with the investment powers and restrictions set out in Appendix A), in order to gain exposure to certain markets at a lower cost or to reduce risk. The Fund does not invest extensively or primarily in financial derivatives instruments for investment purposes.

On an ancillary basis, the Fund may invest in the securities of companies organised and located in countries other than the Latin American countries where the value of the company's securities will reflect principally conditions in a Latin American country or where the principal securities trading market is in a Latin American country, or where 50% of the company's revenue alone or on a consolidated basis is derived from either goods produced, sales made or services performed in Latin American countries.

The investment process takes into account information about ESG issues when making investment decisions. We focus on engaging company management around corporate governance practices as well as what we deem to be materially important environmental and/or social issues facing a company.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Latin American Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

NextGen Emerging Markets Fund

The NextGen Emerging Markets Fund's investment objective is to seek long term capital appreciation, measured in Euro, by investing primarily in equity securities of "NextGen" issuers, including depositary receipts (including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs)), "NextGen" issuers being defined as 1) issuers Located in emerging market countries, including frontier market countries, which are determined based on classification in the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index, or 2) issuers Located in upcoming developing markets outside the "mainstream" emerging markets, whose capital markets have traditionally been overlooked by foreign investors or are in early stages of capital market and/or economic development. These are countries that are not part of the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index, that the International Monetary Fund, the United Nations or the World Bank generally consider to be less economically mature than developed nations. The Fund may invest in the countries under 2) provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law.

The countries that fall within 1) or 2) of the definition of "NextGen" issuers above can include, among others, Albania, Algeria, Argentina, Bahrain, Bangladesh, Bolivia, Botswana, Brazil, Bulgaria, Chile, Colombia, Costa Rica, Cote d'Ivoire, Croatia, Czech Republic, Dominican Republic, Ecuador, Egypt, Estonia, Ethiopia, Georgia, Ghana, Greece, Guatemala, Guinea, Haiti, Honduras, Hungary, Iceland, India, Indonesia, Jamaica, Jordan, Kazakhstan, Kenya, Kuwait, Laos, Latvia, Lebanon, Lithuania, Malaysia, Mauritius, Mexico, Mongolia, Morocco, Mozambique, Namibia, Nigeria, Oman, Pakistan, Panama, Paraguay, Peru, Poland, Philippines, Qatar, Romania, Russia, Rwanda, Saudi Arabia, Serbia, Senegal, Slovenia, Slovakia, South Africa, Sri Lanka, Tanzania, Trinidad & Tobago, Thailand, Turkey, Turkmenistan, Ukraine, United Arab Emirates, Uganda, Uruguay, Uzbekistan, Vietnam, and Zambia. The countries within the investment universe may change from time to time.

The Fund may also invest on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants, and other equity linked instruments. The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect. Investments in securities listed on exchanges which are not Recognised Exchanges shall be treated as investments in non-listed securities (subject to Appendix A – Investment Powers and Restrictions) until such time as such exchanges are deemed to be Recognised Exchanges.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs which are eligible investments for UCITS under the 2010 Law.

The Investment Adviser integrates the consideration of ESG criteria in its investment decision-making, including in the conduct of due diligence and research, investment valuation, asset selection, portfolio construction, and ongoing investment monitoring and portfolio management. In doing so, the Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or trim a security, commence active dialogue/engagement with company management, or make adjustments to the top-down allocations to geographies, sectors, or asset classes. In implementing its integration of Sustainability Risks, the Investment Adviser may utilize a combination of information sources, including company-disclosed information, non-company disclosed information, and third-party research and data.

ESG criteria are considered by the Investment Adviser during both the investment and research process to select investments which limit exposure to Sustainability Risks. These criteria may include, but are not limited to carbon emissions, resource management, biodiversity, labour management, diversity (for example, board diversity), health & safety, product safety, data privacy & security, executive remuneration, board independence and shareholder rights.

The Investment Adviser believes companies with forward-looking management teams that establish proactive strategies on ESG issues will be better positioned from a business and financial perspective over the long term than companies that do not consider them. The Fund will invest in companies that follow good governance practices and that exhibit strong management of materially important environmental and social factors relative to peers. The Fund will seek to exclude certain companies that are highly exposed to Sustainability Risks, applying both a best-in-class and best effort approach to its investment selection in this regard. By applying a best-in-class approach, priority is given to the companies best positioned from a non-financial viewpoint within their sector of activity, without favouring or excluding one sector relative to the stock market index used as a basis for starting. By applying a best-effort approach, priority is given to the issuers demonstrating an improvement in or good prospects for their ESG practices and performance over time.

The Investment Adviser focuses on engaging company management on what it deems to be materially important governance, environmental and/or social issues facing a company. The application of the above ESG criteria will result in a 20% or more reduction of the investible universe, as defined in the first three paragraphs above. The aim of the Investment Adviser will be to ensure that at least 90% of the Fund's portfolio is subject to the analysis of ESG criteria described above.

Additionally, the Fund seeks to achieve a lower carbon footprint than the MSCI Frontier Emerging Markets Index (based on available third party data).

Investments shall not knowingly include any company whose core business activity, according to the Investment Adviser's methodology, involve the following:

- thermal coal mining;
- oil sands;
- Arctic oil and gas;
- tobacco;
- adult entertainment;
- gambling; and
- weapons related: civilian firearms, cluster munitions, weapons and landmines.

For the purposes of all the investment restrictions outlined above, a core business activity is one that accounts for more than 10% of the relevant company's revenue as classified by MSCI, except for oil sands, Arctic oil & gas, and gambling which is considered at a 5% revenue threshold and weapons related which is considered at above 0% revenue.

The Investment Adviser may, in its discretion, elect to apply additional ESG investment restrictions, including companies that do not meet the exclusionary criteria set forth above, over time that it believes are consistent with its investment objective. Such additional restrictions will be disclosed as they are implemented in the exclusion policy document, which will be available on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The exclusions are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusion criteria are applied to all equity investments

within the Fund. The exclusion criteria will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusion criteria are subject to periodic review and any changes will be reflected in the exclusion policy document. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

The Fund is actively managed and references the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index for the purpose of partially defining a geographical allocation of the countries the Fund may invest into. The Investment Adviser has full discretion over the composition of the assets of the Fund. While the Fund will mostly hold assets within the countries referenced in the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index and can invest in securities from such countries in different proportions, it may also hold assets which are not exposed to countries referenced in the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the MSCI Emerging Markets Net Index or MSCI Frontier Emerging Markets Index.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the NextGen Emerging Markets Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors"."

Parametric Emerging Markets Fund

The Parametric Emerging Markets Fund's investment objective is to seek long-term capital appreciation by investing primarily in a diversified portfolio of equity securities of companies which are Located in emerging market countries. Emerging market countries are generally countries not considered to be developed market countries, and therefore not included in the MSCI World Index. The Fund intends to invest primarily in securities of companies located in countries included in the MSCI Emerging Markets Index or the MSCI Frontier Markets Index.

The Fund's primary investments will be limited to depositary receipts (including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs) and Non-Voting Depositary Receipts (NVDRS)), and common and preferred stocks.

The Fund may also invest, on an ancillary basis, in equity securities that do not fall within the Fund's primary investment universe, Fixed Income Securities, warrants on securities, Cash Equivalents and other equity linked securities.

As part of integrating environmental, social and governance ("ESG") factors and sustainability risk into its investment process as well as responding to European legal developments and investor preferences with regard to investing in companies directly involved in the development, production, maintenance or trade of so-called "controversial weapons," the Fund has adopted a policy of excluding such investments from its portfolio. In addition, whilst not pursuing any particular environmental, social or governance objectives, the Fund will exclude investments based on criteria consistent with the UN Global Compact and the Norges Bank exclusion list. A public list of companies included on the Norges Bank exclusion list is available at the following website: https://www.nbim.no/en/responsibility/exclusion-of-companies/. In terms of which companies are excluded, the Fund will give consideration to the severity of company behaviour, pervasiveness within and/or materiality to the business, and management's responsiveness to the issues.

In order to qualify as an "Equity Fund" within the meaning of sec. 2, para. 6 of the German Investment Tax Act, at least 51% of the Fund's Net Asset Value will be invested on an ongoing basis in holdings in equity securities that are authorised for official trading on a stock exchange or included in an organised market.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for the UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange trades and over-the-counter options, futures and other derivatives for efficient portfolio management (including hedging) purposes.

The Fund is actively managed and references the MSCI Emerging Markets Index or the MSCI Frontier Markets Index for the purpose defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the MSCI Emerging Markets Index or the MSCI Frontier Markets Index, it can invest in such securities from countries in different proportions, and it can hold assets which are not exposed to countries referenced in the MSCI Emerging Markets Index or the MSCI Frontier Markets Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from that of the MSCI Emerging Markets Index or the MSCI Frontier Markets Index.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Parametric Emerging Markets Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Parametric Global Defensive Equity Fund

The Parametric Global Defensive Equity Fund's investment objective is to provide a defensive equity exposure that is expected to provide favourable risk adjusted performance relative to the MSCI All Country World Index in USD Net (base). The Fund generates returns through the receipt of option premiums as well as returns on the equity index positions held and Cash Equivalents held. In this regard, the Fund will sell equity index call options in respect of UCITS eligible equity indices which cumulatively will provide an exposure to the markets which are similar to those comprised in the MSCI All Country World Index in USD Net (base) and will sell equity index put options in respect of UCITS eligible equity indices which cumulatively will provide an exposure to the markets which are similar to those comprised in the MSCI All Country World Index in USD Net (base). Thus, the Fund's investment policy is designed to capitalise on volatility risk premium.

The Fund will invest in a base portfolio that is generally comprised of: (i) equity index positions normally comprised of regulated equity index UCITS ETFs; and (ii) Cash Equivalents, which will consist primarily of Fixed Income Securities with a maximum remaining maturity of one year or less. The Fund will generally have an equal exposure to (i) and (ii) (ie. 50% of the Fund's Net Asset Value is intended to be invested in each of (i) and (ii)), however, the base portfolio will be rebalanced periodically where the exposure deviates from the above intended exposure by more than 5% (i.e. where exposure to (i) or (ii) falls below 45% of the Fund's Net Asset Value or above 55% of the Fund's Net Asset Value).

Each such ETF or money market will not charge annual management fees in excess of 3% of its net asset value and will be domiciled in the EEA, the United Kingdom, the United States, Guernsey, Jersey or the Isle of Man.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, fixed income securities, warrants on securities, Cash Equivalents, as well as in regulated equity index ETFs which are alternative investment funds ("AIFs") but which are UCITS eligible and exchange traded equity index futures contracts.

In order to qualify as an "Equity Fund" within the meaning of sec. 2, para. 6 of the German Investment Tax Act, at least 25% of the Fund's Net Asset Value will be invested on an ongoing basis in holdings in equity securities that are authorised for official trading on a stock exchange or included in an organised market.

With a view to enhancing returns and/or as part of the investment the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Investment Adviser does not integrate sustainability risks into its investment decisions as they are not relevant given the investment strategy of the Fund. Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

The Fund will be actively managed and is not designed to track the MSCI All Country World Index in USD Net (base) except to the extent that the ETFs in which the Fund will invest will be selected so as to provide exposure to the markets similar to those comprised in the MSCI All Country World Index in USD Net (base).

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Parametric Global Defensive Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Saudi Equity Fund

The Saudi Equity Fund's investment objective is to seek long-term capital appreciation and growth, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities of issuers listed and traded on the main market of the Saudi Stock Exchange, the Tadawul, as well as issuers in the course of an initial public offering (IPO) (including rights issues, and subject to the restrictions stated below). The Fund's investments in listed shares may include investments in the listed shares of smaller companies (meaning those that have a market value of less than SAR1 billion).

Furthermore, the Fund may invest up to 20% of its net assets in one or more eligible closed-ended Real Estate Investment Trusts (REITs) that are listed on Tadawul, as well as in units of eligible closed-ended REITs being offered in the course of an IPO and which will be listed on Tadawul in due course.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the above requirements, debt securities convertible into common shares, preference shares, Cash Equivalents, warrants and other equity linked instruments.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Saudi Equity Fund's investment objective it may be appropriate for investors who:

- Seek to invest in equity securities.
- Seek capital appreciation over the long term.
- Accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Dealing Days and settlements for the Saudi Equity Fund

Dealing Days for the Saudi Equity Fund take into account both the Saudi Arabian trading calendar and Luxembourg trading calendar. In this Prospectus references to Dealing Day in relation to the Saudi Equity Fund means any Monday, Tuesday, Wednesday or Thursday in which both Luxembourg and Saudi Arabian financial markets are open for trading. The Saudi Equity Fund will not be available for dealing on the Dealing Day immediately prior to Saudi extended holidays, for example Eid holidays. In addition, there is no settlement of subscriptions and redemptions for the Saudi Equity Fund on Fridays and settlement of subscriptions and redemptions will take into account US Dollar and Saudi Riyal bank holidays. The list of expected "non-Dealing Days" and settlement holidays will be available on the Company's website (www.morganstanleyinvestmentfunds.com) and will be updated in advance, at least semi-annually. However, the list may be further updated from time to time in the presence of exceptional circumstances where the Directors believe that it is in the best interests of the Shareholders of the Fund.

Sustainable Asia Equity Fund

The Sustainable Asia Equity Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, through investment primarily in the equity securities of companies domiciled in or exercising the predominant part of their economic activity in Asia, excluding Japan, thereby taking advantage of the dynamic economic growth capabilities of the region. In targeting this investment objective, the Investment Adviser will consider ESG criteria (as detailed below) in order to invest in companies which perform better than their peers on one or more sustainability themes and are positioned to benefit from, or contribute to, the sustainable development of the countries in which they operate based on the Investment Adviser's ESG criteria. The application of these ESG criteria should result in a 20% or more reduction of the investible universe.

The Fund may also invest on an ancillary basis in debt securities convertible into common shares, preference shares, debentures, participatory notes, warrants, and to a limited extent the Fund may also invest in collective investment schemes including closed-end funds and the Company's Funds, ETFs and securities not widely traded.

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainable themes and ESG issues in its investment decision-making. The Fund will invest in companies from the Asia excluding Japan region markets that positively contribute to and/or address one or more sustainability themes including but not limited to climate change, natural capital, human capital, access and affordability, and innovation/infrastructure. The Fund will also strive to align with the objectives of the Paris Agreement on climate change. The Fund seeks to achieve a lower carbon footprint than the MSCI All Country Asia ex-Japan Net Index.

The Investment Adviser believes companies with forward-looking management teams that establish proactive strategies on these sustainability and ESG issues will be better positioned from a business and financial perspective over the long term than companies that do not consider them. The Fund seeks to invest in companies whose product revenues, policies, initiatives, industry leadership, and/or established targets proactively address one or more of the above sustainability themes. The Investment Adviser will support this stock selection process with research and analysis, including direct company engagements and third-party data.

In addition to investing in companies that follow good governance practices and that exhibit strong management of sustainability factors relative to peers, the Fund will invest in companies in Asia excluding Japan that offer scalable and profitable solutions to address pressing sustainability issues including climate change and environmental/resource management. The Fund will be focused on companies that are positively contributing to the sustainable development of countries in Asia excluding Japan, and will seek to exclude certain companies that are highly exposed to Sustainability Risks. Up to 10% of the Fund's portfolio may consist of assets that are not subject to the sustainability analysis described above.

The Investment Adviser will implement a top-down driven approach to identify investment opportunities across the permissible investment universe, and select companies utilizing the Investment Adviser's fundamental bottom up security selection process. ESG criteria are considered by the Investment Adviser during both the investment and research process to limit exposure to Sustainability Risks and seek out investment opportunities that are aligned with the sustainability themes identified above. These criteria may include, but are not limited to carbon emissions, water scarcity, waste management, biodiversity, labour management, gender diversity, health & safety, product safety, data privacy & security, executive remuneration, board independence and shareholder rights. The Investment Adviser focuses on engaging company management on what it deems to be materially important environmental and/or social issues facing a company as well as around corporate governance practices.

Investments shall not knowingly include any company involved in the manufacturing or production of:

- tobacco²⁰;
- adult entertainment;

²⁰ The Fund does not invest in securities of issuers that generate 5% or more of revenue from the manufacture of tobacco products or from the supply of key products necessary for the production of tobacco products, such as filters.

- civilian firearms;
- controversial weapons;
- coal²¹;
- oil sands²²;
- Arctic oil²³; and
- gambling²⁴.

The Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanley.com/im. The exclusions are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusions criteria are applied to all equity investments within the Fund. The exclusion criteria will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusion criteria are subject to periodic review and any changes will be reflected in the exclusion policy document. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

The Investment Adviser integrates Sustainability Risks into its investment decision-making process, including in the conduct of due diligence and research, valuation, asset selection, portfolio construction, and ongoing investment monitoring and portfolio management. In doing so, the Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate these risks, the Investment Adviser may sell or underweight a security, commence active dialogue/engagement with company management, or make adjustments to the top-down allocations to geographies, sectors, or asset classes. In implementing its integration of Sustainability Risks, the Investment Adviser may utilize a combination of information sources, including company-disclosed information, non-company disclosed information, and third-party research and data.

For the avoidance of doubt, equity securities includes depositary receipts (including American Depositary Receipts (ADRs), Global Depositary Receipts (GDRs), International Depository Receipts (IDRs) and European Depositary Receipts (EDRs)), and closed-end Real Estate Investment Trusts (REITS). The Fund invests in markets of the Asia, excluding Japan, region, such as China, Hong Kong, India, Malaysia, Singapore, South Korea, Taiwan, Indonesia and Thailand. Additional opportunities are also sought, whenever regulations permit, in any of the emerging markets and frontier markets in Asia including China A-Shares via Stock Connect. The Fund may invest up to 20% of its net assets in China A-Shares via Stock Connect.

The Fund uses MSCI All Country Asia ex-Japan Net Index as a comparator benchmark, which is representative of the Asia ex Japan equities universe. The benchmark is used for performance comparison purposes only and does not integrate environmental or social characteristics.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

²¹ The Fund does not invest in securities of issuers that generate 10% or more of their revenue from the mining and extraction of thermal coal, or from coal-fired power.

²² The Fund does not invest in securities that derive 5% or more of their revenues from the extraction or production of oil sands.

²³ The Fund does not invest in securities of issuers that generate 5% or more of their revenue from oil extraction or production in the Arctic region, including in the Arctic National Wildlife Refuge (ANWR).

²⁴ The Fund does not invest in securities of issuers that generate 5% or more of revenue from gambling activities.

In light of the Sustainable Asia Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable Emerging Markets Equity Fund

The Sustainable Emerging Markets Equity Fund's investment objective is to seek to maximise total return, measured in US Dollars, through investment primarily in emerging market country equity securities. The Fund will principally invest in those Emerging Market (as defined below) countries in which the Company believes that economies are developing strongly and in which the markets are becoming more sophisticated. "Emerging Market" countries, for the purposes of this Fund, are as defined by the MSCI Emerging Markets Net Index, provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law. To achieve its principal investment in Emerging Market countries, the Fund may invest in the securities of companies organized under the laws of, located in or which have a principal office in, Emerging Market countries and may also invest in the securities of companies organized under the laws of, located in or which have a principal office in Developed Market countries (for the purposes of this Fund, as defined by MSCI World Index) where the company's principal securities trading market is in an Emerging Market country, or where, at the time of purchase, 50% of the company's revenue alone or on a consolidated basis is derived from goods produced, sales made or services performed in one or more Emerging Market countries, or where the company has at least 50% of its assets, core business operations and/or employees in Emerging Market countries, or in other equity linked instruments located in Developed Markets but providing exposure to Emerging Markets (for example depositary receipts) ("EM Exposed Securities").

The Fund may invest on an ancillary basis in eligible Frontier Markets equity (meaning those countries that are neither Developed Markets nor Emerging Markets as such terms are defined above for this Fund) or in equity securities not meeting the criteria of the Fund's primary investments. The Fund may also invest on an ancillary basis in units/shares of other collective investment schemes, including the Company's Funds as well as in closed-end funds, China A-Shares via Stock Connect, Fixed Income Securities, debt securities convertible into common shares, preference shares, participatory notes and warrants, in each case to gain exposure to Emerging Markets or Frontier Markets. The Fund may also invest up to 20% of its net assets in China A-Shares via Stock Connect. For temporary defensive purposes, during periods in which the Company believes changes in economic, financial or political conditions make it advisable, the Fund may reduce its holdings in Emerging Market country equity securities to below 50% of the Fund's assets and invest in equity securities in Developed Market countries (whether EM Exposed Securities or not) or in Fixed Income Securities (whether providing exposure to Emerging Markets or Developed Markets).

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of ESG criteria and sustainable themes in its investment decision-making.

In targeting its investment objective, the Investment Adviser will consider ESG criteria (as detailed below) in order to invest in companies which (i) perform better than their peers on one or more sustainability themes (as detailed below); and (ii) are positioned to benefit from, and/or contribute to, the sustainable development of the countries in which they operate based on the Investment Adviser's ESG criteria. The application of these ESG criteria should result in a 20% or more reduction of the investible universe as defined in the first two paragraphs above. The aim of the Investment Adviser will be to ensure that at least 90% of the Fund's portfolio is subject to the sustainability analysis described above.

ESG criteria may include, but are not limited to, carbon emissions, resource management, waste management, biodiversity, labour management, gender diversity, health & safety, product safety, data privacy & security, executive remuneration, board independence and shareholders rights. Sustainability themes may include, but are not limited to, climate change, natural capital, human capital, access and affordability, and innovation/infrastructure. The Fund seeks to achieve a lower carbon footprint than the MSCI Emerging Markets Net Index in aggregate at the portfolio level. With an increased investment focus on companies that positively contribute to and/or address one or more sustainability themes including climate change, the Fund will strive to align with the objectives of the Paris Agreement.

In terms of investing in companies that perform better than their peers on one or more sustainability themes, the Investment Adviser believes companies with forward-looking management teams that establish proactive strategies on the sustainability and ESG issues as described in this investment policy will be better positioned from a business and financial perspective over the long term than companies that do not consider them. Therefore, the Fund seeks to invest in companies whose product revenues, policies, initiatives, industry leadership, and/or established targets proactively address one or more of the above sustainability themes. The Investment Adviser will support this stock selection process with research and analysis, including direct company engagements and third-party data.

In addition to investing in companies that follow good governance practices and that exhibit strong management of sustainability factors relative to peers, the Fund will invest in companies from Emerging Market countries that offer scalable and profitable solutions to address pressing sustainability issues including climate change and environmental/resource management. In this connection, the Fund will be focused on companies that are positively contributing to the sustainable development of Emerging Market countries, applying both a best-in-class and best effort approach to its investment selection in this regard. By applying a best-in-class approach, priority is given to the companies best positioned from a non-financial viewpoint within their sector of activity, without favouring or excluding one sector relative to the stock market index used as a basis for starting. By applying a best-effort approach, priority is given to the issuers demonstrating an improvement in or good prospects for their ESG practices and performance over time. The Fund will also seek to exclude companies that are highly exposed to Sustainability Risks.

The Investment Adviser will implement a top-down driven approach to identify investment opportunities across the permissible investment universe, and select companies utilizing the Investment Adviser's fundamental bottom up security selection process. ESG criteria are considered by the Investment Adviser during both the investment and research process to limit exposure to Sustainability Risks and seek out investment opportunities that are aligned with the sustainability themes identified above. The Investment Adviser focuses on engaging company management on what it deems to be materially important environmental and/or social issues facing a company as well as around corporate governance practices.

Investments shall not knowingly include any company involved in the manufacturing or production of:

- tobacco²⁵;
- adult entertainment;
- civilian firearms;
- controversial weapons;
- coal²⁶:
- oil sands²⁷;
- Arctic oil²⁸; and
- gambling²⁹.

The Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented in the exclusion policy document, which is available on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. The current exclusions (and any additional restrictions applied over time) are determined by the Investment Adviser's own proprietary analysis rather than reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusions are applied to all equity investments within the Fund. The exclusions will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusions are subject to periodic review and any changes will be reflected in the exclusion policy document. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund references third party ESG data during the security research process, but does not rely on third party ESG data for the purposes of constructing the portfolio. The Investment Adviser relies on its own proprietary analysis for security selection and portfolio construction rather than third party analysis. However, in some cases data on specific issuers or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using reasonable estimates or third-party data.

The Investment Adviser integrates Sustainability Risks into its investment decision-making process, including in the conduct of due diligence and research, valuation, asset selection, portfolio construction, and ongoing investment monitoring and portfolio management. In doing so, the Investment Adviser gives due consideration to the relevance and potential materiality of Sustainability Risks for a particular investment opportunity or for the portfolio as a whole

The Fund does not invest in securities of issuers that generate 5% or more of rever

²⁵ The Fund does not invest in securities of issuers that generate 5% or more of revenue from the manufacture of tobacco products or from the supply of key products necessary for the production of tobacco products, such as filters.

²⁶ The Fund does not invest in securities of issuers that generate 10% or more of their revenue from the mining and extraction of thermal coal, or from coal-fired power.

²⁷ The Fund does not invest in securities that derive 5% or more of their revenues from the extraction or production of oil sands.

²⁸ The Fund does not invest in securities of issuers that generate 5% or more of their revenue from oil extraction or production in the Arctic region, including in the Arctic National Wildlife Refuge (ANWR).

²⁹ The Fund does not invest in securities of issuers that generate 5% or more of revenue from gambling activities.

in the context of the investment objective and intended time horizon for holding a particular security. Sustainability Risks may negatively impact the value of a security or portfolio. In order to mitigate the Sustainability Risks, the Investment Adviser may sell or underweight a security that is subject to such risks, commence active dialogue/engagement with company management, or make adjustments to the top-down allocations to geographies, sectors, or asset classes. In implementing its integration of Sustainability Risks, the Investment Adviser may utilize a combination of information sources, including company-disclosed information, non-company disclosed information, and third-party research and data.

The Fund is actively managed and references the MSCI Emerging Markets Net Index for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the MSCI Emerging Markets Net Index, it can invest in such securities from countries in different proportions to their allocations under the MSCI Emerging Markets Net Index, and it can hold assets which are not exposed to countries referenced in the MSCI Emerging Markets Net Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the MSCI Emerging Markets Net Index. The MSCI Emerging Markets Net Index is used for performance comparison purposes only and does not integrate environmental or social characteristics.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable Emerging Markets Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Advantage Fund

The US Advantage Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in securities issued by US companies and on an ancillary basis in securities issued by companies that are not from the US. An issuer may be considered to be from a particular country (including the US) or geographic region if (i) its principal securities trading market is in that country or geographic region; (ii) alone or on a consolidated basis it derives 50% or more of its annual revenue from goods produced, sales made or services performed in that country or geographic region; or (iii) it is organized under the laws of, or has a principal office in, that country or geographic region. By applying these tests, it is possible that a particular issuer could be deemed to be from more than one country or geographic region. Under normal market conditions, the Fund's investment objective will be pursued by investing primarily in equity securities of established large-capitalization companies. The investment process will emphasize a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in companies it believes have strong name recognition and sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also be invested, on an ancillary basis, in equities of companies not meeting the above requirements, debt securities convertible into common shares, depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), preference shares, warrants on securities, China A-Shares via Stock Connect, Cash Equivalents and other equity linked securities. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening

durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("GICS") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Advantage Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Core Equity Fund

The US Core Equity Fund's investment objective is to seek long-term capital appreciation, measured in US Dollars. The Fund invests primarily in equity securities issued by mid to large capitalization companies, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)) and closed-end Real Estate Investment Trusts (REITS), issued by companies Located in the United States of America.

The investment process uses a combination of quantitative models, to assess market drivers, and fundamental research of individual stocks to identify companies with attractive valuations, above-average appreciation potential and competitive dividend yields.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the above requirements, debt securities convertible into common shares, preference shares, Cash Equivalents, warrants and other equity linked instruments.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. The management of the Fund is not constrained by the composition of any benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the US Core Equity Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Focus Property Fund³⁰

³⁰ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

Consequently, the second and third paragraphs will be amended, as follows:

"Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment.

The Fund may also invest on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments including but not limited to: preference shares and convertible preferred securities, debt securities convertible into common shares, rights and warrants to purchase equity securities, shares of investment companies, limited partnership interests, and other equity linked instruments. "

In addition, the seventh paragraph will be replaced by the following paragraphs:

"The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

The US Focus Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in a focused, concentrated portfolio of equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of companies in the real estate industry or closely related to the real estate industry Located in the United States that the Investment Adviser considers offer higher forecasted total return potential relative to other opportunities within the above mentioned asset classes.

Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property, companies that operate, construct, finance or sell real estate, companies with substantial real estate related holdings and/or services related to the real estate industry, including, but not limited to, real estate management, brokers, and building products, and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment.

The Fund may also invest on an ancillary basis in equity securities not meeting the criteria of the Fund's primary investments, preference shares and convertible preferred securities, debt securities convertible into common shares, rights and warrants to purchase equity securities, shares of investment companies, limited partnership interests, and other equity linked instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Investment Adviser actively manages the Fund using a high-conviction, disciplined, bottom-up, fundamentally driven investment methodology. The Fund will be comprised of those securities which the Investment Adviser considers offer higher forecasted total return potential based upon valuation relative to other opportunities in the investment universe. The Investment Adviser will assess real estate specific factors, broader equity factors, as well as ESG factors in their fundamental analysis in order to calculate appropriate valuation metrics. Top-down considerations are incorporated into the portfolio construction process and seek to achieve exposure across sectors and integrate forecasted fundamental inflections and macroeconomic considerations, among other factors. The Investment Adviser actively selects positions in a limited number of equity securities.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Investment Adviser and Sub-Advisers may engage with company management regarding corporate governance practices as well as what the Investment Adviser and Sub-Advisers deem to be materially important environmental and/or social issues facing a company.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser:
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the US Focus Property Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Growth Fund

The US Growth Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in securities issued by US companies and on an ancillary basis in securities issued by companies that are not from the US. An issuer may be considered to be from a particular country (including the US) or geographic region if (i) its principal securities trading market is in that country or geographic region; (ii) alone or on a consolidated basis it derives 50% or more of its annual revenue from goods produced, sales made or services performed in that country or geographic region; or (iii) it is organized under the laws of, or has a principal office in, that country or geographic region. By applying these tests, it is possible that a particular issuer could be deemed to be from more than one country or geographic region. Under normal market conditions, the Fund's investment objective will be pursued by investing primarily in equity securities of high quality growth oriented companies. The investment process will emphasize a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in unique companies it believes have sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward. Individual security selection will be emphasized.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis, in equities of companies not meeting the above requirements, debt securities convertible into common shares, depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), preference shares, warrants on securities, China A-Shares via Stock Connect, Cash Equivalents and other equity linked securities. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes

are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Growth Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Insight Fund

The US Insight Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in securities issued by US companies and on an ancillary basis in securities issued by companies that are not from the US. An issuer may be considered to be from a particular country (including the US) or geographic region if (i) its principal securities trading market is in that country or geographic region; (ii) alone or on a consolidated basis it derives 50% or more of its annual revenue from goods produced, sales made or services performed in that country or geographic region; or (iii) it is organized under the laws of, or has a principal office in, that country or geographic region. By applying these tests, it is possible that a particular issuer could be deemed to be from more than one country or geographic region.

Under normal market conditions, the Fund's investment objective will be pursued by investing primarily in equity securities of established and emerging companies. The investment process will emphasize a bottom-up stock selection process, seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in unique companies it believes have sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward. Individual security selection will be emphasized.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis, in equities of companies not meeting the above requirements, debt securities convertible into common shares, depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), preference shares, warrants on securities, China A-Shares via Stock Connect, Cash Equivalents and other equity linked securities. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

• **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);

- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including civilian firearms (securities of issuers that generate 10% or more of revenue from the
 manufacture or production of civilian firearms or ammunition) and cluster munitions and anti-personnel
 mines (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel
 landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Insight Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Permanence Fund

The US Permanence Fund's investment objective is to seek long-term capital appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of established companies located in the US, with capitalizations within the range of companies included in the S&P 500 Index.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, China A-Shares via Stock Connect, warrants and other equity linked instruments. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The investment process will emphasize a bottom up stock selection process seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in companies it believes have strong name recognition and sustainable competitive advantages with above average business visibility, the ability to deploy capital at high rates of return, strong balance sheets and an attractive risk/reward.

The Fund will make long-term investments in companies that the Investment Adviser believes have qualities of Permanence including durable long-term competitive advantages. The Fund may also invest in more moderate growth companies, companies with lower earnings volatility and/or companies with some cyclicality in their end markets.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco** (securities of issuers classified as Global Industry Classification Standard ("**GICS**") sub-industry tobacco, or generate 5% or more of revenue from the manufacture of tobacco products);
- **coal mining** (securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate 10% or more of revenue from the mining and extraction of thermal coal); or
- weapons, including **civilian firearms** (securities of issuers that generate 10% or more of revenue from the manufacture or production of civilian firearms or ammunition) and **cluster munitions and anti-personnel mines** (securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines).

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of the S&P 500 Index. The Fund references the S&P 500 Index for the purpose of setting the capitalization range of issuers the Fund will invest in. The Fund's performance is measured against the S&P 500 Index as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Permanence Funds' investment objective, it may be appropriate for investors who;

- seek to invest in equity securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Property Fund³¹

_

Consequently, the first paragraph will be amended, as follows:

"The US Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in equity securities of companies in the US real estate industry Located in the US. Companies in the real estate industry or closely related to the real estate industry may include companies principally engaged in the development and/or ownership of income-producing property; companies that operate, construct, finance or sell real estate; companies with substantial real estate related

³¹ As from 12 December 2022, the investment policy of the Fund will be changed in order to promote, among other characteristics, environmental or social characteristics, or a combination of those characteristics, provided that the companies in which the investments are made follow good governance. The Fund will then fall within the scope of Article 8 of the SFDR.

holdings and/or services or products related to the real estate industry, including, but not limited to, real estate management, brokers, building products and property technology; and collective investment vehicles with exposure to property, such as publicly quoted property unit trusts, all types of eligible closed-end Real Estate Investment Trusts (REITS) and undertakings for collective investment."

In addition, the second paragraph will be replaced by the following paragraphs:

"The investment process utilizes internal proprietary research to invest in public real estate companies that may offer the best relative value relative to their underlying assets and earnings. The Investment Adviser utilizes a bottom-up approach, valuing each security within the investment universe to arrive at an estimate of net asset value and forward cash flows. Real estate specific factors, broader equity factors, and ESG factors are assessed in the fundamental analysis to calculate appropriate valuation metrics.

The Investment Adviser also incorporates a top-down approach in the portfolio construction process by integrating several factors which may include forecasted fundamental inflections, macroeconomic considerations, and geopolitical and country risk assessments, to achieve diversified exposure across countries and/or sectors.

The Investment Adviser actively integrates sustainability into the investment process by assessing key ESG risks and opportunities in the bottom-up stock selection process primarily by leveraging third-party ESG providers to assess and quantify ESG performance for issuers, supplementing third-party research with proprietary research conducted by the Investment Adviser including utilizing a proprietary framework for assessing and quantifying risks and opportunities related to ESG which results in a quantitative adjustment to valuation estimates, and through engagements with company management to discuss ESG-related strengths, weaknesses, and opportunities in an effort to effect positive change within the industry. Key ESG topics may include, but are not limited to: energy usage and renewables, water usage, emissions, diversity and gender equality, labour and human rights, employee and tenant health, wellness and safety and company ESG governance and disclosure. In an effort to drive positive change and encourage companies to improve their performance on material ESG issues, the Investment Adviser may approach company management with competitive insights, financially sound business cases and practical solutions to potentially improve their real estate operations. While ESG considerations are an integrated and fundamental part of the investment process, they are only one of several key determinants used by the Investment Adviser to determine if an investment will be made or size adjusted in the overall portfolio.

Investments shall not knowingly include any company whose primary business activity in any of the following is more than 10% of company revenue:

- owning or operating real estate used for prisons;
- owning or operating real estate used to manufacture cannabis;
- owning, manufacturing or production of tobacco;
- owning, manufacturing or production of coal mining;
- owning, manufacturing or production of controversial weapons and civilian firearms; and
- owning, manufacturing or production of arctic oil and gas.

Investments shall not knowingly include the following companies:

- companies that have a notable controversy related to their operations and/or products, where the severity of the social or environmental impact of the controversy, is judged by the Investment Adviser;
- companies that fail to comply with the UN Global Compact or the ILO fundamental principles, without material remediation and improvement.

The Investment Adviser references third party ESG data and its own proprietary research during the security research process. The Investment Adviser will review controversy cases (such as the exclusions noted above) that it views as being very severe using ratings by relevant ESG data providers and internal research. However, in some cases data on specific issuers or the exclusions noted above may not be readily available and/or may be estimated by the Investment Adviser using reasonable estimates."

Finally, the Taxonomy Regulation disclosure will be amended as follows:

"In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within

The US Property Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, by investing primarily in equity securities of companies in the US real estate industry Located in the US. The Fund intends to invest in the publicly traded securities of closed-end Real Estate Investment Trusts (REITS) and similar Real Estate Operating Companies (REOCS). REITS and REOCS are companies that acquire and/or develop real property for long term investment purposes. They invest the majority of their assets directly in real property and derive their income primarily from rents.

The investment process may consider information about ESG issues in its bottom-up stock selection process when making investment decisions. The Sub-Adviser may engage with company management regarding corporate governance practices as well as what the Sub-Adviser deems to be materially important environmental and/or social issues facing a company.

The Fund may also invest, on an ancillary basis, in preference shares, debt securities convertible into common shares, warrants and other equity-linked instruments.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the US Property Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Value Fund

The US Value Fund's investment objective is to seek long term capital growth, by investing in a broadly diversified selection of transferable equity securities, emphasizing common stocks of leading companies the Investment Adviser believes have been systematically mispriced by the market. Value stocks are common stocks that the Investment Adviser believes are inexpensive relative its estimates of the intrinsic value of the business.

The Fund will invest primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of companies Located in the US.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, Fixed Income Securities, warrants on securities, Cash Equivalents and other equity linked securities.

The Investment Adviser of the Fund employs a bottom-up, research-driven and value-oriented approach that seeks to identify pricing anomalies that occur due to fundamental factors that are perceived to be temporary and not permanent. In selecting securities, the Fund focuses mainly on U.S. issuers with attractive valuations and sound business fundamentals. The Investment Adviser also considers how constructing the portfolio and purchasing or selling an investment impacts the overall portfolio's risk profile (for example, the portfolio's beta relative to its benchmark and each economic sector, its standard deviation of return, its active share and tracking error, as well as how its tracking error is composed on an industry and stock level).

In addition to the above, in making investment decisions, the Investment Adviser will integrate ESG criteria. The Investment Adviser utilizes the extensive proprietary ESG research generated by its affiliate, Calvert, in evaluating investments. The Investment Adviser may consider financially material ESG factors as part of the Fund's securities selection process. Such financially material factors comprise Sustainability Risks or opportunities likely to affect the financial condition or performance of the issuer over a long term investment horizon and may include, without

the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%."

limitation, business ethics, energy use, energy efficiency and renewable energy, workplace diversity, climate change, data security and workplace safety.

The Investment Adviser believes that corporate governance practices of issuers have a direct impact on business performance and results produced by those issuers and incorporating companies' ESG behaviours into investment decisions provides a more comprehensive, holistic approach to investing that the Investment Adviser believes can enhance both stock selection and risk-adjusted returns over the long term.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the US Value Fund's investment objective it may be appropriate for investors who:

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Vitality Fund

The Vitality Fund's investment objective is to seek long-term capital appreciation, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in equity securities, including depositary receipts (including American Depositary Receipts (ADRs) and Global Depositary Receipts (GDRs)), of companies Located in the US that are principally engaged in the discovery, development, production, or distribution of products or services related to advances in healthcare. These companies include, but are not limited to, biotechnology, pharmaceuticals, medical equipment and supplies, healthcare technology, healthcare providers and services, and life sciences tools and services.

In relation to this Fund, Vitality means the power to endure and thrive. This reflects the healthcare sector-oriented focus of the Fund and the view of the Investment Adviser that the sector will offer superior opportunity for long-term capital appreciation.

The Fund may also allocate to industries related to the healthcare industry, but are not primarily engaged in the scientific advancement of healthcare. The allocation among these areas will vary depending on the relative potential the Investment Adviser sees within each area and the outlook for the overall healthcare sector.

The Fund may also invest, on an ancillary basis, in equity securities not meeting the criteria of the Fund's primary investments, debt securities convertible into common shares, preference shares, warrants and other equity linked instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The investment process will emphasize a bottom up stock selection process seeking attractive investments on an individual company basis. In selecting securities for investment, the Investment Adviser typically invests in unique companies it believes have sustainable competitive advantages, strong research and development and productive new product flow, financial strength, and an attractive risk/reward profile. The Fund generally seeks investments in companies that are developing new and effective medicines, enabling novel and more efficient research and development efforts, as well as companies whose business models reduce costs or improve quality in healthcare systems.

The Fund will make long-term investments in companies that the Investment Adviser believes have the most durable long-term competitive advantages. The Fund may also invest in more moderate growth companies, companies with lower earnings volatility and/or companies with some cyclicality in their end markets. The Investment Adviser

generally considers selling a portfolio holding when it determines that the holding no longer satisfies its investment criteria.

The Investment Adviser actively integrates sustainability into the investment process by using ESG factors as a lens for additional fundamental research, which can contribute to investment decision-making. The Investment Adviser seeks to understand how environmental and social initiatives within companies can create value by strengthening durable competitive advantages, creating growth opportunities, driving profitability, and/or aligning with secular growth trends. The Investment Adviser generally engages with company management teams to discuss their ESG practices, with the aim of identifying how sustainability themes present opportunities and risks that can be material to the value of the security over the long-term. Other aspects of the investment process include a proprietary, systematic evaluation of governance policies, specifically focusing on compensation alignment on long-term value creation.

Investments shall not knowingly include any company whose primary business activity involves the following:

- **tobacco**; securities of issuers classified as GICS sub-industry tobacco, or generate >5% or more of revenue from the manufacture of tobacco products;
- **coal mining**; securities of issuers classified as GICS sub-industry coal & consumable fuels, or generate >10% or more of revenue from the mining and extraction of thermal coal; and
- weapons, including **civilian firearms**; securities of issuers that generate >10% or more of revenue from the manufacture or production of civilian firearms or ammunition; and **cluster munitions and anti-personnel mines**; securities of issuers that manufacture, produce or distribute cluster munitions or anti-personnel landmines.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im). Further to the above, the Investment Adviser may, in its discretion, elect to apply additional investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im. Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund is actively managed, not designed to track a benchmark, and therefore not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Vitality Fund's investment objective, it may be appropriate for investors who;

- seek to invest in equity securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

BOND FUNDS

The various Bond Funds have the following objectives:

Calvert Sustainable Global Green Bond Fund

The Calvert Sustainable Global Green Bond Fund's investment objective is to provide an attractive level of total return, measured in US Dollars, through investment in global Fixed Income Securities that meet the Investment Adviser's ESG and Impact criteria (as further described below) and qualify as sustainable investments under the SFDR, with the objective of supporting positive environmental and social impacts and outcomes. The Fund will seek to achieve the investment policy through allocation across fixed income asset classes as well as through market and instrument selection.

The Fund will invest primarily in Green Bonds (as defined below) including the global securities of corporate, government and government related issuers across a spectrum of fixed income asset classes including, investment-grade bonds, high yield bonds, mortgage-backed securities, asset-backed securities, convertibles and currencies, and

subject to applicable law, in other asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised. High yield bonds are considered to be Fixed Income Securities issued by corporations that are rated lower than "BBB-" by S&P or "Baa3" by Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. These investments may include fixed income securities issued in emerging markets.

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainability themes and ESG issues in its investment decision-making on a discretionary basis as detailed below. In addition, the Investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals. The Investment Adviser may engage company management around corporate governance practices and what it deems to be materially important environmental and/or social issues facing a company. It will also seek to support positive environmental and social impacts and outcomes, having regard to sustainability themes such as Decarbonisation & Climate Risk, Circular Economy & Waste Reduction, Diverse & Inclusive Business and Decent Work & Resilient Jobs.

For the purpose of this Fund, 'Green Bonds' include but are not limited to the following instruments:

- Green Use of Proceeds Bonds, with proceeds targeted to environmentally beneficial projects (with or without external Green Bond labelling);
- Sustainability Bonds, with a proportion of the proceeds targeted to environmentally beneficial projects;
- Transition Bonds, with proceeds targeted to transitioning to more environmentally favourable business models;
- Sustainability-linked Bonds, with environmental key performance indicators and targets; and
- Bonds of issuers that seek to provide environmental solutions or that demonstrate environmental sustainability leadership.

The Investment Adviser employs a proprietary assessment framework for Green Bonds (which has regard to, where appropriate, recognised green bond guidelines such as the Green Bonds Principles of the International Capital Market Association), through which the robustness, expected impact and transparency of all such instruments in the Fund are evaluated. The Investment Adviser uses third-party data and ESG research as part of its analysis. Where data may not be available, the Investment Adviser will use internal methodologies or reasonable estimates. The methodologies used by different data providers can vary and may result in different evaluations. Details of the evaluation methodology and Calvert Principles for Responsible Investment is available on the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

On an ancillary basis the Fund may invest in securities which are not identified by the Investment Adviser as Green Bonds, provided the Investment Adviser has assessed that the issuer and/or security makes a significant contribution to positive social impacts or outcomes either through the issuer's products and services or through its practices or through the specific projects or investments financed by the bond issue, and provided the Investment Adviser has established that the activities of the issuer or projects associated with the bond issue, as applicable, do not result in adverse environmental or social impacts, or cause significant harm in accordance with the SFDR sustainable investment requirements. Such investments may include labelled Social Bonds, with proceeds allocated to projects focussed on positive social outcomes and / or target populations, or labelled Sustainability-linked Bonds with social key performance indicators and targets. Such labelled bonds will also be subject to the Investment Adviser's proprietary assessment framework, as described above. The labels "Green", "Social", "Sustainability" and "Sustainability-Linked" attached to the bonds by their issuers stand for the fact that their frameworks are, in most cases, although not exclusively, aligned with the International Capital Market Association (ICMA) Green/Social/Sustainability-Linked Bond Principles, or Sustainability Bond Guidelines, or principles/guidelines established in other jurisdictions.

The aim of the Investment Adviser will be to ensure that at least 90% of the securities in the portfolio are assessed with respect to the sustainability themes and ESG issues described above.

In relation to this Fund, the use of "Calvert" in the name refers to Calvert Research and Management ("Calvert"), whose role in relation to this Fund is limited to the provision of non-discretionary investment advice to the Investment Adviser to assist the Investment Adviser in its management of the Fund. Calvert has no discretion to make discretionary investment decisions on behalf of the Fund, such investment discretion being vested solely in the Investment Adviser.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of derivative instruments such as index futures, contracts for difference (CFDs), TRS, structured notes (including participatory notes), convertible bonds, convertible preferred stocks, options, foreign exchange options, currency forwards and forward contracts and other derivatives for investment or efficient portfolio management (including hedging) purposes.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;
 - o manufacturing or production of civilian firearms;

- manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail;
- o gambling, where the company derives >10% revenue from such business activity; and
- o adult entertainment, where the company derives >10% revenue from such business activity.

The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

Investments that are held by the Fund but subsequently become restricted due to the application of the ESG criteria above, after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

It is not expected that the Fund will invest more than 50% of its net assets in aggregate in securitized instruments. It is not expected that the Fund will invest more than 20% of its net assets in any one type of securitized instrument (e.g., asset-backed securities (ABS), commercial mortgage-backed securities (CMBS), residential mortgage-backed securities (RMBS), agency mortgage-backed securities (MBS)).

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Calvert Sustainable Global Green Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Markets Corporate Debt Fund

The Emerging Markets Corporate Debt Fund's investment objective is to seek to maximize total return, measured in US Dollars, primarily through investments across the credit spectrum of Fixed Income Securities of corporate issuers, together with investing in Fixed Income Securities of government and government related issuers located in each case in Emerging Markets countries. The Fund intends to invest its assets in Emerging Market countries' Fixed Income Securities that provide a high level of current income, while at the same time holding the potential for capital appreciation.

"Emerging Market" countries, for the purposes of this Fund, are as defined by the JP Morgan Corporate Emerging Markets Bond Index – Broad Diversified, provided that the markets of these countries are considered to be recognized exchanges ("Recognized Exchanges") within the meaning of Article 41(1) of the 2010 Law. As markets in other countries develop, the Fund expects to expand and further diversify the emerging markets in which it primarily invests as such markets are added to this index. To achieve its principal investment in Emerging Market countries, the Fund may invest in Fixed Income Securities of companies organised and located in Emerging Market countries and may also invest in Fixed Income Securities of companies organised and located in Developed Market countries (for the purpose of this Fund, as defined by the JP Morgan Government Bond Index) where the value of the company's securities will reflect principally conditions in an Emerging Market country or where the principal securities trading market for such Emerging Market country, or where 50% of the company's revenue alone or on a consolidated basis is derived from either goods produced, sales made or services performed in Emerging Market countries ("EM Exposed Securities"). Fixed Income Securities held by the Fund will take the form of bonds, notes, bills, debentures, convertible securities, bank debt obligations, short-term paper, mortgage and, subject to applicable

law, other asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may also invest on an ancillary basis in the aforementioned Fixed Income Securities where such securities are issued by issuers organised and located (1) in neither Developed Market countries nor Emerging Market countries; or (2) in Developed Market countries but the securities are not EM Exposed Securities.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 20% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes. The Fund may, on an ancillary basis, invest in other emerging markets Fixed Income Securities and in emerging markets Fixed Income Securities denominated in currencies other than US Dollars.

The Fund may also invest, to a limited extent, in warrants on transferable securities issued by issuers in emerging markets.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund is actively managed and references the JP Morgan Corporate Emerging Markets Bond Index – Broad Diversified for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the JP Morgan Corporate Emerging Markets Bond Index – Broad Diversified, it can invest in such securities from countries in different proportions, and it can hold assets which are not exposed to countries referenced in the JP Morgan Corporate Emerging Markets Bond Index – Broad Diversified. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the JP Morgan Corporate Emerging Markets Bond Index – Broad Diversified.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Corporate Debt Fund's investment objective, it may be appropriate for investors who:

- seek to invest in Fixed Income Securities in emerging markets;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Markets Debt Fund

The Emerging Markets Debt Fund's investment objective is to seek to maximise total return, measured in US Dollars, primarily through investment in the Fixed Income Securities of government and government-related issuers located in Emerging Market (as defined below) countries (including, to the extent these instruments are securitised, participations in loans between governments and financial institutions), together with investing in the Fixed Income Securities of corporate issuers located in, organised under the laws of, or which have a principal office in, Emerging Market countries. The Fund intends to invest its assets in Emerging Market country Fixed Income Securities that provide a high level of current income, while at the same time holding the potential for capital appreciation.

"Emerging Market" countries, for the purposes of this Fund, are as defined by the JP Morgan Emerging Market Bond Index Global Diversified provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law. As markets in other countries develop, the Fund expects to expand and further diversify the emerging markets in which it primarily invests as or before such markets are added to the JP Morgan Emerging Market Bond Index Global Diversified. Fixed Income Securities held by the Fund will take the form of bonds, notes, bills, debentures, convertible securities, bank debt obligations, short-term paper, mortgage and, subject to applicable law, other asset-backed securities, loan participations and loan assignments (to the extent that these instruments are securitised) and interests issued by entities organised and operated for the purpose of restructuring the investment characteristics of instruments issued by Emerging Market issuers.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on sovereign issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates

an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as carbon emissions, climate vulnerability, forestry conservation, life expectancy & health, education, living standards, voice & accountability, political stability, effective government, regulatory quality, rule of law, corruption, violence/terrorism. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes good governance and social practices among sovereigns. Consequently, the Investment Adviser will not make any new investments in countries where there is evidence of significant social violations. However, any existing investments in such countries shall not necessarily need to be sold, and any countries exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The Investment Adviser shall disclose the methodology used to evaluate significant social violations on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

To achieve its principal investment in Emerging Market countries, the Fund may invest in the Fixed Income Securities of companies organised under the laws of, located in or which have a principal office in Emerging Market countries and may also invest in the Fixed Income Securities of companies organised under the laws of, located in or which have a principal office in Developed Market countries (for the purpose of this Fund, as defined by JP Morgan Government Bond Index) where the company's principal Fixed Income Securities trading market is in an Emerging Market country, or where 50% of the company's revenue alone or on a consolidated basis is derived from goods produced, sales made or services performed in Emerging Market countries ("EM Exposed Securities").

For investments in such companies specifically, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com are www.morganstanleyinvestmentfunds.com and www.morganstanleyinvestmentfunds.com are www.morganstanleyin

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis in the aforementioned classes of Fixed Income Securities where such Fixed Income Securities are issued by issuers organised under the laws of and located (1) neither in Developed Market countries nor Emerging Market countries; or (2) in Developed Market countries, but where the Fixed Income Securities are not EM Exposed Securities.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 20% of the Fund's assets will be invested in such securities.

The Fund may also invest, to a limited extent, in warrants issued by Emerging Market issuers and in open-ended ETFs provided that any such ETFs are eligible investments for UCITS funds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund is actively managed and references the JP Morgan Emerging Market Bond Index Global Diversified for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the JP Morgan Emerging Market Bond Index Global Diversified, it can invest in such securities from countries in different proportions to their allocations under the JP Morgan Emerging Market Bond Index Global Diversified, and it can hold assets which are not exposed to countries referenced in the JP Morgan Emerging Market Bond Index Global Diversified. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the JP Morgan Emerging Market Bond Index Global Diversified is used for performance comparison purposes only and does not integrate environmental or social characteristics.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Debt Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities in Emerging Markets;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Markets Debt Opportunities Fund

The Emerging Markets Debt Opportunities Fund's investment objective is to generate total return, which is defined as income plus capital appreciation, by establishing long and short investment exposures to Emerging Markets (as defined below) debt pool through currencies, local interest rates, and sovereign and corporate credit, while investing primarily in (i) Fixed Income Securities of corporate, government and government-related issuers Located in Emerging Market countries; and/or (ii) derivative instruments denominated in or based on the currencies, interest rates, or issuers of Emerging Market countries. Fixed Income Securities may include zero coupon bonds, preferred stocks, deferred interest bonds and notes and bonds and notes on which the interest is payable in the form of additional eligible stocks, bonds or notes of the same kind (Payment-in-Kind (PIK) securities).

"Emerging Market" country, for the purposes of this Fund, is any country determined by the Investment Adviser to have an emerging market economy, considering factors such as the country's political and economic stability, and the development of its financial and capital markets. Emerging market countries include so- called frontier market countries. Frontier markets are a sub-category of emerging markets and are generally considered by the Investment Adviser to be countries which are not developed countries and (i) are not included in the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index; or (ii) represent 2% or less of the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index, provided that in each case, the markets of these countries are considered to be recognised exchanges within the meaning of Article 41(1) of the 2010 Law. The Fund does not invest in developed countries except for purposes of hedging, gaining exposure to Fixed Income

Securities denominated in currencies of emerging market countries and investments (including through derivative instruments) in U.S treasuries and U.S. government and agency securities and agency mortgage-backed securities.

The Fund may have significant investment in a geographic region or country and, subject to the foregoing in this paragraph, there is no limit on the Fund's exposure to any geographic region or country. It is not intended that the Fund will have a particular sector or industry focus.

Fixed Income Securities held by the Fund will take the form of sovereign bonds and debentures, mortgage-backed securities and asset-backed securities, convertible debt securities (which may embed derivatives), contingent convertible instruments, municipal obligations, corporate bonds and debentures, fixed income and floating rate debt securities, loan participations and loan assignments, to the extent that these instruments are securitised, and participation notes.

The Fund may invest in instruments of any credit rating, including those rated below investment grade (rated below BBB by either Standard & Poor's Ratings Services or Fitch Ratings, or below Baa by Moody's Investors Service, Inc.) or in unrated instruments considered to be of comparable quality by the Investment Adviser.

In evaluating the Fixed Income Securities issued by emerging market corporates and in its engagement with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, currency and duration / local rates, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as carbon emissions, climate vulnerability, forestry conservation, life expectancy & health, education, living standards, voice & accountability, political stability, effective government, regulatory quality, rule of law, corruption, and violence/terrorism. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers, in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least annually.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes good governance and social practices among corporate issuers. Consequently, the Investment Adviser will not make any new investments in issuers where there is evidence that the relevant issuer has engaged in significant social violations (being activities which infringe significantly on an individual or a group's rights). However, any existing investments in such issuers shall not necessarily need to be sold, and any issuers exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The methodology used by the Investment Adviser to evaluate significant social violations shall be disclosed on the following websites: www.morganstanley.com/im.

With respect to investment in Fixed Income Securities issued by corporates, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above are that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices of issuers on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

The Investment Adviser also promotes good governance and social practices among sovereigns. Consequently, the Investment Adviser will not make any new investments in countries where there is evidence of significant social violations (being significant failures by the government to ensure that an individual or a group's social rights are protected). However, any existing investments in such countries shall not necessarily need to be sold, and any

countries exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The Investment Adviser shall disclose the methodology used to evaluate significant social violations on its website.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with the investment objective of the Fund. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may take exposure to issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the Fund such as hedging instruments, unscreened investments for diversification purposes, or investments for which data is lacking.

The Investment Adviser believes that as sustainability issues are sources of long-term risk and return, the integration of sustainability risks as outlined above may assist in enhancing the returns of the Fund over the long term. Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

In managing the Fund, the Investment Adviser performs macroeconomic and political research and analysis on individual countries. It considers such factors as a country's political system and environment, fiscal policy, monetary policy, incomes policy (i.e. any government policy that affects corporate or individual earnings in the particular country subject to the policy (such as tax policy, subsidies in particular economic sectors, minimum wage laws, or industry regulations)) and trade policy, among other factors. Based on this research and analysis, the Investment Adviser seeks to identify countries and currencies it believes have potential to outperform investments in other countries and currencies, and to anticipate changes in global economies, markets, political conditions and other factors for this purpose. The Investment Adviser selects and adjusts investments in an effort to take advantage of differences in the perceived values of countries' currencies, interest rates and credit spreads. To implement investment decisions, the Investment Adviser selects an asset class and specific instrument that most optimally and efficiently expresses the Investment Adviser's view of that particular country. The Investment Adviser considers the relative risk/return characteristics of prospective investments in determining the most efficient means for achieving desired exposures.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities not meeting the criteria of the Fund's primary investments, equity securities, warrants on securities, Cash Equivalents and other equity linked securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may invest no more than 20% of its assets in debt securities that are not paying current income in anticipation of the receipt of possible future income or capital appreciation which are unrated or categorised as the lowest rated obligations (rated C by Moody's or D by S&P).

The Fund may invest no more than 10% of its assets in participation notes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest no more than 20% of its assets in aggregate asset-backed securities (ABS) and/or mortgage-backed securities (MBS).

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and references the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index for the purposes of defining a geographical allocation of the countries the Fund will invest into and monitoring error relative to the benchmark. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index, it can invest in such securities from countries in different proportions to their allocations under the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index, and it can hold assets which are not exposed to countries referenced in the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index. Hence, there are no

restrictions on the extent to which the Fund's performance may deviate from the J.P. Morgan Emerging Markets Bond (JEMB) Hard Currency/Local Currency 50-50 Index.

The Fund's performance is measured against a benchmark as detailed in the Fund's key information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Debt Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Markets Domestic Debt Fund

The Emerging Markets Domestic Debt Fund's investment objective is to seek to maximise total return, measured in US Dollars, primarily through investment in a portfolio of Emerging Market bonds and other Emerging Market (as defined below) Fixed Income Securities, denominated in the local currency of issue. The Fund intends to invest its assets in Emerging Market Fixed Income Securities that provide a high level of current income, while at the same time holding the potential for capital appreciation.

"Emerging Market" countries, for the purposes of this Fund, are as defined by the JP Morgan Government Bond Index – Emerging Markets Global Diversified, provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law. As emerging markets in other countries develop, the Fund expects to expand and further diversify the emerging markets in which it primarily invests as or before such markets are added to the JP Morgan Government Bond Index – Emerging Markets Global Diversified.

To achieve its principal investment in Emerging Market countries, the Fund may invest in Fixed Income Securities of government and government-related issuers located in Emerging Markets countries (including, to the extent these instruments are securitised, participations in loans between governments and financial institutions) and Fixed Income Securities of corporate issuers located in, organised under the laws of, or which have a principal office in, Emerging Market countries, denominated in the local currency of issue ("EM Exposed Securities").

The Fund may invest in Fixed Income Securities of entities organised to restructure outstanding debt of Emerging Market issuers.

Fixed Income Securities held by the Fund will take the form of bonds, notes, bills, debentures, convertible securities, bank debt obligations, short-term paper, mortgage and, subject to applicable law, other asset-backed securities, loan participations and loan assignments to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on sovereign issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as carbon emissions, climate vulnerability, forestry conservation, life expectancy & health, education, living standards, voice & accountability, political stability, effective government, regulatory quality, rule of law, corruption, violence/terrorism. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes good governance and social practices among sovereigns. Consequently, the Investment Adviser will not make any new investments in countries where there is evidence of significant social violations. However, any existing investments in such countries shall not necessarily need to be sold, and any

countries exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The Investment Adviser shall disclose the methodology used to evaluate significant social violations on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

For investments in corporate issuers specifically, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis in the aforementioned classes of Fixed Income Securities where such securities are issued by issuers organised under the laws of and located (1) neither in Developed Market (for the purposes of this Fund, as defined by the JP Morgan Government Bond Index – Emerging Markets Global Diversified) countries nor Emerging Market countries; or (2) in Developed Market countries, but where the Fixed Income Securities are not EM Exposed Securities.

In addition, the Fund may invest on an ancillary basis in Fixed Income Securities which are not denominated in the local currency of issue, provided that for temporary defensive purposes, during periods in which the Company believes changes in economic, financial or political conditions make it advisable, the Fund may reduce its holdings denominated in the local Emerging Market currency of issue to below 50% of the Fund's assets and invest in eligible Fixed Income Securities denominated in the currencies of Developed Market countries.

The Fund may also invest, to a limited extent, in warrants on transferable securities issued by issuers in Emerging Market countries and in open-ended ETFs provided that any such ETFs are eligible investments for UCITS funds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 20% of the Fund's assets will be invested in such securities.

The Fund is actively managed and references the JP Morgan Government Bond Index – Emerging Markets Global Diversified for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the JP Morgan Government Bond Index – Emerging Markets Global Diversified, it can invest in such securities from countries in different proportions to their allocations under the JP Morgan Government Bond Index – Emerging Markets Global Diversified, and it can hold assets which are not exposed to countries referenced in the JP Morgan Government Bond Index – Emerging Markets Global Diversified. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the JP Morgan Government Bond Index – Emerging Markets Global Diversified is used for performance comparison purposes only and does not integrate environmental or social characteristics.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Domestic Debt Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities in Emerging Markets;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors", specifically
 the risks relating to Emerging Markets and in particular Fixed Income Securities denominated in the currency
 of the countries of investment.

Emerging Markets Fixed Income Opportunities Fund

The Emerging Markets Fixed Income Opportunities Fund's investment objective is to seek to maximise total return, measured in US Dollars, primarily through investment in the Fixed Income Securities of issuers located in Emerging Market Countries or issues denominated in the currencies of Emerging Market Countries. The Fund intends to invest its assets in Emerging Market countries' Fixed Income Securities that provide a high level of current income, while at the same time holding the potential for capital appreciation.

"Emerging Market" countries, for the purposes of this Fund, are any countries included in JP Morgan Emerging Markets Blended Index – Equally Weighted which is an equally weighted index of JP Morgan Emerging Market Bond Index Global Diversified, JP Morgan Corporate Emerging Market Bond Index Broad Diversified and JP Morgan Government Bond Index –Emerging Markets Global Diversified Index, provided that the markets of these countries are considered to be recognised exchanges ("Recognised Exchanges") within the meaning of Article 41(1) of the 2010 Law. For the avoidance of doubt, an investment in an issue included in this index will constitute an investment within the primary investment strategy of the Fund. As markets in other countries develop, the Fund expects to expand and further diversify the emerging markets in which it primarily invests as such markets are added to this index.

To achieve its principal investment in Emerging Market countries, the Fund may also invest in the Fixed Income Securities of companies organised and located in Developed Market countries (for the purpose of this Fund, as defined by the JP Morgan Government Bond Index) where the value of the company's securities will reflect principally conditions in an Emerging Market country or where the principal securities trading market for such is in an Emerging Market country, or where 50% of the company's revenue alone or on a consolidated basis is derived from either goods produced, sales made or services performed in Emerging Market countries ("EM Exposed Securities").

The Fund may invest in Fixed Income Securities of issuers organised to restructure outstanding debt of Emerging Market issuers.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment

process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may invest on an ancillary basis in Fixed Income Securities issued by issuers organised and located (1) in neither Developed Market countries nor Emerging Market countries; or (2) in Developed Market countries but the securities are not EM Exposed Securities.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 20% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund is actively managed and references the JP Morgan Emerging Markets Blended Index – Equally Weighted for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the JP Morgan Emerging Markets Blended Index – Equally Weighted, it can invest in such securities from countries in different proportions, and it can hold assets which are not exposed to countries referenced in the JP Morgan Emerging Markets Blended Index – Equally Weighted. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the JP Morgan Emerging Markets Blended Index – Equally Weighted.

The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Fixed Income Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Emerging Markets Local Income Fund

The Emerging Markets Local Income Fund's investment objective is to generate total return, which is defined as income plus capital appreciation, by establishing a locally denominated Emerging Markets (as defined below) debt pool through currencies and interest rates, while investing primarily in: (i) Fixed Income Securities of corporate, government and government related issuers Located in Emerging Market countries or denominated in currencies of Emerging Markets countries and/or (ii) derivative instruments, denominated in or based on the currencies, interest rates, or issues of, Emerging Market countries.

For the purposes of this Fund, "Emerging Markets" countries are defined to include any country that did not become a member of the OECD prior to 1975 and Turkey. Emerging market countries include so-called frontier market countries, which generally are considered by the Investment Adviser to be less developed countries that: (i) are not included in the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified; or (ii) represent 2% or less of the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified.

The Fund may have significant investment in a geographic region or country. It is not intended that the Fund will have a particular sector or industry focus.

Fixed Income Securities held by the Fund will take the form of sovereign bonds and debentures, mortgage-backed securities and asset-backed securities, convertible debt securities, municipal obligations, corporate bonds and debentures, fixed-income and floating rate debt securities and, to a limited extent, participation notes. Fixed Income Securities may include zero coupon bonds, preferred stocks, deferred interest bonds and notes and bonds and notes on which the interest is payable in the form of additional eligible stocks, bonds or notes of the same kind ("Payment-in-Kind (PIK) securities").

The Fund may invest in instruments of any credit rating, including those rated below investment grade (rated below BBB by either Standard & Poor's Ratings Services or Fitch Ratings, or below Baa by Moody's Investors Service, Inc.) or in unrated instrument considered to be of comparable quality by the Investment Adviser.

The investment process and strategy seeks to identify local emerging markets and their currencies that will outperform on a relative basis other comparable emerging markets and their currencies. The principal risk factors in analysing emerging markets debt are credit, duration, foreign exchange, and sustainability risk. Any or all of these factors may be identified as having the potential to contribute to outperformance. Therefore, in some cases, the Investment Manager will take the view that a country's currency will appreciate and will seek to obtain exposure to

that currency through the derivative instruments referenced above or investment in the securities denominated in such currency.

In evaluating the Fixed Income Securities issued by emerging market corporates and in its engagement with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, currency and duration / local rates, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to environmental, social and governance ("ESG") themes such as carbon emissions, climate vulnerability, forestry conservation, life expectancy & health, education, living standards, voice & accountability, political stability, effective government, regulatory quality, rule of law, corruption, and violence/terrorism. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers, in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least annually.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes good governance and social practices among corporate issuers. Consequently, the Investment Adviser will not make any new investments in issuers where there is evidence that the relevant issuer has engaged in significant social violations (being activities which infringe significantly on an individual or a group's rights). However, any existing investments in such issuers shall not necessarily need to be sold, and any issuers exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The methodology used by the Investment Adviser to evaluate significant social violations shall be disclosed on the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

With respect to investment in Fixed Income Securities issued by corporates, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above are that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices of issuers on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

The Investment Adviser also promotes good governance and social practices among sovereigns. Consequently, the Investment Adviser will not make any new investments in countries where there is evidence of significant social violations (being significant failures by the government to ensure that an individual or a group's social rights are protected). However, any existing investments in such countries shall not necessarily need to be sold, and any countries exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The Investment Adviser shall disclose the methodology used to evaluate significant social violations on its website.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with the investment objective of the Fund. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available

and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may take exposure to issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the Fund such as hedging instruments, unscreened investments for diversification purposes, or investments for which data is lacking.

The Investment Adviser believes that as sustainability issues are sources of long-term risk and return, the integration of sustainability risks as outlined above may assist in enhancing the returns of the Fund over the long term. Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

In managing the Fund, the Investment Adviser performs macroeconomic and political research and analysis on individual countries. It considers such factors as a country's political system and environment, fiscal policy, monetary policy, incomes policy (i.e. any government policy that affects corporate or individual earnings in the particular country subject to the policy (such as tax policy, subsidies in particular economic sectors, minimum wage laws, or industry regulations)) and trade policy, among other factors. Based on this research and analysis, the Investment Adviser seeks to identify countries and currencies it believes have potential to outperform investments in other countries and currencies, and to anticipate changes in global economies, markets, political conditions and other factors for this purpose. The Investment Adviser selects and adjusts investments in an effort to take advantage of differences in the perceived values of countries' currencies, interest rates and credit spreads. To implement investment decisions, the Investment Adviser selects an asset class and specific instrument that most optimally and efficiently expresses the Investment Adviser's view of that particular country. The Investment Adviser considers the relative risk/return characteristics of prospective investments in determining the most efficient means for achieving desired exposures.

The Fund may invest no more than 20% of its assets in debt securities that are not paying current income in anticipation of the receipt of possible future income or capital appreciation which are unrated or categorised as the lowest rated obligations (rated C by Moody's or D by S&P).

The Fund may invest no more than 20% of its assets in aggregate in asset-backed securities (ABS) and/or mortgage-backed securities (MBS).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities not meeting the criteria of the Fund's primary investments, equity securities, warrants on securities, Cash Equivalents and other equity linked securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and references the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified for the purposes of defining a geographical allocation of the countries the Fund will invest into and monitoring error relative to the benchmark. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified, it can invest in such securities from countries in different proportions to their allocations under the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified, and it can hold assets which are not exposed to countries referenced in the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified.

The Fund's performance is measured against a benchmark as detailed in the Fund's key information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Emerging Markets Local Income Fund's investment objective it may be appropriate for investors who:

• seek to invest in Fixed Income Securities;

- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Euro Bond Fund

The Euro Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro.

The Fund will invest primarily in Euro denominated Fixed Income Securities whether issued by corporations, governments or government guaranteed issuers. With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and <a href="https://ww

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Euro Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Euro Corporate Bond Fund

The Euro Corporate Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro.

The Fund will invest primarily in Euro denominated Fixed Income Securities, issued by corporations and other non-government related issuers ("Corporate Bonds"). With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB-in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income

Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may, on an ancillary basis, invest in:

- Non-Euro denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non-Euro currency exposure may be hedged back to the Euro.
- Fixed Income Securities that are not Corporate Bonds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take

into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Euro Corporate Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Euro Corporate Bond - Duration Hedged Fund

The Euro Corporate Bond – Duration Hedged Fund's investment objective is to provide an attractive rate of return, measured in Euro, while seeking to reduce the Fund's exposure to market interest rate movements.

Duration is a measure of the sensitivity of the price of a debt security to changes in interest rates. Duration risk is the risk that an investment's value will change due to changes in interest rates. The value of debt securities will generally increase when interest rates fall and decrease when interest rates rise. The Fund seeks to reduce its exposure to market interest rate movements by hedging the duration of the Fund.

The Fund will invest primarily in high quality issues of Euro denominated Fixed Income Securities, issued by corporations and other non-government related issuers ("Corporate Bonds").

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by

relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

Securities will be deemed to be high quality if at the time of purchase they are rated either "BBB-" or better by S&P or "Baa3" or better by Moody's or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Fund may continue to hold such securities that are downgraded after purchase but it will not hold securitised bonds downgraded below "BBB-" by S&P or "Baa3" by Moody's and/or other securities downgraded below "B-" by S&P or "B3" by Moody's, or, in each case, a similar rating by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Fund may not make additional purchases of securities that are downgraded.

The Fund may, on an ancillary basis, invest in:

- Corporate Bonds which at the time of purchase are rated either lower than "BBB-" by S&P or "Baa3" by
 Moody's or similarly by another internationally recognised rating service or determined to be of similar
 creditworthiness by the Investment Adviser.
- Non-Euro denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non-Euro currency exposure may be hedged back to the Euro.
- Fixed Income Securities that are not Corporate Bonds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest no more that 20% of its assets in mortgage-backed securities or assets backed securities.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Euro Corporate Bond – Duration Hedged Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Euro Strategic Bond Fund

The Euro Strategic Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro.

The Fund will invest primarily in issues of Euro denominated Fixed Income Securities whether issued by corporations, government or government guaranteed issuers, and subject to applicable law, in asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in non-Euro denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non-Euro currency exposure may be hedged back to the Euro.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Euro Strategic Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

European Fixed Income Opportunities Fund³²

The European Fixed Income Opportunities Fund's investment objective is to provide an attractive level of total return, measured in Euro, through the selection of Euro-denominated Fixed Income Securities. The Fund will seek to achieve the investment policy through allocation across fixed income asset classes as well as through market and instrument selection.

The Fund will invest primarily in Euro-denominated Fixed Income Securities of corporate, government and government related issuers across a spectrum of fixed income asset classes including investment-grade bonds, high-yield bonds, mortgage-backed securities, convertibles and currencies and subject to applicable law, in other asset-backed securities as well as loan participations and loan assignments to the extent that these instruments are securitised. High yield bonds are considered to be Fixed Income Securities issued by corporations that are rated lower than "BBB-" by S&P or "Baa3" by Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Investment Adviser may invest in any combination of two or more of the above asset classes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of Asset Backed Securities, including Mortgage Backed Securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB- in the case of Asset Backed Securities, including Mortgage Backed Securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse &

-

³² As from 12 December 2022, the investment policy of the Fund will be changed in order to reflect that it does no longer need to comply with certain German rules *Versicherungsaufsichtsgesetz*. Consequently, the third paragraph will be deleted.

inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities that are not denominated in Euro, emerging markets Fixed Income Securities and additional securities including but not limited to debt securities convertible into common shares, Cash Equivalents, equity or other equity linked securities.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange-traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the European Fixed Income Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

European High Yield Bond Fund

The European High Yield Bond Fund's investment objective is to provide an attractive rate of return, measured in Euro, through investments worldwide primarily in lower rated and unrated Fixed Income Securities issued by governments, agencies and corporations that offer a yield above that generally available on the Fixed Income Securities in the four highest rating categories of S&P or Moody's denominated in European currencies. These investments may include Fixed Income Securities issued by non-European issuers and such securities issued in emerging markets and, subject to applicable law, asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may, on an ancillary basis, invest in non-European currency denominated Fixed Income Securities. In order to optimize European currency exposure, the non-Euro currency exposure may be hedged back to the Euro and the Fund may, on an ancillary basis, use derivatives contracts to create synthetic European currency high yield Fixed Income Securities within the limits set forth in Appendix A – Investment Powers and Restrictions.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the European High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Floating Rate ABS Fund

The Floating Rate ABS Fund's investment objective is to generate an attractive return, whilst minimising exposure to changes in interest rates through investments in a portfolio of floating rate asset-backed securities ("ABS"), measured in Euro.

The Fund will invest primarily in floating-rate Fixed Income Securities with a duration of less than two years that are mortgage-backed securities, commercial mortgage-backed securities, collateralized mortgage obligations, covered bonds that are covered by mortgages including uniform mortgage-backed securities and other ABS rated investment-grade by an internationally recognised rating agency, securities determined to be of similar creditworthiness by the Investment Adviser, or securities backed by the United States of America. Floating-rate securities are defined as securities that are benchmarked to a floating-rate index. The Fund will maintain an average duration between 0-1 year.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund will invest in. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, lenders and servicers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Fund promotes sustainable and responsible business practices by issuers of securitisations. In order to do this, the Fund shall not knowingly purchase any securitisation where the Investment Adviser has perceived any of the following which are directly related to the issuer of a securitisation:

- predatory lending practices;
- breaches in compliance of standards of the Consumer Financial Protection Bureau (CFPB);
- severe malpractice in payment collection processes and foreclosure practices; and
- fraudulent behaviour.

The Investment Adviser uses third-party data and in some cases data on specific issuers. ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates, using best efforts and in good faith. The methodologies used by different data providers may also vary and may result in different scores.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives.

Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures, swaps, options on swaps (swaptions) and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in Fixed Income Securities not meeting the above requirements

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation.

Profile of the typical investor

In light of the Floating Rate ABS Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the short term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Frontier Markets Local Debt Fund

The Frontier Markets Local Debt Fund's investment objective is to generate total return, which is defined as income plus capital appreciation by investing primarily in (i) Fixed Income Securities of corporate, government and government-related issuers Located in Frontier Market countries or denominated in currencies of Frontier Market countries and/or (ii) derivative instruments, denominated in or based on the currencies, interest rates, or issues of, Frontier Market countries.

For the purposes of this Fund, "Frontier Market" are a sub-category of emerging markets and are generally considered by the Investment Adviser to be countries which are not developed countries and (i) are not included in the JP Morgan

Government Bond Index – Emerging Markets Global Diversified (GBI-EM) or (ii) represent 2% or less of the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM), provided that in each case, the markets of these countries are considered to be recognised exchanges within the meaning of Article 41(1) of the 2010 Law. Large Emerging Market countries that are not included in the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM) such as Korea, Taiwan, Hong Kong, Singapore, Israel and India are not considered Frontier Markets countries and the Fund will not have exposure to such countries.

The Fund may have significant investment in a geographic region or country. It is not intended that the Fund will have a particular sector or industry focus.

Fixed Income Securities held by the Fund will take the form of sovereign bonds and debentures, corporate bonds and debentures, fixed-income and floating rate debt securities and, to a limited extent, participation notes. Fixed Income Securities may also include zero coupon bonds, preferred stocks, deferred interest bonds and notes and bonds and notes on which the interest is payable in the form of additional eligible stocks, bonds or notes of the same kind ("Payment-in-Kind (PIK) securities").

The Fund may invest up to 100% of its assets in instruments which are rated at least B- by either Standard & Poor's Ratings Services or Fitch Ratings, or B3 by Moody's Investors Service, Inc. or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment.

In evaluating the Fixed Income Securities issued by frontier market corporates and in its engagement with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, currency and duration / local rates, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to environmental, social and governance ("ESG") themes such as carbon emissions, climate vulnerability, forestry conservation, life expectancy & health, education, living standards, voice & accountability, political stability, effective government, regulatory quality, rule of law, corruption, and violence/terrorism. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers, in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least annually.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated. The proprietary assessment framework is part of the Fund's screening policy available on the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

The Investment Adviser promotes good governance and social practices among corporate issuers. Consequently, the Investment Adviser will not make any new investments in issuers where there is evidence that the relevant issuer has engaged in significant social violations (being activities which infringe significantly on an individual or a group's rights). However, any existing investments in such issuers shall not necessarily need to be sold, and any issuers exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The methodology used by the Investment Adviser to evaluate significant social violations shall be disclosed on the following websites: www.morganstanley.com/im.

With respect to investment in Fixed Income Securities issued by corporates, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above are that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices of issuers on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

The Investment Adviser also promotes good governance and social practices among frontier markets sovereigns. Consequently, the Investment Adviser will not make any new investments in countries where there is evidence of significant social violations (being significant failures by the government to ensure that an individual or a group's social rights are protected). However, any existing investments in such countries shall not necessarily need to be sold, and any countries exhibiting positive momentum with respect to such violations shall not be subject to the purchase restriction. The Investment Adviser shall disclose the methodology used to evaluate significant social violations on its website.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with the investment objective of the Fund. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores." The Fund may take exposure to in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the Fund such as hedging instruments, unscreened investments for diversification purposes, or investments for which data is lacking.

The Investment Adviser believes that as sustainability issues are sources of long-term risk and return, the integration of sustainability risks as outlined above may assist in enhancing the returns of the Fund over the long term. Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

In managing the Fund, the Investment Adviser performs macroeconomic and political research and analysis on individual countries, including on a country's political system and environment, fiscal policy, monetary policy, borrowing rates, the level of national debt, incomes policy (i.e. any government policy that affects corporate or individual earnings in the particular country subject to the policy (such as tax policy, subsidies in particular economic sectors, minimum wage laws, or industry regulations)), and trade policy. Based on this research and analysis, the Investment Adviser seeks to identify countries and currencies it believes have potential to outperform investments in other countries and currencies, and to anticipate changes in global economies, markets, political conditions and other factors for this purpose. The Investment Adviser selects and adjusts investments in an effort to take advantage of differences in the perceived values of countries' currencies, interest rates and credit spreads. To implement investment decisions based on the research undertaken by the Investment Adviser, the Investment Adviser will select securities to which the Fund will obtain exposure, which it believes will provide the Fund with the most optimal and efficient exposure to the relevant country. The Investment Adviser considers the relative risk/return characteristics of prospective investments in determining the most efficient means for achieving desired exposures.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities not meeting the criteria of the Fund's primary investments, equity securities, warrants on securities, Cash Equivalents and other equity linked securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and references the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM) for the purpose of defining a geographical allocation of the countries the Fund will invest into. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will generally hold assets within the countries referenced in the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM), it can invest in such securities in different proportions to their allocations under the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM), and it can hold assets which are not exposed to countries referenced in the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM). Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the JP Morgan Government Bond Index – Emerging Markets Global Diversified (GBI-EM).

The Fund's performance is measured against a benchmark as detailed in the Fund's key information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Frontier Markets Local Debt Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Asset Backed Securities Fund

The Global Asset Backed Securities Fund's investment objective is to provide an attractive rate of total return, measured in USD.

The Fund will invest primarily in Fixed Income Securities that are mortgage-backed securities, commercial mortgage-backed securities, collateralized mortgage obligations and covered bonds that are covered by mortgages including uniform mortgage-backed securities ("Mortgage-Backed Securities"). At least 50% of the Fund's assets will be invested in securities rated investment-grade by an internationally recognised rating agency, securities determined to be of similar creditworthiness by the Investment Adviser, or securities backed by the United States of America.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund will invest in. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, lenders and servicers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Fund promotes sustainable and responsible business practices by issuers of securitisations. In order to do this, the Fund shall not knowingly purchase any securitisation where the Investment Adviser has perceived any of the following which are directly related to the issuer of a securitisation:

- predatory lending practices;
- breaches in compliance of standards of the Consumer Financial Protection Bureau (CFPB);
- severe malpractice in payment collection processes and foreclosure practices; and
- fraudulent behaviour.

The Investment Adviser uses third-party data and in some cases data on specific issuers. ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates, using best efforts and in good faith. The methodologies used by different data providers may also vary and may result in different scores.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures, swaps, options on swaps (swaptions) and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in Fixed Income Securities that are not Mortgage-Backed Securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Asset Backed Securities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Bond Fund

The Global Bond Fund's investment objective is to provide an attractive rate of return, measured in US Dollars, through market, instrument and currency selection. The Fund will invest primarily in domestic, international, emerging market and Euromarket Fixed Income Securities of varying maturities denominated in US Dollars and other currencies, including asset-backed securities (including mortgage-backed securities), loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above are, firstly, that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. Secondly, the Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

Securities will be deemed to be suitable for investment if at the time of purchase they are rated either "BBB-" or better by S&P or "Baa3" or better by Moody's, or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Fund may continue to hold investments in such securities that are downgraded after purchase. In addition, the Fund may invest, up to 10% of its Net Asset Value, in securities which at the time of purchase are rated either lower than "BBB-" by S&P or "Baa3" by either Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such Fixed Income Securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any such ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Convertible Bond Fund

The Global Convertible Bond Fund's investment objective is to seek long term capital appreciation, measured in US Dollars, through investment primarily in convertible bonds issued by companies organised or operating in either the developed or emerging markets which will be denominated in global currencies.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including ESG assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued

by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above are, firstly, that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. Secondly, the Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in other Fixed Income Securities as well as a combination of equities and warrants on transferable securities either as a result of exercising the conversion option in the convertible bonds in the Fund or as an alternative to convertible bonds when it deems it appropriate.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take

into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Convertible Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- seek to invest in Fixed Income Securities;
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Credit Fund

The Global Credit Fund's investment objective is to provide an attractive rate of relative return, measured in USD.

The Fund will invest primarily in Fixed Income Securities issued by corporations and other non-government issuers ("Corporate Bonds") organised or operating in both developed and emerging markets and denominated in global currencies. With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focused on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund but subsequently become restricted due to the application of the ESG criteria above, after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Fund may, on an ancillary basis, invest in Fixed Income Securities that are not Corporate Bonds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Credit Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Credit Opportunities Fund

The Global Credit Opportunities Fund's investment objective is to provide an attractive level of total return, measured in US Dollars.

The Fund will invest primarily in Fixed Income Securities issued by corporations ("Corporate Bonds") and other non-government related issuers Located in developed markets, including investment-grade bonds, high-yield bonds. High yield bonds are considered to be Fixed Income Securities issued by corporations that are rated lower than "BBB-" by Standard & Poor's Corporation ("S&P") or "Baa3" by Moody's Investors Services, Inc. or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Fund may also invest primarily in mortgage-backed securities, and convertibles subject to applicable law and in other asset-backed securities as well as loan participations and loan assignments to the extent that these instruments are securitised. The Investment Adviser may invest in any combination of two or more of the above asset classes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by S&P or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities

held by the Fund are subsequently downgraded to a rating below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six (6) months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Fund promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human health and wellbeing, through exclusion of tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may, on an ancillary basis, invest in Fixed Income Securities not meeting the criteria of the Fund's primary investments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange-traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

Taxonomy Regulation disclosures

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation.

Profile of the typical investor

In light of the Global Credit Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Fixed Income Opportunities Fund

The Global Fixed Income Opportunities Fund's investment objective is to provide an attractive level of total return, measured in US Dollars, through investment in global fixed income securities. The Fund will seek to achieve the investment policy through allocation across fixed income asset classes as well as through market and instrument selection.

The Fund will invest primarily in the global securities of corporate, government and government related issuers across a spectrum of fixed income asset classes including high yield bonds, investment-grade bonds, mortgage-backed securities, convertibles and currencies, and subject to applicable law, in other asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised. High yield bonds are considered to be Fixed Income Securities issued by corporations that are rated lower than "BBB-" by S&P or "Baa3" by Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. These investments may include Fixed Income Securities issued in emerging markets. The Investment Adviser may invest in any combination of two or more of the above asset classes.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

manufacturing or production of controversial weapons;

- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanlevinvestmentfunds.com and <a href="https://ww

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may also invest, on an ancillary basis, additional securities including but not limited to Cash Equivalents, equity or other equity linked securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Fixed Income Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";

• accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global High Yield Bond Fund

The Global High Yield Bond Fund's investment objective is to provide an attractive rate of return, measured in US Dollars, through investments worldwide primarily in Fixed Income Securities issued by corporations that are rated lower than "BBB-" by S&P or "Baa3" by Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. These investments may include Fixed Income Securities issued in emerging markets and, subject to applicable law, asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above

may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in other Fixed Income Securities (including securities issued by governments and agencies, and securities which are rated "BBB-" or higher by S&P or "Baa3" or higher by Moody's, or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser).

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Macro Fund

The Global Macro Fund's investment objective is to deliver a positive absolute return in all market conditions, in the form of income plus capital appreciation over a rolling three-year period, through exposure to currencies, and U.S. and non-U.S. interest rates and issuers.

The Fund will seek to achieve its investment objective by investing primarily in a portfolio of transferable securities and Fixed Income Securities globally and through the use of financial derivative instruments.

The Fund's primary investment universe will include Fixed Income Securities (issued by governments and government related, as well as corporate issuers Located in both developed and Emerging Market countries, which include, without limitation, mortgage-backed securities and other asset-backed securities, inflation-indexed bonds issued by governmental, government-related, and corporate issuers, corporate debt securities (including, without limitation, convertible securities and corporate commercial paper)), units in open and closed-ended collective investment schemes, structured products, deposits, and, to a limited extent, participation notes.

The Fund will not have any particular sectoral or industry focus. The Fund's investments may be highly concentrated in a geographic region or country, and there is no limit on the Fund's exposure to any geographic region or country. The Fund may invest up to 100% of its net assets in the securities of issuers Located in Emerging Market countries (for the purposes of this Fund, defined as any country which did not become a member of the OECD prior to 1975 and Turkey).

The Investment Adviser adjusts investments to take advantage of differences in securities, countries, currencies and credits based on its perception of various factors, including the most favourable markets, interest rates and issuers, the relative yield and appreciation potential of a particular country's securities, and the relationship of a country's currency to the USD. The Investment Adviser identifies countries and currencies where the Fund's investments will out-perform comparable investments in other countries and currencies and in many cases seeks to predict changes in economies, markets, political conditions, and other factors.

In evaluating the debt securities issued by governments, government agencies and government-sponsored enterprises in which the Fund may invest, the Investment Adviser will integrate environmental, social and governance ("ESG") criteria. The Investment Adviser, as part of its research and asset selection process in respect of debt securities, will consider the ESG issues in respect of each country and will carry out a qualitative analysis of each country, rating the ESG factors for each country based on information gathered by the research staff of the Investment Adviser, subadviser (if applicable) and/or their affiliates as well as third party sources. Such qualitative analysis will be supplemented by quantitative ESG scores attributed by the Investment Adviser to each country and the ESG scores allow for cross-country comparisons. As part of the Investment Adviser's analysis, the Investment Adviser may have regard to ESG factors comprising Sustainability Risks or opportunities likely to affect the financial condition or performance of the country over a long term investment horizon, including, without limitation, pollution/environmental/ecosystem health, risk of social unrest, human development, democracy and rights and freedoms.

The Investment Adviser believes that as sustainability issues are sources of long-term risk and return, the integration of Sustainability Risks as outlined above may assist in enhancing the returns of the Fund over the long term.

The Fund may invest no more than 20% of its assets in aggregate in asset-backed securities (ABS) and/or mortgage-backed securities (MBS).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may also invest, on an ancillary basis, in warrants.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Macro Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Short Duration US Government Income Fund

The Short Duration US Government Income Fund's investment objective is to seek total return, which is defined as income plus capital appreciation, by investing at least 90% of its net assets in Fixed Income Securities issued, backed or otherwise guaranteed by the U.S. Government or its agencies or instrumentalities. The dollar weighted average duration of the Fund will normally be less than 3 years, although this may be higher under certain market conditions. In general, the longer the average duration of the portfolio the greater the exposure to interest rate risk and as such the more volatile the value of the fund. Due to the Fund's relative short duration, the Fund is likely to generate reasonably steady returns and maintain relatively low levels of volatility in the value of the Fund. Since the Fund's return will primarily be generated by income payments from the underlying investments rather than by capital appreciation of the portfolio.

The Fund may enter into forward commitments to purchase mortgage-backed securities ("MBS") issued by Ginnie Mae, Fannie Mae and Freddie Mac (to-be-announced transactions, or "TBAs"). The Fund may also enter into generic MBS coupon swaps, agency swaps, and term swaps by entering into forward commitments to both purchase and sell MBS. Such forward commitments may be entered into for the purposes of increasing returns or reducing risk and may result in investment leverage.

The Fund may also enter into Mortgage Dollar Rolls, whereby the Fund sells MBS for delivery in the current month and simultaneously contracts to repurchase substantially similar (same type, coupon and maturity) MBS on a specified future date. During the roll period, the Fund forgoes principal and interest paid on the MBS.

The Fund's Investment Adviser utilizes the information provided by, and the expertise of, the research staff of the Investment Adviser, sub-adviser (if applicable) and/or their affiliates in making investment decisions. When selecting which securities to invest in, the Investment Adviser seeks to identify the securities which it believes to be the best relative value with regard to price, yield, and expected total return in relation to other available instruments. Securities are analysed using various prepayment models (i.e., models which analyse the expected level of prepayments on a portfolio of assets) and scenarios which are proprietary to the Investment Adviser, and evaluated versus the overall market on a risk/return basis. Securities are also reviewed in the context of the Fund's overall portfolio to determine if the security fits the portfolio and to make sure that allocation does not overweight the portfolio to a particular risk factor. As part of the selection process, the Investment Adviser will analyse a number of factors in relation to each security, including (i) the underlying collateral characteristics of the security, (ii) their prepayment expectations for the collateral, and (iii) the compatibility of the security's structure with those prepayment expectations. Collateral analysis involves an in-depth examination of borrower characteristics that may influence a security's propensity to prepay and includes factors such as agency, borrower geographic dispersion, mortgage servicer mix, loan size, etc. Security structure selection involves identifying structures that are expected to be enhanced by the Investment Adviser's prepayment expectations. The Investment Adviser makes its investment decisions primarily on the basis of fundamental research, including the factors outlined above, and relative value and may sell a security when it believes it no longer represents the best relative value based on the fundamental research. Based on the analysis and research undertaken by the Investment Adviser, which will include the relative risk / return characteristics of each prospective investment, the Investment Adviser will select securities which it believes will provide the Fund with the most appropriate relative value.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange trades and over-the-counter options, futures and other derivatives for efficient portfolio management (including hedging) purposes.

In addition to the above, in making investment decisions, the Investment Adviser may integrate environmental, social and governance ("ESG") criteria. The Investment Adviser may utilize the extensive proprietary ESG research generated by its affiliate, Calvert Research and Management, as well as its own ESG research in evaluating investments. Given the investment focus of the Fund on debt instruments issued, backed or otherwise guaranteed by the U.S. Government or its agencies or instrumentalities, it is not generally expected that ESG considerations will be an significant factor in the Fund's investment decisions. The Investment Adviser believes that as sustainability issues are sources of long-term risk and return, the integration of sustainability risks as outlined above may assist in enhancing the returns of the Fund over the long term.

The Fund may invest more than 20% of its assets in aggregate in mortgage-backed securities (MBS).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may also invest, to a limited extent, in Fixed Income Securities not meeting the criteria of the Fund's primary investments, warrants on securities, Cash Equivalents and other equity linked securities.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Short Duration US Government Income Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Short Maturity Euro Bond Fund

The Short Maturity Euro Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro.

The Fund will invest primarily in Euro denominated Fixed Income Securities whether issued by corporations, government or government guaranteed issuers. With a view to enhancing returns and/or as part of the investment

strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund will invest primarily, in order to reduce volatility, in individual securities with maturity dates having a maximum unexpired term of five years.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB-in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Short Maturity Euro Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the short term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Short Maturity Euro Corporate Bond Fund

The Short Maturity Euro Corporate Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro.

The Fund will invest, in order to reduce volatility, primarily in Euro denominated Fixed Income Securities, issued by corporations("Corporate Bonds") with maturity dates or call dates (whichever is earlier) having a maximum unexpired term of three (3) years.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB-in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six (6) months of its downgrade.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include but are not limited to ESG themes such as decarbonisation and climate risk, circular economy and waste reduction, diverse and inclusive business and decent work and resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Fund promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human health and wellbeing through exclusion of tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

• manufacturing or production of controversial weapons;

- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www. morganstanley.com/im. Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund may, on an ancillary basis, invest in Fixed Income Securities not meeting the criteria of the Fund's primary investments.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest no more than 20% of its assets in aggregate asset-backed securities (ABS) and/or mortgage-backed securities (MBS).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and is not designed to track a benchmark. Therefore, the management of the Fund is not constrained by the composition of a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

In light of the Short Maturity Euro Corporate Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the short term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable Euro Corporate Bond Fund

The Sustainable Euro Corporate Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro, while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores, and maintaining a lower carbon footprint than the corporate bond component of the Bloomberg European Corporate Index (see below) taking into account the long-term global warming objectives of the Paris Agreement, as further described below.

The Fund will invest primarily in Euro denominated Fixed Income Securities, issued by corporations and other non-government related issuers ("Corporate Bonds").

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainable themes and ESG issues in its investment decision-making on a discretionary basis as further detailed below. In addition, the investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices as well as what it deems to be materially important environmental and/or social issues facing a company.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at on an annual basis.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;
 - o manufacturing or production of civilian firearms;
 - o manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail; and
 - o gambling, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, investments shall not knowingly include any company whose business activity involves the following:

- o mining and extraction of thermal coal;
- \circ coal-fired power generation, where the company derives >10% revenue from such business activity;
- extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- \circ oil or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled Sustainable Bonds (see below), or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

The Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe, including those related to the OECD Guidelines for Multinational Enterprises, based on ratings by relevant ESG data providers, where the Investment Adviser believes appropriate remedial action has not been taken.

The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

The Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

The Fund will only invest in the top 80% of rated ESG-scoring corporates in each sub-sector of the Bloomberg European Corporate Index. The ESG scores are determined by the Investment Adviser, using third party ESG data as a base having regard to ESG themes such as Decarbonisation & Climate Risk, Circular Economy & Waste Reduction, Diverse & Inclusive Business and Decent Work & Resilient Jobs. The aforementioned ESG criteria should result in at least 20% reduction of the investible universe of corporates, and the Investment Adviser will use its best efforts to ensure that at least 90% of the corporates in the portfolio are assessed through its ESG scoring methodology. The positive screening process and methodology is available on the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund but subsequently become restricted due to the application of the ESG criteria above, after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

For the proportion of the Fund invested in bonds issued by corporates, the Fund will maintain a lower carbon footprint than the corporate bond component of the Bloomberg European Corporate Index, taking into account the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average Carbon Intensity, defined as tonnes of CO2 per \$1 million revenue. Sustainable Bonds with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds. In such cases, the Investment Adviser shall disclose its methodology for determining the relevant carbon footprint of such instruments.

As part of its strategy, the Investment Adviser may invest a proportion of the Fund in Sustainable Bonds, defined as labelled Sustainable Bonds where the issuer has committed to financing or attaining specific environmental and/or social objectives, including but not limited to the below based on the main labels of bonds issued to date, including those recognised by the International Capital Market Association (ICMA):

- Green Bonds;
- Social Bonds;
- Sustainability Bonds;
- Transition Bonds; and
- Sustainability-linked Bonds.

The Investment Adviser deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

As noted above, the Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of derivative instruments such as index futures, contracts for difference (CFDs), TRS, structured notes (including participatory notes), convertible bonds, convertible preferred stocks, options, foreign exchange options, currency forwards and forward contracts and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Fund may, on an ancillary basis, invest in:

- Non-Euro denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non-Euro currency exposure may be hedged back to the Euro.
- Fixed Income Securities that are not Corporate Bonds.

The Fund may invest no more than 20% of its assets in asset-backed securities.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and references the Bloomberg European Corporate Index for the purpose of defining the investment universe. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will primarily invest in securities that are within the Bloomberg European Corporate Index, it may invest in such components in different weights to those of the Bloomberg European Corporate Index, and it may hold securities which are not included within the Bloomberg European Corporate Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the Bloomberg European Corporate Index.

In addition, the Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the corporate bond component of the Bloomberg European Corporate Index for the purposes of comparing the Fund's carbon footprint.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%. Profile of the typical investor

In light of the Sustainable Euro Corporate Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable Euro Strategic Bond Fund

The Sustainable Euro Strategic Bond Fund's investment objective is to provide an attractive rate of relative return, measured in Euro, by investing in securities that meet the Investment Adviser's Fixed Income Securities' ESG criteria while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores, and maintaining a lower carbon footprint than the corporate bond component of the Bloomberg Euro Aggregate Index (see below), taking into account the long-term global warming objectives of the Paris Agreement, as further described below.

The Fund will invest primarily in issues of Euro denominated Fixed Income Securities whether issued by corporations, government or government guaranteed issuers, and subject to applicable law, no more than 20% of its assets in asset-backed securities to the extent that these instruments are securitised.

In relation to this Fund, "Sustainable" means that the Investment Adviser, integrates the consideration of sustainable themes and ESG issues in its investment decision-making on a discretionary basis as further detailed below. In addition, the investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices as well as what it deems to be materially important environmental and/or social issues facing a company.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at on an annual basis.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;
 - o manufacturing or production of civilian firearms;
 - o manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail; and
 - o gambling, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, investments shall not knowingly include any company whose business activity involves the following:

- o mining and extraction of thermal coal;
- o coal-fired power generation, where the company derives >10% revenue from such business activity;
- o extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- o oil or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled Sustainable Bonds (see below), or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

The Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe, including those related to the OECD Guidelines for Multinational Enterprises, based on ratings by relevant ESG data providers, where the Investment Adviser believes appropriate remedial action has not been taken. The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

The Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

For its investments in Fixed Income Securities issued by corporations, the Fund will only invest in the top 80% of rated ESG-scoring corporates in each sub-sector of the Bloomberg Euro Aggregate Index. The ESG scores are determined by the Investment Adviser, using third party ESG data as a base having regard to ESG themes such as such as Decarbonisation & Climate Risk, Circular Economy & Waste Reduction, Diverse & Inclusive Business and Decent Work & Resilient Jobs. For the Fund's investments in Fixed Income Securities issued by sovereigns, the Fund will only invest in sovereigns within the top 80% of ESG scoring sovereigns that we have evaluated, unless the Investment Adviser perceives that a sovereign in the bottom 20% is demonstrating positive sustainability momentum, and/or issues a Sustainable Bond (see below). The Fund will not invest in bonds from sovereigns which we have not scored. The aforementioned ESG criteria should result in at least 20% reduction of the investible universe of corporates and sovereigns, and the Investment Adviser will use its best efforts to ensure that at least 90% of the corporates in the portfolio are assessed through its ESG scoring methodology. The positive screening process and methodology is available on the following websites: www.morganstanley.com/im.

Investments that are held by the Fund but subsequently become restricted due to the application of the ESG criteria above, after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

For the proportion of the Fund invested in bonds issued by corporates, the Fund will maintain a lower carbon footprint than the corporate bond component of the Bloomberg Euro Aggregate Index, taking into account the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average Carbon Intensity, defined as tonnes of CO2 per \$1 million revenue. Sustainable Bonds with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds. In such cases, the Investment Adviser shall disclose its methodology for determining the relevant carbon footprint of such instruments.

As part of its strategy, the Investment Adviser may invest a proportion of the Fund in Sustainable Bonds, defined as labelled Sustainable Bonds where the issuer has committed to financing or attaining specific environmental and/or social objectives, including but not limited to the below based on the main labels of bonds issued to date, including those recognised by the International Capital Market Association (ICMA):

- Green Bonds:
- Social Bonds;
- Sustainability Bonds;
- Transition Bonds; and
- Sustainability-linked Bonds.

The Investment Adviser deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

As noted above, the Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of derivative instruments such as index futures, contracts for difference (CFDs), TRS, structured notes (including participatory notes), convertible bonds, convertible preferred stocks, options, foreign exchange options, currency forwards and forward contracts and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may, on an ancillary basis, invest in non-Euro denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non-Euro currency exposure may be hedged back to the Euro.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and references the Bloomberg Euro Aggregate Index for the purpose of defining the investment universe. The Investment Adviser has full discretion over the composition of the assets in the Fund. While the Fund will primarily invest in securities that are within the Bloomberg Euro Aggregate Index, it may invest in such components in different weights to those of the Bloomberg Euro Aggregate Index, and it may hold securities which are not included within the Bloomberg Euro Aggregate Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the Bloomberg Euro Aggregate Index.

In addition, the Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document. In the context of the Fund's ESG integration, the Investment Adviser uses the corporate bond component of the Bloomberg Euro Aggregate Index for the purposes of comparing the Fund's carbon footprint.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable Euro Strategic Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable European High Yield Bond Fund

The Sustainable European High Yield Bond Fund's investment objective is to generate current income and total return, while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores, with a view to achieving the long-term global warming objectives of the Paris Agreement adopted under the United Nations Framework Convention on Climate Change (the "Paris Agreement"), as further described below.

The Fund will invest primarily in a diversified portfolio of Fixed Income Securities from issuers that are Located in Europe and which may include zero coupon bonds, deferred interest bonds and bonds on which the interest is payable in the form of additional eligible bonds of the same kind (Payment-in-Kind (PIK) bonds).

The Fund will invest in Fixed Income Securities that are rated Baa3 or lower by Moody's Investors Service, Inc. ("Moody's"), BBB- or lower by Standard & Poor's Ratings Group ("S&P"), BBB- or lower by Fitch's Investors Service, Inc. or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as of the date of investment

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainable themes and ESG issues in its investment decision-making on a discretionary basis as further detailed below. In addition, the Investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices as well as what it deems to be materially important environmental and/or social issues facing a company.

The Investment Adviser utilizes a screening process, the extensive proprietary ESG research generated by its affiliate, Calvert Research and Management, as well as its own ESG research and data from third party providers in evaluating investments. The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focused on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least annually. This assessment has regard to ESG themes such as decarbonisation and climate risk, circular economy and waste reduction, diverse and inclusive business and decent work and resilient jobs.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;
 - o manufacturing or production of civilian firearms;
 - o manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail; and
 - o gambling, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, investments shall not knowingly include any company whose business activity involves the following:

- o mining and extraction of thermal coal;
- o coal-fired power generation, where the company derives >10% revenue from such business activity;
- o extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- o oil or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled Sustainable Bonds (as defined below), or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

The Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe based on ratings by relevant ESG data providers, where the Investment Adviser believes appropriate remedial action has not been taken. The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

The Investment Adviser will also elect to apply positive screening criteria in its investment process that it believes are consistent with its objectives. The positive screening process and methodology are available at the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and www.worganstanleyi

The aforementioned ESG criteria will result in a reduction of the investible universe by at least 20%, and the Investment Adviser will ensure that at least 90% of the Net Asset Value of the Fund is assessed through its ESG research. The investment universe for these purposes is defined as the Index. The Investment Adviser will also monitor core sustainability indicators, including ESG assessments from third-party providers, and carbon footprint (for the proportion of the Fund invested in fixed income debt securities issued by corporates) in order to measure and evaluate the contribution of the fixed income debt securities to the ESG themes described above. The investible universe will be reviewed on an ongoing basis. The ESG criteria will be reviewed in respect of each individual company at least annually, but this may be carried out more frequently. The Investment Adviser will make adjustments and changes to the methodology as necessary. The indicators referred to above will be measured and evaluated at least annually.

Investments that are held by the Fund but subsequently become restricted after they are acquired for the Fund due to the application of the ESG criteria above, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Fund will maintain a lower carbon footprint than the ICE BofA European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index, taking into account the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average Carbon Intensity, defined as tonnes of CO2 per \$1million revenue. Sustainable Bonds with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds. In such cases, the Investment Adviser shall disclose its methodology for determining the relevant carbon footprint of such instruments.

As part of its strategy, the Investment Adviser may invest a proportion of the Fund in "**Sustainable Bonds**", defined as labelled sustainable bonds where the issuer has committed to financing or attaining specific environmental and/or social objectives, including but not limited to the below based on the main labels of bonds issued to date, including those recognised by the International Capital Market Association (ICMA):

- Green Bonds;
- Social Bonds;
- Sustainability Bonds;
- Transition Bonds; and
- Sustainability-linked Bonds.

The Investment Adviser deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

As noted above, the Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund will not have any particular sectoral or industry focus and the investment by the Fund is generally not intended to exceed 25% of its net assets in any one industry but investment may exceed this from time to time.

The Fund may invest no more than 20% of its assets in debt securities that are not paying current income in anticipation of the receipt of possible future income or capital appreciation which are unrated or categorized as the lowest rated obligations (rated C by Moody's or D by S&P).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may also invest, on an ancillary basis, in equity securities or Fixed Income Securities not meeting the criteria of the Fund's primary investments, warrants on securities, Cash Equivalents and other linked securities.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document, and this benchmark will also be designated as the ICE BofA European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index and will be used to monitor error relative to the benchmark. While the Fund will generally hold assets referenced in the ICE BofA European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index, it can invest in such securities in different proportions to their allocations under the ICE BofA

European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index, and it can hold assets which are not referenced in the ICE BofA European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the ICE BofA European Currency Developed Markets High Yield excluding Subordinated Financials Constrained Index.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable European High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable Global Fixed Income Opportunities Fund

The Sustainable Global Fixed Income Opportunities Fund's investment objective is to provide an attractive level of total return, measured in US Dollars, through investment in global Fixed Income Securities that meet the Investment Adviser's ESG criteria while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores and maintaining a lower carbon footprint than the corporate bond component of the Bloomberg Global Aggregate Index, taking into account the long-term global warming objectives of the Paris Agreement, as further described below. The Fund will seek to achieve the investment policy through allocation across fixed income asset classes as well as through market and instrument selection.

The Fund will invest primarily in the global securities of corporate, government and government related issuers across a spectrum of fixed income asset classes including high yield bonds, investment-grade bonds, mortgage-backed securities, convertibles and currencies, and subject to applicable law, in other asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised. High yield bonds are considered to be Fixed Income Securities issued by corporations that are rated lower than "BBB-" by S&P or "Baa3" by Moody's or similarly by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. These investments may include Fixed Income Securities issued in emerging markets.

In relation to this Fund, "Sustainable" means that the Investment Adviser, integrates the consideration of sustainable themes and ESG issues in its investment decision-making on a discretionary basis as detailed below. In addition, the Investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices as well as what it deems to be important environmental and/or social issues facing a company.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;

- o manufacturing or production of civilian firearms;
- o manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail;
- gambling, where the company derives >10% revenue from such business activity; and
- o adult entertainment, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, investments shall not knowingly include any company whose business activity involves the following:

- mining and extraction of thermal coal;
- coal-fired power generation, where the company derives >10% revenue from such business activity;
- extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- oil or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

The Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe based on ratings by relevant ESG data providers, where the Investment Adviser believes appropriate remedial action has not been taken. The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

The Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

The Fund will invest in the top 80% of rated ESG-scoring corporates in each sub-sector of any of the Bloomberg Global Aggregate Corporate Index, the Bloomberg Global High Yield Corporate Index and the J.P. Morgan CEMBI Broad Diversified Index. Where any corporates appear in both (A) the J.P. Morgan CEMBI Broad Diversified Index and (B) either the Bloomberg Global Aggregate Corporate Index or the Bloomberg Global High Yield Corporate Index, the ESG scoring for those corporates will be considered by reference to the J.P. Morgan CEMBI Broad Diversified Index only. The ESG scores are determined by the Investment Adviser, using third party ESG data as a base having regard to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs.

For the Fund's investments in Fixed Income Securities issued by sovereigns, the Fund will only invest in sovereigns within the top 80% of ESG scoring sovereigns evaluated by the Investment Adviser, unless the Investment Adviser perceives that a sovereign in the bottom 20% is demonstrating positive sustainability momentum, and/or issues a Sustainable Bond (see below). The Fund will not invest in bonds from sovereigns which the Investment Adviser has not scored.

The aforementioned ESG criteria will seek to achieve a selection of corporates and sovereigns that are better rated than the other corporates and sovereigns of the investment universe (the corporate component of which is defined as corporates in any one or more of the Bloomberg Global Aggregate Corporate Index, the Bloomberg Global High Yield Corporate Index and the J.P. Morgan CEMBI Broad Diversified Index, subject to a maximum 10% of the Fund's portfolio that may be invested in corporates not covered by these indices). In this respect, the ESG-scoring of corporates should result in at least a 20% reduction of the corporate component of the investment universe of the Fund, and the aim of the Investment Adviser will be to ensure that at least 90% of the corporate issuers in the portfolio are assessed through its ESG scoring methodology. The positive screening process and methodology is available on the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

For the Fund's investments in non-corporate and non-sovereign Fixed Income Securities, ESG criteria are considered by the Investment Adviser during the research and investment process to limit exposure to Sustainability Risks and evaluate investment opportunities.

Investments that are held by the Fund but subsequently become restricted due to the application of the ESG criteria above, after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

For the proportion of the Fund invested in bonds issued by corporates, the Fund will maintain a lower carbon footprint than the corporate bond component of the Bloomberg Global Aggregate Index, taking into account the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average Carbon Intensity, defined as tonnes of CO2 per \$1 million revenue. Sustainable Bonds with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds. In such cases, the Investment Adviser shall disclose its methodology for determining the relevant carbon footprint of such instruments.

As part of its strategy, the Investment Adviser may invest a proportion of the Fund in Sustainable Bonds, defined as labelled Sustainable Bonds where the issuer has committed to financing or attaining specific environmental and/or social objectives, including but not limited to the below based on the main labels of bonds issued to date, including those recognised by the International Capital Market Association (ICMA):

- Green Bonds;
- Social Bonds;
- Sustainability Bonds;
- Transition Bonds; and
- Sustainability-linked Bonds.

The Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities not meeting the above requirements, Cash Equivalents, equity securities or other equity linked securities.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

It is not expected that the Fund will invest more than 50% of its net assets in aggregate in securitized instruments. It is not expected that the Fund will invest more than 20% of its net assets in any one type of securitized instrument (*e.g.*, asset-backed securities (ABS), commercial mortgage-backed securities (CMBS), residential mortgage-backed securities (RMBS), agency mortgage-backed securities (MBS)).

The Fund may invest in Fixed Income Securities acquired on the China Interbank Bond Market. No more than 10% of the Fund's assets will be invested in such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark, and therefore the composition of the portfolio holdings is not constrained by any benchmarks. In the context of the Fund's ESG integration, the Investment Adviser uses (i) the corporate bond component of the Bloomberg Global Aggregate Index for the purpose of comparing the Fund's carbon footprint and (ii) the sub-sectors of the Bloomberg Global Aggregate Corporate Index, the Bloomberg Global High Yield Corporate Index and the J.P. Morgan CEMBI Broad Diversified Index for the purposes of defining the top rated ESG-scoring corporates the Fund will invest into.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable Global Fixed Income Opportunities Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable Global High Yield Bond Fund

The Sustainable Global High Yield Bond Fund's investment objective is to generate current income and total return, while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores, with a view to achieving the long-term global warming objectives of the Paris Agreement adopted under the United Nations Framework Convention on Climate Change (the "Paris Agreement"), as further described below.

The Fund will invest primarily in a diversified portfolio of high yielding, Fixed Income Securities issued by corporations and other non-government issuers ("Corporate Bonds") organised or operating in both developed and emerging markets and denominated in global currencies, which may include zero coupon bonds, preferred stocks, deferred interest bonds and notes and bonds and notes on which the interest is payable in the form of additional eligible stocks, bonds or notes of the same kind (Payment-in-Kind (PIK) securities).

The Fund will invest in securities that are rated Baa3 or lower by Moody's, BBB- or lower by Standard & Poor's, BBB- or lower by Fitch's, or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser as at the date of investment.

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainable themes and ESG issues into its investment decision-making on a discretionary basis as further detailed below. In addition, the Investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices, as well as what it deems to be materially important environmental and/or social issues facing a company. The Fund will maintain a lower carbon footprint than the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index with a view to achieving the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average carbon intensity (WACI), defined as tonnes of CO2 per \$1million revenue. Sustainable Bonds (as defined below) with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds, such as for example where the bond is issued as part of the climate transition strategy of an issuer.

The Investment Adviser utilizes a screening process, the extensive proprietary ESG research generated by its affiliate, Calvert Research and Management, as well as its own ESG research and data from third party providers in evaluating investments. The Investment Adviser also applies proprietary assessment and scoring methodologies that are bespoke to the Fixed Income Securities in which the Fund may invest, focused on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. This assessment has regard to ESG themes such as decarbonisation and climate risk, circular economy and waste reduction, diverse and inclusive business and decent work and resilient jobs.

The Investment Adviser will also elect to apply positive screening criteria in its investment process that it believes are consistent with its objectives. The positive screening process and methodology are available at the following websites: www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

The aforementioned ESG criteria will result in a reduction of the investible universe by at least 20%, and the Investment Adviser will ensure that at least 90% of the Net Asset Value of the Fund is assessed through its ESG research. The investment universe for these purposes is defined as the Index. The Investment Adviser will also monitor core sustainability indicators, including ESG assessments from third-party providers, and carbon footprint (for the proportion of the Fund invested in fixed income debt securities issued by corporates) in order to measure and evaluate the contribution of the fixed income debt securities to the ESG themes described above. The investible universe will be reviewed on an ongoing basis. The ESG criteria will be reviewed in respect of each individual company at least annually, but this may be carried out more frequently. The Investment Adviser will make adjustments and changes to the methodology as necessary. The indicators referred to above will be measured and evaluated at least annually.

As part of the investment and screening process, the Fund shall not knowingly invest in any company whose business activity involves the following:

- manufacturing or production of:
 - o controversial weapons;
 - o manufacturing or production of civilian firearms;
 - o manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail; and
 - \circ gambling, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, the Fund shall also not knowingly invest in any company whose business activity involves the following:

o mining and extraction of thermal coal;

- o coal-fired power generation, where the company derives >10% revenue from such business activity;
- extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- oil or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

An exception to the fossil fuel exclusions in the preceding paragraph is that the Fund may invest in labelled sustainable bonds or other Fixed Income Securities that are issued to raise capital specifically for climate-related or environmental projects ("Sustainable Bonds"), which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer of its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. Examples of Sustainable Bonds include, but are not limited to:

- Green bonds;
- Social bonds;
- Sustainability bonds;
- Transition bonds; and
- Sustainability-linked bonds.

Prior to investing in Sustainable Bonds, the Investment Adviser deploys a proprietary assessment framework for labelled Sustainable Bonds, through which the robustness, impact and transparency of such instruments are evaluated.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

In addition to the above exclusions, the Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe based on ratings by relevant ESG data providers, and where the Investment Adviser believes appropriate remedial action has not been taken. The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund but subsequently become restricted after they are acquired for the Fund due to the application of the ESG criteria above, will be sold.

As noted above, the Investment Adviser uses third-party data and ESG scores and, in some cases, data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest, on an ancillary basis, in Fixed Income Securities not meeting the criteria of the Fund's primary investments such as fixed-income and floating rate transferable debt securities issued or guaranteed by the U.S. Government, any state or territory of the United States, any non-U.S. government or any of their respective political subdivisions, agencies or instrumentalities, equity securities, warrants on securities, Cash Equivalents and other equity linked securities. Under normal circumstances, the Fund may invest no more than 10% of its assets in such equity securities, warrants on securities and other equity linked securities.

The Fund generally intends to invest no more than 5% of its net assets in emerging markets, however, investment in emerging markets may comprise up to 20% of the net assets of the Fund.

In addition to exposure to sustainable investments, all investments made by the Fund will be deemed to do no significant harm to environmental or social factors, as determined by the Investment Adviser. The Investment Adviser assesses this according to an internal methodology (as amended from time to time) which considers a representative sub-set of principal adverse sustainability impact indicators.

The Fund will not have any particular sectoral or industry focus and the investment by the Fund is generally not intended to exceed 25% of its net assets in any one industry but investment may exceed this from time to time.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

The Fund may invest no more than 20% of its assets in unrated or lowest rated categorised debt securities (rated C by Moody's or D by S&P).

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document, and this benchmark will also be designated as the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index and used to monitor error relative to the benchmark. While the Fund will generally hold assets referenced in the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index, it can invest in such securities in different proportions to their allocations under the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index, and it can hold assets which are not referenced in the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the ICE BofA Developed Markets High Yield Index Excluding Subordinated Financial Index.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable Global High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Sustainable US High Yield Bond Fund

The Sustainable US High Yield Bond Fund's investment objective is to generate current income and total return, while integrating ESG characteristics by screening out sectors and issuers with low overall sustainability scores, with a view to achieving the long-term global warming objectives of the Paris Agreement adopted under the United Nations Framework Convention on Climate Change (the "Paris Agreement"), as further described below.

The Fund will invest primarily in a diversified portfolio of Fixed Income Securities from issuers that are Located in the U.S. and which may include zero coupon bonds, deferred interest bonds and bonds on which the interest is payable in the form of additional eligible bonds of the same kind (Payment-in-Kind (PIK) bonds).

The Fund will invest in Fixed Income Securities that are rated Baa3 or lower by Moody's Investors Service, Inc. ("Moody's"), BBB- or lower by Standard & Poor's Ratings Group ("S&P"), BBB- or lower by Fitch's Investors Service, Inc. or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as of the date of investment.

In relation to this Fund, "Sustainable" means that the Investment Adviser integrates the consideration of sustainable themes and ESG issues into its investment decision-making on a discretionary basis, as further detailed below. In addition, the Investment Adviser may include evaluation of positive and negative contributions towards the UN Sustainable Development Goals and may engage company management around corporate governance practices, as well as what it deems to be materially important environmental and/or social issues facing a company. The Fund will maintain a lower carbon footprint than the ICE BofA U.S. High Yield Index with a view to achieving the long-term global warming objectives of the Paris Agreement. Carbon footprint shall be measured as weighted average carbon intensity (WACI), defined as tonnes of CO2 per \$1million revenue. Sustainable Bonds (as defined below) with climate-related objectives may be determined to have different carbon footprints than the issuers of the bonds, such as for example where the bond is issued as part of the climate transition strategy of an issuer.

The Investment Adviser utilizes a screening process, the extensive proprietary ESG research generated by its affiliate, Calvert Research and Management, as well as its own ESG research and data from third party providers in evaluating

investments. The Investment Adviser also applies proprietary assessment and scoring methodologies that are bespoke to the Fixed Income Securities in which the Fund may invest, focused on corporate, sovereign and securitized issuance. Additionally, as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. This assessment has regard to ESG themes such as decarbonisation and climate risk, circular economy and waste reduction, diverse and inclusive business and decent work and resilient jobs.

The Investment Adviser will also elect to apply positive screening criteria in its investment process that it believes are consistent with its objectives. The positive screening process and methodology are available at the following websites: www.morganstanley.com/im.

The aforementioned ESG criteria will result in a reduction of the investible universe by at least 20%, and the Investment Adviser will ensure that at least 90% of the Net Asset Value of the Fund is assessed through its ESG research. The investment universe for these purposes is defined as the Index. The Investment Adviser will also monitor core sustainability indicators, including ESG assessments from third-party providers, and carbon footprint (for the proportion of the Fund invested in fixed income debt securities issued by corporates) in order to measure and evaluate the contribution of the fixed income debt securities to the ESG themes described above. The investible universe will be reviewed on an ongoing basis. The ESG criteria will be reviewed in respect of each individual company at least annually, but this may be carried out more frequently. The Investment Adviser will make adjustments and changes to the methodology as necessary. The indicators referred to above will be measured and evaluated at least annually.

As part of the investment and screening process, the Fund shall not knowingly invest in any company whose business activity involves the following:

- manufacturing or production of:
 - controversial weapons;
 - o manufacturing or production of civilian firearms;
 - manufacturing or production of tobacco, or where the company derives >10% revenue from tobacco distribution or retail; and
 - o gambling, where the company derives >10% revenue from such business activity.

With respect to fossil fuel activities specifically, the Fund shall also not knowingly invest in any company whose business activity involves the following:

- o mining and extraction of thermal coal;
- coal-fired power generation, where the company derives >10% revenue from such business activity;
- extraction or production from oil sands, where the company derives >5% revenue from such business activity; and
- o il or gas extraction or production in the Arctic region, where the company derives >5% revenue from such business activity.

An exception to the fossil fuel exclusions in the preceding paragraph is that the Fund may invest in labelled sustainable bonds or other Fixed Income Securities that are issued to raise capital specifically for climate-related or environmental projects ("Sustainable Bonds"), which are issued by issuers that would otherwise be subject to the fossil fuel exclusions above, so long as the Investment Adviser has determined that the objectives of such instruments are consistent with a reduction by the issuer of its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. Examples of Sustainable Bonds include, but are not limited to:

- Green bonds;
- Social bonds;
- Sustainability bonds;
- Transition bonds; and
- Sustainability-linked bonds.

Prior to investing in Sustainable Bonds, the Investment Adviser deploys a proprietary assessment framework for labelled Sustainable Bonds, through which the robustness, impact and transparency of such instruments are evaluated.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers.

In addition to the above exclusions, the Fund will not invest in securities of issuers with controversy cases that the Investment Adviser views as being very severe based on ratings by relevant ESG data providers, and where the Investment Adviser believes appropriate remedial action has not been taken. The Fund will also not invest in securities of issuers that fail to comply with the UN Global Compact or the ILO fundamental principles, and where the Investment Adviser believes appropriate remedial action has not been taken.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund but subsequently become restricted after they are acquired for the Fund due to the application of the ESG criteria above, will be sold.

As noted above, the Investment Adviser uses third-party data and ESG scores and, in some cases, data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

Further information in respect of the integration of sustainability risks in relation to the Company is included in the section of the Prospectus headed "General information relating to Sustainability Risks integration".

The Fund will not have any particular sectoral or industry focus.

The Fund may also invest, on an ancillary basis, in equity securities or Fixed Income Securities not meeting the criteria of the Fund's primary investments, warrants on securities, Cash Equivalents and other linked securities.

The Fund may invest no more than 10% of its assets in debt securities that are not paying current income in anticipation of the receipt of possible future income or capital appreciation which are unrated or categorized as the lowest rated obligations (rated C by Moody's or D by S&P).

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under the 2010 Law.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

In addition to exposure to sustainable investments, all investments made by the Fund will be deemed to do no significant harm to environmental or social factors, as determined by the Investment Adviser. The Investment Adviser assesses this according to an internal methodology (as amended from time to time) which considers a representative sub-set of principal adverse sustainability impact indicators.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document, and this benchmark will also be designated as the ICE BofA U.S. High Yield Index and will be used to monitor error relative to the benchmark. While the Fund will generally hold assets referenced in the ICE BofA U.S. High Yield Index, it can invest in such securities in different proportions to their allocations under the ICE BofA U.S. High Yield Index, and it can hold assets which are not referenced in the ICE BofA U.S. High Yield Index. Hence, there are no restrictions on the extent to which the Fund's performance may deviate from the ICE BofA U.S. High Yield Index.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Sustainable US High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the long term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Dollar Corporate Bond Fund

The US Dollar Corporate Bond Fund's investment objective is to provide an attractive rate of relative return, measured in US Dollars.

The Fund will seek to achieve its investment objective by investing primarily in US Dollar denominated Fixed Income Securities issued by corporations and other non-government issuers ("Corporate Bonds").

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs. The Investment Adviser will monitor core sustainability indicators, including environmental, social and governance assessments from third-party providers, and carbon footprint (measured by carbon intensity, defined as tonnes of CO2 per \$1 million revenue for the proportion of the Fund invested in bonds issued by corporates) in order to measure and evaluate the contribution of the Fixed Income Securities to the ESG themes described above. The indicators will be measured and evaluated at least on an annual basis.

The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

The Investment Adviser promotes the low carbon transition through excluding corporate issuers for which thermal coal production is a core business, and promotes human life through excluding products that are detrimental to health and wellbeing, specifically tobacco, controversial weapons and civilian firearms manufacturing. Accordingly, investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exceptions to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has been determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser. The Fund may invest in issuers who may not themselves contribute to the specific environmental or social characteristics promoted by the financial product such as hedging instruments, unscreened investments for diversification purposes, or investments for which data are lacking.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, at its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund which as a result of the application of the ESG criteria above become restricted after they are acquired for the Fund, will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may invest in derivatives (in accordance with the investment powers and restrictions set out in Appendix A) such as exchange traded and overthe-counter options, futures, swaps and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund will not invest in any securities that are rated below B- (or below BBB- in the case of asset-backed securities, including mortgage-backed securities) by Standard & Poor's Corporation ("S&P") or an equivalent rating from another rating agency or an equivalent internal rating from the Investment Adviser as at the date of investment. In the event that any securities held by the Fund are subsequently downgraded to a rating below B- (or below BBB-in the case of asset-backed securities, including mortgage-backed securities), the Investment Adviser may maintain a maximum total exposure of 3% of the Fund's Net Asset Value to such downgraded securities but will divest any such security that has not been upgraded to a rating of at least B- within six months of its downgrade.

The Fund may, on an ancillary basis, invest in:

- Non-US Dollar denominated Fixed Income Securities. In accordance with Appendix A "Investment Powers and Restrictions" non US Dollar currency exposure may be hedged back to US Dollars;
- Fixed Income Securities that are not Corporate Bonds.

The Fund may invest no more than 20% of its assets in asset-backed securities and mortgage-backed securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Dollar Corporate Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Dollar High Yield Bond Fund

The US Dollar High Yield Bond Fund's investment objective is to provide an attractive rate of total return, measured in US Dollars. The Fund will seek to achieve its investment objective by investing primarily in high yield and unrated US Dollar denominated Fixed Income Securities, including without limitation those issued by governments, agencies and corporations located in emerging markets and for the avoidance of doubt and subject to applicable law, including asset-backed securities and loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs, The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity. The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or

environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on <a href="https:/

Investments that are held by the Fund but, due to the application of the ESG criteria above, become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

High yield Fixed Income Security means a Fixed Income Security that has a rating less than "BBB-" by S&P or "Baa3" by Moody's or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser.

The Fund may, on an ancillary basis, invest in higher quality Fixed Income Securities and non-USD denominated Fixed Income Securities.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Dollar High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Dollar Short Duration Bond Fund

The US Dollar Short Duration Bond Fund's investment objective is to provide an attractive rate of return, measured in US Dollars. The Fund will seek to achieve its investment objective by investing primarily in high quality USD denominated Fixed Income Securities, and for the avoidance of doubt including asset-backed securities (including mortgage-backed securities). The weighted average duration of the Fund will normally be less than 1 year, although this may be higher under certain market conditions.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the

Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs, The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- · manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity. The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on www.morganstanleyinvestmentfunds.com and on

Investments that are held by the Fund but, due to the application of the ESG criteria above, become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

Securities will be deemed to be high quality if at the time of purchase they are rated either "BBB-" or better by Standard & Poor's Corporation ("S&P") or "Baa3" or better by Moody's Investors Service, Inc. ("Moody's"), or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser. The Fund may continue to invest in such securities that are downgraded after purchase but may not make additional purchases of such securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any ETFs are eligible investments for UCITS funds.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Dollar Short Duration Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the short term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

US Dollar Short Duration High Yield Bond Fund

The US Dollar Short Duration High Yield Bond Fund's investment objective is to provide an attractive rate of total return, measured in USD. The Fund will seek to achieve its investment objective by investing primarily in high yield and unrated USD denominated High Yield Fixed Income Securities, including without limitation those issued by governments, agencies and corporations located in emerging markets and for the avoidance of doubt and subject to applicable law, including asset-backed securities, loan participations and loan assignments, to the extent that these instruments are securitised.

The Investment Adviser applies proprietary assessment and scoring methodologies that are bespoke to Fixed Income Securities the Fund may invest in, focussed on corporate, sovereign and securitized issuance. Additionally as part of the Investment Adviser's bottom-up, fundamental research process, and in its engagements with issuers, the Investment Adviser incorporates an assessment of sustainability-related risks and opportunities into the assessment process to determine impacts on credit fundamentals, implications for valuation and spreads, and any material aspects that may affect the trading technicalities of the Fixed Income Securities. These criteria may include, but are not limited to ESG themes such as decarbonisation & climate risk, circular economy & waste reduction, diverse & inclusive business and decent work & resilient jobs, The Investment Adviser also deploys a proprietary assessment framework for labelled sustainable bonds, through which the robustness, impact and transparency of such instruments are evaluated.

Investments shall not knowingly include any company whose business activity involves the following:

- manufacturing or production of controversial weapons;
- manufacturing or production of civilian firearms; and
- manufacturing or production of tobacco.

With respect to fossil fuel activities specifically, the Investment Adviser may engage company management around the topics of decarbonisation and climate risk, as well as corporate governance practices and what it deems to be materially important other environmental and/or social issues facing a company. Investments shall not knowingly include any company whose business activity involves the mining and extraction of thermal coal, where the company derives >5% revenue from such business activity.

The exception to the above is that the Fund may invest in labelled sustainable bonds, or other debt securities that are issued to raise capital specifically for climate-related or environmental projects, which are issued by issuers that would otherwise be subject to this fossil fuel exclusion, so long as it has determined that the objectives of such instruments are consistent with a reduction by the issuer in its carbon emissions. Investment in such instruments will be subject to diligence by the Investment Adviser.

In addition to the above sectoral exclusions, the Investment Adviser monitors business practices on an ongoing basis, through data on ESG controversies and standards screening sourced from third party providers. The Investment Adviser will consider controversy cases that it views as being very severe based on ratings by relevant ESG data providers, and failures to comply with the UN Global Compact or the ILO fundamental principles, although such incidents will not automatically result in exclusion from the portfolio.

Further to the above, the Investment Adviser may, in its discretion, elect to apply additional ESG-related investment restrictions over time that it believes are consistent with its investment objectives. Such additional restrictions will be disclosed as they are implemented on www.morganstanleyinvestmentfunds.com and on www.morganstanley.com/im.

Investments that are held by the Fund but, due to the application of the ESG criteria above, become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund. The Investment Adviser uses third-party data and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores. With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The weighted average duration of the Fund will normally be less than 3 years, although this may be higher under certain market conditions.

High Yield Fixed Income Security means a Fixed Income Security that has a rating less than "BBB-" by S&P or "Baa3" by Moody's or similarly rated by another internationally recognised rating service or determined to be of similar creditworthiness by the Investment Adviser.

The Fund may, on an ancillary basis, invest in:

- Higher quality Fixed Income Securities than High Yield Fixed Income Securities; and
- Non-USD denominated Fixed Income Securities.

The Fund may invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds and open-ended ETFs provided that any such ETFs are eligible investments for UCITS funds.

The Fund may invest no more than 20% of its assets in Contingent Convertible Instruments.

The Fund is actively managed and is not designed to track a benchmark. The Fund's performance is measured against a benchmark as detailed in the Fund's key investor information document.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the US Dollar Short Duration High Yield Bond Fund's investment objective it may be appropriate for investors who:

- seek to invest in Fixed Income Securities;
- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

ASSET ALLOCATION FUNDS

The following Asset Allocation Funds have the following objectives:

Global Balanced Defensive Fund

The Global Balanced Defensive Fund's investment objective is to provide an attractive level of total return, measured in Euro, through investing primarily in a portfolio of transferable securities and money market instruments globally and through the use of financial derivative instruments. The Fund is managed within a clearly-defined, risk-controlled framework, actively managing the total portfolio risk, by managing the portfolio volatility level in line with a predefined target volatility below that of the Global Balanced Fund. The Investment Adviser dynamically adjust the portfolio positioning in a balanced mix of equity, fixed income, Cash Equivalents, and commodity-linked securities, aiming to maintain a stable risk profile using an ex-ante approach.

The Fund will implement top-down, tactical views on global asset classes across (1) equity including closed-end Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents.

The Investment Adviser's tactical view will be implemented by accessing the following investment universe:

- Primary investments: equity securities, including closed-end Real Estate Investment Trusts (REITS); Fixed Income Securities (including investment grade, non-investment grade and unrated); and Cash Equivalents. This investment may be direct, or via derivatives. The Fund may invest in financial derivative instruments to achieve its investment objective and derivatives may be used for both investment and hedging purposes. These derivative instruments may include but are not limited to futures, options, warrants, contracts for difference, forward contracts on financial instruments and options on such contracts, credit linked instruments and swap contracts, whether traded on or off exchange.
- Ancillary investments: other transferable securities or, to a limited extent, to collective investment schemes
 including the Company's Funds and ETFs as well as China A-Shares via Stock Connect. The Fund may
 invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser anticipates it will predominantly employ the following investment strategies:

Global asset allocation: investing in securities in line with individual asset class indices. The Investment
Adviser will select the indices which will be used to make tactical views globally across (1) equity securities
including Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents.
The Investment Adviser may use on or off exchange traded derivatives such as futures or swaps to gain
diversified and/or more efficient exposure to specific global markets where due to the intended allocation

- size or due to inefficiencies in transacting directly in the underlying securities, investing via derivatives is a preferable investment option in the Investment Adviser's opinion.
- ESG screening: in addition, the Investment Adviser may consider ESG factors on a bottom-up basis; where investing directly in transferable securities, the Investment Adviser may consider the relative ranking of issuers for ESG factors as part of assessing the potential for return and as part of managing total portfolio risk

The Investment Adviser may use additional investment strategies such as:

- Commodity-linked investments: the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) and/or commodity linked notes.
- Use of derivatives for efficient portfolio management: the Fund may also invest in Cash Equivalents, warrants, exchange traded and over-the-counter options, and other derivatives for efficient portfolio management (including hedging) purposes as well as for the investment purposes as outlined above.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Balanced Defensive Fund's investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek a balanced investment that focuses on those asset classes the Investment Adviser finds tactically most attractive;
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Balanced Fund

The Global Balanced Fund's investment objective is to provide an attractive level of total return, measured in Euro, by investing in a broad and balanced asset mix of both equity and Fixed Income Securities while incorporating ESG considerations. The Fund is managed within a clearly-defined, risk-controlled framework, actively managing the total portfolio risk, by managing the portfolio volatility level in line with a pre-defined target volatility. The Investment Adviser dynamically adjust the portfolio positioning in a balanced mix of equity, fixed income, Cash Equivalents, and commodity-linked securities, aiming to maintain a stable risk profile using an ex-ante approach.

The Investment Adviser will implement a flexible top-down driven approach to identify investment opportunities across the permissible investment universe.

Sustainability factors are considered by the Investment Adviser during both the investment and research process to limit exposure to Sustainability Risks and seek out investment opportunities. These criteria may include, but are not limited to, climate change, carbon emissions, water scarcity, waste management, biodiversity, labour management, gender diversity, health & safety, product safety, data privacy & security, executive remuneration, board independence, shareholder rights, and bribery and corruption.

In relation to this Fund, "Sustainable" means that the Investment Adviser, in its discretion, integrates ESG considerations in its investment decision-making as further detailed below. The Investment Adviser may engage company management around corporate governance practices as well as what it deems to be materially important environmental and/or social issues facing a company in order to ensure that the target companies follow good governance practices.

The Investment Adviser shall not knowingly include the following companies in the investment universe, companies:

- associated with severe ESG controversies;
- involved with controversial weapons (including civilian firearms);
- involved in gambling;
- associated with tobacco; and
- that derive revenue from thermal coal and oil sands.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanley.com/im). The exclusions (which comprise, but are not limited to, the above exclusions) are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG

controversies analysis and business involvement research. The exclusions criteria is applied to all direct investments within the Fund. The exclusion criteria will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusion criteria is subject to periodic review and any changes will be reflected in the exclusion policy document.

Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Investment Adviser integrates ESG data including ratings and scores from third parties when constructing security baskets to implement asset class views, subject to a tracking error and other constraints. The Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Investment Adviser tilts the portfolio of the Fund towards high ESG performing stocks, and aims to improve the aggregated fund ESG performance versus that of the underlying equity index. In addition to ESG factors, the Investment Adviser treats climate change as a separate consideration, and also applies an additional climate change score during the tilting process.

The Investment Adviser creates – for all 11 GICS sectors within each regional equity market – individual baskets that tilt towards the top quartile ESG and low-carbon transition leaders.

The Fund's primary investment universe includes equity securities, including eligible closed-end Real Estate Investment Trusts (REITS); Fixed Income Securities (including investment grade, non-investment grade and unrated); and Cash Equivalents. This investment may be direct, or via derivatives. The Fund may invest in financial derivative instruments for both investment and hedging purposes. These derivative instruments may include but are not limited to futures, options, warrants, contracts for difference, forward contracts on financial instruments and options on such contracts, credit linked instruments and swap contracts, whether traded on or off exchange.

The Fund's ancillary investment universe may also invest to a limited extent in units/shares of other collective investment schemes including the Company's Funds and open-ended ETFs, provided that any ETFs are eligible investments for UCITS funds. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser may use additional investment strategies such as commodity-linked investments which qualify as transferable securities in accordance with article 1(34) of the 2010 Law and article 2 of the Grand-Ducal Regulation of 8 February 2008. The Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) and/or commodity linked notes.

The Fund may also invest in Cash Equivalents, warrants, exchange traded and over-the-counter options, and other derivatives for efficient portfolio management (including hedging) purposes as well as for the investment purposes as outlined above.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Balanced Fund's investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek a balanced investment that focuses on those asset classes the Investment Adviser finds tactically most attractive:
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Balanced Income Fund

The Global Balanced Income Fund's investment objective is to provide regular income and capital appreciation, measured in Euro, through investing primarily in a portfolio of transferable securities and money market instruments globally and through the use of financial derivative instruments. The Fund is managed within a clearly-defined, risk-controlled framework, actively managing the total portfolio risk, by managing the portfolio volatility level in line with a pre-defined target volatility. The Investment Adviser dynamically adjust the portfolio positioning in a balanced mix of equity, fixed income, Cash Equivalents, and commodity-linked securities, aiming to maintain a stable risk profile using an ex-ante approach.

The Fund will implement top-down, tactical views on global asset classes across (1) equity including closed-end Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents. The Investment Adviser's tactical view will be implemented by accessing the following investment universe:

- Primary investments: equity securities, including closed-end Real Estate Investment Trusts (REITS); Fixed Income Securities (including investment grade, non-investment grade and unrated); and Cash Equivalents. This investment may be direct, or via derivatives. The Fund may invest in financial derivative instruments to achieve its investment objective and derivatives may be used for both investment and hedging purposes. These derivative instruments may include but are not limited to futures, options, warrants, contracts for difference, forward contracts on financial instruments and options on such contracts, credit linked instruments and swap contracts, whether traded on or off exchange.
- Ancillary investments: other transferable securities or, to a limited extent, to UCIs including the Company's
 Funds and ETFs as well as China A-Shares via Stock Connect. The Fund may invest up to 10% of its net
 assets in China A-Shares via Stock Connect.

Investment in equity securities will not be less than 10% of the Fund's net asset value, nor more than 90% of the Fund's net asset value.

Investment in non-investment grade bonds will not exceed 30% of the Fund's net asset value.

The Investment Adviser anticipates it will predominantly employ the following investment strategies:

- Global asset allocation: investing in securities in line with individual asset class indices. The Investment Adviser will select the indices which will be used to make tactical views globally across (1) equity securities including Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents. The Investment Adviser may use on or off exchange traded derivatives such as futures or swaps to gain diversified and/or more efficient exposure to specific global markets where due to the intended allocation size or due to inefficiencies in transacting directly in the underlying securities, investing via derivatives is a preferable investment option in the Investment Adviser's opinion.
- Income generation: in addition to purchasing income generating transferable securities, additional income will be sought through the receipt of premiums as a result of writing on or off exchange traded options, which may be written on indices; single securities; or currencies.

The Investment Adviser may use additional investment strategies such as:

- Commodity-linked investments: the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) and/or commodity linked notes.
- Use of derivatives for efficient portfolio management: the Fund may also invest in cash warrants, exchange traded and over-the-counter options, and other derivatives for efficient portfolio management (including hedging) purposes as well as for the investment purposes as outlined above.
- Use of open and closed ended UCIs, including the Company's Funds and ETFs; the Investment Adviser
 would primarily use UCIs and ETFs to gain diversified and/or more efficient exposure to specific global
 markets where due to the intended allocation size or due to inefficiencies in transacting directly in the
 underlying securities an UCI or ETF is a preferable investment option in the Investment Adviser's opinion.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Balanced Income Fund's investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";

accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Balanced Risk Control Fund of Funds

The Global Balanced Risk Control Fund of Funds' investment objective is to invest in units/shares of other collective investment schemes to provide an attractive level of total return, measured in Euro, while actively managing total portfolio risk. The Fund is managed within a clearly-defined, risk-controlled framework, actively managing the volatility level in line with a pre-defined target volatility. The Investment Adviser adjust the portfolio positioning in a balanced mix of equity, fixed income, Cash Equivalents, and commodity-linked securities, aiming to maintain a stable risk profile using an ex-ante approach. The Fund will implement tactical views on global fixed income and equity asset classes by investing primarily in undertakings for collective investments ("UCI") including the Company's Funds, ETFs, futures, investment funds managed by the Investment Adviser or its affiliates and other collective investment schemes.

On an ancillary basis and with a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) and/or commodity linked notes, and may also invest in Cash Equivalents, warrants, exchange traded and over-the-counter options, and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also invest on an ancillary basis in equity securities, including eligible closed-end Real Estate Investment Trusts (REITS); Fixed Income Securities (including investment grade, non-investment grade and unrated); China A-Shares via Stock Connect. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Balanced Risk Control Fund of Funds' investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek income whether in the form of capital appreciation or distributions, as outlined in "Dividend Policy";
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Balanced Sustainable Fund

The Global Balanced Sustainable Fund's investment objective is to provide an attractive level of total return, measured in Euro, by investing in a broad and balanced asset mix of both equity and Fixed Income Securities while incorporating ESG considerations and taking into account the long-term global warming objectives of the Paris Agreement. The Fund is managed within a clearly-defined, risk-controlled framework, actively managing the volatility level in line with a pre-defined target volatility. The managers adjust the portfolio positioning in a balanced mix of equity, fixed income, Cash Equivalents, and commodity-linked securities, aiming to maintain a stable risk profile using an ex-ante approach.

The Investment Adviser will implement a flexible top-down driven approach to identify investment opportunities across the permissible investment universe.

Sustainability factors are considered by the Investment Adviser during both the investment and research process to limit exposure to Sustainability Risks and seek out investment opportunities. These criteria may include, but are not limited to, climate change, carbon emissions, water scarcity, waste management, biodiversity, labour management, gender diversity, health & safety, product safety, data privacy & security, executive remuneration, board independence, shareholder rights, and bribery and corruption.

In relation to this Fund, "Sustainable" means that the Investment Adviser, in its discretion, integrates ESG considerations in its investment decision-making as further detailed below. The Investment Adviser may engage company management around corporate governance practices as well as what it deems to be materially important environmental or social issues facing a company in order to ensure that the target companies follow good governance practices.

The Investment Adviser shall not knowingly include the following companies in the investment universe companies:

- associated with severe ESG controversies;
- involved with controversial weapons (including civilian firearms);
- involved in gambling;
- associated with tobacco;
- that derive revenue from thermal coal and oil sands.

The details of the above exclusions can be found in the Fund's exclusion policy which is available on the Company's website (www.morganstanley.com/im). The exclusions (which comprise, but are not limited to, the above exclusions) are determined by the Investment Adviser's own proprietary analysis rather than the reliance on third party analysis. However, the analysis may be supported by third party ESG controversies analysis and business involvement research. The exclusions criteria is applied to all direct investments within the Fund. The exclusion criteria will not be applied to investments in which the Investment Adviser does not have direct control of the underlying holdings, for example collective investment schemes or open-ended ETFs. The exclusion criteria is subject to periodic review and any changes will be reflected in the exclusion policy document.

Investments that are held by the Fund but become restricted after they are acquired for the Fund will be sold. Such sales will take place over a time period to be determined by the Investment Adviser, taking into account the best interests of the Shareholders of the Fund.

The Investment Adviser integrates ESG rating scores data from third parties when constructing security baskets to implement asset class views, subject to a tracking error and other constraints. The Investment Adviser tilts the portfolio of the Fund towards high ESG performing stocks, and aims to improve the aggregated fund ESG performance versus that of the underlying equity index. In addition to ESG factors, the Investment Adviser treats climate change as a separate consideration, and also applies an additional climate change score during the tilting process.

The Fund will take into account the long-term global warming objectives of the Paris Agreement. To reach this environmental objective the Investment Adviser's ESG integration process has two distinct climate tilts: 1) stock-specific and 2) sector-specific.

For the stock-specific, the Investment Adviser creates – for all 11 GICS sectors within each regional equity market – individual baskets that tilt towards the top quartile ESG and low-carbon transition leaders. For the sector-specific, the Investment Adviser aggregates the sector baskets subject to an overall carbon budget, aligned with a science-based climate trajectory subject to tracking error.

The Investment Adviser will also allocate 5% to 30% of the Fund for investment in assets with a measurable positive social or environmental impact (based on the Investment Adviser's methodology), alongside a competitive long term financial return. These investments can be held directly through equity or Fixed Income Securities, or indirectly through collective investment schemes.

The aforementioned ESG criteria should result in a reduction of the investible universe (as described below) in a significantly engaging manner, and the Investment Adviser will use its best efforts to ensure that at least 90% of the issuers in the portfolio are assessed against these ESG considerations.

As noted above, the Investment Adviser uses third-party data and ESG scores and in some cases data on specific issuers, ESG themes or the exclusions noted above may not be available and/or may be estimated by the Investment Adviser using internal methodologies or reasonable estimates. The methodologies used by different data providers may also vary and may result in different scores.

The Fund's Primary investment universe includes equity securities, including eligible closed-end Real Estate Investment Trusts (REITS); Fixed Income Securities (including investment grade, non-investment grade and unrated); and Cash Equivalents. This investment may be direct, or via derivatives. The Fund may invest in financial derivative instruments for both investment and hedging purposes, including derivative instruments that may not, meet the ESG criteria noted above. These derivative instruments may include but are not limited to futures, options, warrants, contracts for difference, forward contracts on financial instruments and options on such contracts, credit linked instruments and swap contracts, whether traded on or off exchange.

The Fund's ancillary investment universe may also invest to a limited extent in units/shares of other collective investment schemes, including the Company's Funds, and open-ended ETFs, provided that any ETFs are eligible investments for UCITS funds. The Fund may invest up to 10% of its net assets in China A-Shares via Stock Connect.

The Investment Adviser may use additional investment strategies such as Commodity-linked investments which qualify as transferable securities in accordance with article 1(34) of the 2010 Law and article 2 of the Grand-Ducal Regulation of 8 February 2008. The Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) and/or commodity linked notes.

The Fund may also invest in Cash Equivalents, warrants, exchange traded and over-the-counter options, and other derivatives for efficient portfolio management (including hedging) purposes as well as for the investment purposes as outlined above.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

In light of the Taxonomy Regulation, investors' attention is drawn to the fact that the "Do No Significant Harm" principle does not apply to any of the investments of the Fund, since the investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities within the meaning of the Taxonomy Regulation. Therefore the Fund's alignment with the Taxonomy Regulation is 0%.

Profile of the typical investor

In light of the Global Balanced Sustainable Fund's investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek a balanced investment that focuses on those asset classes the Investment Adviser finds tactically most attractive;
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Global Multi-Asset Opportunities Fund

The Global Multi-Asset Opportunities Fund's investment objective is to provide an absolute return, measured in Euros, while actively managing total portfolio risk. The Investment Adviser seeks to manage downside risk and targets below market volatility.

The Fund will seek to achieve its investment objective by investing in baskets of stocks, ETFs, bonds and derivatives.

The Fund may take both long and short positions, either directly or (specifically in the case of short positions) through the use of derivatives as described below, in a diversified range of equity and equity related securities of any market capitalisation, Fixed Income Securities, currencies and commodities. The Fund will only take indirect exposure to commodities, by investing into eligible structured products such as commodity-linked notes the underlying of which are commodity indices and/or sub-indices and/or derivative instruments referencing one or more commodity indices qualifying as financial indices under UCITS Regulations. When using such indices, the Fund must comply with the diversification limits set out in the UCITS Regulations: each component of a financial index may represent up to 20% of the index, except that in exceptional market circumstances one single component may represent up to 35% of the index.

With a view to enhancing returns and/or as part of the investment strategy, the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) make use of exchange traded and over-the-counter options, futures, swaps and other derivatives for investment or efficient portfolio management (including hedging) purposes.

The Fund may also make the following investments or use the following financial instruments:

- i. currency spot transactions, currency forward transactions, non-deliverable currency forward transactions and Money Market Funds, the assets of which may be managed by the Investment Adviser or any of its related, associated or affiliated companies; and/or
- ii. deposits, fixed or floating rate instruments (including but not limited to commercial paper), floating rate notes, certificates of deposit, debentures, asset-backed securities and government or corporate bonds, and Cash Equivalents; and/or
- iii. Fixed Income Securities acquired on the China Interbank Bond Market which will account for no more than 20% of the Fund's net assets; and/or
- iv. China A-Shares via Stock Connect which will account for no more than 10% of the Fund's net assets; and/or
- v. up to 10% of the Fund may be invested in accordance with paragraph 2.2. of Appendix A.

The Investment Adviser will utilize a top-down, global macroeconomic and thematic investment approach that focuses on asset class, sector, region and country selection, as opposed to individual security selection. The Fund's allocations will be based upon the Investment Adviser's evaluations and analyses, taking into account its fundamental investment research, which is primarily driven by quantitative macroeconomic and market data.

Investment decisions will be made without regard to any particular limit, subject to Appendix A, as to geographical location, sector, credit rating, maturity, currency denomination or market capitalization. The Fund may invest in any

country, including developing or emerging market countries. The Fund's investments may be denominated in Euros or in currencies other than the Euro.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

Taxonomy Regulation disclosure

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Global Multi-Asset Opportunities Fund's investment objective it may be appropriate for investors who:

- seek a return over the medium term;
- seek a balanced investment that focuses on those asset classes the Investment Adviser finds tactically most attractive;
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

Multi-Asset Risk Control Fund

The Multi-Asset Risk Control Fund's investment objective is to provide an attractive level of total return, measured in USD, through investing primarily in a portfolio of transferable securities and money market instruments globally and through the use of financial derivative instruments.

The Fund will implement top-down, tactical views on global asset classes across (1) equity including closed-end Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents.

The Investment Adviser's tactical view will be implemented by accessing the following investment universe:

- Primary investments: equity securities, including closed-end Real Estate Investment Trusts (REITS) which are eligible for investment by a UCITS under the 2010 Law; Fixed Income Securities (including investment grade, non-investment grade and unrated); and Cash Equivalents. This investment may be direct, or via derivatives. The Fund may invest in financial derivative instruments to achieve its investment objective and derivatives may be used for both investment and hedging purposes. These derivative instruments may include but are not limited to futures, options, warrants, contracts for difference, forward contracts on financial instruments and options on such contracts and swap contracts, whether traded on or off exchange.
- Ancillary investments: other transferable securities or, to a limited extent, to collective investment schemes
 including the Company's Funds and open-ended ETFs, which are eligible investments for UCITS under
 the 2010 Law, as well as China A-Shares via Stock Connect. The Fund may invest up to 10% of its net
 assets in China A-Shares via Stock Connect.

The Investment Adviser anticipates it will predominantly employ the following investment strategies:

- Global asset allocation: investing in securities in line with individual asset class indices. The Investment Adviser will select the indices which will be used to make tactical views globally across (1) equity securities including Real Estate Investment Trusts (REITS); (2) Fixed Income Securities; and (3) Cash Equivalents. The Investment Adviser may use on or off exchange traded derivatives such as futures or swaps to gain diversified and/or more efficient exposure to specific global markets where due to the intended allocation size or due to inefficiencies in transacting directly in the underlying securities, investing via derivatives is a preferable investment option in the Investment Adviser's opinion.
- ESG screening: in addition, the Investment Adviser may consider ESG factors on a bottom-up basis; where investing directly in transferable securities, the Investment Adviser may consider the relative ranking of issuers for ESG factors as part of assessing the potential for return and as part of managing total portfolio risk.

The Investment Adviser may use additional investment strategies such as:

- Commodity-linked investments: the Fund may (in accordance with the investment powers and restrictions set out in Appendix A) implement tactical views on commodities via exchange traded commodities (ETCs) which qualify as transferable securities in accordance with article 1(34) of the 2010 Law and article 2 of the Grand-Ducal Regulation of 8 February 2008.
- Use of derivatives for efficient portfolio management: the Fund may also invest in Cash Equivalents, warrants, exchange traded and over-the counter options, and other derivatives for efficient portfolio management (including hedging) purposes as well as for the investment purposes as outlined above.

The Fund's investments may result in exposure to emerging markets and to higher yielding securities. The Fund aims to manage total portfolio risk by managing the volatility level of the portfolio.

The Fund is actively managed and the composition of the portfolio holdings is not constrained by any benchmarks.

The investments underlying the Fund do not take into account the EU criteria for environmentally sustainable economic activities.

Profile of the typical investor

In light of the Multi-Asset Risk Control Fund's investment objective it may be appropriate for investors who:

- seek capital appreciation over the medium term;
- seek a balanced investment that focuses on those asset classes the Investment Adviser finds tactically most attractive:
- accept the risks associated with this type of investment, as set out in Section 1.5 "Risk Factors".

1.3 Methodology for Calculating the Global Exposure

The Management Company has implemented a risk management process for each Fund in order to comply with its obligations under the UCITS Directive. The Management Company may calculate global exposure for the Fund using the commitment approach, relative VaR or absolute VaR.

The selection of the appropriate methodology for calculating global exposure is made by the Management Company based upon a consideration of the following factors:

- (i) whether the Fund engages in complex investment strategies which represent a significant part of the Fund's investment policy;
- (ii) whether the Fund has a significant exposure to exotic derivatives; and/or
- (iii) whether the commitment approach adequately captures the market risk of the Fund's portfolio.

The selection of relative VaR or absolute VaR will depend on whether the Fund has a leverage free Reference Portfolio which reflects its investment strategy. The Reference Portfolios adopted by the Funds are standard, widely-used industry indices.

Classification of a Fund will depend on a consideration of each of these factors and the fact that a Fund is authorised to use derivative instruments for investment purposes will not automatically, in isolation, mean that the global exposure of that Fund will be calculated using either relative or absolute VaR approach.

The table below sets out the methodology which the Management Company has adopted for each Fund in order to calculate the global exposure and the expected leverage for the Funds using the VaR approach, as defined by the UCITS Regulations.

Expected Leverage is provided using the 'sum of the notionals of the derivatives used' methodology as per the ESMA Guidelines 10-788.

Expected leverage is provided at a Fund level, however the Fund's actual level of leverage may be higher or lower than the disclosed expected leverage in the table below. Specific hedged share classes may have higher or lower levels of expected leverage than indicated at the Fund level.

Expected leverage is not a regulatory limit on the Fund and there may be no action as a result of the actual leverage being higher or lower than the disclosed expected leverage.

Derivatives usage is consistent with the investment objective and the risk profile of the Fund. The 'sum of notionals' calculation does not allow netting or hedging. The expected leverage using the 'sum of notionals' calculation may not reflect the investment risk associated with the derivative positions held in the Fund.

Additional information about the realized range of leverage employed by the relevant Funds can be found in the Annual Report.

Fund Name	Global Exposure Methodology	Reference Portfolio	Expected Gross Leverage		
American Resilience Fund	Commitment	N/A	N/A		
Asia Opportunity Fund	Commitment	N/A	N/A		
Asian Property Fund	Commitment	N/A	N/A		
Calvert Sustainable Climate Aligned Fund	Commitment	N/A	N/A		
Calvert Sustainable Climate Transition Fund	Commitment	N/A	N/A		

Fund Name	Global Exposure Methodology	Reference Portfolio	Expected Gross Leverage					
Calvert Sustainable Developed Europe Equity Select Fund	Commitment	N/A	N/A					
Calvert Sustainable Developed Markets Equity Select Fund	Commitment	N/A	N/A					
Calvert Sustainable Diversity, Equity and Inclusion Fund	Commitment	N/A	N/A					
Calvert Sustainable Emerging Markets Equity Select Fund	Commitment	N/A	N/A					
Calvert Sustainable Global Equity Engagement Fund	Commitment	N/A	N/A					
Calvert Sustainable Global Green Bond Fund	Relative VaR	ICE BofA Green Bond Index	100%					
Calvert Sustainable US Equity Select Fund	Commitment	N/A	N/A					
China A-shares Fund	Commitment	N/A	N/A					
China Equity Fund	Commitment	N/A	N/A					
Counterpoint Global Fund	Commitment	N/A	N/A					
Developing Opportunity Fund	Commitment	N/A	N/A					
Emerging Leaders Equity Fund	Commitment	N/A	N/A					
		JP Morgan Corporate	- 1/ - 2					
Emerging Markets Corporate Debt Fund	Relative VaR	Emerging Markets Bond Index – Broad Diversified	100%					
Emerging Markets Debt Fund	Relative VaR	JP Morgan Emerging Market Bond Index Global Diversified	100%					
Emerging Markets Debt Opportunities Fund	Relative VaR	J.P. Morgan Emerging Markets Bond Hard Currency/Local Currency 50- 50 Index	150%					
Emerging Markets Domestic Debt Fund	Relative VaR	JP Morgan Government Bond Index- Emerging Markets Global Diversified Index	100%					
Emerging Markets Fixed Income Opportunities Fund	Relative VaR	JP Morgan Emerging Markets Blended Index – Equally Weighted	100%					
Emerging Markets Local Income Fund	Relative VaR	J.P. Morgan Government Bond Index: Emerging Market (JPM GBI-EM) Global Diversified (unhedged) Index	300%					
Euro Bond Fund	Relative VaR	Bloomberg Euro Aggregate A- or Better Index	100%					
Euro Corporate Bond Fund	Relative VaR	Bloomberg Euro-Aggregate Corporates Index	100%					
Euro Corporate Bond – Duration Hedged Fund	Absolute VaR	N/A	150%					
Euro Strategic Bond Fund	Relative VaR	Bloomberg Euro Aggregate Index	100%					
Europe Opportunity Fund	Commitment	N/A	N/A					
European Fixed Income Opportunities Fund	Absolute VaR	N/A	160%					
European High Yield Bond Fund	Relative VaR	ICE BofA ML European Currency High Yield 3% Constrained Ex-Sub Financials Index	100%					
European Property Fund	Commitment	N/A	N/A					
Floating Rate ABS Fund	Absolute VaR	N/A	160%					
Frontier Markets Local Debt Fund	Absolute VaR Absolute VaR	N/A N/A	150%					
Global Asset Backed Securities Fund	Absolute VaR	N/A	160%					
Global Balanced Defensive Fund	Absolute VaR	N/A	100%					
	158							

Fund Name	Global Exposure Methodology	Reference Portfolio	Expected Gross Leverage				
Global Balanced Fund	Absolute VaR	N/A	100%				
Global Balanced Income Fund	Absolute VaR	N/A	100%				
Global Balanced Risk Control Fund of Funds	Absolute VaR	N/A	50%				
Global Balanced Sustainable Fund	Absolute VaR	N/A	100%				
Global Bond Fund	Relative VaR	Bloomberg Global Aggregate Index	100%				
Global Brands Equity Income Fund	Relative VaR	MSCI World Index	200%				
Global Brands Fund	Commitment	N/A	N/A				
Global Convertible Bond Fund	Relative VaR	Refinitiv Global Convertible Index Global Focus Hedged USD	100%				
Global Credit Fund	Relative VaR	Bloomberg Global Aggregate Corporate Index	160%				
Global Credit Opportunities Fund	Absolute VaR	N/A	160%				
Global Core Equity Fund	Commitment	N/A	N/A				
Global Endurance Fund	Commitment	N/A	N/A				
Global Fixed Income Opportunities Fund	Absolute VaR	N/A	160%				
Global Focus Property Fund	Commitment	N/A	N/A				
Global High Yield Bond Fund	Relative VaR	Bloomberg Global High Yield - Corporate Index	100%				
Global Infrastructure Fund	Commitment	N/A	N/A				
Global Insight Fund	Commitment	N/A	N/A				
Global Macro Fund	Absolute VaR	N/A	450%				
Global Multi-Asset Opportunities Fund	Absolute VaR	N/A	500%				
Global Opportunity Fund	Commitment	N/A	N/A				
Global Permanence Fund	Commitment	N/A	N/A				
Global Property Fund	Commitment	N/A	N/A				
Global Quality Fund	Commitment	N/A	N/A				
Global Sustain Fund	Commitment	N/A	N/A				
Indian Equity Fund	Commitment	N/A	N/A				
International Equity (ex US) Fund	Commitment	N/A	N/A				
Japanese Equity Fund	Commitment	N/A	N/A				
Japanese Small Cap Equity Fund	Commitment	N/A	N/A				
Latin American Equity Fund	Commitment	N/A	N/A				
NextGen Emerging Markets Fund	Commitment	N/A	N/A				
Multi-Asset Risk Control Fund	Absolute VaR	N/A	100%				
Parametric Emerging Markets Fund	Commitment	N/A	N/A				
Parametric Global Defensive Equity Fund	Commitment	N/A	90%				
Saudi Equity Fund	Commitment	N/A	N/A				
Short Duration US Government Income Fund	Absolute VaR	N/A	125%				
Short Maturity Euro Bond Fund	Commitment	N/A	N/A				
Short Maturity Euro Corporate Bond Fund	Relative VaR	Bloomberg Euro Aggregate Corporates 1-3 Year Index	110%				
Sustainable Asia Equity Fund	Commitment	N/A	N/A				
Sustainable Emerging Markets Equity Fund	Commitment	N/A	N/A				
Sustainable Euro Corporate	Relative VaR	Bloomberg Euro-Aggregate	100%				

Fund Name	Global Exposure Methodology	Reference Portfolio	Expected Gross Leverage			
Sustainable Euro Strategic Bond Fund	Relative VaR	Bloomberg Euro Aggregate Index	130%			
Sustainable European High Yield Bond Fund	Commitment	N/A	25%			
Sustainable Global Fixed Income Opportunities Fund	Absolute VaR	N/A	160%			
Sustainable Global High Yield Bond Fund	Commitment	N/A	25%			
Sustainable US High Yield Bond Fund	Commitment	N/A	10%			
US Advantage Fund	Commitment	N/A	N/A			
US Core Equity Fund	Commitment	N/A	N/A			
US Dollar Corporate Bond Fund	Relative VaR	Bloomberg US Corporate Index	110%			
US Dollar High Yield Bond Fund	Relative VaR	Bloomberg US Corporate High Yield Index	100%			
US Dollar Short Duration Bond Fund	Commitment	N/A	N/A			
US Dollar Short Duration High Yield Bond Fund	Relative VaR	Bloomberg US High Yield 1- 5 Year Cash Pay 2% Issuer Capped Index	100%			
US Focus Property Fund	Commitment	N/A	N/A			
US Growth Fund	Commitment	N/A	N/A			
US Insight Fund	Commitment	N/A	N/A			
US Permanence Fund	Commitment	N/A	N/A			
US Property Fund	Commitment	N/A	N/A			
US Value Fund	Commitment	N/A	N/A			
Vitality Fund	Commitment	N/A	N/A			

1.4 Financial Indices

Funds may, in accordance with their investment policy and the investment restrictions under Appendix A paragraph 2.6, invest in financial derivative instruments based on financial indices which are eligible in accordance with article 9 of the Grand-Ducal Regulation of 8 February 2008.

To the extent consistent with the prevailing investment policy, Funds may obtain exposure, in the discretion of the Investment Adviser, to indices which may not be named specifically in the investment policy, subject always to compliance with the ESMA guidelines 2014/937 on ETF and other UCITS issues. In particular, the Funds will not invest in financial derivative instruments based on financial indices with a daily or intra-day rebalancing frequency or in indices whose methodology for the selection and the re-balancing of their components is not based on a set of pre-determined rules and objective criteria.

1.5 Risk Factors

This Section of the Prospectus explains the risks that apply to the Funds – investors must read these risk considerations before investing in any of the Company's Funds.

The risk factors set out in Section 1.5.1 General Risk Factors may apply to all of the Company's Funds. The table in Section 1.5.2 sets out which of the risk factors set out the Specific Risk Factors section that may potentially also apply to the Company's Funds at the time of the issue of this Prospectus based on their respective investment objectives. Specific risk factors may apply to each Fund to varying degrees, and the exposure to specific risk factors will also vary over time. Additionally, risks may arise in the future which could not have been foreseen in advance. This Prospectus will be regularly updated to reflect any changes to the risk factors set out in Section 1.5.1 General Risk Factors and Section 1.5.2 Specific Risk Factors.

1.5.1. General Risk Factors

General Risk

Past performance is not a guide to future performance and the Funds should be regarded as medium to long-term investments. Investments in Funds are subject to market fluctuations and other risks inherent to investing in securities and other financial instruments. The price of the Shares can go down as well as up. An investor may not get back the

amount he has invested, particularly if Shares are redeemed soon after they are issued and the Shares have been subject to a Sales Charge, as further disclosed in Section 2.1 "Share Class Description" or transaction charge.

Investment Objective

There is no guarantee or representation that the investment objective of the Company or a Fund will be achieved. Depending on market conditions and the macroeconomic environment, it may become more difficult or even impossible to achieve investment objectives.

Market Risk

Investors may experience losses due to changes in the level of one or more market prices, rates, indices, or other market factors. Market risk cannot be eliminated through diversification, though it can be hedged against. Sources of market risk include, but are not limited to, recessions, political turmoil, changes in monetary policies, outbreak of disease epidemics or pandemics, etc.

Currency Risk

The Funds may invest in investments denominated in a number of different currencies other than the Reference Currency in which the Funds are denominated. Changes in foreign currency exchange rates between the Reference Currency and the currency in which the investments are denominated will cause the value of the investments expressed in the Reference Currency to differ.

Funds that have the ability to invest in overseas assets may be subject to currency volatility including currency devaluation. Currency movements may impact the value of the Fund's assets. The Funds may use derivatives to reduce this risk. However, certain market conditions may make it impossible or uneconomical to hedge against currency risk. The Company may in its discretion choose not to hedge against currency risk within the Funds.

Share Class Currency

Certain Share Classes of certain Funds may be denominated in a currency other than the Reference Currency of the relevant Funds. Therefore changes in foreign currency exchange rates between the Reference Currency and the currency in which these Share Classes are denominated (the "Share Class Currency") will cause the value of Shares held in such Funds to differ.

Liquidity Risk

Liquidity risk exists when some of the Funds' investments may be difficult to sell due to unforeseen economic or market conditions, such as the deterioration in the creditworthiness of an issuer. In case of a large redemption request, the Funds may consequently not be able to sell certain assets to meet the redemption requirement or may not be able to sell certain assets at levels close to current valuation price.

Counterparty Risk

The Funds may enter into transactions with counterparties (which could be a company, government or other institution), thereby exposing them to the counterparties' creditworthiness and their ability to perform and fulfil their financial obligations. There exists a risk that the obligation of such counterparties will not be satisfied. This risk may arise at any time the Funds' assets are deposited, extended, committed, invested or otherwise exposed through actual or implied contractual agreements. The weaker the financial strength of a counterparty, the greater the risk of that party failing to satisfy its obligations. The Net Asset Value of the Funds could be affected by any actual or anticipated breach of the party's obligations, while the income of the Fund would be affected only by an actual failure to pay, which is known as a default.

In addition, the Funds may enter into contracts with service providers and other third party contractors (the "Service Providers"). This risk means that in certain circumstances (including but not limited to force majeure events) the Service Providers may not be able to perform or fulfil their contractual obligations to the Funds. This could result in periods where the normal trading activity of the Funds may be affected or disrupted.

Depositary Risk

Country risk linked to the custody

The Management Company may decide from time to time to invest in a country where the Depositary has no correspondent. In such a case, the Depositary will have to identify and appoint after due diligence a local custodian. This process may take time and deprive in the meantime the Management Company of investment opportunities.

In the same manner, the Depositary shall assess on an ongoing basis the custody risk of the country where the Company's assets are safe-kept. The Depositary may identify from time to time a custody risk in a jurisdiction and

recommends to the Management Company to realize the investments immediately. In doing so, the price at which such assets will be sold may be lower than the price the Company would have received in normal circumstances, potentially affecting the performance of the relevant Funds.

Central Securities Depositaries

In accordance with the UCITS Directive, entrusting the custody of the Company's assets to the operator of a securities settlement system ("SSS") is not considered as a delegation by the Depositary and the depositary is exempted from the strict liability of restitution of assets. A central securities depositary ("CSD") being a legal person that operates a SSS and provides in addition other core services should not be considered as a delegate of the Depositary irrespective of the fact that the custody of the Company's assets have been entrusted to it. There is however some uncertainty around the meaning to be given to such exemption, the scope of which may be interpreted narrowly by some supervisory authorities, notably the European supervisory authorities.

Custody Risk

The Assets of the Company are entrusted to the Depositary for safekeeping and are identified in the Depositary's books as belonging to the Company. Securities held by the Depositary are segregated from other assets of the Depositary which mitigates but does not exclude the risk of non-restitution in case of bankruptcy of the Depositary. However, no such segregation applies to cash which increases the risk of non-restitution in case of bankruptcy.

The Depositary may not keep all the assets of the Company itself but may use a network of sub-custodians which are not always part of the same group of companies as the Depositary. Investors are exposed to the risk of bankruptcy of the sub-custodians in the same manner as they are to the risk of bankruptcy of the Depositary.

The Funds may invest in markets where custodial and/or settlement systems are not fully developed. The assets of the Funds that are traded in such markets and which have been entrusted to such sub-custodians may be exposed to risk in circumstances where the Depositary will have no liability.

Inflation/Deflation Risk

Inflation risk refers to the possibility of a reduction in the value of the income or assets as inflation decreases the value of money. The real value of a Fund's portfolio could decline as inflation increases. Deflation risk is the risk that prices throughout the economy may decline over time. Deflation may have an adverse effect on the creditworthiness of issuers and may make issuer default more likely, which may result in a decline in the value of a Fund's portfolio.

Regulatory Risk

The Funds are domiciled in Luxembourg and investors should note that all the regulatory protections provided by their local regulatory authorities may not apply. Additionally, some of the Funds may be registered in non-EU jurisdictions and, as a result, may be subject, without any notice to the Shareholders in the Funds concerned, to more restrictive regulatory regimes. In such cases the Funds will abide by these more restrictive requirements. This may prevent the Funds from making the fullest possible use of the investment limits. Regulators are authorised to take extraordinary actions in the event of market emergencies. The effect of any future regulatory action on the Funds could be substantial and adverse.

The withdrawal of the UK from the EU

As at the date of this Prospectus, the exit by the United Kingdom from the EU ("Brexit") has resulted in global economic and political uncertainty and it is unknown what the impact shall be on the economic or political environment of each of the United Kingdom and the EU.

On 29 March 2017, the UK's government gave notice of its intention to withdraw from the EU pursuant to Article 50 on the Treaty of the EU. On 31 January 2020 at 11 p.m. (London Time), the UK exited from the EU. On the basis of the agreement for an orderly withdrawal of the UK from the EU, the UK benefited from a transitionary period, pursuant to which all EU Treaties and EU legislation still applied to the UK. This transitionary period ended on 31 December 2020. Since the end of this transitionary period, the UK is considered a third country.

An agreement determines the terms of the UK's relationship with the EU, including the terms of trade between the UK and the EU, after such transitionary period. In addition, the UK is required to negotiate with other countries with which the UK previously traded on the basis of agreements concluded with the EU (having been members thereof).

The UK's exit from the EU may result in regulatory change for the UK since a significant portion of the UK regulatory regime is derived from EU directives and regulations. Such uncertainty could lead to a high degree of economic and market disruption and uncertainty. It is not possible to ascertain how long this period will last and the impact it will have within the EU markets, including market value and liquidity, as well as the assets held by the

Funds. Such conditions could have a material adverse effect on the business, financial condition, results of operations and prospects of the Company, the Management Company, the Investment Advisers, and other transaction parties. The Company, the Management Company, and the Investment Advisers cannot predict when political stability will return, or when the market conditions relating to the assets held by the Funds will stabilise.

MiFID II

The EU's second Markets in Financial Instruments Directive ("MiFID II"), laws and regulations introduced by Member States of the EU to implement MiFID II, and the EU's Markets in Financial Instruments Regulation ("MiFIR"), came into force on 3 January 2018. MiFID II and MiFIR imposed new regulatory obligations and costs on the Management Company and the Investment Advisers. This may have a negative impact on the Management Company, the Investment Advisers, the Company and/or the Funds.

In particular, MiFID II and MiFIR require certain standardised OTC derivatives to be executed on regulated trading venues. The overall impact of MiFID II and MiFIR on the Company remains highly uncertain and it is unclear how the OTC derivatives markets will adapt to these new regulatory regimes.

In addition, MiFID II introduces wider transparency regimes in respect of trading on EU trading venues and with EU counterparties. MiFID II extends the pre- and post-trade transparency regimes from equities traded on a regulated market to cover equity-like instruments such as DRs, ETFs and certificates that are traded on regulated trading venues as well as to cover non-equities such as bonds, structured finance products, emission allowances and derivatives. The increased transparency regime under MiFID II, together with the restrictions on the use of "dark pools" and other trading venues, will mean a wealth of new information relating to price discovery becoming available. Such increased transparency and price discovery may have macro effects on trading globally, which may have an adverse effect on the Net Asset Value.

Share Classes

Suspension of Share Class Dealing

Investors are reminded that in certain circumstances their right to redeem or switch Shares may be suspended (see Section 2.3 Redemption of Shares and Section 2.4 Conversion of Shares).

Currency Hedged Share Classes

The Management Company may decide from time to time for some or all of the Funds to issue Currency Hedged Share Classes.

Currency Hedged Share Classes utilise hedging strategies to seek to limit exposure to currency movements between a Fund's Reference Currency, Investment Currencies or Index Currencies and the Currency Hedged Share Class is denominated in. The hedging strategy of the Currency Hedged Share Class does not seek to eliminate all currency exposure. Exchange rate risk exists as a result of movements between the currency of denomination of the Currency Hedged Share Class and the valuation currencies of the assets in which the Fund invests where these currencies differ from the Reference currency of the Fund.

Such hedging strategies used by the Investment Adviser(s) (or any agent appointed by the Investment Adviser(s)) may not completely eliminate exposure to such currency movements. There can be no guarantee that hedging strategies will be successful. Mismatches may result between a Fund's currency position and the Currency Hedged Share Classes issued for that Fund. Investors should be aware that certain market events or circumstances could result in the Investment Adviser(s) no longer being able to perform hedging transactions or that such strategies may no longer be economically viable.

The use of hedging strategies may substantially limit Currency Hedged Share Class shareholders from benefiting if the currency of the Currency Hedged Share Class falls against a Fund's Reference Currency, Investment Currencies or Index Currencies. The costs of hedging and all gains/losses from hedging transactions are borne separately by the shareholders of the respective Currency Hedged Share Classes. Investors should also note that the hedging of Currency Hedged Share Classes is distinct from any hedging strategies that the Investment Adviser(s) may implement at the Fund level.

Non-deliverable Currencies

Several Emerging Markets, Frontier Markets and other non-developed markets' currencies are traded as cash settled, non-deliverable forwards, because they are either thinly traded or non-convertible. As such, where the Hedged Share Class Currency is non-deliverable the share class will be denominated, for subscription and redemption purposes, in a currency other than the Hedged Share Class Currency. For example, a Colombian Peso NAV Hedged Share Class may be subscribed or redeemed in US Dollar or Euro but the Fund's Reference Currency would be hedged to the Columbian Peso, notwithstanding it being denominated in US Dollar or Euro. Investors should note that additional

exchange rate risk exists as a result of the movements between the non-deliverable currency denomination of the Currency Hedged Share Class and the valuation currencies of the assets in which the Fund invests where these currencies differ from the Reference currency of the Fund and the currency in which the shareholder subscribes or redeems.

Renminbi ("RMB") Currency Risk

Since 2005, the RMB exchange rate is no longer pegged to the US dollar. RMB has now moved to a managed floating exchange rate based on market supply and demand with reference to a basket of foreign currencies. RMB exchange rate is also subject to exchange control policies. The daily trading price of the RMB against other major currencies in the inter-bank foreign exchange market is allowed to float within a narrow band around the central parity published by the People's Republic of China. As the exchange rates are influenced by government policy and market forces, the exchange rates for RMB against other currencies, including US dollar and HK dollar, are susceptible to movements based on external factors. Accordingly, the investment in Classes denominated in RMB may be adversely affected by the fluctuations in the exchange rate between RMB and other foreign currencies.

RMB is currently not freely convertible and RMB convertibility from offshore RMB (CNH) to onshore RMB (CNY) is a managed currency process subject to foreign exchange control policies of and restrictions imposed by the Chinese government. The value of CNH could differ, perhaps significantly, from that of CNY due to a number of factors including without limitation those foreign exchange control policies and repatriation restrictions.

Classes denominated in RMB, both hedged and unhedged, participate in the offshore RMB (CNH) market, which allows investors to freely transact RMB (CNH) outside of mainland China. Classes denominated in RMB will have no requirement to remit RMB (CNH) to onshore RMB (CNY). Non-RMB based investors (e.g. Hong Kong investors) in Classes denominated in RMB may have to convert HK dollar or other currency(ies) into RMB when investing in Classes denominated in RMB and subsequently convert the RMB redemption proceeds and/or dividend payment (if any) back to HK dollar or such other currency(ies). Investors will incur currency conversion costs and you may suffer losses depending on the exchange rate movements of RMB relative to HK dollar or such other currencies.

Even if the Fund aims to pay redemption monies and/ or distributions of RMB denominated Classes in RMB, the Management Company may, under extreme market conditions when there is not sufficient RMB for currency conversion and with the approval of the Directors, pay redemption monies and/or distributions in US dollar. There is also a risk that payment of redemption monies and/or distributions in RMB may be delayed when there is not sufficient amount of RMB for currency conversion for settlement of the redemption monies and distributions in a timely manner due to the exchange controls and restrictions applicable to RMB. In any event, redemption monies will be paid within one calendar month upon receipt of all properly completed documentation.

Discretionary Distributing Share Class

The Discretionary Distributing Share Class may pay distributions from capital or may charge all or part of the Fund's fees and expenses to the capital of the Fund. Dividends paid out of capital amount to a return or withdrawal of part of a Shareholder's original investment or from any capital gains attributable to that original investment. Such dividends may result in an immediate decrease of the Net Asset Value per Share. Where part or all of the fees and expenses of a Share Class is charged to the capital of that Share Class the result will be a reduction of the capital that the Fund has available for investment in the future and capital growth may be reduced. In periods of declining markets, the compounding effect of such dividend distributions may lead to a material erosion of the Share Class Net Asset Value.

Automatic Exchange of Information

Common Reporting Standards ("CRS")

The Company should be subject to the Standard for Automatic Exchange of Financial Account Information in Tax matters (the "Standard") and its Common Reporting Standard (the "CRS") as set out in the Luxembourg law dated 18 December 2015 implementing the CRS (the "CRS Law").

Under the terms of the CRS Law, the Company is likely to be treated as a Luxembourg Reporting Financial Institution.

Without prejudice to other applicable data protection provisions as set out in the Company documentation, since 30 June 2017, the Company has been required to annually report to the Luxembourg tax administration personal and financial information related, inter alia, to the identification of, holdings by and payments made to (i) certain investors as per the CRS Law (the "Reportable Persons") and (ii) Controlling Persons of certain non-financial entities ("NFEs") which are themselves Reportable Persons. This information, as exhaustively set out in Annex I of the CRS Law (the "CRS Information"), will include personal data related to the Reportable Persons.

The Company's ability to satisfy its reporting obligations under the CRS Law will depend on each investor providing the Company with the CRS Information, along with the required supporting documentary evidence. In this context, the investors are hereby informed that, as data controller, the Company will process the CRS Information for the purposes as set out in the CRS Law. The investors undertake to inform their Controlling Persons, if applicable, of the processing of their CRS Information by the Company.

The investors are further informed that the CRS Information related to Reportable Persons within the meaning of the CRS Law will be disclosed to the Luxembourg tax administration annually for the purposes set out in the CRS Law. The Luxembourg tax administration may in turn, acting as data controller, further disclose such information to foreign tax authorities.

Any investor that fails to comply with the Company's documentation requests may be charged, for non-compliance under the CRS Law, with any penalties imposed on the Company attributable to such investor's failure to provide the CRS Information and the Company may, in its sole discretion, redeem the Shares of such investor.

Taxation

Investors should note that the proceeds from the sale of securities in some markets or the receipt of any dividends or other income may be or may become subject to withholding or other taxes imposed by the authorities in that market. Tax and law practice in certain countries into which a Fund invests or may invest in the future may not be clearly established, may be subject to change or may be subject to change with retrospective effect. It is possible therefore that the Funds could become subject to additional taxation in such countries that is not anticipated either at the date of the Prospectus or when investments are made, valued or disposed of.

Foreign Account Tax Compliance ("FATCA")

The Foreign Account Tax Compliance provisions of the 2010 Hiring Incentives to Restore Employment Act ("HIRE Act") generally impose a new reporting and 30% withholding tax regime with respect to certain U.S. source income (including dividends and interest) and gross proceeds from the sale or other disposal of property that can produce U.S. source interest or dividends. As a general matter, the new rules are designed to require U.S. persons' direct and indirect ownership of non-U.S. accounts and non-U.S. entities to be reported to the Internal Revenue Service ("IRS"). The 30% withholding tax regime applies if there is a failure to provide required information regarding U.S. ownership.

Under the terms of FATCA, the Company will be treated as a Foreign Financial Institution. As such, the Company may require all investors to provide documentary evidence of their tax residence and all other information, including personal data, deemed necessary to comply with the above mentioned regulations (the "FATCA Information").

Although the Company will attempt to satisfy any obligation imposed on it to avoid imposition of FATCA withholding tax, no assurance can be given that the Company will be able to satisfy these obligations. If the Company becomes subject to a withholding tax as result of the FATCA regime, the value of the Shares held by the investor may suffer material losses.

Therefore and despite anything else herein contained and as far as permitted by Luxembourg law, the Company shall have the right to:

- require any Shareholder or beneficial owner of the Shares to promptly furnish the FATCA Information as may be required by the Company in its discretion in order to comply with any law and/or to promptly determine the amount of withholding to be retained;
- divulge any element of the FATCA Information to any tax authority, as may be required by law or such authority;
- withhold any taxes or similar charges that it is legally required to withhold, whether by law or otherwise, in respect of any shareholding in the Company; and
- withhold the payment of any dividend or redemption proceeds to a Shareholder until the Company holds sufficient information to enable it to determine the correct amount to be withheld.

As described in Section 2.2 "Issue of Shares, Subscription and Payment Procedure" the Directors have resolved to prevent the ownership of Shares by any U.S. Person (as defined pursuant to Regulation S under the U.S. Securities Act of 1933, as amended), provided however that no provisions of this Prospectus shall prevent a Morgan Stanley entity or any of its affiliates or subsidiaries from owning Shares.

Each prospective investor should consult its own tax advisers regarding the requirements under FATCA with respect to its own situation.

Potential Conflicts of Interest

The Management Company, the Investment Adviser(s) and other affiliates may effect transactions in which they have, directly or indirectly, an interest which may involve a potential conflict with the Management Company's duty

to the Company. Neither the Management Company nor the Investment Adviser(s) nor other affiliates shall be liable to account to the Company for any profit, commission or remuneration made or received from or by reason of such transactions or any connected transactions and none of their fees shall be adjusted (unless otherwise provided), provided that the Management Company and the Investment Adviser(s) (where applicable) will ensure that:

- such transactions are effected on terms which are not less favourable to the Company than if the potential conflict had not existed;
- such transactions are on arm's length terms;
- the Management Company uses due care in the selection of brokers or dealers and that they are suitably qualified in the circumstances;
- the fee or commission paid to any such broker or dealer in respect of a transaction must not be greater than that which is payable at the prevailing market rate for a transaction of that size and nature;
- the Management Company monitors such transactions to ensure compliance with its obligations; and
- the nature of such transactions and the total commissions and other quantifiable benefits received by such broker or dealer are disclosed in the Company's annual report.

The Depositary shall not carry out activities with regard to the Company or the Management Company that may create conflicts of interest between the Company, the Shareholders, the Management Company and itself, unless the Depositary has functionally and hierarchically separated the performance of its depositary tasks from its other potentially conflicting tasks, and the potential conflicts of interest are properly identified, managed, monitored and disclosed to the Shareholders.

Potential conflicting interests or duties may arise because the Management Company or the Investment Adviser(s) may have invested directly or indirectly in the Company. More specifically, the Management Company and Investment Adviser, under the organisational and conduct rules applicable to them, must take all appropriate steps to identify and to prevent or manage conflicts of interest.

Cyber Security

The Funds and their service providers are susceptible to cyber security risks that include, among other things, theft, unauthorized monitoring, release, misuse, loss, destruction or corruption of confidential and highly restricted data; denial of service attacks; unauthorized access to relevant systems, compromises to networks or devices that the Funds and their service providers use to service the Funds' operations; or operational disruption or failures in the physical infrastructure or operating systems that support the Funds and their service providers. Cyber-attacks against or security breakdowns of the Funds or their service providers may adversely impact the Funds and their shareholders, potentially resulting in, among other things, financial losses; the inability of Funds' shareholders to transact business and the Funds to process transactions; inability to calculate the Funds' NAV; violations of applicable privacy and other laws; regulatory fines, penalties, reputational damage, reimbursement or other compensation costs; and/or additional compliance costs. The Funds may incur additional costs for cyber security risk management and remediation purposes. In addition, cyber security risks may also impact issuers of securities in which the Funds invest, which may cause the Funds' investments in such issuers to lose value. There can be no assurance that the Funds or their service providers will not suffer losses relating to cyber-attacks or other information security breaches in the future.

Epidemics / Pandemics / Outbreaks Risk

The performance of the Shares depends on the performance of the investments of the Funds, which could also be adversely affected by the effects of epidemics, pandemics or outbreaks of communicable diseases. In response to intensifying efforts to contain epidemics, pandemics or outbreaks of communicable diseases, governments around the world may take a number of actions, such as prohibiting residents' freedom of movement, encouraging or ordering employees to work remotely from home, and banning public activities and events, among others. Any prolonged disruption of businesses could negatively impact financial conditions. The performance of the Shares could be adversely affected to the extent that any of these epidemics, pandemics or outbreaks harms the economy in general.

IBOR Reform

The term "IBOR" refers generally to any reference rate or benchmark rate that is an "interbank offered rate" intended to reflect, measure or estimate the average cost to certain banks of borrowing or obtaining unsecured short-term funds in the interbank market in the relevant currency and maturity. IBORs have been used extensively as reference rates across the financial markets for many years. A Fund may invest in securities or derivatives whose value or payments are derived from an IBOR.

Pursuant to recommendations of the Financial Stability Board (FSB), financial institutions and other market participants have been working to promote the development of alternative reference rates (ARRs). ARRs are in response to concerns over the reliability and robustness of IBORs. In July 2017, the UK Financial Conduct Authority (FCA) announced that the FCA would no longer use its influence or powers to persuade or compel contributing banks to make IBOR submissions after the end of 2021. Following this statement, other regulators across the globe have

made announcements encouraging financial institutions and other market participants to transition from the use of IBORs to the use of new ARRs by the end of 2021. This has raised concerns about the sustainability of IBORs beyond 2021.

Regulatory and industry initiatives concerning IBORs may result in changes or modifications affecting investments referencing IBORs.

Sustainability Risk

Sustainability Risk means an ESG event or condition that, if it occurs, could cause an actual or a potential material negative impact on the value of a Fund. Such Sustainability Risks are integrated into the investment decision making and risk monitoring to the extent that they represent a potential or actual material risk and/or an opportunity to maximize the long-term risk-adjusted returns.

The impacts following the occurrence of a Sustainability Risk may be numerous and vary depending on the specific risk, region and asset class. Sustainability Risks generally revolve around the following factors including but not limited to:

- Climate change risks including both global warming driven by human emissions of greenhouse gases and
 the resulting large scale shifts in weather patterns. Risks associated with climate change include transition
 risks (policy changes, reputational impacts and shifts in market preferences, norms and technology) and
 physical risk (physical impacts of climate change such as droughts, floods or thawing ground).
- Natural resource risks including rising costs from resource scarcity or resource usage taxes and systemic risk from biodiversity loss.
- Pollution and waste risks including liabilities associated with contamination and waste management costs.
- Human capital risks including declining employee productivity, attrition and turnover costs, pandemics and supply chain reputational risks or disruption.
- Community risks factors including loss of license to operate, operational disruptions caused by protests or boycotts and systematic inequality and instability.
- Security and safety risks such as consumer security, data privacy and security.

In general, where a Sustainability Risk occurs in respect of an asset, there could be a negative impact on, or an entire loss of, its value. Such a decrease in the value of an asset may occur for a company in which a Fund invests as a result of damage to its reputation resulting in a consequential fall in demand for its products or services, loss of key personnel, exclusion from potential business opportunities, increased costs of doing business and/or increased cost of capital. A company may also suffer the impact of fines and other regulatory sanctions. The time and resources of the company's management team may be diverted from furthering its business into dealing with the Sustainability Risk event, including changes to business practices and dealing with investigations and litigation. Sustainability Risk events may also give rise to loss of assets and/or physical loss including damage to real estate and infrastructure. The utility and value of assets held by companies to which the relevant Fund is exposed may also be adversely impacted by a Sustainability Risk event.

A Sustainability Risk trend may arise and impact a specific investment or may have a broader impact on an economic sector (*e.g.* IT or health care), geography (*e.g.* emerging market) or political region or country.

1.5.2. Specific Risk Factors

In addition to the general risks, as set out above, that should be considered for all Funds, there are other risks that investors should also bear in mind when considering investment into specific Funds. The table below shows which specific risk warnings may apply to each of the Funds based on their respective investment objective and policy. Investors' attention is drawn to the fact that these risks will only arise in the event the Funds invest in the relative instruments as may be decided from time to time by the Investment Adviser.

The specific risk factors that have been listed as relevant to a Fund in the table below are those identified as the material risks applicable to each of the Funds at the time of the issue of this Prospectus. Specific risk factors may apply to each Fund to varying degrees, and the exposure to specific risk factors will also vary over time. Additionally, risks may arise in the future which could not have been foreseen in advance. This Prospectus will be regularly updated to reflect any changes to the risk factors set out in Section 1.5.1 General Risk Factors and Section 1.5.2 Specific Risk Factors.

Strategy	Fund Name	Equity Risk	Fixed Income Risk	Asset Allocation Funds	Absolute Return Strategies	Financial Derivative Instruments	EPM Techniques	Investmen t Funds	Emerging Markets	Eurozone	Real Estate Industry	Infra- structure	Commodities	Volatility Strategies	High Yield	ABS/M BS	CoCos	SPACs	Investments in Mainland China	Investment s in Saudi Arabia
Equity	American Resilience Fund	✓	✓			✓			✓	✓										
Equity	Asia Opportunity Fund	✓				✓	✓	✓	✓									✓	✓	
Equity	Asian Property Fund	✓	✓			✓		✓	✓	✓	✓								✓	
Equity	Calvert Sustainable Climate Aligned Fund	✓								✓	✓	✓								
Equity	Calvert Sustainable Climate Transition Fund	✓	✓			~		√	~	√	>	√	√					✓	✓	
Equity	Calvert Sustainable Developed Europe Equity Select Fund	✓								✓	✓	√								
Equity	Calvert Sustainable Developed Markets Equity Select Fund	✓								✓	✓	✓								
Equity	Calvert Sustainable Diversity, Equity and Inclusion Fund	✓								✓	✓	✓								
Equity	Calvert Sustainable Emerging Markets Equity Select Fund	√						✓	✓		√	✓								
Equity	Calvert Sustainable Global Equity Engagement Fund	✓								✓										
Equity	Calvert Sustainable US Equity Select Fund	✓									✓	✓								
Equity	China A-shares Fund	✓	✓			✓	✓	✓	✓										✓	
Equity	China Equity Fund	✓	✓			✓	✓	✓	✓										✓	
Equity	Counterpoint Global Fund	✓	✓			✓	✓	✓	✓	✓	✓							✓	✓	
Equity	Developing Opportunity Fund	✓	✓			✓	✓	✓	✓		✓							✓	✓	
Equity	Emerging Leaders Equity Fund	✓	✓			✓	✓		✓										✓	
Equity	Europe Opportunity Fund	✓	✓			✓	✓	✓	✓	✓								✓		
Equity	European Property Fund	✓	✓			✓		✓		✓	✓									

Strategy	Fund Name	Equity Risk	Fixed Income Risk	Asset Allocation Funds	Absolute Return Strategies	Financial Derivative Instruments	EPM Techniques	Investmen t Funds	Emerging Markets	Eurozone	Real Estate Industry	Infra- structure	Commodities	Volatility Strategies	High Yield	ABS/M BS	CoCos	SPACs	Investments in Mainland China	Investment s in Saudi Arabia
Equity	Global Brands Equity Income Fund	✓	✓			✓	✓		✓	✓									✓	
Equity	Global Brands Fund	✓	✓			✓			✓	✓									✓	
Equity	Global Core Equity Fund	✓	✓			✓	✓		✓	✓	✓									
Equity	Global Endurance Fund	✓	✓			✓	✓		✓	✓	√							√	✓	
Equity	Global Focus Property Fund	✓	✓			✓		✓	✓	✓	√								✓	
Equity	Global Infrastructure Fund	✓	✓			✓			✓	✓	√	✓								
Equity	Global Insight Fund	✓	✓			✓	✓		✓	✓								✓	✓	
Equity	Global Opportunity Fund	✓	✓			✓	✓	✓	✓	✓	✓							✓	✓	
Equity	Global Permanence Fund	✓	✓			✓	✓	✓	✓	✓	✓							✓	✓	
Equity	Global Property Fund	✓	✓			✓		✓	✓	✓	✓									
Equity	Global Quality Fund	✓	✓			✓			✓	✓									✓	
Equity	Global Sustain Fund	✓	✓			✓			✓	✓									✓	
Equity	Indian Equity Fund	✓	✓			✓			✓											
Equity	International Equity (ex US) Fund	✓	✓			✓			✓	✓									✓	
Equity	Japanese Equity Fund	✓	✓			✓														
Equity	Japanese Small Cap Equity Fund	✓	✓			✓														
Equity	Latin American Equity Fund	✓				✓	✓		✓	✓										
Equity	NextGen Emerging Markets Fund	✓	✓			✓			✓	✓										
Equity	Parametric Emerging Markets Fund	√				✓			✓		✓								_	√
Equity	Parametric Global Defensive Equity Fund	√	✓			√		√	√	✓	√			√						
Equity	Saudi Equity Fund	✓				✓	✓		✓		✓	✓								✓
Equity	Sustainable Asia Equity Fund	✓	✓			✓		✓	✓	✓	✓								✓	

Strategy	Fund Name	Equity Risk	Fixed Income Risk	Asset Allocation Funds	Absolute Return Strategies	Financial Derivative Instruments	EPM Techniques	Investmen t Funds	Emerging Markets	Eurozone	Real Estate Industry	Infra- structure	Commodities	Volatility Strategies	High Yield	ABS/M BS	CoCos	SPACs	Investments in Mainland China	Investment s in Saudi Arabia
Equity	Sustainable Emerging Markets Equity Fund	✓	✓			✓		√	√	✓									✓	
Equity	US Advantage Fund	✓	✓			✓	✓		✓	✓								✓	✓	
Equity	US Core Equity Fund	✓	✓			✓	✓				✓									
Equity	US Focus Property Fund	~	✓			√		✓		✓	✓									
Equity	US Growth Fund	✓	✓			✓	✓		✓	✓								✓	✓	
Equity	US Insight Fund	✓	✓			✓	✓	✓	✓									✓	✓	
Equity	US Permanence Fund	✓	✓			✓	✓	✓	✓		✓							✓	✓	
Equity	US Property Fund	✓	✓			✓				✓	✓									
Equity	US Value Fund	✓									✓									
Equity	Vitality Fund	✓	✓			✓	✓	✓	✓	✓								✓		
Bond	Calvert Sustainable Global Green Bond Fund		√			√	√		✓	√					√	√	✓			
Bond	Emerging Markets Corporate Debt Fund		✓			✓	✓	✓	✓						✓	✓	>			
Bond	Emerging Markets Debt Fund		✓			✓	✓		✓	✓					✓	✓	✓			
Bond	Emerging Markets Debt Opportunities Fund		✓			√	√		✓	✓				√	✓					
Bond	Emerging Markets Domestic Debt Fund		✓			✓	✓		✓	✓					✓	✓	✓			
Bond	Emerging Markets Fixed Income Opportunities Fund		✓			√	√	✓	✓						√	✓	√			
Bond	Emerging Markets Local Income Fund		✓			✓	✓		✓	✓				✓	✓					
Bond	Euro Bond Fund		✓			✓	✓	✓		✓						✓	~			
Bond	Euro Corporate Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Euro Corporate Bond – Duration Hedged Fund		✓			√	✓	✓	✓	√					√	✓	√			
Bond	Euro Strategic Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			

Strategy	Fund Name	Equity Risk	Fixed Income Risk	Asset Allocation Funds	Absolute Return Strategies	Financial Derivative Instruments	EPM Techniques	Investmen t Funds	Emerging Markets	Eurozone	Real Estate Industry	Infra- structure	Commodities	Volatility Strategies	High Yield	ABS/M BS	CoCos	SPACs	Investments in Mainland China	Investment s in Saudi Arabia
	European Fixed Income Opportunities Fund	√	√		√	√	√	✓	✓						✓	√	✓			
Bond	European High Yield Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Floating Rate ABS Fund		✓		✓	✓	✓	✓		✓				✓		✓				
Bond	Frontier Markets Local Debt Fund		✓			✓			✓	✓				✓	✓					
	Global Asset Backed Securities Fund		✓			✓	✓	✓	✓	✓					✓	✓				
Bond	Global Bond Fund		✓			✓	✓		✓	✓					✓	✓	✓			
Bond	Global Convertible Bond Fund	✓	✓			✓	✓		✓	✓					✓	✓	✓			
Bond	Global Credit Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Global Credit Opportunities Fund		✓		✓	✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Global Fixed Income Opportunities Fund	✓	✓			✓	✓	✓	✓	✓					✓	✓	✓			
	Global High Yield Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Global Macro Fund	✓	✓		✓	✓	✓		✓	✓			✓	✓	✓	✓				
	Short Duration US Government Income Fund		√			√	√	✓								√				
	Short Maturity Euro Bond Fund		✓			✓	✓	✓		✓						✓	✓			
Bond	Short Maturity Euro Corporate Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Sustainable Euro Corporate Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓			
Bond	Sustainable Euro Strategic Bond Fund		✓			✓	✓	✓	✓	✓					✓	✓	✓		_	
Bond	Sustainable European High Yield Bond Fund		√			✓				√					√					
Bond	Sustainable Global Fixed Income Opportunities Fund	✓	✓			√	√	✓	✓	✓					✓	✓	✓			

Strategy	Fund Name	Equity Risk	Fixed Income Risk	Asset Allocation Funds	Absolute Return Strategies	Financial Derivative Instruments	EPM Techniques	Investmen t Funds	Emerging Markets	Eurozone	Real Estate Industry	Infra- structure	Commodities	Volatility Strategies	High Yield	ABS/M BS	CoCos	SPACs	Investments in Mainland China	Investment s in Saudi Arabia
Rond	Sustainable Global High Yield Bond Fund		✓			✓				✓					✓					
Rond	Sustainable US High Yield Bond Fund		✓			✓				✓					✓					
Rond	US Dollar Corporate Bond Fund		✓			✓	✓	✓							✓	✓				
Rond	US Dollar High Yield Bond Fund		✓			✓	✓	✓	✓						✓	✓	✓			
Bond	US Dollar Short Duration Bond Fund		✓			✓	✓	✓							✓	✓				
Bond	US Dollar Short Duration High Yield Bond Fund		✓			✓	✓	✓	✓						✓	✓	√			
	Global Balanced Defensive Fund	✓	✓	✓		✓	✓	✓	✓	✓	✓		✓	✓	✓	✓			✓	
	Global Balanced Fund	✓	✓	✓		✓	✓	✓	✓	✓	✓		✓	✓	✓	✓			✓	
	Global Balanced Income Fund	✓	✓	✓		✓	✓	✓	✓	✓	✓		✓	✓	✓	✓			✓	
Asset Allocation	Global Balanced Risk Control Fund of Funds	✓	✓	✓		✓	✓	✓	✓	✓			✓	√	✓	✓				
	Global Balanced Sustainable Fund	✓	✓	✓		✓	✓	✓	✓	✓	✓		✓	✓	✓	✓			✓	
	Global Multi-Asset Opportunities Fund	✓	✓	✓	✓	✓	✓	✓	✓	✓			✓		✓	✓			✓	
	Multi-Asset Risk Control Fund	✓	✓	✓		✓	✓	✓	✓	✓	✓		✓	✓	✓	✓			✓	

Fixed Income Risk

Funds which invest in Fixed Income Securities will be subject to interest rate and credit risk, and the additional risks associated with securities such as high-yield Fixed Income Securities or asset-backed securities.

Fixed Income Securities are subject to the risk of an issuer's ability to meet principal and interest payments on the obligation (credit risk), and may also be subject to price volatility due to such factors as interest rate sensitivity, market perception of the creditworthiness of the issuer and general market liquidity (market risk). A Fund may invest in Fixed Income Securities which are interest rate sensitive. An increase in interest rates will generally reduce the value of Fixed Income Securities, while a decline in interest rates will generally increase the value of Fixed Income Securities. The performance of such Funds will therefore depend in part on the ability to anticipate and respond to such fluctuations on market interest rates, and to utilise appropriate strategies to maximise returns, while attempting to minimise the associated risks to investment capital.

Interest Rate Risk

The values of Fixed Income Securities held by the Funds will vary with changes in interest rates and such variation may affect Share prices accordingly. The value of Fixed Income Securities will generally increase when interest rates fall and decrease when interest rates rise. Fixed Income Securities with greater interest rate sensitivity and longer maturities are usually subject to greater fluctuations in value in response to interest rate changes.

Certain sub-funds seek to reduce their exposure to interest rate risk through duration hedging. Duration hedged sub-funds utilise hedging strategies to seek to limit their exposure to interest rate movements. Sensitivity to interest rate movements may vary by Fund. The sensitivity of the Funds to interest rate movements may affect the result of the duration hedging.

Such hedging strategies used by the Investment Adviser(s) (or any agent appointed by the Investment Adviser) may not completely eliminate exposure to such interest rate movements. There can be no guarantee that hedging strategies will be successful. Investors should be aware that certain market events or circumstances could result in the Investment Adviser(s) no longer being able to perform hedging transactions or that such strategies may no longer be economically viable.

The use of hedging strategies may substantially limit shareholders in duration hedged Funds from benefiting if interest rates fall.

Credit Risk

Funds which invest in Fixed Income Securities are subject to the risk that an issuer will fail to make timely payments of interest and principal. Issuers with higher credit risk typically offer higher yields for this added risk. Conversely, issuers with lower credit risk typically offer lower yields. Generally, government securities are considered to be the safest in terms of credit risk, while corporate debt, especially those with poorer credit ratings, have the highest credit risk. Changes in the financial condition of an issuer, changes in economic and political conditions in general, or changes in economic and political conditions specific to an issuer, are all factors that may have an adverse impact on an issuer's credit quality and security values.

High Yield Securities

Funds may invest in higher yielding Fixed Income Securities which are subject to greater credit and market risk than lower yielding securities. Generally, lower rated securities pay higher yields than more highly rated securities to compensate investors for the higher risk. Such securities are subject to the risk of an issuer's inability to meet principal and interest payments on its obligations (credit risk) and may also be subject to price volatility due to such factors as interest rate sensitivity, market perception of the creditworthiness of the issuer and general market liquidity and as a result may be less liquid than lower yielding securities.

Negative Yields

Certain Funds may invest in fixed income instruments which, in certain cases, may trade at a negative yield. When a Fund invests in a negative yielding fixed income instrument, the value of the Fund's investment will reduce on a daily basis by the amount of the negative yield and the Fund may not get back its full investment.

Downgrading Risk

The credit ratings given to Fixed Income Securities may be subject to changes. The downgrading of a rated Fixed Income Security could decrease the value and liquidity of the security, particularly in a thinly traded market, and also increase the price volatility. The Company may continue to invest in securities that are downgraded after purchase.

Non-Investment Grade Securities

Non-investment grade securities have a lower credit rating than investment grade securities or are unrated and are generally considered to have a higher credit risk than more highly rated securities. In addition, non-investment grade securities tend to be more volatile than higher rated securities, so that adverse economic events may have a greater impact on the prices of non-investment grade securities than on higher rated securities. The market for securities which are rated below investment grade, have a lower credit rating or are unrated generally has lower liquidity and is less active than that for higher rated securities and a Fund's ability to liquidate its holdings in response to changes in the economy or the financial markets may be further limited by such factors as adverse publicity and investor perceptions. Certain Funds may invest in securities rated below investment grade.

Unrated Securities

Some Funds may invest in securities that are not rated. As they are unrated these securities may be subject to greater price volatility and Funds investing in these securities must rely on the Investment Adviser(s)' credit assessment of such securities and are in particular subject to a high credit risk.

Sovereign Debt

Certain countries and government entities rely more heavily than others upon foreign investment and the international markets for funding. Investment in sovereign debt issued or guaranteed by such countries or government entities involves a high degree of risk as the issuing entity may be unable or unwilling to repay the principal or interest when due in accordance with the terms of the debt. As a result, there may be a risk that the issuing entity will reschedule repayment or default on the debt.

Asset-Backed Securities

Some Funds may invest in Asset-Backed Securities (ABS) which are Fixed Income Securities backed or collateralised by the income stream from an underlying pool of assets such as credit cards, automobile loans, student loans, small business loans, mortgages and receivables. An ABS may be usually issued in a number of different tranches, or classes, with varying characteristics depending on the riskiness of the underlying assets assessed by reference to their credit quality and term and can be issued at a fixed or a floating rate.

The higher the risk contained in the tranche, the more the ABS is likely to have to pay by way of income. The obligations associated with these securities may be subject to greater credit, liquidity and interest rate risk compared to other Fixed Income Securities. ABS are often exposed to extension risk (where obligations on the underlying assets are not paid on time) and prepayment risks (where obligations on the underlying assets are paid earlier than expected), these risks may have a substantial impact on the timing and size of the cash flows paid by the securities and may negatively impact the returns of the securities. The average life of each individual security may be affected by a large number of factors such as the existence and frequency of exercise of any optional redemption and mandatory prepayment, the prevailing level of interest rates, the actual default rate of the underlying assets, the timing of recoveries and the level of rotation in the underlying assets.

Mortgage-Backed Securities

Some Funds may invest in Mortgage-Backed Securities (MBS) which are Fixed Income Securities backed or collateralised by the income stream from an underlying pool of commercial and/or residential mortgages. This type of security is commonly used to redirect the interest and principal payments from the pool of mortgages to investors. An MBS may be issued in a number of different tranches, or classes, with varying characteristics depending on the riskiness of the underlying mortgages assessed by reference to their credit quality and term and can be issued at a fixed or a floating rate. The higher the risk contained in the tranche, the more the MBS pays by way of income. MBS may be subject to prepayment risk which is the risk that, in a period of falling interest rates, borrowers may refinance or otherwise repay principal on their mortgages earlier than scheduled. When this happens, certain types of MBS will be paid off more quickly than originally anticipated and the Funds will have to invest the proceeds in securities with lower yields. MBS may also be subject to extension risk, which is, the risk that, in a period of rising interest rates, certain types of MBS will be paid off more slowly than originally anticipated and the value of these securities will fall. As a result, the average duration of the Funds' portfolios may increase. The value of longer-term securities generally changes more in response to changes in interest rates than that of shorter-term securities. Because of prepayment risk and extension risk, MBS may react differently to changes in interest rates than other Fixed Income Securities. Small movements in interest rates (both increases and decreases) may quickly and significantly reduce the value of certain MBS. In some circumstances investments in MBS may become less liquid and in the case of a large redemption or change in market liquidity the Investment Adviser may not be able to sell the securities to meet the redemption requirement or may only be able to sell the securities at a price which negatively affects the Funds' Net Asset Value. In addition, the market price for MBS may be volatile.

On June 3, 2019, under the Federal Housing Finance Agency's "Single Security Initiative" intended to maximize liquidity for both Fannie Mae and Freddie Mac mortgage-backed securities in the To Be Announced ("TBA") security market, Fannie Mae and Freddie Mac expect to start issuing uniform mortgage-backed securities ("UMBS") in place of their current separate offerings of TBA-eligible mortgage-backed securities. The effects of the issuance of UMBS on the market for mortgage-backed securities are uncertain even though UMBS are not new instruments but rather a harmonisation of already existing instruments. The Funds' ability to invest in UMBS to the same degree that the Funds currently invests in Fannie Mae and Freddie Mac MBS is also uncertain. While Fannie Mae and Freddie Mac have taken steps to have a smooth transition to the issuance of UMBS, there may be factors which affect the timing of the transition or the ability of market participants, including the Funds, to adapt to the issuance of UMBS.

Coupon swaps, agency swaps, and term swaps

Certain Funds may use coupon swaps, agency swaps and/or term swaps.

For example, in a coupon swap, a Fund may buy a Fannie Mae TBA trade and simultaneously sell a differently priced Fannie Mae TBA trade, which will result in the Fund having exposure to the spread between the price of each of those trades.

Whereas by using an agency swap, a Fund will gain exposure to the spread between the prices of the same coupon and maturity of one agency (e.g. Fannie Mae) and another agency (e.g. Ginnie Mae).

Finally, a term swap is a paired TBA trade that gives a Fund exposure to the spread in prices between different terms of MBS.

Consequently, in addition to the risks related to the underlying investments in MBS, the use of such instruments also presents the risks linked to the use of swaps, as further described below in the sub-section "Financial Derivative Instruments".

Non-Agency Mortgage-Backed Securities

Non-agency Mortgage-Backed Securities are MBS issued by private institutions. These securities have no credit guarantee other than the quality of the loans behind them, and any other structural credit protection provided by the terms of the bond deal they belong to. Investing in non-agency mortgage-backed securities generally entails credit, prepayment, extension, liquidity and default risk.

Mortgage Dollar Rolls

Mortgage Dollar Rolls are instruments whereby a Fund sells MBS for delivery in the current month and simultaneously contracts to repurchase substantially similar (same type, coupon and maturity) MBS on a specified future date. During that roll period, the Fund forgoes principal, and interest paid on the MBS. The Fund is compensated by the difference between the current sales price and the lower forward price for the future purchase as well as by the interest earned on the cash proceeds of the initial sales. Cash proceeds may be invested in instruments that are permissible investments for the Fund.

The use of mortgage rolls is a speculative technique involving the risk that the market value of the MBS the Fund is committed to buy may decline below the price of the MBS the Fund has sold. Such transactions may furthermore involve leverage.

Funds are only allowed to enter into covered Mortgage Dollar Rolls for which there is an offsetting cash position or permissible liquid assets earmarked or in a segregated account to secure the obligation for the forward commitment to buy MBS or a Cash Equivalent security position that matures on or before the forward settlement date of the dollar roll transaction.

In addition to the risks related to the underlying investments in MBS, the use of such instrument also presents the risks linked to the use of forwards, as further described below in the sub-section "Financial Derivative Instruments".

Convertible Bonds

Convertible bonds are subject to a number of risks including risk arising from both debt and equity securities, and to convertible securities specific risks. Convertible bond valuations are sensitive to macro-economic risk, interest rate risk, spread risk, default risk, and equity risk. In addition, convertible bonds issuers may be downgraded. In certain market conditions convertible bonds may be less liquid than other asset classes.

Characteristics of the contingent convertible debt securities

Certain Funds may invest in contingent convertible debt securities which are Fixed Income Securities that may pay an attractive coupon and which may be converted into equity securities or suffer capital losses by decreasing the face value if pre-specified events occur ("trigger events"), depending in particular of the capital ratio levels of the issuer of such contingent convertible debt securities ("trigger levels"). Contingent convertible debt securities may be issued as perpetual instruments which may (or may not) be called at pre-determined date.

Specific risks associated with the related contingent convertible debt securities

Trigger levels and conversion risks: contingent convertible debt securities are complex financial instruments in respect of which, trigger levels (and thus exposure to conversion risk) differ widely. In particular, conversion may cause the value of the investment to fail significantly and irreversibly, and in some cases even to zero.

Unknown and yield related risks: contingent convertible debt securities are also innovative financial instruments and their behaviour under a stressed financial environment is thus unknown. This increases uncertainty in the valuation of contingent convertible debt securities and the risks of potential price contagion, as well as the volatility and also the liquidity risks of the entire contingent convertible securities asset class. In certain circumstances finding a ready buyer for contingent convertible bonds may be difficult and the seller may have to accept a significant discount to the expected value of the bond in order to sell it. Furthermore, because of the often attractive yield of contingent convertible debt securities, it still remains unclear whether holders of contingent convertible debt securities have fully considered the underlying risks of these instruments.

Write-down and capital structure inversion risks: the investment in contingent convertible debt securities may also result in material losses to the relevant Fund as the contingent convertible debt security may suffer capital market loss by decreasing the face value ("write-down") on the occurrence of certain trigger events. In this event, holders of contingent convertible debt securities will suffer losses ahead of holders of equity securities issued by the same issuer, contrary to the classic order of capital structure hierarchy where equity holders are expected to suffer the loss before debt holders.

Call extension risk: as contingent convertible debt securities may be perpetual instruments which may not be called on the predefined call date and investors may not receive return of principal on the call date or at any date.

Coupon cancellation risk: In addition, some contingent convertible debt securities are also subject to the risk of discretionary cancellation of coupon payments by the issuer at any point, for any reason, and for any length of time.

Odd Lots

Certain of the Funds' securities may be valued by an outside pricing service approved by the Management Company. The pricing service may utilize a matrix system or other model incorporating attributes such as security quality, maturity and coupon as the evaluation model parameters, and/or research evaluations by its staff, including review of broker-dealer market price quotations in determining what it believes is the fair valuation of the portfolio securities valued by such pricing service. Pricing services generally value securities assuming orderly transactions of an institutional round lot size, but the Company's Funds may hold or transact in such securities in smaller, odd lot sizes ("odd lots"). Odd lots often trade at lower prices than institutional round lots.

Equity Risk

Funds which invest in equity securities are subject to the volatility of the capital markets on which these securities are traded and may incur significant losses. The price of equities can be influenced by many factors at the individual company level, as well as by broader economic and political developments, including trends in economic growth, inflation and interest rates, corporate earnings reports, demographic trends and catastrophic events.

Depositary Receipts

Depositary receipts (ADRs, GDRs and EDRs) are instruments that represent shares in companies trading outside the markets in which the depositary receipts are traded. Accordingly whilst the depositary receipts are traded on Recognised Exchanges, there may be other risks associated with such instruments to consider- for example the shares underlying the instruments may be subject to political, inflationary, exchange rate or custody risks.

Small and mid-sized companies

The stock prices of small and mid-sized companies tend to be more volatile than the stock prices of larger companies. Smaller companies may have limited resources and product ranges and therefore may be more sensitive to changes

in market conditions. The stocks of smaller companies are traded less frequently and in lower volumes than those of larger companies and this may contribute to greater stock price volatility.

Participatory Notes

Equity funds may generate exposure to certain equity securities in certain countries by purchasing a participatory note. A participatory note, while generating the desired equity security exposure, adds counterparty risk exposure to the issuer of the participatory note.

Asset Allocation Funds

The Asset Allocation Funds provide the Investment Adviser with wide discretion to allocate between different asset classes. From time to time, the Asset Allocation Funds may have significant exposure to a single or limited number of fixed income or equity asset classes. Accordingly, the relative relevance of the risks associated with equity securities, Fixed Income Securities and derivatives will fluctuate over time.

Absolute Return Strategies

The intention of an absolute return strategy is to deliver positive returns through a market cycle. However there can be no guarantee that such returns or capital will be achieved.

Financial Derivative Instruments

Funds may, in accordance with their investment policy, invest in financial derivative instruments including but not limited to European and American options including single security, currency, basket and index calls and puts; single security, equity index and volatility futures; interest rate, Eurodollar and treasury futures; contract for differences (CFDs); single currency swaps; credit default swaps; interest rate swaps; Consumer Price Index (CPI) swaps, TRS, coupon swaps, agency swaps, term swaps, structured notes, warrants, currency forwards, Mortgage Dollar Rolls and participatory notes.

While the prudent use of derivatives may be beneficial, derivatives also involve risks different from, and, in certain cases, greater than, the risks presented by more traditional investments. If so provided in its investment policy, a Fund may engage various investment strategies with a view to reducing certain of its risks and/or enhancing return. These strategies may include the use of derivative instruments such as options, warrants, swaps and/or futures. Such strategies may be unsuccessful and incur losses for the Fund.

Derivatives also involve specific risks. These risks relate specifically to market risks, management risk, counterparty risk, liquidity risk, the risk of mispricing or improper valuation of derivatives and the risk that derivatives may not correlate perfectly with underlying assets, interest rates and indices.

The following is a general discussion of important risk factors and issues concerning the use of derivatives that investors should understand before investing in a Fund.

Market Risk

This is a general risk that applies to all investments, including derivatives, meaning that the value of a particular derivative may go down as well as up in response to changes in market factors. A Fund may also use derivatives to gain or short exposure to some investments. In extreme market conditions the use of derivatives may, theoretically, give rise to unlimited losses for the Fund. However, an investor's loss is always limited to the amount invested in the Fund. Should such extreme market conditions occur, investors could, in certain circumstances, therefore face minimal or no returns, or may even suffer a loss on their investment in that particular Fund.

Liquidity Risk

Liquidity risk exists when a particular instrument is difficult to purchase or sell. If a derivative transaction is particularly large or if the relevant market is illiquid, it may not be possible to initiate a transaction or liquidate a position at an advantageous price (however, the Company will only enter into OTC derivatives if it is allowed to liquidate such transactions at any time at fair value).

Counterparty Risk

The Funds may enter into transactions in OTC markets, which will expose the Funds to the credit of its counterparties and their ability to satisfy the terms of such contracts. In the event of a bankruptcy or insolvency of a counterparty, the Funds could experience delays in liquidating the position and significant losses, including declines in the value of its investment during the period in which the Company seeks to enforce its rights, inability to realise any gains on its investment during such period and fees and expenses incurred in enforcing its rights. There is also a possibility that the above agreements and derivative transactions may be terminated due, for instance, to bankruptcy,

supervening illegality or change in the tax or accounting laws relative to those at the time the agreement was originated. However this risk is limited in view of the investment restrictions laid down in Section 3 – "Derivatives and efficient portfolio management techniques" of Appendix A – Investment Powers and Restrictions.

Collateral Management Risk

Counterparty risk arising from investments in OTC financial derivative instruments is generally mitigated by the transfer or pledge of collateral in favour of the Fund. However, transactions may not be fully collateralised. Fees and returns due to the Fund may not be collateralised. If a counterparty defaults, the Fund may need to sell non-cash collateral received at prevailing market prices. In such a case the Fund could realise a loss due, inter alia, to inaccurate pricing or monitoring of the collateral, adverse market movements, deterioration in the credit rating of issuers of the collateral or illiquidity of the market on which the collateral is traded. Difficulties in selling collateral may delay or restrict the ability of the Fund to meet redemption requests.

A Fund may also incur a loss in reinvesting cash collateral received, where permitted. Such a loss may arise due to a decline in the value of the investments made. A decline in the value of such investments would reduce the amount of collateral available to be returned by the Fund to the counterparty as required by the terms of the transaction. The Fund would be required to cover the difference in value between the collateral originally received and the amount available to be returned to the counterparty, thereby resulting in a loss to the Fund.

Margin posted by Funds

Where a Fund enters into a derivatives transaction, it will generally be required as a matter of law and/or contract to deliver cash or assets as margin (often referred to as 'collateral') to protect the relevant broker from the risk of a potential default by the Fund. Where the broker receives margin on a title transfer basis (ie the broker becomes the owner of the margin outright), or exercises a right of reuse, on the broker's default or insolvency the Fund will be an unsecured creditor and may not be able to recover the full value of the amount owed to it in full or at all. The Fund will not be entitled to exercise voting, consent or other similar rights attached to assets it provides as margin on a title transfer basis or in respect of which a right of use has been exercised, unless and until equivalent assets have been returned. In the event that a broker fails to return equivalent assets when due, the Fund may be unable to perform its settlement obligations under a hedging or other transaction it has entered into in relation to such assets.

Counterparty Replacement Risk

The Global Brands Equity Income Fund is expected to take exposure via a swap on options on equity securities and/or equity indices. This swap is highly customised and gaining this exposure is dependent on the availability of a counterparty. This Fund intends to use Morgan Stanley International Plc, a company affiliated to the Investment Adviser, as a counterparty to the swap. If Morgan Stanley International Plc is unable to continue as counterparty to the swap, the Fund may be unable to contract with another counterparty for an equivalent swap. This may mean that the Fund is unable to implement its investment strategy fully and, in particular, the income enhancement component of the strategy may be affected resulting in a reduction of income. This may also mean that the Fund cannot meet its investment objective to deliver regular income.

Leverage Risk

Derivative instruments allow the Fund to gain a larger exposure to asset values than the amount the Fund invests. As a result, losses on derivative instruments can exceed the amount invested in them which may significantly reduce the value of the Fund as a whole.

Other Risks

Other risks in using derivatives include the risk of differing valuations of derivatives arising out of different permitted valuation methods and the inability of derivatives to correlate perfectly with underlying securities, rates and indices. Many derivatives, in particular OTC derivatives, are complex and often valued subjectively and the valuation can only be provided by a limited number of market professionals who often are acting as counterparties to the transaction to be valued. The Management Company will seek to obtain independent valuations for OTC derivatives in order to limit this risk.

Derivatives do not always perfectly or even highly correlate to or track the value of the securities, rates or indices they are designed to track. Consequently, a Fund's use of derivative techniques may not always be an effective means of following a Fund's investment objective. In cases where derivatives are being used to hedge risk, it is possible that the offsetting investments will not experience price changes that are perfectly inversely correlated. As a result, hedged portfolios may be exposed to basis risk – the risk that the portfolio will realize excess gains or losses in the execution of the hedging strategy.

An OTC derivative is a derivative instrument which is not listed and traded on a formal exchange such as FTSE or NYSE but is traded by counterparties who negotiate directly with one another over computer networks and by telephone. The counterparty risk on any transaction involving OTC derivative instruments may not exceed 10% of the assets of a Fund when the counterparty is a credit institution domiciled in the EU or in a country where the CSSF considers that supervisory regulations are equivalent to those prevailing in the EU. This limit is set at 5% in any other case.

The Management Company ensures that appropriate risk monitoring is in place for any OTC transactions.

Clearing

When a Fund enters into cleared derivatives transactions (whether exchange traded or over-the-counter), and a clearing broker it uses for such transactions is declared to be in default by an EU central counterparty ("EU CCP"), the EU CCP will try to transfer ("port") the Fund's transactions and assets to another clearing broker or, if this cannot be achieved, the EU CCP will terminate the Fund's transactions. The early termination of transactions in this context may result in significant losses to the Fund. In the event that other parties in the clearing structure default (e.g., a central counterparty, a custodian, settlement agent or any clearing broker instructed by the Fund's broker), the Fund may not receive all of its assets back and its rights may differ depending on the law of the country in which the party is incorporated and the specific protections that that party has put in place.

Risks associated with the Control and Monitoring of Derivatives

Derivative products are highly specialised instruments that require investment techniques and risk analysis different from those associated with equity securities and Fixed Income Securities. The use of derivative techniques requires an understanding not only of the underlying assets of the derivative but also of the derivative itself, without the benefit of observing the performance of the derivative under all possible market conditions.

In particular, the use and complexity of derivatives require the maintenance of adequate controls to monitor the transactions entered into, the ability to assess the risk that a derivative adds to a Fund and the ability to forecast the relative price, interest rate or currency rate movements correctly.

There is no guarantee that a particular forecast will be correct or that an investment strategy which deploys derivatives will be successful.

Warrants

Certain Funds may invest in equity linked securities or equity linked instruments such as warrants. The gearing effect of investment in warrants and the volatility of warrant prices make the risk attached to the investment in warrants higher than in the case with investment in equities.

Efficient Portfolio Management Techniques

A Fund may enter into repurchase agreements and reverse repurchase agreements as a buyer or as a seller subject to the conditions and limits set out in section 3 – "Derivatives and efficient portfolio management techniques" of Appendix A – Investment Powers and Restrictions. If the other party to a repurchase agreement or reverse repurchase agreement should default, the Fund might suffer a loss to the extent that the proceeds from the sale of the underlying securities and/or other collateral held by the Fund in connection with the repurchase agreement or reverse repurchase agreement are less than the repurchase price or, as the case may be, the value of the underlying securities. In addition, in the event of bankruptcy or similar proceedings of the other party to the repurchase agreement or reverse repurchase agreement or its failure otherwise to perform its obligations on the repurchase date, the Fund could suffer losses, including loss of interest on or principal of the security and costs associated with delay and enforcement of the repurchase agreement or reverse repurchase agreement.

A Fund may enter into securities lending transactions subject to the conditions and limits set out in Section 3 – "Derivatives and efficient portfolio management techniques" of Appendix A – Investment Powers and Restrictions. If the other party to a securities lending transaction should default, the Fund might suffer a loss to the extent that the proceeds from the sale of the collateral held by the Fund in connection with the securities lending transaction are less than the value of the securities lent. In addition, in the event of the bankruptcy or similar proceedings of the other party to the securities lending transaction or its failure to return the securities as agreed, the Fund could suffer losses, including loss of interest on or principal of the securities and costs associated with delay and enforcement of the securities lending agreement.

Securities lending transactions, repurchase agreements and reverse repurchase agreements also entail liquidity risks due, inter alia, to locking cash or securities positions in transactions of excessive size or duration relative to the liquidity profile of the Fund or delays in recovering cash or securities paid to the counterparty. These circumstances

may delay or restrict the ability of the Fund to meet redemption requests. The Fund may also incur operational risks such as, inter alia, non-settlement or delay in settlement of instructions, failure or delays in satisfying delivery obligations under sales of securities, and legal risks related to the documentation used in respect of such transactions.

The Funds will only use repurchase agreements, reverse repurchase agreements or securities lending transactions for the purpose of either reducing risks (hedging) or generating additional capital or income for the relevant Fund. When using such techniques, the Funds will comply at all times with the provisions set out section 3 – "Derivatives and efficient portfolio management techniques" of Appendix A – Investment Powers and Restrictions. The risks arising from the use of repurchase agreements, reverse repurchase agreements and securities lending transactions will be closely monitored and techniques (including collateral management) will be employed to seek to mitigate those risks. The use of repurchase agreements, reverse repurchase agreements and securities lending transactions is generally not expected to have a material adverse impact on a Fund's performance, subject to the Risk Factors described above.

Counterparty risk arising from investments in EPM techniques is generally mitigated by the transfer or pledge of collateral in favour of the Fund. However, transactions may not be fully collateralised. Fees and returns due to the Fund may not be collateralised. If a counterparty defaults, the Fund may need to sell non-cash collateral received at prevailing market prices. In such a case the Fund could realise a loss due, *inter alia*, to inaccurate pricing or monitoring of the collateral, adverse market movements, deterioration in the credit rating of issuers of the collateral or illiquidity of the market on which the collateral is traded. Difficulties in selling collateral may delay or restrict the ability of the Fund to meet redemption requests.

A Fund may incur custody risk, due to the risk that the value of the collateral held with the Depositary or a sub-custodian decreases through the operational risk attributed to the Depositary or sub-custodian. These operational risks include but are not limited to, insolvency, negligence, misuse of the collateral, poor administration or inadequate record keeping.

A Fund may also incur a loss in reinvesting cash collateral received, where permitted. Such a loss may arise due to a decline in the value of the investments made. A decline in the value of such investments would reduce the amount of collateral available to be returned by the Fund to the counterparty as required by the terms of the transaction. The Fund would be required to cover the difference in value between the collateral originally received and the amount available to be returned to the counterparty, thereby resulting in a loss to the Fund.

The Company may enter into securities lending transactions, repurchase agreements and reverse repurchase agreements with other companies in the same group of companies as the Investment Adviser. Affiliated counterparties, if any, will perform their obligations under any securities lending transactions, repurchase agreements and buy-sell back transactions concluded with a Fund in a commercially reasonable manner. In addition, the Investment Adviser will select counterparties and enter into transactions in accordance with best execution principles. However, investors should be aware that the Investment Adviser may face conflicts between its role and its own interests or that of affiliated counterparties.

Investment Funds

Open-ended and Closed-end Collective Investment Vehicles

Some Funds may invest in other collective investment vehicles. By investing in collective investment vehicles indirectly through the Fund, the investor will bear not only his proportionate share of the management fee of the Fund, but also indirectly, the management and administration expenses of the underlying collective investment vehicles.

In the case of investment in closed-end investment vehicles, shares may at times be acquired only at market prices representing premiums to their net asset values or disposed of at market prices representing discounts to their net asset value. Shares of such closed-end collective investment vehicles will be valued at their last available stock market value.

Collective investment vehicles established in different jurisdictions may not always provide an equivalent level of protection. This may expose the Funds investing in them to additional risks, for instance, less frequent opportunities for disposal, delayed payment or non-receipt of settlement monies, or less protective judicial structures.

Investment in Third Party Funds

Certain Funds may invest in shares of collective investment schemes including ETFs, Closed Ended Funds and UCITS funds – collectively, underlying funds. These Funds may be advised or sub-advised by the Investment Adviser, an affiliate adviser or by an unaffiliated adviser. The ability of a Fund which invests in shares of an underlying fund or funds to achieve its investment objective may be directly related to the ability of the underlying funds to meet their investment objectives. Funds which invest in underlying funds will be exposed to the risks to which the underlying funds are exposed. These risks may include liquidity risk where the ability of the Fund to meet

the liquidity requirements of its investment is directly linked to the ability of the underlying funds to meet their liquidity requirements.

Emerging Markets, Frontier Markets and other non-developed markets

In certain countries, there is the possibility of expropriation of assets, confiscatory taxation, political or social instability or diplomatic developments which could affect investment in those countries. There may be less publicly available information about certain financial instruments than some investors would find customary and entities in some countries may not be subject to accounting, auditing and financial reporting standards and requirements comparable to those to which certain investors may be accustomed. Certain financial markets, while generally growing in volume, have, for the most part, substantially less volume than more developed markets, and securities of many companies may be less liquid and their prices more volatile than securities of comparable companies in more sizeable markets. There are also varying levels of government supervision and regulation of exchanges, financial institutions and issuers in various countries. In addition, the manner in which foreign investors may invest in securities in certain countries, as well as limitations on such investments, may affect the investment operations of certain of the Funds.

Settlement systems in Emerging Markets, Frontier Markets and other non-developed markets may be less well organised than in developed markets. Thus there may be a risk that settlement may be delayed and that cash or securities of the Funds may be in jeopardy because of failures of or defects in the systems. In particular, market practice may require that payment shall be made prior to receipt of the security which is being purchased, or that delivery of a security must be made before payment is received. In such cases, default by a broker or bank (the "Counterparty") through whom the relevant transaction is effected might result in a loss being suffered by Funds investing in non-developed market securities.

The Company will seek, where possible, to use Counterparties whose financial status is such that this risk is reduced. However, there can be no certainty that this risk will be successfully eliminated for the Funds, particularly as Counterparties operating in Emerging Markets, Frontier Markets and other non-developed markets frequently lack the substance or financial resources of those in developed countries.

There may also be a danger that, because of uncertainties in the operation of settlement systems in individual markets, competing claims may arise in respect of securities held by or to be transferred to the Funds. Furthermore, compensation schemes may be non-existent or limited or inadequate to meet the Company's claims in any of these events.

Investments in the Russian Federation are subject to certain heightened risks with regard to the ownership and custody of securities. In these countries the ownership is evidenced by entries in the books of a company or its registrar (which is neither an agent nor responsible to the Depositary). No certificates representing ownership of such companies will be held by the Depositary or any of its local correspondents or in an effective central depository system. As a result of this system and the lack of effective state regulation and enforcement, the Company could lose its registration and ownership of such securities through fraud, negligence or even mere oversight. However, in recognition of such risks, the relevant correspondent to the Depositary follows increased "due diligence" procedures. The correspondent has entered into agreements with company registrars and will only permit investment in those companies that have adequate registrar procedures in place. In addition, the settlement risk is minimised as the correspondent will not release cash until registrar extracts have been received and checked. In addition, Fixed Income Securities in these countries have an increased custodial risk associated with them as such securities are, in accordance with market practice, held in custody with institutions which may not have adequate insurance coverage to cover loss due to theft, destruction or default whilst such assets are in its custody.

Other risks could include, by way of example, controls on foreign investment and limitations on the repatriation of capital and the exchange of local currencies for global reserve currencies such as US\$, the impact on the economy as a result of religious or ethnic unrest.

In addition investments in India may be subject to the withdrawal or non-renewal of the Investment Adviser's Foreign Institutional Investor licence.

Corporate and Sovereign Debt

Both corporate and sovereign debt will be subject to high risk in emerging markets, frontier markets and other non-developed markets will not be required to meet a minimum rating standard and may not be rated for creditworthiness by any internationally recognised credit rating organisation.

The issuer or governmental authority that controls the repayment of such a non-developed country's debt may not be able or willing to repay the principal and/or interest when due in accordance with the terms of such debt. As a result of the foregoing, a government obligor may default on its obligations. If such an event occurs, the Company and/or the Management Company may have limited legal recourse against the issuer and/or guarantor. Remedies must, in

some cases, be pursued in the courts of the defaulting party itself, and the ability of the holder of foreign government debt securities to obtain recourse may be subject to the political climate in the relevant country.

In addition, no assurance can be given that the holders of commercial debt will not contest payments to the holders of other foreign government debt obligations in the event of default under their commercial bank loan agreements.

Investments in China

Certain Funds may invest in securities or instruments which have exposure to the Chinese market (where "China" or the "PRC" means the People's Republic of China (except where the context requires, and only for the purpose of the Prospectus, references to the "PRC" or "China" do not include Hong Kong, Macau and Taiwan)). The exposure may be obtained via the Qualified Foreign Investor ("QFI") scheme or the Stock Connect. Other than risks involved in investments in emerging markets, as well as other risks of investments generally, as described in this section, which are applicable to investments in China, investors should also note the additional specific risks below. Investments in China involve a risk of a total loss due to factors such as government action or inaction, market volatility and reliance on primary trading partners.

Investments via QFI

Under current China law and regulations, investments in the Chinese domestic securities market (China A-Shares and other domestic securities as permitted) can be made by or through holders of a QFI licence subject to applicable Chinese regulatory requirements (the "QFI Regulations"). The Funds may invest in China indirectly via access products such as participation notes, equity linked notes or similar financial instruments where the underlying assets consist of securities issued by companies quoted on regulated markets in China, and/or the performance of which is linked to the performance of securities issued by companies quoted on regulated markets in China ("Access Products"). The relevant Funds will not satisfy the criteria to qualify as a QFI and to gain direct exposure to the China A-Shares market, investment will be made through managers or issuers of such schemes, notes or instruments who possess QFI licenses.

Access Products are designed to mirror the returns on the underlying China A-Shares and are generally subject to the terms and conditions which reflect the underlying QFI Regulations and may also be subject to the terms and conditions imposed by the issuers. These terms may lead to delays in implementing the Investment Adviser's investment strategy due to the restrictions they may place on the issuer acquiring or disposing of the securities underlying the Access Products or on the implementation of realisations and payment of realisation proceeds to the Fund.

Furthermore, Access Products can be illiquid as there may be no active market in such securities. In the case of a default, the Fund could become subject to adverse market movements while replacement transactions are executed. In addition, there is a risk that the issuer will not settle a transaction due to a credit or liquidity problem, thus causing the Fund to suffer a loss.

In addition, upon request from CSRC, QFIs should report information about its offshore hedging positions and other information related to its securities and futures investment in the PRC. PRC stock exchanges are also entitled to require QFIs to report the positions held by its underlying investors in securities, derivatives and shares if there is any abnormal trading which may seriously affect the normal trading order or suspected violations of applicable laws and regulations. Therefore, the position of the Fund in the Access Products may be disclosed to the PRC regulators or PRC stock exchanges upon their requests.

QFI Regulatory Risks

Actions of the relevant manager or issuer which violate the QFI Regulations could result in the revocation of, or other regulatory action against, the relevant QFI licence as a whole, and may impact on the Fund's exposure to Chinese securities as the relevant scheme, note or instrument may be required to dispose its holdings in Chinese securities. In addition, a Fund may also be impacted by the rules and restrictions under the QFI Regulations (including rules on permissible investment scope, shareholding restrictions, and repatriation of principal and profits), which may consequently have an adverse impact on the liquidity and/or investment performance of the Fund.

The QFI Regulations which regulate investments by QFIs in China may be subject to further revisions in the future. The application and interpretation of the QFI Regulations are relatively untested before PRC courts and there is limited certainty as to how they will be applied. There is no assurance whether future revisions to the QFI Regulations or application of the QFI Regulations may or may not adversely affect a Fund's investments in China.

QFI Custody Risks

Where a Fund invests in China A-Shares or other securities in China through a QFI, such securities will be maintained by one or more custodian bank(s) (the "QFI Custodian") appointed by the QFI in accordance with the QFI Regulations and the relevant China A-Shares will be held through a securities account with the China Securities

Depository and Clearing Corporation Limited ("ChinaClear"). Such account may be opened based on the naming convention of "QFI – Clients' Money", "QFI – Fund Name" or "QFI – Client Name", and not in the sole name of such Fund, and the assets within such account may be held for and on behalf of clients of the QFI including but not limited to such Fund. The Chinese regulators have affirmed their recognition of the concepts of nominee holders and beneficiary owners, and the QFI Regulations also specify that the assets held within such account belongs to the client or the fund and should be independent from the assets of the QFI or QFI Custodian. However, if the QFI does not open a designated account specifically for the Fund and only put Fund's money in an omnibus account (i.e., the account named as "QFI-Clients' Money"), the assets of such Fund held within such account may be subject to a risk of being mingled with other clients and cannot be segregated from each other. If the Fund purchases Access Products issued by the QFI, the purchase proceeds will be treated as part of the assets of the QFI, rather than client money under the management of the QFI.

Investors should also note that cash deposited in the cash account of the relevant Funds with the QFI Custodian may not be segregated but may be a debt owing from the QFI Custodian to the relevant Funds as a depositor. Such cash may be co-mingled with cash belonging to other clients of the QFI Custodian.

Investments in China A-Shares via Stock Connect

The Stock Connect (currently comprising of the Shanghai Stock Connect and the Shenzhen Stock Connect), is a securities trading and clearing linked program developed by the Hong Kong Exchanges and Clearing Limited ("HKEX"), the Shanghai Stock Exchange ("SSE"), the Shenzhen Stock Exchange ("SZSE") and ChinaClear with an aim to achieve the mutual stock market access between the PRC and Hong Kong. The Shanghai Stock Connect and the Shenzhen Stock Connect are operated independently from each other, but are similar in respect to the fundamental principles, operational mechanism and regulatory framework.

The Stock Connect comprises a Northbound trading link and a Southbound trading link. Under the Northbound trading link, Hong Kong and overseas investors, such as the Company, through its Hong Kong brokers and a securities trading service company established by the Stock Exchange of Hong Kong Limited ("SEHK"), respectively in Shanghai (for trading under the Shanghai Stock Connect) and Shenzhen (for trading under the Shenzhen Stock Connect), may be able to trade certain eligible China A-Shares listed on SSE/SZSE by routing orders to SSE/SZSE. Under the Southbound trading link, investors in the PRC will be able to trade certain stocks listed on SEHK. Under a joint announcement issued by the Securities and Futures Commission ("SFC") and the China Securities Regulatory Commission ("CSRC") on 10 November 2014, the Shanghai Stock Connect commenced trading on 17 November 2014. The Shenzhen Stock Connect commenced trading on 5 December 2016.

Under the Stock Connect, the Company, through its Hong Kong brokers may trade certain eligible shares listed on SSE/SZSE. As for trading on SSE, the eligible China A-Shares include all the constituent stocks from time to time of the SSE 180 Index and the SSE 380 Index, and all the SSE listed A-Shares that are not included as constituent stocks of the relevant indices but which have corresponding H-Shares listed on SEHK (companies that issue both A-Shares on SSE/SZSE and H-Shares on SEHK are referred to as "A+H Shares Companies"). As for trading on SZSE, the eligible China A-Shares include all constituent shares of the SZSE Constituent Index and the SZSE Small/Mid Cap Innovation Index issued by a company with a market capitalisation of RMB6 billion or above, all eligible shares on the ChiNext markets, and China A-Shares issued by A+H Shares Companies listed on SZSE. SSE/SZSE-listed shares which are not traded in Renminbi ("RMB") and SSE/SZSE-listed shares which are included in the "risk alert board" or under suspension of listing are explicitly excluded from the eligible shares under the Stock Connect. It is expected that the list of eligible securities will be subject to review and adjustment (in particular, the adjustment along with the changes of the constituent China A-Shares in the relevant indices).

The Hong Kong Securities Clearing Company Limited ("**HKSCC**"), a wholly-owned subsidiary of HKEX, and ChinaClear are responsible for the clearing, settlement and the provision of depository, nominee and other related services of the trades executed by their respective market participants and investors. The China A-Shares traded through the Stock Connect are issued in scripless form, and investors will not hold any physical China A-Shares.

Although HKSCC does not claim proprietary interests in the SSE/SZSE securities held in its omnibus stock account in ChinaClear, ChinaClear as the share registrar for SSE/SZSE listed companies will still treat HKSCC as one of the shareholders when it handles corporate actions in respect of such SSE/SZSE securities. Stock Connect trades are settled in RMB and investors must have timely access to a reliable supply of RMB in Hong Kong, which cannot be guaranteed.

In addition to paying trading fees, levies and stamp duties in connection with trading in the China A-Shares, the Funds investing via the Stock Connect may be subject to new fees arising from trading of the China A-Shares via the Stock Connect which are yet to be determined and announced by the relevant authorities.

Liquidity and Volatility Risk

The existence of a liquid trading market for China A-Shares may depend on whether there is supply of, and demand for, China A-Shares. The price at which securities may be purchased or sold by the relevant Funds and the Net Asset

Value of such Funds may be adversely affected if trading markets for China A-Shares are limited or absent. The China A-Share market may be more volatile and unstable (for example, due to the risk of suspension of a particular stock or government intervention). Market volatility and settlement difficulties in the China A-Share markets may also result in significant fluctuations in the prices of the securities traded on such markets and thereby may affect the value of the relevant Funds.

Suspension Risk

It is contemplated that both SEHK and SSE/SZSE have the right to suspend or limit trading in any security traded on the relevant exchange if necessary for ensuring an orderly and fair market and that risks are managed prudently. In particular, trading in any China A-Share security on SSE/SZSE is also subject to the trading band limits applicable to each China A-Share. Any trading suspension and/or trading band limit may render it impossible for the relevant Funds to liquidate positions and could thereby expose the Funds to significant losses. Further, when the suspension is subsequently lifted, it may not be possible for the Funds to liquidate positions at a favourable price, which could thereby expose the affected Funds to significant losses. Finally, where a suspension is effected, the relevant Funds' ability to access the PRC market will be adversely affected.

Quota and Other Limitations

Although the Stock Connect is the first program allowing non-Chinese investors to trade the China A-Shares without a license and there is no longer an aggregate quota limitation, trading of China A-Shares through the Stock Connect is still subject to a daily quota ("Daily Quota"), which limits the maximum net buy value of cross-boundary trades under the Stock Connect each day. Northbound trading and Southbound trading under each of the Shanghai Stock Connect and the Shenzhen Stock Connect will be subject to a separate set of Daily Quota. The Northbound Daily Quota for each of the Shanghai Stock Connect and the Shenzhen Stock Connect is currently and respectively set at RMB52 billion. Quota limitations may prevent the Funds from purchasing the Stock Connect securities when it is otherwise advantageous to do so. In particular, once the remaining balance of the relevant Daily Quota drops to zero or the Daily Quota is exceeded, buy orders will be rejected (although investors will be permitted to sell their cross-boundary securities regardless of the quota balance).

Differences in Trading Day

Because Stock Connect trades are routed through Hong Kong brokers and the SEHK, Stock Connect will only operate on days when both the PRC and Hong Kong markets are open for trading and when banks in both markets are open on the corresponding settlement days. Therefore, it is possible that there are occasions when it is a normal trading day for the PRC market but the relevant Funds cannot carry out any China A-Shares trading via the Stock Connect. As a result, prices of the relevant China A-Shares may fluctuate at times when the Funds are unable to add to or exit its position.

Additionally, an investor cannot sell the securities purchased on the current trading day on SSE/SZSE, which may restrict the Funds' ability to invest in China A-Shares through Stock Connect and to enter into or exit trades where it is advantageous to do so on the same trading day.

Eligibility of Shares

Only certain China A-Shares are eligible to be accessed through Stock Connect. Such securities may lose their eligibility at any time. When a China A-Share is recalled from the scope of eligible shares for trading via the Stock Connect, the China A-Share can only be sold but restricted from being bought. This may affect the investment portfolio or strategies of the relevant Funds, for example, if the Adviser wishes to purchase a China A-Share which is recalled from the scope of eligible shares.

Operational Uncertainty

Because Stock Connect is relatively new, its effects on the market for trading China A-Shares are uncertain. In addition, the trading, settlement and IT systems required to operate Stock Connect are relatively new and continuing to evolve. In particular, the Stock Connect is premised on the functioning of the operational systems of the relevant market participants. Market participants are permitted to participate in this program subject to meeting certain information technology capability, risk management and other requirements as may be specified by the relevant exchange and/or clearing house. The securities regimes and legal systems of the two markets differ significantly and market participants may need to address issues arising from the differences on an on-going basis. There is no assurance that the systems of the SEHK and market participants will function properly or will continue to be adapted to changes and developments in both markets. In the event that the relevant systems do not function properly, trading through Stock Connect could be disrupted and the relevant Funds' ability to access the China A-Share market may be adversely affected and the Fund may not be able to effectively pursue its investment strategy.

Other Legal and Regulation Risks

Stock Connect is subject to regulation by both Hong Kong and China. The current regulations are untested and there is no certainty as to how they will be applied. In addition, the current regulations are subject to change and there can be no assurance that the Stock Connect will not be abolished. There can be no assurance that further regulations will not affect the availability of securities in the program, the frequency of redemptions or other limitations. Additional shareholder restrictions and disclosure requirements might also be applicable to the Company as a result of their investments in China A-Shares via Stock Connect.

Legal/Beneficial Ownership

In China, Stock Connect securities are held on behalf of ultimate investors (such as the Company) by the HKSCC as nominee. HKSCC in turn holds the SSE/SZSE shares, as the nominee holder, through an omnibus securities account in its name registered with ChinaClear. While Chinese regulators have affirmed that the ultimate investors hold a beneficial interest in Stock Connect securities, the law surrounding such rights is in its early stages and the mechanisms that beneficial owners may use to enforce their rights are untested and therefore pose uncertain risks. Further, courts in China have limited experience in applying the concept of beneficial ownership and the law surrounding beneficial ownership will continue to evolve as they do so. There is accordingly a risk that as the law is tested and developed, the Company's ability to enforce its ownership rights may be negatively impacted. Because of this uncertainty, in the unlikely event that HKSCC becomes subject to winding up proceedings in Hong Kong it is not clear if the SSE/SZSE shares will be regarded as held for the beneficial ownership of the Funds or as part of the general assets of HKSCC available for general distribution to its creditors. Furthermore, the Company may not be able to participate in corporate actions affecting Stock Connect securities due to time constraints or for other operational reasons. Similarly, the Company will not be able to vote in shareholders' meetings except through HKSCC and will not be able to attend shareholders' meetings.

Clearing and Settlement Risk

ChinaClear and HKSCC have established the clearing links and each becomes a participant of each other to facilitate clearing and settlement of cross-boundary trades. For cross-boundary trades initiated in a market, the clearing house of that market will on one hand clear and settle with its own clearing participants, and on the other hand undertake to fulfil the clearing and settlement obligations of its clearing participants with the counterparty clearing house.

As the national central counterparty of the PRC's securities market, ChinaClear operates a comprehensive network of clearing, settlement and stock holding infrastructure. ChinaClear has established a risk management framework and measures that are approved and supervised by CSRC. The chances of ChinaClear default are considered to be remote. In the remote event of a ChinaClear default, HKSCC's liabilities in SSE/SZSE shares under its market contracts with clearing participants will be limited to assisting clearing participants in pursuing their claims against ChinaClear. HKSCC should in good faith, seek recovery of the outstanding stocks and monies from ChinaClear through available legal channels or through ChinaClear's liquidation. In that event, the relevant Fund may suffer delay in the recovery process or may not fully recover its losses from ChinaClear.

Pre-Trade Requirements and Special Segregated Accounts

PRC regulations require that before an investor sells any share, there should be sufficient shares in the account; otherwise SSE/SZSE will reject the sell order concerned. SEHK will carry out pre-trade checking on China A-Share sell orders of its participants (i.e. the stock brokers) to ensure there is no over-selling.

If a Fund intends to sell certain China A-Shares it holds, it must transfer those China A-Shares to the respective accounts of its broker(s) before the market opens on the day of selling (the "**trading day**"). If it fails to meet this deadline, it will not be able to sell those shares on the trading day. Because of this requirement, a Fund may not be able to dispose of its holdings of China A-Shares in a timely manner.

In addition, as the broker(s) of the Fund will hold and safekeep the Chinese A-Shares before the trading day, there is a risk that the creditors of the broker(s) will seek to assert that the Chinese A-Shares are owned by the brokers rather than the Funds if it is not made clear that the broker(s) act as a custodian in respect of the Chinese A-Shares for the benefit of the Funds.

Alternatively, if the relevant Fund maintains its SSE/SZSE shares with a custodian which is a custodian participant or general clearing participant participating in the Hong Kong Central Clearing and Settlement System ("CCASS"), the Fund may request such custodian to open a special segregated account ("SPSA") in CCASS to maintain its holdings in SSE/SZSE shares under the enhanced pre-trade checking model. Each SPSA will be assigned a unique "Investor ID" by CCASS for the purpose of facilitating the Stock Connect system to verify the holdings of an investor such as a Fund. Provided that there is sufficient holding in the SPSA when a broker inputs the relevant Fund's sell order, the Fund will only need to transfer SSE/SZSE shares from its SPSA to its broker's account after execution and not before placing the sell order and the Fund will not be subject to the risk of being unable to dispose of its holdings of China A-Shares in a timely manner due to failure to transfer of China A-Shares to its brokers in a timely manner.

In addition, these pre-trade requirements may, as a practical matter, limit the number of brokers that the Funds may use to execute trades. In relation to transactions executing through a SPSA order, the Fund, as the investor, may at most designate 20 brokers currently. While the Funds may use SPSA in lieu of the pre-trade check, many market participants have yet to fully implement IT systems necessary to complete trades involving securities in such accounts in a timely manner. Market practice as well as governmental policies with respect to SPSA is continuing to evolve.

Chinese Interbank Bond Market ("CIBM")

CIBM is an OTC market outside the two main stock exchanges in the PRC (i.e. SSE and SZSE) and was established in 1997. On CIBM, institutional investors (including domestic institutional investors but also QFIs as well as other offshore institutional investors, subject to authorization) trade sovereign, government and corporate bonds on a one-to-one quote-driven basis. CIBM accounts for more than 95% of outstanding bond values of total trading volume in the PRC as of 2019.

The main debt instruments traded on CIBM include government bonds, financial bonds, corporate bonds, bond repo, bond lending, People's Bank of China ("**PBOC**") bills, and other financial debt instruments.

CIBM is regulated and supervised by PBOC. PBOC is responsible inter alia for establishing listing, trading, functioning rules applying to CIBM and supervising the market operators of CIBM, and CSRC is responsible for taking enforcement actions against illegal activities on CIBM. CIBM facilitates two trading models: (i) bilateral negotiation and (ii) click-and-deal. Under the China Foreign Exchange Trading System ("CEFTS"), which is the unified trading platform for CIBM, negotiation is applied to all inter-bank products while one-click trading is only applied to spot bonds and interest rate derivatives.

The market-maker mechanism, whereby an entity ensures bilateral quotations for bonds, was officially introduced in 2001 to improve market liquidity and enhance efficiency. Deals through market making can enjoy benefits such as lower trading and settlement costs.

Bond transactions must be conducted by way of bilateral trading through independent negotiations and be concluded on a transaction by transaction basis. Bid and ask prices for primary bond transactions and repurchase interest rates must be determined independently by the parties to the transaction. Both parties to a transaction shall typically, in accordance with the contract, promptly send instructions for delivery of bonds and funds, and shall have sufficient bonds and funds for delivery on the agreed delivery date.

China Central Depository & Clearing Co., Ltd ("CCDC") or Shanghai Clearing House ("SHCH"), depending on where bonds are deposited, will deliver bonds on time according to the instructions matching with elements sent by both parties to a transaction. Fund clearing banks (e.g. settlement agent banks of foreign institutional investors) will handle the transfer and settlement of bond transaction payments on behalf of participants in a timely manner.

Investors should be aware that trading on CIBM exposes the Fund to increased counterparty and liquidity risks.

Risks associated with the investment through Bond Connect

In addition to opening an account in China to access the CIBM (the so called "direct CIBM access"), some Funds may invest in the bonds tradable in the PRC ("Bond Connect Securities") through connection between the PRC and Hong Kong financial infrastructure institutions ("Bond Connect").

Regulatory risk

Any laws, rules, regulations, policies, notices, circulars or guidelines published or applied by any of the Bond Connect Authorities (as defined below) are subject to change from time to time in respect of Bond Connect or any activities arising from Bond Connect (the "Applicable Bond Connect Laws and Rules") and there can be no assurance that Bond Connect will not be abolished. The relevant Fund may be adversely affected as a result of any change in the Applicable Bond Connect Laws and Rules. "Bond Connect Authorities" refers to the exchanges, trading systems, settlement systems, governmental, regulatory or tax bodies which provide services and/or regulate Bond Connect and activities relating to Bond Connect, including, without limitation, the PBOC, the Hong Kong Monetary Authority ("HKMA"), the Hong Kong Exchanges and Clearing Limited, CFETS, the Central Moneymarkets Unit of the HKMA ("CMU"), CCDC and SHCH and any other regulator, agency or authority with jurisdiction, authority or responsibility in respect of Bond Connect.

No off-market transfer

Pursuant to the Applicable Bond Connect Laws and Rules, the transfer of Bond Connect Securities between two members of CMU and between two CMU sub-accounts of the same CMU Member is not allowed.

No amendment of orders, limited cancellation of orders

Pursuant to the Applicable Bond Connect Laws and Rules, instructions relating to sell and buy orders for Bond Connect Securities may only be cancelled in limited circumstances pursuant to the Applicable Bond Connect Laws and Rules and that instructions may not be amended.

Hedging Activities

Hedging activities are subject to the Applicable Bond Connect Laws and Rules and any prevailing market practice and there is no guarantee that the Fund will be able to carry out hedging transactions at terms which are satisfactory to the Management Company, the relevant Investment Adviser and the relevant Sub-Adviser. The Fund may also be required to unwind its hedge in unfavourable market conditions.

Tax

The treatment of tax under the Applicable Bond Connect Laws and Rules is not entirely clear. Accordingly, where the Applicable Bond Connect Laws and Rules require a custodian/ clearing house / any other agent stipulated by such rules to withhold any tax, or where such custodian / clearing house / any other agent has a reasonable basis for believing that such withholding may be required, the custodian / clearing house / any other agent may do so at the rate required by the regulation, or if in the custodian's opinion the Applicable Bond Connect Laws and Rules are not very clear on the rate, at such rate as the custodian/ clearing house / any other agent may, reasonably determine to be appropriate. Tax may be withheld on a retroactive basis.

Nominee Holding Structure

Bond Connect Securities will be held by CMU, opening two nominee accounts with CCDC and SHCH. While the distinct concepts of "nominee holder" and "beneficial owner" are generally recognized under the Applicable Bond Connect Laws and Rules, the application of such rules is untested, and there is no assurance that PRC courts will recognise such rules, e.g. in liquidation proceedings of PRC companies or other legal proceedings.

Variable Interest Entities

A Fund may gain economic exposure to certain operating companies in China through legal structures known as variable interest entities ("VIEs"). From time to time, a Fund's investments in US-listed special purpose companies relying on VIE structures to consolidate China-based operations may be significant. In a VIE structure, a China-based operating company ("Operating Company") typically establishes an offshore special purpose company ("Offshore Company") in another jurisdiction, such as the Cayman Islands, which then enters into service and other contracts with the Operating Company to replicate the rights and obligations similar to equity ownership in such Operating Company, and issues shares on a foreign exchange, like the New York Stock Exchange or Hong Kong Exchange. Under VIEs, investors hold stock in the Offshore Company rather than directly in the Operating Company and the Offshore Company may not own stock or other equity in the Operating Company. Certain Chinese companies have adopted VIEs to facilitate foreign investment access where they are subject to foreign investment prohibition or restriction. Through a VIE arrangement, the Operating Companies may indirectly access capital from foreign investors without distributing ownership of the Operating Companies to foreign investors.

Investments in VIEs are subject to risks in addition to those generally associated with investments in China. The VIE structure has not been formally recognized or approved by any Chinese regulatory body. The validity and enforceability of VIE agreements remain untested in the Chinese courts. For example, local government authorities may determine that such structures do not comply with applicable laws and regulations, including those relating to restrictions on foreign ownership. In such event, the Offshore Company and/or the Operating Company may be subject to penalties, revocation of business and operating licenses or forfeiture of foreign ownership interests. Further, breaches of the contractual arrangements, changes in Chinese law with respect to enforceability or permissibility of these arrangements or failure of these contracts to function as intended would likely adversely affect an investment in a VIE. In addition, VIEs are also subject to the risk of inconsistent and unpredictable application of Chinese law, that the Offshore Company may lose effective control over the Operating Company and that the equity owners of the Operating Company may have interests conflicting with those of the Offshore Company's investors. There is also uncertainty related to the Chinese taxation of VIEs and the Chinese tax authorities may take positions which result in increased tax liabilities. Thus, investors, such as a Fund, faces risks and uncertainty about future actions or intervention by the government of China at any time and without notice that could suddenly and significantly adversely affect VIEs and the enforceability of the Offshore Company's contractual arrangements with the Operating Company. If these risks materialize, the value of investments in VIEs could be significantly adversely affected and a Fund could incur significant losses with no recourse available.

Exposure to the Euro and the Eurozone

The "Eurozone" is an economic and monetary union of 19 European member states that have adopted the Euro as their common currency and sole legal tender. The success of the Euro and the Eurozone is therefore dependant on

the general economic and political condition of each member state, as well as each state's credit worthiness and the willingness of the members to remain committed to monetary union and support for the other members. Currently, there are widely held concerns in the market regarding the credit risk associated with certain sovereigns, including some member states of the Eurozone, and the continued viability of the Eurozone. Risk to the Company includes the possibility of exit of individual countries from the Euro, full breakup of the Eurozone or other circumstances which may result in the emergence or re-introduction of national currencies.

Default by any state on its Euro debts or a material decline in the credit rating of any Eurozone state could have a material negative impact on the Company and its investments. A number of the Funds of the Company may operate in Euro and/or may hold Euro denominated assets either directly or as collateral and may experience a reduction of the value and/or liquidity of their investments as a result of events in the Eurozone regardless of the measures the Investment Adviser(s) or Board may seek to take to reduce this risk.

In addition, the Management Company and/or the Company's counterparties, banks, custodians and service providers may have direct or indirect exposure to these countries or currency and a default or credit decline could impact their ability to meet their obligations to and/or perform services for the Company. In the event of one or more member states exiting the Eurozone, or the abandonment of the Euro entirely, there may be material negative impact on some or all Funds of the Company and the value of investments, including risk of redenomination from Euro into another currency, possible capital controls and legal uncertainty as to the ability to enforce obligations and debts.

Prospective shareholders should inform themselves as to the risks surrounding the Eurozone crisis and the associated risk of an investment in the Company, taking into account the uncertainty as to how the Eurozone crisis and more general global economic situation will continue to evolve.

Holdings Concentration

Some Funds may invest in a relatively small number of investments. Concentrated portfolios may be more volatile than more diversified portfolios with a larger number of investments and may be more significantly impacted by a decline in the value or circumstance of any one stock, asset classes or sector.

Real Estate Industry

There are special risk considerations associated with investing in the real estate industry. Securities such as closed-end Real Estate Investment Trusts (REITS), Real Estate Operating Companies (REOCS) and the securities of companies principally engaged in the real estate industry may be subject to these risks. REITS and REOCS are companies that acquire and/or develop real property for long term investment purposes. They invest the majority of their assets directly in real property and derive their income primarily from rents. These risks include: the cyclical nature of real estate values, risks related to general and local economic conditions, overbuilding and increased competition, increases in property taxes and operating expenses, demographic trends and variations in rental income, changes in zoning laws, casualty or condemnation losses, environmental risks, regulatory limitations on rents, changes in neighbourhood values, related party risks, changes in the appeal of properties to tenants, increases in interest rates and other real estate capital market influences. Generally, increases in interest rates will increase the costs of obtaining financing, which could indirectly decrease the value of a Fund investing in the Real Estate Industry.

By investing in REITS and REOCS through the Fund the investor will bear not only his proportionate share of the management fees of the Fund but also indirectly, management expenses of the underlying REITS and REOCS. REITS and REOCS and distributions by REITS and REOCS may be subject to withholding or other taxes imposed by the authorities in that market.

Infrastructure

There are special risk considerations associated with investing in the securities of companies principally engaged in the infrastructure industry. Infrastructure-related companies are subject to a variety of factors that may adversely affect their business or operations including high interest costs in connection with capital construction programs, costs associated with compliance with and changes in environmental and other regulations, difficulty in raising capital in adequate amounts on reasonable terms in periods of high inflation and unsettled capital markets, the effects of surplus capacity, increased competition from other providers of services in a developing deregulatory environment, uncertainties concerning the availability of fuel at reasonable prices, the effects of energy conservation policies and other factors.

Additionally, infrastructure-related entities may be subject to regulation by various governmental authorities and may also be affected by governmental regulation of rates charged to customers, government budgetary constraints, service interruption due to environmental, operational or other mishaps and the imposition of special tariffs and changes in tax laws, regulatory policies and accounting standards. Other factors that may affect the operations of infrastructure-related companies include innovations in technology that could render the way in which a company delivers a product or service obsolete, significant changes to the number of ultimate end-users of a company's products, increased susceptibility to terrorist acts or political actions, risks of environmental damage due to a

company's operations or an accident, and general changes in market sentiment towards infrastructure and utilities assets.

In the event that any of the risks associated with the infrastructure industry materialise, the value of securities issued by companies engaged in the infrastructure business may decline. To the extent that a Fund is invested in such securities, this may result in a corresponding decline in the Net Asset Value per share of that Fund, potentially uncorrelated to the rest of the equity market.

Companies engaged in the infrastructure business may also include Real Estate Investment Trusts ("closed-end REITS") and collective investment vehicles with exposure to infrastructure assets. Investors should refer to special risk considerations applicable to the Real Estate Industry and Collective Investment Vehicles.

Commodities

Certain Funds may invest in commodity-related assets, in accordance with their investment objective. Such Funds will only take indirect exposure to commodities, by investing into eligible structured products such as commodity-linked notes, commodity ETFs and/or derivative instruments referencing one or more commodity indices qualifying as financial indices under UCITS regulations.

Commodity related assets are highly volatile. Commodity markets are influenced by, among other things, changing supply and demand relationships, weather, governmental, agricultural, commercial and trade programmes and policies designed to influence commodity prices, world political and economic events, and changes in interest rates.

Volatility Strategies

Certain Funds may invest, either directly or indirectly via investment into collective investment schemes in strategies which seek to take advantage of increases or decreases in volatility. Investment in such strategies may result in an increase in total portfolio volatility of the Fund. The risks of such investments are related to the risks associated with the underlying asset class on which the strategy invests (e.g. equity risk, commodity risk, liquidity risk, etc.). In periods of high market volatility the value of investments in volatility strategies may be subject to drawdowns which exceed those experienced in the markets and asset classes upon which the volatility strategy draws. Exposure to extreme market volatility may not be fully hedged which may result in a decrease in portfolio value. Funds which invest in volatility strategies will be exposed to derivative risk and basis risk, which is the risk that exposure gained through derivatives to an underlying asset or assets will not be perfectly correlated to the price movements of the assets themselves.

Special Purpose Acquisition Companies

Certain Funds may invest in special purpose acquisition companies ("SPACs"). A SPAC is a publicly traded company that raises investment capital for the purpose of acquiring or merging with an existing company. Typically, the acquisition target is an existing private company that wants to trade publicly, which it accomplishes through an acquisition by, or combination with, a SPAC rather than by conducting a traditional initial public offering ("IPO"). A SPAC is a blank check company, meaning it does not have any operating history or ongoing business other than seeking to acquire an ongoing business. The identity of the acquisition target is typically not known at the time the SPAC seeks investors.

A SPAC may raise additional funds for a range of purposes, including in order to fund the acquisition, provide post-acquisition working capital, redeem the publicly traded shares as requested by its existing shareholders or some combination of these purposes. This additional fundraising may be in the form of a private placement of a class of equity securities or the issuance of debt. Where in the form of equity, the equity securities sold in this kind of fundraising are generally the same class of securities that trade on the exchange on which the shares of the SPAC are listed. Where in the form of debt, the debt could be secured by the assets of the SPAC, by the operating company existing after the acquisition, or it could be unsecured. The debt may also be investment grade debt or below investment grade debt.

A Fund would typically invest in equity securities of SPACs or similar special purpose entities in a private placement transaction designed to help fund an acquisition by the SPAC. In connection with the private placement, the SPAC usually agrees to file a registration statement covering the shares purchased in the private placement on an expedited basis after the transaction is complete. Until that registration statement becomes effective, the securities purchased by a Fund will not be publicly tradable unless an exemption from registration is available. Exemptions in these types of transactions become available one year after the date of the business combination, so it is critical that the registration statement become effective soon after the investment in order to create liquidity for the securities purchased by the Fund in the private placement.

An investment in a SPAC prior to an acquisition is subject to the risks that the proposed acquisition or merger may not obtain the requisite approval of SPAC shareholders, may require governmental or other approvals that it fails to obtain or that an acquisition or merger, once effected, may prove unsuccessful and lose value.

Investments in SPACs are also subject to the risks that apply to investing in any IPO, including the risks associated with companies that have little operating history as public companies, including unseasoned trading, a limited number of shares available for trading (i.e. "free float") and limitations to the availability of information about the issuer. In addition, like IPO issuers, the market for newly-public may be volatile, and share prices of newly-public companies have historically fluctuated significantly over short periods of time. Although some IPOs may produce high returns, such returns are not typical and may not be sustainable. Any equity investments made in the SPAC in connection with a proposed business combination will be diluted by the acquisition itself and any further fundraising post-acquisition by the acquired operating business.

Investments in Saudi Arabia

Special Risk Considerations Relating to the QFI Regime and Investments in Saudi Arabian Equities

A Fund's ability to achieve its investment objective depends on the ability of the Investment Adviser as a QFI, and the Fund as a QFI Client, to obtain and maintain their respective authorizations from the Capital Market Authority ("CMA"), and thereby allow the Fund to invest. A failure to obtain or maintain such authorization could limit the Fund's ability to gain exposure to Saudi securities and could increase the cost to the Fund of gaining such exposure. Even if it obtains QFI Client approval, the Fund will not have an exclusive investment quota and will be subject to foreign investment limitations and other regulations imposed by the CMA on QFIs and QFI Clients (individually and in the aggregate), as well as local market participants. If such limitations are reached or other market factors impact the Fund's ability to invest in Saudi securities or otherwise, the Fund may not be able to meet investor demand for Fund shares. If the Fund is unable to invest in a manner consistent with its investment objective, the Investment Adviser may reject new creation orders for Fund shares. Rejection of new creation orders could cause the Fund's shares to trade in the secondary market at a higher than usual premium or discount to Net Asset Value. QFI regulations and local market infrastructure are relatively new and have not been tested through multiple business cycles or market events, and the CMA may discontinue the QFI regime and other avenues for foreign investment in Saudi Arabian equities at any time, or impose additional barriers or restrictions on foreign investment. Any change in the QFI system generally, including the possibility of the Investment Adviser or the Fund losing their respective QFI and QFI Client status, may affect the Fund's ability to invest in Saudi securities.

Risk of Investing in Saudi Arabia

The ability of foreign investors (such as a Fund) to invest in Saudi Arabian issuers is new and untested. Such ability could be restricted or revoked by the Saudi Arabian government at any time, and unforeseen risks could materialize due to foreign ownership in such securities. In addition, the CMA places investment limitations on the ownership of Saudi Arabian issuers by foreign investors, including a limitation on the Fund's ownership of any single issuer listed on the Saudi Arabian Stock Exchange, which may prevent the Fund from investing in accordance with its strategy. Saudi Arabia is highly reliant on income from the sale of petroleum and trade with other countries involved in the sale of petroleum, and its economy is therefore vulnerable to changes in foreign currency values and the market for petroleum. As global demand for petroleum fluctuates, Saudi Arabia may be significantly impacted. Like most Middle Eastern governments, the government of Saudi Arabia exercises substantial influence over many aspects of the private sector. Although liberalization in the wider economy is underway, in many areas it has lagged significantly: restrictions on foreign ownership persists, and the government has an ownership stake in many key industries. The situation is exacerbated by the fact that Saudi Arabia is governed by an absolute monarchy. Saudi Arabia has historically experienced strained relations with economic partners worldwide, including other countries in the Middle East due to geopolitical events. Governmental actions in the future could have a significant effect on economic conditions in Saudi Arabia, which could affect private sector companies and the Fund, as well as the value of securities in the Fund's portfolio. Any economic sanctions on Saudi Arabian individuals or Saudi Arabian corporate entities, or even the threat of sanctions, may result in the decline of the value and liquidity of Saudi Arabian securities, a weakening of the Saudi riyal or other adverse consequences to the Saudi Arabian economy. In addition, Saudi Arabia's economy relies heavily on cheap, foreign labor, and changes in the availability of this labor supply could have an adverse effect on the economy.

Investments in securities of Saudi Arabian issuers involves risks not typically associated with investments in securities of issuers in more developed countries that may negatively affect the value of the Fund's investments. Such heightened risks may include, among others, expropriation and/or nationalization of assets, restrictions on and government intervention in international trade, confiscatory taxation, political instability, including authoritarian and/or military involvement in governmental decision making, armed conflict, crime and instability as a result of religious, ethnic and/or socioeconomic unrest. Although the political situation in Saudi Arabia is largely stable, Saudi Arabia has historically experienced political instability, and there remains the possibility that instability in the larger Middle East region could adversely impact the economy of Saudi Arabia. Political instability in the larger Middle East region has caused significant disruptions to many industries. Continued political and social unrest in these areas may negatively affect the value of securities in the Fund's portfolio.

Saudi Arabia Broker Risk

There are a number of different ways of conducting transactions in equity securities in the Saudi Arabian market. A Fund generally expects to conduct its transactions in a manner in which it would not be limited by Saudi Arabian regulations to a single broker. However, there may be a limited number of brokers who can provide services to the Fund, which may have an adverse impact on the prices, quantity or timing of Fund's transactions. In addition, the limited number of brokers available to the Fund may make the Fund more susceptible to credit loss or trading disruptions in the event of a default or business disruption by one or more of the available brokers. Should the Fund's ability to use one or more brokers be affected for any reason, this could disrupt the operations of the Fund and/or cause the Fund's shares to trade at a premium or discount to Net Asset Value. In addition, brokers in the Saudi Arabian market are not typically able to separate execution commissions from research costs, which means that the Fund may incur execution costs that exceed those in other markets where execution commissions and research costs are separated. The Fund may also incur losses due to the acts or omissions of its brokers in the execution or settlement of any transaction or in the transfer of any funds or securities.

Section 2

2.1 Share Class Description

The Management Company may create one or more different classes of Shares ("Share Classes") in each Fund. Each Share Class is represented by a code composed by a share class indicator and a share class sub-indicator. The first letter or number of that code denotes the Share Class Indicator (*e.g.* "A"), while further letters or numbers represent additional characteristics and in combination denote the Share Class Sub-Indicator (*e.g.* "H").

Not all Share Classes may be available in all Funds. Details on the available Share Classes can be obtained from the Management Company and at the Company's registered office.

Share Class Indicator	Target audience*	Minimum Initial Subscriptio n per Fund:	Minimum Holding
A	Retail investors who receive either investment advice or execution services from an intermediary and have agreed that intermediary may receive inducements.	N/A	N/A
B***	Retail investors who receive investment advice from an intermediary and have agreed that intermediary may receive inducements. The intermediary may receive an up-front payment from the distributor, which will be recouped from the investor over four years through an additional distribution fee. If an investor redeems within four years of its initial subscription a contingent deferred sales charge will be charged.	N/A	N/A
С	Retail investors who receive investment advice from an intermediary and have agreed that intermediary may receive inducements. The intermediary may receive an up-front payment from the distributor. If an investor utilising this model redeems within one year of its initial subscription a contingent deferred sales charge will be charged.	N/A	N/A
F	Retail investors who either receive advice from an intermediary or utilise an intermediary's order execution platform to invest in the Company. For intermediary business within the EEA and the UK, with effect from 3 January 2018 this share class is available for retail investors using intermediaries in either of the following situations: (1) where the intermediary is not allowed to accept and retain inducements due to the application of MiFID; or (2) where the intermediary has agreed with its client that it is not allowed to accept and retain inducements.	N/A	N/A
I	Investors, including retail investors, who invest either directly or through an intermediary. For intermediary business within the EEA and the UK, with effect from 3 January 2018 this share class is available for retail investors using intermediaries in either of the following situations: (1) where the intermediary is not allowed to accept and retain inducements due to the application of MiFID; or (2) where the intermediary has agreed with its client that it is not allowed to accept and retain inducements.	N/A	N/A

Share Class Indicator	Target audience*	Minimum	Minimum
		Initial	Holding
		Subscriptio	
		n per	
		Fund:	
J	At the discretion of the Management Company,	10,000,000*	10,000,000*
	institutional investors, including discretionary	*	*
	portfolio managers acting on behalf of their		
	clients, who invest when a Fund is newly		
	launched within a period determined by the		
	Investment Adviser in its absolute discretion and		
	invest no less than USD 10,000,000 in the		
	relevant Fund.		
N	The Management Company and its affiliates	N/A	N/A
	(whether for themselves or on behalf of clients)		
	or, at the discretion of the Management Company,		
	clients of the Management Company or its		
	affiliates. This share class is reserved for		
	institutional investors only.		
S	Pension funds or other Institutional Investors, as	40,000,000*	40,000,000*
	selected by the Management Company, and who	*	*
	invest at least USD 40,000,000 in the relevant		
	Fund.		
Z	Institutional Investors, including discretionary	N/A	N/A
	portfolio managers acting on behalf of their		
	clients.		

For the purposes of the table above, "Institutional Investor" means an institutional investor within the meaning of Article 174 of the Luxembourg Law of 2010. For investors who are subject to MiFID, Eligible Counterparties (as defined under MiFID) and per se Professional Clients (as defined under item I of annex II of MiFID) are expected to be eligible to invest in institutional share classes. Retail Clients and elective Professional Clients (as defined under MiFID) are typically not eligible for institutional share classes although some entities, such as local authority pension funds, are eligible. Please consult the application form for more information on eligibility and, in the case of doubt, please contact csilix@morganstanley.com.

* Where the Management Company in its absolute discretion determines that an investor does not fall within the target audience for a Share Class Indicator as described above, including where the investor has ceased to be part of the target audience after initial subscription, the Management Company may, upon providing such investor with one month's written notice, elect to either compulsorily redeem their holding or convert their holding into another Share Class. The Management Company may also reject subscriptions to any share class where the investor does not fall within the target audience or where, in relation to share classes A, B and C, the relevant intermediary has not been approved by the Management Company to subscribe for that share class.

This is at the Management Company's absolute discretion and the Management Company may choose not to exercise this power. For example, the Management Company may determine that an investor in Class E who has ceased to be employed by the Investment Adviser may retain their existing investment.

- ** These amounts may be in US Dollars (or the Euro, Yen or Sterling equivalent of the US Dollar amount). These minima may be waived or varied, in any particular case or generally, at the Management Company's discretion.
- *** With effect from 1 July 2018, Class B Shares are automatically converted free of charge into corresponding Class A Shares on or shortly after the fourth anniversary of the original subscription date for those Class B Shares. Further information is set out below under the heading "Share Class Indicator B Shares".

If the Net Asset Value of an investor's holding of a Class of Shares falls below the Minimum Holding amount shown above, the Management Company may, upon providing such investor with one month's written notice, elect to either compulsorily redeem their holding or convert their holding into another Share Class.

Restrictions apply to the purchase of Share Class Indicators A, B, C, F, I, J, N, S and Z Shares. First time applicants should contact the Management Company before submitting an Application Form for these Share Classes.

List of Share Class Sub-Indicators

Share Class Sub-Indicator	Characteristics (see section below)
The following Share Class Sub-Indicators are explained further in this Section 2.1 below	
Н	NAV Hedged Share Class

H1	Portfolio Hedged Share Class
H2	Index Hedged Share Class
Н3	NAV Hedged (Non-Deliverable Currency) Share Class
The following Share Class Sub-Indicators are	
explained further in the Section 2.10 "Dividend	
Policy"	
X	Distributing Share Class
R	Discretionary Distributing Share Class
M	Monthly Distributing Share Class

Currency Hedged Share Classes

Currency Hedged Share Classes seek to reduce the currency exposure of the Shareholder to currencies other than the "Hedged Share Class Currency". The Investment Adviser(s) will decide on the most appropriate hedging strategies for each Fund and a Fund-by-Fund summary is available from the registered office of the Company and is included in the Company's Annual and Semi-Annual Reports. Currency Hedged Share Classes are subdivided as follows:

NAV Hedged Share Classes (Share Class Sub-Indicator "H")

NAV Hedged Share Classes utilise hedging strategies that seek to reduce exposure to currency movements between the Hedged Share Class Currency and the Fund's Reference Currency. This involves hedging the Fund's Reference Currency to the Hedged Share Class Currency generally without reference to the currencies represented in the Fund's underlying investment portfolio (the "Investment Currencies").

Portfolio Hedged Share Classes (Share Class Sub-Indicator "H1")

Portfolio Hedged Share Classes utilise hedging strategies that seek to reduce exposure to currency movements between the Hedged Share Class Currency and the Fund's Investment Currencies. This involves hedging the Fund's Investment Currencies back to the Hedged Share Class Currency without reference to its Reference Currency.

Index Hedged Share Classes (Share Class Sub-Indicator "H2")

Index Hedged Share Classes utilise hedging strategies that seek to reduce exposure to currency movements between the Hedged Share Class Currency and the currencies of a Fund's benchmark index ("Index Currencies"). This involves hedging the Fund's Index Currencies back to the Hedged Share Class Currency without reference to the Fund's Reference Currency or its Investment Currency.

NAV Hedged (Non-deliverable Currency) Share Class (Share Class Sub-Indicator "H3")

The NAV Hedged (Non-deliverable Currency) Share Classes utilise similar hedging strategies to the NAV Hedged Share Classes, however, where the Hedged Share Class Currency is non-deliverable, the share class will be denominated, for subscription and redemption purposes, in a currency other than the Hedged Share Class Currency. For example, a Colombian Peso NAV Hedged Share Class may be subscribed or redeemed in US Dollar or Euro but the Fund's Reference Currency would be hedged to the Columbian Peso, notwithstanding it being denominated in US Dollar or Euro.

Sales Charge

The Sales Charge is an initial charge levied on subscription amounts, which is not contained within the price at which the Shares are issued.

Share Class Indicator	Bond Funds (excluding Emerging Markets Debt Fund, Emerging Markets Debt Opportunities Fund, Emerging Markets Domestic Debt Fund, Emerging Markets Local Income Fund, and Frontier Markets Local Debt Fund)	Equity Funds (and Emerging Markets Debt Fund, Emerging Markets Debt Opportunities Fund, Emerging Markets Domestic Debt Fund, Emerging Markets Local Income Fund, and Frontier Markets Local Debt Fund):	Asset Allocation Funds	Alternativ e Investmen t Funds
A	Up to 4.00%	Up to 5.75%	Up to 5.75%	Up to 5.75%
В	0.00%	0.00%	0.00%	0.00%
С	Up to 3.00%	Up to 3.00%	Up to 3.00%	Up to 3.00%
F	0.00%	0.00%	0.00%	0.00%
I	Up to 3.00%	Up to 3.00%	Up to 3.00%	Up to 3.00%
J	0.00%	0.00%	0.00%	0.00%
N	0.00%	0.00%	0.00%	0.00%
S	0.00%	0.00%	0.00%	0.00%
Z	Up to 1.00%	Up to 1.00%	Up to 1.00%	Up to 1.00%

The Sales Charge shall revert to the Management Company or to the relevant distributor through which the initial purchase was made. If in any country in which the Share Class Indicators A, C, I, and Z Shares are offered, local law or practice requires or permits a lower initial Sales Charge than that listed above for any individual purchase order, the Management Company may sell Shares and may authorise distributors to sell Shares within such country at a total price less than the applicable price set forth above, but in accordance with the maximum amounts permitted by the law or practice of such country.

Further information on Share Class Indicator B Shares

As described in more detail in Section 2.5 of this Prospectus, Share Class Indicator B Shares are subject to a Distribution Fee, accrued daily and paid monthly, at the annual rate of 1.00% of their average daily Net Asset Value of Shares. The Distribution Fee ultimately accrues to the Management Company, which may pay part or all of the Distribution Fee to those institutions involved in the distribution of Share Class Indicator B Shares.

Share Class Indicator B Shares that are redeemed prior to the fourth anniversary of the date on which they were subscribed are subject to Contingent Deferred Sales Charge, which is described in detail below under the heading "Contingent Deferred Sales Charge".

With effect from 1 July 2018, Share Class Indicator B Shares are automatically converted into the corresponding Share Class Indicator A Shares or on the fourth anniversary of the date on which the Share Class Indicator B Shares were issued or the next Dealing Day if that day is not a Dealing Day. This conversion may give rise to a tax liability for shareholders in certain jurisdictions. Shareholders should consult their tax adviser for advice about their own position.

Until 22 June 2018, holders of Share Class Indicator B Shares retained the ability to voluntarily convert such Shares into Share Class Indicator A Shares or on after the fourth anniversary of the date on which the relevant Share Class Indicator B Shares were issued, or the next Dealing Day if that is not a Dealing Day. This conversion may give rise to a tax liability for shareholders in certain jurisdictions. Shareholders should consult their tax adviser for advice about their own position.

All conversions of Share Class Indicator B Shares into Share Class Indicator A Shares are made on the basis set out in section 2.4 of this Prospectus.

No Contingent Deferred Sales Charge will be applicable to Share Class Indicator B Shares on either (i) automatic conversions, or (ii) voluntary conversions made on or after the fourth anniversary of the date on which the Share Class Indicator B Shares were issued. No Sales Charge will be applicable to the Share Class Indicator A Shares issued as a result of the conversion of Share Class Indicator B Shares.

Contingent Deferred Sales Charge

Contingent Deferred Sales Charge for all Funds

Time Period Since Subscription*	Share Class Indicator B	Share Class Indicator C
0-365 days	4.00%	1.00%
1-2 years	3.00%	None
2-3 years	2.00%	None
3-4 years	1.00%	None
4 years and thereafter	None	None

^{*}Subscription for these purposes means the date on which Shares were issued, rather than the date of application for subscription.

The calculation is determined in a manner that results in the lowest possible rate being charged. Therefore, it is assumed that, unless otherwise specified, the redemption is applied to Share Class Indicators B and C Shares held longest by the Shareholder. The rate will be determined based on the Fund in which the Shareholder's subscription was first made. Conversion from one Fund to another will not affect the initial purchase date not the rate to be applied upon redemption for the purposes of calculating the Contingent Deferred Sales Charge.

To provide an example, assume a Shareholder purchased 100 Class B Shares in a Fund at \in 25 per Share (at a cost of \in 2,500) and in the third year after issuance the Net Asset Value per Share is \in 27. If at such time the Shareholder makes his first redemption of 50 Shares (proceeds of \in 1,350), the charge is applied only to the original cost of \in 25 per Share and not to the increase in Net Asset Value of \in 2 per Share. Therefore \in 1,250 of the \in 1,350 redemption proceeds will be charged a Contingent Deferred Sales Charge at a rate of 2.00% (the applicable rate in the third year after issuance).

The Contingent Deferred Sales Charge, if any, accrues to the Management Company and is used in whole or in part by the Management Company to defray its expenses in providing distribution-related services to the Fund relating to the sale, promotion and marketing of Share Class Indicators B and C Shares and the furnishing of services to Shareholders by sales and marketing personnel of the Management Company.

The Contingent Deferred Sales Charge (in combination with the Distribution Fee (see Section 2.5 "Charges and Expenses") in the case of Share Class Indicator B Shares) is designed to finance the distribution of Share Class Indicators B and C Shares through the Management Company and distributors without a dealer mark-up or Sales Charge assessed at the time of purchase.

Waiver of Contingent Deferred Sales Charge

The Management Company will waive any applicable Contingent Deferred Sales Charge which may be payable on the redemption of Share Class Indicators B and C Shares where a redemption is effected pursuant to the right of the Management Company to liquidate a Shareholder's account as described in more detail under "Compulsory Redemption" (*i.e.* where it is not the fault of the Shareholder which has resulted in the compulsory redemption).

The Management Company will waive any applicable Contingent Deferred Sales Charge which may be payable on the redemption of distributing Share Class Indicators B and C Shares which result from the automatic reinvestment of dividends.

In addition the Management Company may also waive all or part of any applicable Contingent Deferred Sales Charge at the Management Company's discretion.

The alternative Sales Charge arrangements permit an investor to choose the method of purchasing Shares that is most beneficial given the amount of the purchase, the length of time the investor expects to hold the Shares and individual circumstances. Investors should determine whether under their particular circumstances it is more advantageous to incur an initial Sales Charge and not be subject to Distribution Fees and a Contingent Deferred Sales Charge, or to have the entire subscription amount invested in a Fund with the investment thereafter being subject to Distribution Fees and a Contingent Deferred Sales Charge.

2.2 Issue of Shares, Subscription and Payment Procedure

The Directors are authorised without limitation to issue fully paid up Shares of any Class at any time.

Any request for subscription, redemption or conversion shall be irrevocable except in the event of a suspension of the calculation of the Net Asset Value per Share of the relevant Class.

Issue of Shares

Shares of all Classes will be issued at a price corresponding to the Net Asset Value per Share of the relevant Class in the relevant currency. For the currencies in which the Shares for the relevant Funds will be issued, please refer to the table in Section 2.7 "Net Asset Value Determination" below. Up-to-date information is available on www.morganstanleyinvestmentfunds.com or from the Management Company. Share Class Indicator A Shares of all Equity Funds, Bond Funds, Asset Allocation Funds and Alternative Investment Funds are subject to a Sales Charge calculated on the entire subscription amount in the subscription currency. Details of the minimum initial and subsequent subscriptions for a Fund and the Sales Charge are set out in Section 2.1 "Share Class Description".

Initial applications for Shares must be made on the Company's Application Form or in a format acceptable to the Management Company containing the information required by the Management Company and must be forwarded to the Transfer Agent in Luxembourg, or to any distributor indicated on the Application Form. Subsequent applications may be made in writing or by fax. The Management Company may also decide that initial and subsequent subscription applications may be made by electronic or other means (provided that a duly completed Application Form is received for initial subscription applications).

The Management Company or its delegate may request an investor to provide additional information to substantiate any representation made by the investor in its application. The Management Company reserves in any case the right to reject any application for Shares, in whole or in part.

All applications are made subject to this Prospectus, the latest annual report and semi-annual report, if available, the Company's Articles of Incorporation and the Application Form.

Joint applicants must each sign the Application Form unless an acceptable Power of Attorney or other written authority is provided.

Applications for Shares in all Funds received by the Transfer Agent before the Cut-Off Point for any Dealing Day will be processed on that Dealing Day based on the Net Asset Value per Share determined for that Dealing Day.

Any applications received by the Transfer Agent after the Cut-Off Point for any Dealing Day will be processed on the next Dealing Day on the basis of the Net Asset Value per Share next determined. Applications for Distributing Share Classes (Share Class Sub-Indicators "X" and "M") and Discretionary Distributing Share Classes (Share Class Sub-Indicators "R" and "RM") shall begin accruing dividends on the Dealing Day on which such orders are processed.

No distributor is permitted to withhold subscription orders to benefit themselves by a price change. Investors should note that they may be unable to purchase or redeem Shares through a distributor on days that such distributor is not open for business.

A collection agent (that is, an entity located in a Financial Action Task Force country or equivalent country, which collects subscription, redemption and conversion requests on behalf of the Management Company but does not process them) must receive an application before the Cut-Off Point for any Dealing Day in order for the Transfer Agent to process the application on the basis of the Net Asset Value per Share determined for that Dealing Day. Applications received by a collection agent after the Cut-Off Point for any Dealing Day will be processed by the Transfer Agent on the basis of the Net Asset Value per Share determined for the next Dealing Day. Where collection agents appointed by the Management Company receive subscription monies, those monies will have been subject to the anti-money laundering checks of an appointed distributor or the relevant collection agent.

The Company determines the principles of the calculation of the price or net asset value of its Shares, which are implemented by the Management Company on a forward basis. This means that it is not possible to know in advance the Net Asset Value per Share at which Shares will be bought or sold (exclusive of any Sales Charge). The Net Asset Value per Share is calculated at the valuation point following the Cut-Off Point.

The Company's Funds are not designed for investors with short term investment horizons. Activities which may adversely affect the interests of the Company's Shareholders (for example that disrupt investment strategies or impact expenses) are not permitted. Specifically, market timing is not permitted.

Whilst recognising that Shareholders may have legitimate needs to adjust their investments from time to time, the Management Company in its discretion may, if deems such activities adversely affect the interests of the Company's Shareholders, take action as appropriate to deter such activities.

Accordingly if the Management Company determines or suspects that a Shareholder has engaged in such activities, it may suspend, cancel, reject or otherwise deal with that Shareholder's subscription or conversion applications and take any action or measures as appropriate or necessary to protect the Company and its Shareholders. Please refer to "Redemption of Shares" and "Conversion of Shares" for further information in relation to the measures which may be taken.

The Company may restrict or prevent the legal or beneficial ownership of Shares in the Company by any person (individual, corporation, partnership or other entity), and take any of the measures as further set forth in the Articles of Incorporation, if in the opinion of the Company such ownership or practices may (i) result in a breach of any provisions in the Articles of Incorporation, this Prospectus or law or regulations of any jurisdiction, or (ii) requires the Company, the Management Company or the Investment Adviser to be registered under any laws or regulations whether as an investment fund or otherwise, or cause the Company to be required to comply with any registration requirements in respect of any of its shares, whether in the United States of America or any other jurisdiction; or (iii) may cause the Company, the Management Company, the Investment Advisers or Shareholders any legal, regulatory, taxation, administrative, financial or any other disadvantages which they would not have otherwise incurred (such person being herein referred to as "Prohibited Persons"). In particular, the Directors have resolved to prevent the ownership of Shares by any US Person (as defined pursuant to Regulation S under the U.S. Securities Act of 1933, as amended), provided however that no provisions of this Prospectus shall prevent a Morgan Stanley entity or any of its affiliates or subsidiaries from owning Shares.

The Company retains the right to offer only one Class of Shares for purchase by investors in any particular jurisdiction in order to conform to local law, custom or business practice. The Company also reserves the right to adopt standards applicable to classes of investors or transactions that permit or require the purchase of a particular Class of Shares.

In case it would be detrimental to the existing Shareholders to accept a cash application for Shares of any Fund which, either singly or when aggregated with other applications so received in respect of any Dealing Day (the "First Dealing Day") represents more than 10% of such Fund, the Directors may decide that all or part of such applications for Shares be deferred until the next Dealing Day so that not more than 10% of the Net Asset Value of the relevant Fund be subscribed for on the First Dealing Day.

If the Directors decide to defer all or part of such application, the applicant shall be informed prior to the deferral taking place. To the extent that any application is not given full effect on such First Dealing Day by virtue of the exercise of the power to pro-rate applications, it shall be treated with respect to the unsatisfied balance thereof as if a further request had been made by the Shareholder in respect of the next Dealing Day and, if necessary, subsequent Dealing Days, until such application shall have been satisfied in full. With respect to any application received in respect of the First Dealing Day, to the extent that subsequent applications shall be received in respect of following Dealing Days, such later applications shall be postponed until after the satisfaction of applications relating to the First Dealing Day, but subject thereto shall be dealt with as set out in the preceding sentence.

The Management Company, in its sole and absolute discretion, retains the right to accept or refuse applications for payment for Shares in whole or in part by an in kind subscription of suitable investments. The transaction costs incurred in connection with the acceptance by the Management Company of an in-kind subscription will be borne directly by the incoming Shareholder. Any applicable Sales Charge will be deducted before investment commences. The investments forming the in-kind subscription will be valued and a report will be issued by the Company's auditors following their review of the methods of the valuation used by the Management Company for accepting the in-kind subscription.

Such review will be conducted in accordance with the professional recommendations of the *Institut des Réviseurs d'Entreprises*. The value determined, together with the Net Asset Value per Share calculated for the Class of Shares concerned in the relevant Fund, will determine the number of Shares to be issued to the incoming Shareholder. The purpose of the foregoing policy is to ensure that the existing Shareholders in a Fund do not bear the transaction costs of acquiring additional assets for a large incoming Shareholder.

Applicants for Shares may make payment in US Dollars, Euro, Yen or Sterling. Applicants can only make payment for Currency Hedged Share Classes (Share Class Sub-Indicators "H", "H1", "H2", "H3", "O") in the currency in which that share class is denominated, depending on the specific Class this may require payment in US Dollars, Euro, Yen, Sterling, Australian Dollars, Mexican Peso, Singapore Dollars, Swedish Krona, Swiss francs, South African Rand or any other currency which the Management Company may decide. Where payment is made for any other Class in a currency in which the relevant Class does not issue a Net Asset Value per Share, the Administrator will arrange for the necessary currency transaction to convert the subscription monies into the Reference Currency of the relevant Fund. Any such currency transaction will be effected with the Depositary or the Management Company at the applicant's cost. Currency exchange transactions may delay any dealing in Shares as the Administrator may choose at its option to delay executing any foreign exchange transaction until cleared funds have been received.

In the case of subscriptions for Shares with Share Class Indicators A, B, C and F of all Funds unless prior arrangements have been made, cleared funds must be received by the Transfer Agent on the Business Day prior to the relevant Dealing Day in order for the Applicant to receive the Net Asset Value per Share determined for that Dealing Day.

In the case of subscription for Shares with Share Class Indicators I, J, N, S and Z for all Funds, cleared funds must be received by the Transfer Agent by 1 p.m. CET, within three Business Days after the Dealing Day.

Full payment instructions are set out in the Application Form and may also be obtained through the Management Company, a distributor or the Transfer Agent. Applicants for Shares should note that cheques are not accepted as means of payment.

Where Shares are subscribed through a distributor, different payment arrangements to those set out above may apply and will be available from the relevant distributor.

If timely settlement is not made (or a completed Application Form is not received for an initial subscription), the relevant allotment of Shares may be cancelled and the subscription monies returned to the applicant without interest. Alternatively, if payment is received in respect of any application after the settlement date, the Management Company will consider the application as being an application for the number of Shares which can be purchased or subscribed with such payment on the next Dealing Day. An applicant may be required to compensate the relevant distributor and/or the Company for the costs of late or non-payment. The Management Company will have the power to redeem all or part of the applicant's holding of Shares in order to meet such costs.

Confirmation Notes

A Confirmation Note is sent to the applicant by ordinary post (or by fax, electronic or other means) on the Dealing Day the order is processed, providing full details of the transaction.

It is recommended that applicants check Confirmation Notes on receipt.

All Shares are issued in registered form and the share register is conclusive evidence of ownership. The Company treats the registered owner of a Share as the absolute and beneficial owner thereof.

Shares are issued in uncertificated form. The uncertificated form enables the Management Company to effect redemption instructions without undue delay.

The Management Company may permit any distributor to be involved in the collection of subscription, redemption and conversion orders on behalf of the Company and any of the Funds and may itself or arrange for a third party (including a distributor) to, in that case, provide a nominee service for applicants purchasing Shares through them. Applicants may elect, but are not obliged, to make use of such nominee service pursuant to which the nominee will hold Shares in its name for and on behalf of the applicants who shall be entitled at any time to claim direct title to the Shares and who, in order to empower the nominee to vote at any general meeting of Shareholders, shall provide the nominee with specific or general voting instructions to that effect. Applicants retain the ability to directly invest in the Company without using a nominee service.

Applicants are allocated a Shareholder number on acceptance of their application and this together with the Shareholder's personal details are proof of identity. This Shareholder number should be used for all future dealings by the Shareholder with the Management Company or the Transfer Agent.

Any changes to the Shareholder's personal details or loss of Shareholder number must be notified immediately to the Transfer Agent in writing. Failure to do so may result in delay upon redemption. The Management Company on behalf of the Company reserves the right to require an indemnity or verification countersigned by a bank, stockbroker or other party acceptable to it before accepting such instructions.

If any application is not accepted in whole or in part the application monies or the balance outstanding will be returned to the applicant by post or bank transfer at the applicant's risk.

General Provisions

The Management Company reserves the right to reject any application or to accept the application in part only. Furthermore, the issue and sale of Shares of any Class in any or all Funds may be discontinued without notice.

Subscriptions will be accepted upon verification by the Management Company that the relevant investors have received a key investor information document of the Class of Share into which they intend to subscribe.

Pursuant to the Law of 19 February 1973 on the sale of medicinal substances and the defence against drug addiction, as last amended by the Law of 11 August 1998, to the Law of 12 November 2004 relating to the fight against money

laundering and financing of terrorism, and the relevant Circulars issued by the Luxembourg Supervisory Authority, Luxembourg has regulations in place for the prevention of money laundering from drug trafficking proceeds. As a result, evidence of the identity of subscribers, including the provision of supporting documentation, may be required by the Management Company. Such information may be collected at the time an application for Shares is made.

No Shares will be issued by the Company during any period when the calculation of the Net Asset Value per Share of the relevant Fund is suspended by the Company pursuant to the powers reserved to it by its Articles of Incorporation and as discussed herein under "Temporary Suspension of Calculation of Net Asset Value".

Notice of any such suspension will be given to applicants for Shares and applications made or pending during such suspension may be withdrawn by notice in writing received by the Management Company prior to the Cut-Off Point in respect of the first Dealing Day following the end of such suspension. Applications which are not withdrawn will be considered on the first Dealing Day following the end of the suspension period.

Directors' Powers

The Directors may, at any time and in their discretion, resolve to close a Fund or one or more Class of Shares to subscriptions or conversions for a period of time. The circumstances in which they may do so include, but are not limited to, circumstances where the strategy run by the relevant Investment Adviser or Sub-Adviser, of which the Fund forms part, has reached a size where, in the opinion of the Investment Adviser or Sub-Adviser, the universe of securities in which the strategy may invest may become too small to enable the Investment Adviser or Sub-Adviser to continue to invest the assets of the strategy effectively if the Fund, and therefore the strategy, continues to grow. Such Funds may be re-opened at any time by a resolution of the Directors.

In exercising the discretion provided above, the Directors may, at any time, decide to close a Fund or a Share Class and not to accept any further investment into the relevant Fund or Share Class (i) from investors who have not yet invested into the said Funds or into the said Share Classes ("Soft Closure") or (ii) from any investor including investors already invested in the relevant Fund or Share Class ("Hard Closure").

Decisions taken by the Directors on Soft Closure and Hard Closure may have immediate or non-immediate effect and may be effective for an unspecified period of time.

In relation thereto, a notification will be displayed on the website <u>www.morganstanleyinvestmentfunds.com</u> and if applicable on other Morgan Stanley Investment Management websites, and will be updated according to the status of the said Shares Classes or Funds.

Anti-Dilution Measures

As investors enter or exit a fund, the potential purchase and sale of securities may incur trading costs such as bid/ask spreads, brokerage fees, transaction charges and taxes. These costs are charged to the fund and are borne by all remaining shareholders of the fund, an effect known as dilution which can impact the remaining shareholders' returns on their investment in the fund.

In order to protect remaining shareholders from dilution, the Management Company may adjust the NAV of a Fund to reflect these estimated trading costs, a mechanism known as swing pricing.

When net investor activity in a Fund exceeds a certain threshold ("Swing Threshold") on a given Dealing Day, the NAV is adjusted by a factor ("Swing Factor"), upwards in case of net subscriptions and downwards in case of net redemptions. In either case, the swung NAV applies to all transactions, regardless of direction, not the specific circumstances of each individual investor transaction.

For each Fund, the Swing Factor shall not exceed 2% of the NAV in normal market conditions, however, the Management Company may decide to temporarily increase this limit in exceptional circumstances (e.g. stressed or dislocated markets resulting in increased trading costs beyond the 2% cap) to protect Shareholders' interests. This decision would be communicated to Shareholders through the usual communication channels, as outlined in Section 2.6 "Publication of Net Asset Value".

2.3 Redemption of Shares

Shares may be redeemed on any Dealing Day subject to the limitations described below. Shares of all Classes will be redeemed at a price corresponding to the Net Asset Value per Share of the relevant Class in the relevant currency.

Procedure for Redemption

Shareholders wishing to have all or some of their Shares redeemed may do so by fax or by letter to the Transfer Agent or to a distributor. The Management Company may also decide that applications for redemptions may be made by electronic or other means.

The application for redemption must include (i) the monetary amount the Shareholder wishes to redeem, or (ii) the number of Shares the Shareholder wishes to redeem. In addition, the application for redemption must include the Shareholder's personal details together with the Shareholder's account number. Failure to provide any of this information may result in delay of the application for redemption whilst verification is being sought from the Shareholder.

Upon redemption of Distributing Share Classes (Share Class Sub-Indicators "X" and "M") and Discretionary Distributing Share Classes (Share Class Sub-Indicators "R" and "RM"), all dividends will accrue up to, and including, the Dealing Day on which such orders are processed. Holders of Share Class Indicators B and C Distributing Share Classes and Discretionary Distributing Share Classes Shares may be subject to a Contingent Deferred Sales Charge in accordance with Section 2.1 "Share Class Description".

Such applications for redemption will be considered as binding and irrevocable by the Company except in the event of a suspension of the calculation of the Net Asset Value per Share. Written confirmations may be required by the Management Company and must be duly signed by all registered holders, unless in the case of joint registered holders, each such holder has sole signing authority.

Applications for redemption from all Funds received by the Transfer Agent before the Cut-Off Point for any Dealing Day will be processed on that Dealing Day based on the Net Asset Value per Share determined for that Dealing Day. Any applications received by the Transfer Agent after the Cut-Off Point for any Dealing Day will be processed on the next Dealing Day on the basis of the Net Asset Value per Share next determined.

No distributor is permitted to withhold redemption orders received to benefit themselves by a price change. Investors should note that they may be unable to redeem Shares through a distributor on days that such distributor is not open for business.

A collection agent (that is, an entity located in a Financial Action Task Force country or equivalent country, which collects subscription, redemption and conversion requests on behalf of the Management Company but does not process them) must receive an application for redemption before the Cut-Off Point for any Dealing Day in order for the Transfer Agent to process the application on the basis of the Net Asset Value per Share determined for that Dealing Day. Applications received by a collection agent in respect of a Dealing Day after the Cut-Off Point will be processed by the Transfer Agent on the basis of the Net Asset Value per Share determined for the next Dealing Day.

The Company determines the principles of the calculation of the price or net asset value of its Shares, which are implemented by the Management Company on a forward basis. This means that it is not possible to know in advance the Net Asset Value per Share at which Shares will be bought or sold (exclusive of any Sales Charge). The Net Asset Value per Share for a Dealing Day is calculated at the valuation point following the Cut-Off Point.

The Company's Funds are not designed for investors with short term investment horizons. Activities which may adversely affect the interests of the Company's Shareholders (for example that disrupt investment strategies or impact expenses) are not permitted. Specifically, market timing is not permitted.

Whilst recognising that Shareholders may have legitimate needs to adjust their investments from time to time, the Management Company in its discretion may, if it deems such activities adversely affect the interests of the Company's Shareholders, take action as appropriate to deter such activities.

Accordingly if the Management Company determines or suspects that a Shareholder has engaged in such activities, it may suspend, cancel, reject or otherwise deal with that Shareholder's application and take any action or measures as appropriate or necessary to protect the Company and its Shareholders. A redemption fee of up to 2% of an individual redemption may be applied where the Management Company, in its discretion, determines that the shareholder has engaged in trading practices which adversely affect the interests of the Company's shareholders or where it is otherwise appropriate to protect the interest of the Company. The fee is retained for the benefit of continuing Shareholders of the relevant Fund.

A Confirmation Note providing full details of the transaction and the redemption proceeds is sent to the applicant by ordinary post (or by fax, electronic or other means) on the Dealing Day the order is processed.

It is recommended that applicants check Confirmation Notes on receipt.

Payment for Shares redeemed will be effected during the Settlement Period.

Unless otherwise specified by the Shareholder on the application for redemption, such redemption will be paid in the Reference Currency of the Fund or, if applicable, in the currency which the Currency Hedged Share Class is denominated from which the Shareholder is redeeming. If necessary, the Administrator will arrange the currency transaction to convert the redemption monies from the Reference Currency of the relevant Fund into the relevant currency. Such currency transaction will be effected with the Depositary or the Management Company at the

Shareholder's cost if the currency is not one in which the Net Asset Value per Share of the Class from which the Shareholder is redeeming is issued. The Directors reserve the right to delay payment for up to ten Business Days from the relevant Dealing Day if market conditions are unfavourable where it considers such action to be in the best interests of the remaining Shareholders.

Temporary Suspension of Redemption

The redemption of Shares of the Company will be suspended during any period when the calculation of the Net Asset Value per Share of the relevant Class is suspended in accordance with the section "Temporary Suspension of Calculation of Net Asset Value" herein. Any Shareholder tendering Shares for redemption will be notified of such period of suspension. The Shares in question will be redeemed on the first Dealing Day following the end of the suspension period.

If a period of suspension lasts for more than one calendar month after the date of an application for redemption, the application may be cancelled by the Shareholder by notice in writing to a distributor or to the Management Company, provided that the notice is received by the distributor or the Management Company prior to any relevant deadline notified to the Shareholder on the last Dealing Day of the suspension period.

Compulsory Redemption

If the Net Asset Value of any Fund or Class of Shares at any valuation point shall become at any time less than 100 million Euros, or its equivalent in the Reference Currency of the relevant Fund, the Company may, at its discretion, redeem all but not less than all of the Shares of the applicable Classes of Shares in compliance with the procedure set forth under *Dissolution* of Section 3.1 "General Information".

A Contingent Deferred Sales Charge, if applicable, will not be levied on the redemption proceeds of Share Class Indicators B and C Shares subject to this type of compulsory redemption.

If the Management Company receives a request for redemption of Shares relating to: (i) part of a holding of Shares with a value of less than \$2,500 or its equivalent, or (ii) if after redemption the holder would be left with a balance of Shares having a value of less than the current minimum holding amount or less than US\$100 or its equivalent, the Company may treat this as a request to redeem such Shareholder's entire holding or may at a later date, upon providing them with one month's notice, elect to either compulsorily redeem their holding or convert their holding into another Share Class.

If it shall come to the attention of the Company at any time that Shares are beneficially owned by a Prohibited Person, either alone or in conjunction with any other person, and the Prohibited Person fails to comply with the direction of the Company to sell his Shares and to provide the Company with evidence of such sale within thirty days of being so directed by the Company, the Company may in its discretion take any of the measures as further set forth in the Articles of Incorporation, including compulsorily redeem such Shares at their Redemption Price in accordance with the Articles of Incorporation. Immediately after the close of business specified in the notice given to the Prohibited Person of such compulsory redemption, the Shares will be redeemed and such investors will cease to be the owners of such Shares. Share Class Indicators B and C Shareholders should note that in these circumstances, if applicable, a Contingent Deferred Sales Charge will be levied on the redemption proceeds. In particular, if a Shareholder holds less than the minimum holding amount, or does not fall within the target audience for a Class of Shares as prescribed under Section 2.1 "Share Class Description", the Company may, upon providing them with one month's notice, compulsorily redeem their holding in compliance with the procedure as mentioned above.

Any Shareholder or prospective Shareholder may be required to furnish with any information which it may consider necessary for the purpose of determining whether or not the beneficial owner of such Shares is or will be a Prohibited Person.

Procedures for Redemptions and Conversions Representing Ten Percent or More of any Fund

If any application for redemption or conversion is received in respect of any one Dealing Day (the "First Dealing Day") which either singly or when aggregated with other applications so received, is more than 10% of the Net Asset Value of any one Fund, the Company reserves the right in its sole and absolute discretion (and acting in the best interests of the remaining Shareholders) to scale down pro rata each application with respect to such First Dealing Day so that not more than 10% of the Net Asset Value of the relevant Fund be redeemed or converted on such First Dealing Day. In circumstances where the 10% threshold is met as a result of the aggregation of a number of applications, only those applications exceeding a threshold, currently 2%, of the Net Asset Value of the relevant Fund, will be scaled down pro rata. For example, if applications representing 1%, 3%, 5% and 6% of the Net Asset Value of a Fund are received, only the applications representing 3%, 5% and 6% will be scaled down pro rata. The Company may in its discretion change the threshold as it may decide is appropriate in which case the Prospectus will be amended accordingly.

To the extent that any application is not given full effect on such First Dealing Day by virtue of the exercise of the power to pro-rate applications, it shall be treated with respect to the unsatisfied balance thereof as if a further request had been made by the Shareholder in respect of the next Dealing Day and, if necessary, subsequent Dealing Days, until such application shall have been satisfied in full. With respect to any application received in respect of the First Dealing Day, to the extent that subsequent applications shall be received in respect of following Dealing Days, such later applications shall be postponed until after the satisfaction of applications relating to the First Dealing Day, but subject thereto shall be dealt with as set out in the preceding sentence.

Alternatively, the Management Company, in its sole and absolute discretion, may ask such Shareholder to accept payment in whole or in part by an in-kind distribution of securities in lieu of cash. The securities forming the in-kind distribution will be valued and a valuation report obtained from the Company's auditors in connection with the in-kind distribution. Investors who receive securities in lieu of cash upon redemption should note that they may incur brokerage and/or local tax charges on the sale of the securities. In addition, the net proceeds from the sale by the redeeming Shareholder of the securities may be more or less than the Redemption Price due to market conditions and/or the difference between the prices used to calculate the Net Asset Value per Share and bid prices received on the sale of the securities. In the event that a Contingent Deferred Sales Charge is payable on the redemption proceeds of the Share Class Indicators B and C Shares, securities will be retained by the Company to cover such Contingent Deferred Sales Charge before the remaining securities are transferred to the redeeming Shareholder. No Shareholder can be required to accept an in-kind distribution against their will. To the extent that a Shareholder refuses payment in whole or in part by an in-kind distribution of securities in lieu of cash, such redemption will be paid in the Reference Currency of the Fund or, if applicable, in the currency which the Currency Hedged Share Class is denominated from which the Shareholder is redeeming.

Anti-Dilution Measures

As investors enter or exit a fund, the potential purchase and sale of securities may incur trading costs such as bid/ask spreads, brokerage fees, transaction charges and taxes. These costs are charged to the fund and are borne by all remaining shareholders of the fund, an effect known as dilution which can impact the remaining shareholders' returns on their investment in the fund.

In order to protect remaining shareholders from dilution, the Management Company may adjust the NAV of a Fund to reflect these estimated trading costs, a mechanism known as swing pricing.

When net investor activity in a Fund exceeds a certain threshold ("Swing Threshold") on a given Dealing Day, the NAV is adjusted by a factor ("Swing Factor"), upwards in case of net subscriptions and downwards in case of net redemptions. In either case, the swung NAV applies to all transactions, regardless of direction, not the specific circumstances of each individual investor transaction.

For each Fund, the Swing Factor shall not exceed 2% of the NAV in normal market conditions, however, the Management Company may decide to temporarily increase this limit in exceptional circumstances (e.g. stressed or dislocated markets resulting in increased trading costs beyond the 2% cap) to protect Shareholders' interests. This decision would be communicated to Shareholders through the usual communication channels, as outlined in the Section 2.6 "Publication of Net Asset Value".

2.4 Conversion of Shares

Shareholders are entitled to convert all or part of their Shares of one Fund into Shares of the same Class of other Funds and may also convert from one Class of Shares of a Fund into other Classes of Shares of that Fund or other Funds in accordance with the table below, without charge, except as otherwise provided below and provided that the Shareholder meets the eligibility criteria for the Class of Share into which they are converting, as detailed in Section 2.1 "Share Class Description" above.

Conversions are always processed in a common currency. Where conversion is between Share Classes whose net asset values are issued in a common currency, the conversion will be processed in the common currency specified by the investor. Where there is not a common currency between the net asset values of the converting Share Classes, then no conversion is possible except with the specific approval of the Management Company and the investor will need to redeem their holding and subscribe for their chosen Share Class in the appropriate currency.

No conversions will be accepted out of the Saudi Equity Fund.

The Management Company may, at its discretion, accept other conversions than those detailed as permissible in the table below. In certain jurisdictions, where conversions are submitted via a third party agent, different arrangements may apply and the permitted conversions may differ from those set out in the following table. Investors should contact their third party agent for more information.

INTO

	Share Class Indicat or	A	В	С	F	I	J	N	S	Z
	A	✓	X	X	✓	✓	X	X	X	✓
	В	✓	✓	X	✓	✓	X	X	X	X
	C	X	X	✓	✓	✓	X	X	X	X
O U	F	X	X	X	✓	✓	X	X	X	X
T	I	✓	X	X	✓	✓	X	X	✓	✓
	J	X	X	X	X	X	X	X	X	X
0	N	X	X	X	X	X	X	✓	X	X
F	S	X	X	X	X	X	X	X	✓	✓
	Z	X	X	X	X	X	X	X	✓	✓

Applications may be sent in writing by fax to the Transfer Agent or to a distributor stating which Shares are to be converted. The Management Company may also decide that applications for conversion may be made by electronic or other means. The application for conversion must include (i) the monetary amount the Shareholder wishes to convert or (ii) the number of Shares the Shareholder wishes to convert, together with the Shareholder's personal details and Shareholder's account number. Failure to provide any of this information may result in delay of the application for conversion whilst verification is being sought from the Shareholder. The period of notice is the same as for applications for redemption. The application for conversion must be accompanied, as appropriate, by the registered share certificate or by a form of transfer, duly completed, or by any other document providing evidence of transfer.

The Management Company may refuse to accept a conversion application if it is detrimental to the interests of the Company or the Shareholders taking into account the monetary amount or number of Shares to be converted, market conditions or any other circumstances. The Management Company may, for example, at its discretion, decide to refuse a conversion application to protect a Fund and the Shareholders from the effects of short term trading or may limit the number of conversions between Funds which are permitted.

A conversion fee of up to 2% may be applied by the Management Company, in its discretion, determine that the shareholder has engaged in trading practices which adversely affect the interests of the Company's shareholders or where it is otherwise appropriate to protect the interest of the Company and its shareholders. The fee is retained for the benefit of continuing Shareholders of the relevant Fund.

Conversion within Share Class Indicators A, C, I and Z Shares from one Fund to another Fund, will not result in a Sales Charge being levied on the amount to be converted.

Conversion within Share Class Indicators B and C Shares from one Fund to another Fund will not affect the initial purchase date nor the rate which will be applied upon redemption from the new Fund, as the rate which will be applied will be determined based on the first Fund in which the Shareholder purchased Shares. Any conversion of Class B Shares to any other Class of Shares of any Fund within four years of the subscription date will be treated as a redemption and may be subject to a Contingent Deferred Sales Charge as detailed in Section 2.1 "Share Class Description". Any conversion of Share Class Indicator C Shares to any other Class of Shares of any Fund within less than one year of the subscription date will be treated as a redemption and may be subject to a Contingent Deferred Sales Charge as detailed in Section 2.1 "Share Class Description".

Shareholders should note that if an application for conversion relates to a partial conversion of an existing holding and the remaining balance within the existing holding is below the minimum requirement (which is the current minimum initial subscription amount as detailed in this Prospectus), the Company is not bound to comply with such application for conversion.

If a Shareholder holds less than the minimum holding amount for a class of Shares as prescribed under Section 2.1 "Share Class Description", the Management Company may, upon providing them with one month's notice, compulsorily convert such Shareholder's holding into another class of Shares.

Any request for subscription, redemption or conversion shall be irrevocable except in the event of a suspension of the calculation of the Net Asset Value per Share of a relevant Class.

For all Funds, applications for conversion received by the Transfer Agent on any Dealing Day before the Cut-Off Point will be processed on that Dealing Day based on the Net Asset Value per Share determined on that Dealing Day using the relevant valuation methodology for the particular Fund. Any applications received after the Cut-Off Point will be processed on the next Dealing Day on the basis of the Net Asset Value per Share next determined.

Applications for conversion received before the Cut-Off Point in respect of a Dealing Day shall begin accruing dividends on that Dealing Day.

A collection agent (that is, an entity located in a Financial Action Task Force country or equivalent country, which collects subscription, redemption and conversion requests on behalf of the Management Company but does not process them) must receive an application for conversion in respect of a Dealing Day before the Cut-Off Point in order for the Transfer Agent to process the application on the basis of the Net Asset Value per Share determined for that Dealing Day. Applications received by a collection agent after the Cut-Off Point will be processed by the Transfer Agent on the basis of the Net Asset Value per Share determined for the next Dealing Day.

Investors should note that they may be unable to convert Shares through a distributor on days that such distributor is not open for business.

The Company determines the principles of the calculation of the price or net asset value of its Shares, which are implemented by the Management Company on a forward basis. This means that it is not possible to know in advance the Net Asset Value per Share at which Shares will be bought or sold (exclusive of any Sales Charge). The Net Asset Value per Share is calculated at the valuation point following the Cut-Off Point.

The Company's Funds are not designed for investors with short term investment horizons. Activities which may adversely affect the interests of the Company's Shareholders (for example that disrupt investment strategies or impact expenses) are not permitted. Specifically, market timing is not permitted.

Whilst recognising that Shareholders may have legitimate needs to adjust their investments from time to time, the Management Company in its discretion may, if it deems such activities adversely affect the interests of the Company's Shareholders, take action as appropriate to deter such activities.

Accordingly if the Management Company determines or suspects that a Shareholder has engaged in such activities, it may suspend, cancel, reject or otherwise deal with that Shareholder's subscription or conversion applications and take any action or measures as appropriate or necessary to protect the Company and its Shareholders.

Applications for conversion in respect of any one Dealing Day which either singly or when aggregated with other applications for conversion or redemption so received, represent more than 10% of the Net Asset Value of any one Fund, may be subject to additional procedures set forth herein under "Procedures for Redemptions and Conversions Representing Ten Percent or more of any Fund".

The rate at which all or part of the Shares in a given Fund (the "Original Fund") are converted into Shares in another Fund (the "New Fund"), or all or part of the Shares of a particular Class (the "Original Class") are converted into another Class within the same Fund (the "New Class"), is determined in accordance with the following formula:

$$A = \underbrace{B \times C \times E}_{D}$$

where:

- A is the number of Shares to be allocated in the New Fund or New Class;
- B is the number of Shares of the Original Fund or Original Class which is to be converted;
- C is the Net Asset Value per Share of the Original Class or the relevant Class within the Original Fund at the relevant valuation point;
- D is the Net Asset Value per Share of the New Class or the relevant Class within the New Fund at the relevant valuation point; and
- E is the actual rate of exchange on the day concerned applied to conversions between Funds denominated in different currencies, and is equal to 1 in relation to conversions between Funds or Classes denominated in the same currency.

After conversion of the Shares, the Transfer Agent will inform the Shareholder of the number of Shares of the New Fund or New Class obtained by conversion and the price thereof. A Confirmation Note will be sent to the Shareholder by ordinary post (or by fax, electronic or other means) on the Dealing Day, providing full details of the transaction.

It is recommended that applicants check Confirmation Notes on receipt.

Requests for conversions will be accepted upon verification by the Management Company that the relevant shareholders have received a key investor information document of the Class of Share into which they intend to convert.

Any fractions of Shares can be allotted and issued unless the Shareholder holds Shares through Euroclear (see Section 3.1 "General Information").

Shareholder should be aware that the Directors have the discretion to cease accepting any conversions in to any particular Share Class or Fund in the circumstances set out under the heading "Directors Powers" in section 2.2 above.

2.5 Charges and Expenses

Under the terms of the Management Company Services Agreement the Management Company is entitled to receive a Management Fee, accrued daily and paid monthly, at the annual rates set out below, based on the average daily Net Asset Value of each Class of Shares with the relevant Share Class Indicator. Under the Investment Advisory Agreements the Investment Advisers are entitled to receive a fee for their discretionary investment management and investment advisory services. The fees of the Investment Adviser will be paid out of the Management Fee.

Management Fee	Share Class Indicators: A and B	Share Class Indicator: C	Share Class Indicators: F, I, Z	Share Class Indicator: J	Share Class Indicator: S
Equity Funds					
American Resilience Fund	1.60%	2.40%	0.70%	0.50%	0.70%
Asia Opportunity Fund	1.60%	2.40%	0.75%	0.65%	0.75%
Asian Property Fund	1.40%	2.20%	0.75%	N/A	0.75%
Calvert Sustainable Climate Aligned Fund	1.25%	1.65%	0.50%	0.25%	0.50%
Calvert Sustainable Climate Transition Fund	1.90%	2.50%	0.75%	0.45%	0.75%
Calvert Sustainable Developed Europe Equity Select Fund	1.0%	1.35%	0.40%	0.20%	0.40%
Calvert Sustainable Developed Markets Equity Select Fund	1.0%	1.35%	0.40%	0.20%	0.40%
Calvert Sustainable Diversity, Equity and Inclusion Fund	1.25%	1.65%	0.50%	0.25%	0.50%
Calvert Sustainable Emerging Markets Equity Select Fund	1.25%	1.65%	0.50%	0.25%	0.50%
Calvert Sustainable Global Equity Engagement Fund	1.90%	2.50%	0.75%	0.40%	0.75%
Calvert Sustainable US Equity Select Fund	1.0%	1.35%	0.40%	0.20%	0.40%
China A-shares Fund	1.70%	2.40%	0.70%	0.55%	0.70%
China Equity Fund	1.70%	2.40%	0.70%	0.50%	0.70%
Counterpoint Global Fund	1.60%	2.40%	0.80%	0.60%	0.80%
Developing Opportunity Fund	2.00%	2.65%	0.80%	0.60%	0.80%
Emerging Leaders Equity Fund	1.90%	2.60%	0.75%	N/A	0.75%
Europe Opportunity Fund	1.50%	2.20%	0.75%	0.40%	0.75%
European Property Fund	1.40%	2.20%	0.75%	N/A	0.75%

Management Fee	Share Class Indicators: A and B	Share Class Indicator: C	Share Class Indicators: F, I, Z	Share Class Indicator: J	Share Class Indicator: S
Global Brands Equity Income Fund	1.75%	2.50%	0.85%	0.55%	0.85%
Global Brands Fund	1.60%	2.40%	0.75%	N/A	0.75%
Global Core Equity	1.35%	1.80%	0.55%	N/A	0.55%
Fund					
Global Endurance Fund	1.60%	2.40%	0.75%	N/A	N/A
Global Focus Property Fund	1.90%	2.50%	0.75%	0.45%	0.75%
Global Infrastructure Fund	1.50%	2.30%	0.85%	N/A	0.85%
Global Insight Fund	1.60%	2.40%	0.75%	N/A	0.75%
Global Opportunity	1.60%	2.40%	0.75%	N/A	0.75%
Fund					
Global Permanence Fund	1.60%	2.40%	0.75%	N/A	N/A
Global Property Fund	1.50%	2.30%	0.85%	N/A	0.85%
Global Quality Fund	1.60%	2.40%	0.75%	N/A	0.75%
Global Sustain Fund	1.50%	2.30%	0.65%	0.50%	0.65%
Indian Equity Fund	1.60%	2.40%	0.75%	N/A	0.75%
International Equity (ex US) Fund	1.60%	2.40%	0.75%	N/A	0.75%
Japanese Equity Fund	1.40%	2.20%	0.75%	N/A	0.75%
Japanese Small Cap Equity Fund	1.90%	2.50%	0.75%	0.45%	0.75%
Latin American Equity Fund	1.60%	2.40%	1.00%	0.35%	1.00%
NextGen Emerging Markets Fund	1.60%	2.40%	1.10%	N/A	1.10%
Parametric Emerging Markets Fund	1.40%	1.85%	0.55%	N/A	0.55%
Parametric Global Defensive Equity Fund	1.35%	1.60%	0.45%	N/A	0.45%
Saudi Equity Fund	1.90%	2.50%	0.75%	0.40%	N/A
Sustainable Asia Equity Fund	1.40%	2.20%	0.75%	N/A	0.75%
Sustainable Emerging Markets Equity Fund	1.60%	2.40%	1.10%	N/A	1.10%
US Advantage Fund	1.40%	2.200/	0.70%	N/A	0.70%
US Core Equity Fund	1.10%	2.20% 1.50%	0.45%	N/A N/A	0.45%
US Focus Property	1.75%	2.35%	0.70%	0.40%	0.70%
Fund					
US Growth Fund	1.40%	2.20%	0.70%	N/A	0.70%
US Insight Fund	1.50%	2.20%	0.70%	N/A	0.70%
US Permanence Fund	1.40%	2.20%	0.70%	N/A	N/A
US Property Fund	1.40%	2.20%	0.75%	N/A	0.75%
US Value Fund Vitality Fund	1.50%	2.00%	0.60%	N/A 0.45%	0.60%
Bond Funds					
Calvert Sustainable Global Green Bond Fund	1.15%	1.50%	0.45%	0.25%	0.45%
Emerging Markets Corporate Debt Fund	1.45%	2.15%	0.75%	N/A	0.75%
Emerging Markets Debt Fund	1.40%	2.10%	0.65%	N/A	0.60%

Management Fee	Share Class Indicators: A and B	Share Class Indicator: C	Share Class Indicators: F, I, Z	Share Class Indicator: J	Share Class Indicator: S
Emerging Markets Debt Opportunities Fund	1.40%	2.10%	0.65%	0.30%	0.65%
Emerging Markets Domestic Debt Fund	1.40%	2.10%	0.65%	N/A	0.65%
Emerging Markets Fixed Income Opportunities Fund	1.40%	1.90%	0.70%	0.35%	0.70%
Emerging Markets Local Income Fund	1.40%	2.10%	0.65%	0.30%	0.65%
Euro Bond Fund	0.80%	1.45%	0.45%	N/A	0.45%
Euro Corporate Bond Fund	0.80%	1.45%	0.45%	N/A	0.40%
Euro Corporate Bond – Duration Hedged Fund	0.80%	1.45%	0.45%	N/A	0.40%
Euro Strategic Bond Fund	0.80%	1.45%	0.45%	N/A	0.40%
European Fixed Income Opportunities Fund	1.00%	1.45%	0.45%	0.25%	0.45%
European High Yield Bond Fund	0.85%	1.50%	0.50%	N/A	0.50%
Floating Rate ABS Fund	0.75%	1.00%	0.30%	0.15%	0.30%
Frontier Markets Local Debt Fund	2.10%	N/A	1.25%	0.65%	1.25%
Global Asset Backed Securities Fund	1.00%	1.60%	0.50%	N/A	0.50%
Global Bond Fund	0.80%	1.45%	0.45%	N/A	0.45%
Global Convertible Bond Fund	1.00%	1.60%	0.60%	N/A	0.55%
Global Credit Fund	0.80%	1.45%	0.45%	N/A	0.45%
Global Credit Opportunities Fund	1.15%	1.50%	0.45%	0.23%	0.45%
Global Fixed Income Opportunities Fund	1.00%	1.45%	0.45%	N/A	0.45%
Global High Yield Bond Fund	1.00%	1.60%	0.60%	N/A	0.60%
Global Macro Fund	2.00%	2.65%	0.80%	N/A	0.80%
Short Duration US Government Income Fund	1.00%	1.35%	0.40%	N/A	0.40%
Short Maturity Euro Bond Fund	0.80%	1.45%	0.45%	N/A	0.45%
Short Maturity Euro Corporate Bond Fund	0.45%	0.55%	0.17%	0.09%	0.17%
Sustainable Euro Corporate Bond Fund	1.15%	1.50%	0.45%	0.20%	0.45%
Sustainable Euro Strategic Bond Fund	1.15%	1.50%	0.45%	0.20%	0.45%
Sustainable European High Yield Bond Fund	1.50%	N/A	0.60%	N/A	0.60%
Sustainable Global Fixed Income Opportunities Fund	1.15%	1.50%	0.45%	0.25%	0.45%
Sustainable Global High Yield Bond Fund	1.50%	2.00%	0.60%	N/A	0.60%

Management Fee	Share Class Indicators: A and B	Share Class Indicator: C	Share Class Indicators: F, I, Z	Share Class Indicator: J	Share Class Indicator: S
Sustainable US High Yield Bond Fund	1.25%	1.75%	0.60%	N/A	0.60%
US Dollar Corporate Bond Fund	0.80%	1.35%	0.35%	0.20%	0.35%
US Dollar High Yield Bond Fund	1.25%	1.70%	0.50%	0.25%	0.50%
US Dollar Short Duration Bond Fund	0.50%	0.70%	0.20%	N/A	0.20%
US Dollar Short Duration High Yield Bond Fund	1.00%	1.30%	0.40%	0.20%	0.40%
Asset Allocation Funds					
Global Balanced Defensive Fund	1.00%	1.70%	0.40%	N/A	0.40%
Global Balanced Fund	1.50%	2.20%	0.60%	N/A	0.60%
Global Balanced Income Fund	1.50%	2.20%	0.60%	N/A	0.60%
Global Balanced Risk Control Fund of Funds	2.00%	2.70%	0.80%	N/A	0.80%
Global Balanced Sustainable Fund	1.50%	2.00%	0.60%	0.35%	0.60%
Global Multi-Asset Opportunities Fund	1.65%	2.40%	0.70%	N/A	0.70%
Multi-Asset Risk Control Fund	1.50%	2.00%	0.60%	0.35%	0.60%

No Management Fee is payable in respect of Share Class Indicator N Shares.

Where the Management Company and its affiliates invest on behalf of clients, or their clients invest directly into Share Class Indicator N Shares, such clients may pay to the Management Company or its affiliates a base management fee, a performance fee or a mixture of both a base management fee and a performance fee.

Pursuant to the Management Company Services Agreement, the Equity Funds, Bond Funds, Asset Allocation Funds and Alternative Investment Funds will pay the Management Company a <u>Distribution Fee</u>, accrued daily and paid monthly, at the annual rate of 1.00% of the average daily Net Asset Value of Shares with a Share Class Indicator B as compensation for providing distribution-related services to the Funds with respect to such Shares.

The Management Company may pay an amount of any Distribution Fee it receives to distributors with whom it has distribution agreements. Additionally, the Management Company or Investment Adviser(s) may pay a portion of the Management Fee related to any Share Class to distributors, dealers or other entities that assist the Management Company or Investment Adviser(s) in the performance of their duties or provide services, directly or indirectly, to the Funds or their Shareholders, and may pay a portion of the Management Fee related to any Share Class on a negotiated basis in a private arrangement with a holder or prospective holder of Shares. The selection of holders or prospective holders of Shares with whom such private arrangements may be made and the terms on which the Management Company or Investment Adviser(s) or their affiliates, designees or placement agents may enter into such private arrangements are a matter for the Management Company or Investment Adviser(s) except that the Investment Adviser(s) may not enter into any such arrangements unless, as a condition of any such arrangements, neither the Company nor the Management Company will thereby incur any obligation or liability whatsoever.

The Funds will also be subject to a monthly <u>Administration Charge</u> at a current maximum annual rate of 0.25% of the Funds' aggregate average daily Net Asset Value attributable to all Classes of Shares of the Funds which will be applicable as follows:

Share Class Indicator	Administration Charge	
A, B, C, and F	0.19%	

I and J	0.14%
S, N and Z	0.10%

The Administration Charge will remain in force for an unlimited period and may be changed by the Management Company upon no less than one month's prior written notice. The Administration Charge will be utilised by the Company to pay the fees of the Administrator, the Transfer Agent, the Domiciliary Agent and the Management Company. The Funds will also pay other expenses incurred in the operation of the Company out of the Administration Charge including, without limitation, formation expenses of new Funds, the fees of the Depositary, expenses for legal and auditing services, cost of any proposed listings, maintaining such listings, expenses which relate to the administrative or operational support of distribution (including the fees of local representatives or paying agents, platform fees and related administrative services), reporting and publishing expenses (including preparing, printing, advertising and distributing Shareholders' reports and prospectuses), the remuneration of the Directors including all reasonable out-of-pocket expenses, registration fees and other expenses due to supervisory authorities in various jurisdictions, insurance, interest, and the costs of publication of the Net Asset Value per Share, whether such publication is required for regulatory purposes or otherwise. As the amount of such expenses varies, the Administration Charge may generate a surplus or a loss for the Management Company. The Management Company may retain surpluses but will not seek to recover losses.

Certain Funds and Share Classes will also pay <u>Specific Additional Costs</u> in addition to the Administration Charge, including, without limitation, *taxe d'abonnement*, additional custody fees applicable to investment in emerging markets, hedging expenses and the costs relating to the subsidiaries. These Specific Additional Costs will remain in force for an unlimited period and may be changed by the Management Company upon no less than one month's prior written notice. These Specific Additional Costs are applicable as follows:

Taxe d'abonnement

Share Class Indicator	taxe d'abonnement		
A, B, C, F, and I	0.05%		
J, N, S, and Z	0.01%		

Hedging Expenses

Share Class Sub-Indicator	Hedging Expenses	
H, H1, H2, and H3	Up to 0.03%	

Specific Additional Costs are made on the following Funds in addition to the share class costs above:

	Emerging Market Custody		
China A-shares Fund, China	Un to 0 100%		
Equity Fund	Up to 0.10%		
Asia Opportunity Fund,			
Developing Opportunity Fund,			
Parametric Emerging Markets			
Fund, Emerging Leaders Equity	Up to 0.05%		
Fund, Indian Equity Fund, Latin			
American Equity Fund,			
NextGen Emerging Markets			
Fund, Sustainable Asia Equity			
Fund, Sustainable Emerging			
Markets Equity Fund.			
Emerging Markets Debt			
Opportunities Fund, Emerging			
Markets Local Income Fund,	Up to 0.25%		
Frontier Markets Local Debt			
Fund, Saudi Equity Fund			

The Management Company may pay an amount of the Fees and Charges it receives to distributors, dealers or other entities that assist the Management Company in the performance of its duties or provide services, directly or indirectly to the Funds or their Shareholders.

The Management Company may instruct the Company to pay a proportion of any Fee, Charge or Cost directly out of the assets of the Company to any service providers. In such case the Fee, Charge or Cost is reduced accordingly.

In the case of any individual Fund or Class, the Management Company and/or Investment Adviser(s) may choose to waive or rebate (to the investors) all of their Fees and Charges or any portion thereof at their absolute discretion for an indefinite period. The Management Company or any person acting on behalf of the Company or the Management Company may not obtain an incentive allocation or commission on any fees or charges levied by an underlying scheme or its management company, or any quantifiable monetary benefits in connection with investments in any underlying fund.

All Fees, Charges and Costs are payable monthly in arrears and calculated on the average daily net assets (before the deduction of any Fees, Charges and Costs).

The allocation of costs and expenses to be borne by the Company among the various Funds and Classes will be made in accordance with Article 11 of the Articles of Incorporation. It is possible that Specific Additional Costs charged to the Funds and Classes may generate a surplus or a loss for the Management Company as actual expenses paid may vary from the amounts set out above. The Management Company may retain surpluses but will not seek to recover losses.

In certain jurisdictions, where subscriptions, redemptions and conversions are made through a third party agent, additional fees and charges may be imposed by that third party. These may be charged to the local investor or to the Company, and paid out of the Administration Charge.

Other Charges

Transaction Fees

Each Fund bears the costs and expenses of buying and selling portfolio securities and financial instruments, brokerage fees and commissions, interest or taxes payable, and other transaction related expenses. These transaction fees are accounted for on a cash basis and are paid when incurred or invoiced from the net assets of the Fund to which they are attributable. Transaction fees are allocated across each Fund's Share Classes.

Extraordinary Expenses

The Company bears any extraordinary expenses including, without limitation, litigation expenses and the full amount of any tax, levy, duty or similar charge and any unforeseen charges imposed on the Company or its assets.

Costs relating to the use of efficient portfolio management techniques

The Company has entered into securities lending arrangements with the Depositary to participate in the securities lending programme operated by the Depositary, under normal commercial terms. Under the terms relating to the relevant securities lending arrangement and for services provided, the Depositary retains 20% on the first US\$ 5 million of gross revenue generated from securities lending transactions and 15% thereafter. The Funds retain 80% on the first US\$ 5 million of gross revenue generated from securities lending transactions and 85% thereafter.

All revenues arising from repurchase and reverse repurchase agreements, net of direct and indirect operational costs, will be accrued to the relevant Fund.

Underlying Fees

As described in Section 1.5 "Risk Factors", where a Fund invests into another collective investment scheme the Fund will bear a proportionate share of the fees and expenses of such collective investment scheme. These are in addition to the fees and expenses above but will be reported as part of the ongoing charges figure disclosed in the Annual Report and Key Investor Information Documents.

Transfer of shares

The Management Company may, where it determines it is in the best interests of the Company and its Shareholders, charge a fee to any Shareholder who requests that their investment is re-registered into a different Shareholder account. Such fee will be payable to the Management Company to compensate it for the costs of processing such request, out of the Shareholder's investment, such amount not to exceed $\[oldsymbol{\in} 50\]$ per transfer.

2.6 Publication of Net Asset Value

The Net Asset Value per Share of each Class within each Fund is made public at the registered office of the Management Company and is available at the offices of the Transfer Agent. The Management Company will arrange for the Net Asset Value per Share of each Class within each Fund to be published as required. Neither the Company nor the Management Company can accept any responsibility for any error or delay in publication or for inaccurate or non-publication of prices. Shareholders may view the Net Asset Value per Share on the Company's web-site (http://www.morganstanleyinvestmentfunds.com).

2.7 Net Asset Value Determination

The Net Asset Value per Share will be calculated in the following currencies:

	US Dollar	Euro	Yen	Sterling	Currency in which the Shares are denominated
The Net Asset Value per Share of each Class in each Fund will be calculated in	х	X			
Except for the following Funds and Share Classes:					
 Japanese Equity Fund 	X	X	X		
Share Class Indicator A Shares of the Global Brands Fund	X	X	X		
• Share Class Indicator Z Shares of the Global Brands Fund	X	X		X	
Share Class Indicator Z Shares of the Asian Property Fund	X	X		X	
Share Class Indicator N and Z Shares of the Sustainable Emerging Markets Equity Fund	X	X		х	
• Share Class Indicator Z shares of the US Growth Fund	X	X		X	
• Share Class Indicator Z Shares of the Global Property Fund	X	X		X	
Where Distributing Share Classes (Share Class Sub-Indicators "X" and "M") and Discretionary Distributing Share Classes (Share Class Sub-Indicators "R" and "RM") are issued by a Fund, the Net Asset Value per Share of the Class may also be calculated in:	x	x		X	
Where Currency Hedged Share Classes (Share Class Sub-Indicators "H", "H1", "H2", "H3", and "O") are issued by a Fund, the Net Asset Value per Share of the Class will be calculated only in:					X
Where currency indicators are referenced within a Share Class, the Net Asset Value per Share of the Class will be calculated in that currency only (e.g. Share Class A (GBP) or Share Class B (AUD)):					Х

In addition to the currency combinations detailed above, the Management Company reserve the right to publish the Net Asset Value per Share of a particular Class of a Fund in additional combinations not specified above and in additional currencies not detailed in the table. Any such currency Net Asset Value per Share published after the date of this Prospectus would be listed at www.morganstanleyinvestmentfunds.com and included in the next update of this Prospectus.

The Company determines the principles of the calculation of the price or net asset value of its Shares, which are implemented by the Management Company on a forward basis. This means that it is not possible to know in advance the Net Asset Value per Share at which Shares will be bought or sold (exclusive of any Sales Charge). The Net Asset Value per Share is calculated at the valuation point following the Cut-Off Point that applications for Shares must be received in respect of a Dealing Day by the Transfer Agent to be processed on that Dealing Day.

The Funds are valued daily and the Net Asset Value per Share is calculated at the valuation point on each Dealing Day save where a particular Fund has a different definition of Dealing Day included in the investment policy section relating to that Fund. The Net Asset Value per Share for all Funds will be determined on the basis of the last available prices at the valuation point from the markets on which the investments of the various Funds are principally traded. The last available price and valuation point for collective schemes in which the Funds invest may vary and may include prior dealing days of those collective investment schemes. The valuation point of the Funds is available upon written request.

Events may occur between the determination of an investment's last available price and the determination of a Fund's Net Asset Value per Share at the valuation point that may, in the opinion of the Directors, mean that the last available price does not truly reflect the fair market value of the investment. In such circumstances the price of such investment shall be adjusted in accordance with the procedures adopted from time to time by the Directors in their discretion. The process and conduct of fair value adjustments shall be done by the Management Company in consultation with the Depositary.

In certain circumstances, subscriptions, redemptions and conversions in a Fund may have a negative impact on the Net Asset Value of the Fund as a result of the transaction costs incurred in the purchase and sale of the underlying investments and the spread between the buying and selling prices of such investments. In order to protect existing or remaining Shareholders from this negative impact on the Net Asset Value and to the extent that the Management Company considers that it is in the best interests of Shareholders, taking into account factors including the prevailing market conditions, the level of subscriptions and redemptions in a particular Fund and the size of the Fund, the Management Company may adjust the Net Asset Value of a Fund ("Swing Pricing") to reflect the estimated dealing spreads, costs and charges to be incurred by the Fund in liquidating or purchasing investments ("Swing Factor") to satisfy the net transactions received in respect of a particular Dealing Day.

The Swing Factor shall not exceed 2% of the Net Asset Value of the relevant Fund on the relevant Dealing Day in normal market conditions. However, whilst the Swing Factor is normally not expected to exceed 2% of the Net Asset Value of the relevant Fund on the relevant Dealing Day, the Directors may decide to temporarily increase this limit in exceptional circumstances (*e.g.* stressed or dislocated markets resulting in increased trading costs beyond the 2% cap) to protect Shareholders' interests, although it is not possible to accurately predict whether it will occur at any future point in time and consequently how frequently it will need to be made. Up-to-date information on the increased Swing Factor actually applied to the Fund will be made available on the Company's website (www.morganstanley.com/im) and may also be made available to Shareholders free of charge upon request. Shareholders will also be informed on these websites when the market conditions no longer require that the adjustment limit exceeds the level disclosed in the Prospectus.

When net subscriptions in a Fund exceed a certain threshold on a given Dealing Day, the Net Asset Value is adjusted upwards by the Swing Factor. Similarly, when net redemptions in a Fund exceed a certain threshold on a given Dealing Day, the Net Asset Value of the Fund is adjusted downwards by the Swing Factor. The Management Company will review and, as the case may be, adapt the threshold(s) from time to time. The Swing Pricing will apply to all Funds except the Global Balanced Risk Control Fund of Funds.

The Net Asset Value per Share is calculated at the valuation point on each Dealing Day in the Reference Currency of the relevant Fund and is then calculated as appropriate in US Dollars, Yen, Euro or Sterling using the last available rate of exchange prevailing in a recognised market at the valuation point.

The Net Asset Value per Share for Funds with investments with a known short term maturity date will be determined using an amortised cost method for those investments with a known short term maturity date. This involves valuing an investment at its cost and thereafter assuming a constant amortisation to maturity of any discount or premium, regardless of the impact of fluctuating interest rates on the market value of the investments.

While this method provides certainty in valuation, it may result in periods during which value, as determined by amortised cost, is higher or lower than the price the relevant Fund would receive if it sold the investment. The Directors will continually assess this method of valuation and recommend changes, where necessary, to ensure that the Funds' investments will be valued at their fair value as determined in good faith by the Directors. If the Directors believe that a deviation from the amortised cost per Share may result in material dilution or other unfair results to Shareholders, the Directors shall take such corrective action, if any, as they deem appropriate to eliminate or reduce, to the extent reasonably practicable, the dilution or unfair results.

The Net Asset Value per Share of each Class of Shares of each Fund is determined by dividing the value of the assets of the Fund properly allocable to such Class of Shares less the value of the liabilities of the Fund properly allocable to such Class of Shares by the total number of Shares of such Class outstanding at any Dealing Day.

The Net Asset Value per Share of each Class of Shares of each Fund is determined in accordance with Article 11 of the Articles of Incorporation of the Company which, *inter alia*, sets out the following rules to be applied in determining such value:

- a) the value of any cash on hand or on deposit, bills and demand notes and accounts receivable, prepaid expenses, cash dividends and interest declared or accrued as aforesaid and not yet received is deemed to be the full amount thereof, unless in any case the same is unlikely to be paid or received in full, in which case the value thereof is arrived at after making such discount as may be considered appropriate in such case to reflect the true value thereof;
- b) securities listed on a recognised stock exchange or dealt on any Other Regulated Market (as defined in Appendix A hereinafter) will be valued at their last available prices, or, in the event that there should be several such markets, on the basis of their last available prices on the main market for the relevant security. In the event that the last available price does not truly reflect the fair market value of the relevant securities, the value of such securities will be based on the reasonably foreseeable sales price determined prudently and in good faith pursuant to procedures established by the Directors;
- securities not listed or traded on a stock exchange or not dealt on any Other Regulated Market will be valued
 on the basis of the reasonably foreseeable sales price determined prudently and in good faith pursuant to
 procedures established by the Directors;
- d) the liquidating value of futures, or options contracts not traded on exchanges or on other organised markets shall mean their net liquidating value determined, pursuant to the policies established by the Directors, on a basis consistently applied for each different variety of contracts. The liquidating value of futures or options contracts traded on exchanges or on other organised markets shall be based upon the last available prices of these contracts on exchanges and organised markets on which the particular futures or options contracts are traded by the Company; provided that if a futures or options contract could not be liquidated on the day with respect to which net assets are being determined, the basis for determining the liquidating value of such contract shall be such value as the Directors may deem fair and reasonable;
- e) all other securities and other assets will be valued at fair market value as determined in good faith pursuant to procedures established by the Directors;
- f) interest rate swaps will be valued at their market value established by reference to the applicable interest rates curve. Index and financial instruments related swaps will be valued at their market value established by reference to the applicable index or financial instrument. The valuation of the index or financial instrument related swap agreement shall be based upon the market value of such swap transaction established in good faith pursuant to procedures established by the Directors. The value of the credit default swaps shall be determined by applying a recognised and transparent valuation method on a regular basis and by reference to the applicable debt instrument.

The relevant Funds shall, in principle, keep in their respective portfolios the investments determined by the amortisation cost method until their respective maturity dates or sell dates. Any assets held in a particular Fund not expressed in the Reference Currency will be translated into the Reference Currency at the last available rate of exchange prevailing in a recognised market at the valuation point.

The net asset value of the Company is at any time equal to the total of the Net Asset Values of the various Funds, converted, where necessary, into US Dollars at the last available rate of exchange prevailing in a recognised market at the relevant valuation point.

Temporary suspension of calculation of Net Asset Value

Pursuant to Article 12 of the Articles of Incorporation, the Company may suspend the calculation of the Net Asset Value of one or more Funds and the issue, redemption and conversion of Shares:

- a) during any period when any of the principal stock exchanges or other markets on which a substantial portion of the investments of the Company attributable to such Fund from time to time is quoted or dealt in is closed otherwise than for ordinary holidays, or during which dealings therein are restricted or suspended, provided that such restriction or suspension affects the valuation of the investments of the Company attributable to such Fund quoted thereon;
- b) during the existence of any state of affairs which constitutes an emergency in the opinion of the Directors as a result of which disposal or valuation of assets owned by the Company attributable to such Fund would be impracticable;
- during any breakdown in the means of communication or computation normally employed in determining the
 price or value of any of the investments of such Fund or the current price or value on any stock exchange or
 other market in respect of the assets attributable to such Fund;
- d) during any period when the Company is unable to repatriate funds for the purpose of making payments on the redemption of Shares of such Fund or during which any transfer of funds involved in the realisation or

acquisition of investments or payments due on redemption of Shares cannot, in the opinion of the Directors, be effected at normal rates of exchange;

- e) when for any other reason the prices of any investments owned by the Company attributable to such Fund cannot promptly or accurately be ascertained;
- f) any period when the net asset value of any subsidiary of the Company may not be determined accurately;
- g) upon the publication of a notice convening a general meeting of Shareholders for the purpose of resolving the winding-up of the Company;
- h) following the suspension of the calculation of the Net Asset Value per Share, the issue, redemption and/or conversion of Shares, at the level of a master fund in which a Fund invests in its quality of feeder fund of such master fund.

The suspension of calculation of the Net Asset Value of any Fund shall have no effect on the calculation of the Net Asset Value per Share, the issue, redemption and conversion of Shares of any other Fund.

Any request for subscription, redemption or conversion shall be irrevocable except in the event of a suspension of the calculation of the Net Asset Value per Share of any relevant Class.

Notice of the beginning and of the end of any period of suspension will be published on the Company's website (www.morganstanleyinvestmentfunds.com). Notice will likewise be given to any applicant or Shareholder as the case may be applying for purchase, conversion or redemption of Shares in the Fund(s) concerned.

2.8 Best Execution

The Management Company must act in the best interests of each Fund when executing decisions to deal on behalf of the relevant Fund. The Management Company's best execution policy sets out the basis upon which the Management Company, and the Investment Adviser on its behalf, will effect transactions and place orders in relation to the Funds to obtain the best possible result for the Company. The Investment Adviser's best execution policy sets out the systems and controls that have been put in place and the oversight carried out by the Investment Adviser to ensure that notwithstanding any delegation to the Sub-Advisers, who may be non-EU entities, they undertake transactions to achieve the best result for the relevant Funds. All transactions carried out by or on behalf of the Fund must be executed at arm's length and in the best interests of the Shareholders.

2.9 Conflicts of Interests

Potential Conflicts of Interests of the Investment Advisers and the Management Company

The Management Company and the Investment Advisers undertake business for other clients. The Investment Advisers, their employees and their other clients face conflicts with the interests of the Management Company and its clients. The Management Company and the Investment Advisers maintain Conflicts of Interest Policies. It is not always possible for the risk of detriment to a client's interests to be entirely mitigated such that, on every transaction when acting for clients, a risk of detriment to their interests does not remain.

The Management Company and the Investment Advisers acknowledge that there may be some situations where the organisational or administrative arrangements in place for the management of conflicts are not sufficient to ensure, with reasonable confidence, that risks of damage to the interests of the Company or its shareholders will be prevented. Should any such situations arise the Management Company will disclose these to Shareholders in the Company's annual report or other appropriate format.

Conflict Scenarios

Because the Investment Advisers or their affiliates may manage assets for other investment companies, pooled investment vehicles, and/or other accounts (including institutional clients, pension plans and certain high net worth individuals), there may be an incentive to favour one of their clients over another resulting in potential conflicts of interest. For instance, the relevant Investment Adviser or its affiliates may receive fees from certain accounts that are higher than the fee it receives for a particular Fund, or it may receive a performance-based fee on certain accounts. In those instances, the portfolio manager(s) may have an incentive to favour the higher and/or performance-based fee accounts over a particular Fund. In addition, a potential conflict of interest could exist to the extent such Investment Adviser or its affiliates has proprietary investments in certain accounts, where portfolio managers have personal investments in certain accounts or when certain accounts are investment options in the Investment Adviser's and/or its affiliates employee benefits and/or deferred compensation plans. The relevant portfolio manager may have an incentive to favour these accounts over others. If the Investment Advisers and/or one of their affiliates manages accounts that engage in short sales of securities of the type in which a Fund invests, the Investment Advisers and/or their affiliates could be seen as harming the performance of the relevant Fund for the benefit of the accounts engaging in short sales if the short sales cause the market value of the securities to fall.

The Management Company may pay third parties for distribution and related services. Such payments could incentivise third parties to promote the Company to investors against that client's best interests. The Management Company complies with all legal and regulatory requirements in the jurisdictions in which such payments are made.

The Investment Advisers also may utilize an affiliated broker/dealer to execute a portion, and on occasion to execute a substantial portion, of the purchases and sales of securities for the portfolio of a Fund. Primary consideration will be given to such broker/dealer's ability to effect prompt and efficient executions at competitive rates. The Investment Advisers will only utilize an affiliated broker/dealer where it believes it is able to obtain best execution at competitive commission rates. An affiliated broker/dealer may also furnish to the Investment Advisers research and related services.

The Investment Advisers may purchase securities from underwriters, distributors or placement agents in which an affiliate is a member of a syndicate or selling group, as a result of which an affiliate might benefit from the purchase through receipt of a fee or otherwise. The Investment Advisers will not purchase securities from an affiliate that is acting as a manager of a syndicate or selling group. Purchases by the Investment Advisers from an affiliate acting as a distributor or placement agent must meet the requirements of applicable law. Otherwise, the Investment Advisers will not engage in any purchase or sale with an affiliate on a principal basis except as set forth below.

Other Potential Conflicts of Interest

As part of the normal course of global custody business, the Depositary may from time to time have entered into arrangements with other clients, funds or other third parties for the provision of safekeeping and related services. Within a multi-service banking group such as JPMorgan Chase Group, from time to time conflicts may arise between the Depositary and its safekeeping delegates, for example, where an appointed delegate is an affiliated group company and is providing a product or service to a fund and has a financial or business interest in such product or service or where an appointed delegate is an affiliated group company which receives remuneration for other related custodial products or services it provides to the funds, for instance foreign exchange, securities lending, pricing or valuation services. In the event of any potential conflict of interest which may arise during the normal course of business, the Depositary will at all times have regard to its obligations under applicable laws including Article 25 of the UCITS Directive.

In relation to any efficient portfolio management techniques used by the Funds, securities lending, repurchase and reverse repurchase transactions may be entered into on normal commercial terms with entities which are affiliates of the Investment Adviser. In particular, the Funds have entered into securities lending arrangements with the Depositary to participate in the securities lending programme operated by the Depositary, under normal commercial terms. From time to time, affiliates of the Investment Adviser (including but not limited to Morgan Stanley & Co. International PLC) may be named as approved borrowers to which the Depositary could lend securities on behalf of the Funds under the securities lending programme.

The above is not necessarily a comprehensive list of all potential conflicts of interest.

2.10 Dividend Policy

The Company offers Accumulating Share Classes, Distributing Shares Classes and Discretionary Distributing Share Classes.

Accumulating Share Classes

Income and capital gains arising in each Fund in relation to Accumulating Share Classes will be reinvested in such Fund. The value of the Shares of each such Class will reflect the capitalisation of income and gains. The Directors currently intend to propose to the Annual General Meeting the reinvestment of the net results of the year for all such Classes of Shares. However, should payment of a dividend in respect of any such Classes of Shares be considered to be appropriate, the Directors will propose to the general meeting of Shareholders that a dividend be declared out of the net investment income attributable to such Class of Shares and available for distribution and/or realized capital gains after deduction of realized capital losses.

Distributing Share Classes (Share Class Sub-Indicators "X" and "M") and Discretionary Distributing Share Classes (Share Class Sub-Indicators "R" and "RM").

Approach to declaring dividends

Funds – Distributing Share Classes (Share Class Sub-Indicators "X" and "M")

For the Distributing Share Classes of the Bond Funds, Equity Funds, Asset Allocation Funds and Alternative Investment Funds, the Company intends to declare dividends which will be equal to the net investment income

attributable to such Classes (i.e. 100% of the net investment income). For the Global Asset Backed Securities Fund, net investment income is considered to include net gains and losses from prepaid principal on mortgaged-backed and asset-backed securities for the purpose of the dividend policy. For the Global Balanced Income Fund, net investment income is considered to include net premium arising as a result of writing options for the purpose of the dividend policy.

Funds - Discretionary Distributing Share Classes (Share Class Sub-Indicators "R" and "RM")

For the Discretionary Distributing Share Classes of the Bond Funds, Equity Funds, Asset Allocation Funds and Alternative Investment Funds, the Company intends to declare dividends which will be set at the discretion of the Directors. The Discretionary Distributing Share Class may pay distributions from capital or may charge all or part of the Fund's fees and expenses to the capital of the Fund. Dividends paid out of capital amount to a return or withdrawal of part of a Shareholder's original investment or from any capital gains attributable to that original investment. Such dividends may result in an immediate decrease of the Net Asset Value per Share. Where part or all of the fees and expenses of a Share Class are charged to the capital of that Share Class the result will be a reduction of the capital that the Share Class has available for investment in the future and capital may be eroded. Further details on the Discretionary Distributing Share Classes will be included in the relevant key investor information document of the applicable Fund.

Frequency of Dividends

Equity Funds and Alternative Investment Funds - semi-annual dividend (Share Class Sub-Indicators "X" and "R")

Both the Distributing Share Classes (Share Class Sub-Indicator "X") and Discretionary Distributing Share Classes (Share Class Sub-Indicator "R") aim to declare semi-annual dividends for Equity Funds (other than the Global Brands Equity Income Fund) and Alternative Investment Funds. Such dividends, if any, will be accrued on the last Dealing Day of June and December and declared on the next Dealing Day. The dividend declaration will be made public at the registered office of the Company and made available at the offices of the Transfer Agent on the first Dealing Day of July and January.

Bond Funds, Asset Allocation Funds and the Global Brands Equity Income Fund – quarterly dividend (Share Class Sub-Indicators "X" and "R")

Both the Distributing Share Classes (Share Class Sub-Indicator "X") and Discretionary Distributing Share Classes (Share Class Sub-Indicator "R") aim to declare quarterly dividends for Bond Funds, Asset Allocation Funds and the Global Brands Equity Income Fund. Such dividends, if any, will be accrued on the last Dealing Day of March, June, September and December and declared on the next Dealing Day. The dividend declaration, where issued, will be made public at the registered office of the Company and made available at the offices of the Transfer Agent on the first Dealing Day of January, April, July and October as applicable.

Monthly Distributing Share Class (Share Class Sub-Indicators "M" and "RM")

Both the Distributing Share Classes (Share Class Sub-Indicator "M") and Discretionary Distributing Share Classes ("Share Class Sub-Indicator "RM") aim to declare monthly dividends. Such dividends, if any, will be accrued on the last Dealing Day of the month and declared on the next Dealing Day. The dividend declaration for these Distributing Share Classes and Discretionary Distributing Share Classes, where issued, will be made public at the registered office of the Company and made available at the offices of the Transfer Agent on the first Dealing Day of the following month as applicable.

Discretionary Distributing Share Class Examples

In these examples, we assume the following for the period:

- 1) No change in NAV as a result of capital appreciation
- 2) Investment income over the period is equal to 2.50%
- 3) The total management and non-management expenses of the fund for this period are equal to 0.50% of NAV
- 4) Therefore net investment income is 2%

Example – covering a period of 90 days from January 1 to the first Dealing Day in April for a given Bond Fund where net investment income is greater than or equal to the declared dividend.

In this example, we assume the following for the period:

1) Declared dividend is 1.50% (less than net investment income)

NAV on the first Dealing Day of the Period	NAV on Dealing Day prior to the declaration of the dividend	Declared Dividend as a Percentage of NAV	Declared Dividend	NAV on the first Dealing Day following the declaration of dividend
€100.00	€102.00	1.50%	€1.53	€100.47

In this example, the dividend declared is equal to $\[\in \]$ 1.53 and the NAV on the Dealing Day following the declaration of the dividend is equal to $\[\in \]$ 100.47 ($\[\in \]$ 102.00 – (1.50%*102.00) = $\[\in \]$ 100.47). In this scenario, the declared dividend was less than the net investment income over the period and there was no decrease in capital.

Example – covering a period of 90 days from January 1 to the first Dealing Day in April for a given Bond Fund where the declared dividend is greater than the net investment income.

In this example, we assume the following for the period:

2) Declared dividend is 3.00% (greater than net investment income)

NAV on the first Dealing Day of the Period	NAV on Dealing Day prior to the declaration of the dividend	Declared Dividend as a Percentage of NAV	Declared Dividend	NAV on the first Dealing Day following the declaration of dividend
€100.00	€102.00	3.00%	€3.06	€98.94

In this example, the dividend declared is equal to $\[\in \] 3.06$ and the NAV on the Dealing Day following the declaration of the dividend is equal to $\[\in \] 98.94$ ($\[\in \] 102.00 - (3.00\%*102.00)$). The declared dividend has resulted in a decrease in investor capital of $\[\in \] 1.06$.

Figures are for illustration purposes only and are not indicative of the actual return received by Shareholders.

Dividend payment and reinvestment

Dividends will be distributed within three Business Days of the date such dividends are declared, in the case of the Equity Funds, Bond Funds, Asset Allocation Funds and Alternative Investment Funds.

For all Shares other than Share Class Indicator B Shares

Dividends will be automatically reinvested in additional Shares of the relevant Class free of any charge unless (i) the Shareholder has nominated on the Application Form to receive such dividends in cash and (ii) the value of the dividends to be distributed for that month, quarter or half year, as applicable, is in excess of US\$ 100.00, or the Euro or the Sterling equivalent of US\$ 100.00. Dividends are paid in the currency in which the Share is denominated, or, where appropriate for the Share, in Sterling, Euro, or US Dollar. For those Shareholders whose dividend accrual for that month, quarter or half year, as applicable, is less than US\$ 100.00, or the Euro or the Sterling equivalent of US\$ 100.00, the dividends will be automatically reinvested in additional Shares of the relevant Class free of any charge. The Directors may, following a Shareholder's request, at their discretion, decide to waive or vary the limit below which a dividend accrual is automatically reinvested.

The Dealing Day applicable for such automatic reinvestment will be the first Dealing Day of January, April, July or October in the case of the Bond Funds and Asset Allocation Funds, and the first Dealing Day of January or July in the case of the Equity Funds and Alternative Investment Funds.

In the case of the Monthly Distributing Share Class (Share Class Sub-Indicator "M"), such automatic reinvestment will be the first Dealing Day of each month.

For Share Class Indicator B Shares

All dividends earned on Share Class Indicator B Shares will be distributed to the relevant Shareholder in cash, regardless of the value of the dividend. Reinvestment of dividends is not permitted for Share Class Indicator B Shares. Dividends are paid in the currency in which the Share is denominated, or, where appropriate for the Share, in Sterling, Euro, or US Dollar.

For all Shares

Income equalisation is operated in respect of all Distributing Share Classes of all Funds. For such Share Classes, equalisation ensures that the income per Share which is distributed in respect of a distribution period is not impacted

by changes in the number of Shares in issue in that Share Class during the period. Equalisation is operated by the Administrator, who allocates a portion of the proceeds from sales and costs of redemption of Shares – equivalent on a per Share basis to the amount of undistributed net investment income on the date of the subscription or redemption – to undistributed income.

In the event that a dividend is paid in one or several Funds, such dividend will be paid to Shareholders by cheque mailed to their address as shown on the register of Shareholders, or by bank transfer. Dividend cheques not cashed within five years will be forfeited and will accrue for the benefit of the Class of the Fund out of which the dividend is payable.

2.11 Taxation

The statements on taxation set out below are by way of a general guide to potential investors and Shareholders only regarding the law and practice in force in the relevant jurisdiction at the date of this Prospectus and do not constitute tax or legal advice and the investors should consult their own counsel and make their own enquiries, as to tax matters concerning their investment. There can be no guarantee that the tax position or proposed tax position prevailing at the time an investment is made will endure indefinitely. Neither the Company, the Management Company nor their advisors are liable for any loss which may arise as a result of current, or changes in, applicable tax laws, practice and their interpretation by any relevant authority.

Prospective Shareholders should inform themselves of, and where appropriate take advice on, the laws and regulations (such as those relating to taxation and exchange controls) applicable to the subscription, purchase, holding, redemption, conversion and otherwise disposing of Shares in the country of their citizenship, residence, domicile or incorporation. Prospective Shareholders should note that the information contained in this Section of the Prospectus is a general guide only and does not discuss the local tax consequences that all prospective Shareholders may be subject to.

The conversion of Share Class Indicator B Shares to Share Class Indicator A Shares as described in Section 2.1 "Share Class Description" may give rise to a tax liability for Shareholders in certain jurisdictions. Shareholders should consult their tax adviser about their own position.

Luxembourg Taxation

Please be aware that the residence concept used under the respective headings below applies for Luxembourg income tax assessment purposes only. Any reference in the present section to a tax, duty, levy, impost or other charge or withholding of a similar nature refers to Luxembourg tax law and/or concepts only. Also, please note that a reference to Luxembourg income tax generally encompasses corporate income tax (*impôt sur le revenu des collectivités*), municipal business tax (*impôt commercial communal*), a solidarity surcharge (*contribution au fonds pour l'emploi*) and personal income tax (*impôt sur le revenu*). Corporate taxpayers may further be subject to net wealth tax (*impôt sur la fortune*), as well as other duties, levies or taxes. Corporate income tax, municipal business tax, net wealth tax as well as the solidarity surcharge invariably apply to most corporate taxpayers resident of Luxembourg for tax purposes. Individual taxpayers are generally subject to personal income tax and to the solidarity surcharge. Under certain circumstances, where an individual taxpayer acts in the course of the management of a professional or business undertaking, municipal business tax may apply as well.

Taxation of the Company in Luxembourg

Under current law, the Company is not liable to any Luxembourg income tax, nor are dividends paid by the Company liable to any Luxembourg withholding tax.

Subscription tax

As a rule the Company is liable in Luxembourg to a subscription tax (*taxe d'abonnement*) of 0.05% per annum. The taxable basis of the subscription tax is the aggregate net assets of the Company valued on the last day of each quarter of the civil year.

The rate is however reduced to 0.01% per annum for:

- undertakings whose sole object is the collective investment in money market instruments and the placing of deposits with credit institutions;
- undertakings whose sole object is the collective investment in deposits with credit institutions;
- individual compartments of UCIs with multiple compartments as well as for individual classes of securities issued within a UCI or within a compartment of a UCI with multiple compartments, provided that the securities of such compartments or classes are reserved to one or more institutional investors.

Furthermore, exempt from the subscription tax are:

- the value of the assets represented by units held in other UCIs, provided such units have already been subject to the subscription tax provided for in Article 174 of the 2010 Law or in Article 68 of the amended law of 13 February 2007 on specialised investment funds, or by Article 46 of the amended law of 23 July 2016 on reserved alternative investment funds;
- UCIs, as well as individual compartments of UCIs with multiple compartments (i) whose securities are reserved for institutional investors, and (ii) whose sole object is the collective investment in money market instruments and the placing of deposits with credit institutions, and (iii) whose weighted residual portfolio maturity does not exceed 90 days, and (iv) that have obtained the highest possible rating from a recognised rating agency. Where several classes of securities exist within the UCI or the compartment, the exemption only applies to classes whose securities are reserved for institutional investors;
- UCIs whose securities are reserved for (i) institutions for occupational retirement pension or similar investment vehicles, set up on one or more employers' initiative for the benefit of their employees and (ii) companies of one or more employers investing funds they hold, to provide retirement benefits to their employees;
- UCIs as well as individual compartments of UCIs with multiple compartments whose main objective is the investment in microfinance institutions;
- ETFs as defined by Article 175 e) of the 2010 Law.

Provided the continuous compliance with the above mentioned conditions for a reduced subscription tax rate of 0.01% the Net Asset Value of Shares with Share Class Indicator J, N, S or Z is likely to benefit from such a reduced subscription tax rate. However, there can be no guarantee that the benefit of such reduced rate will be obtained or that, once obtained, it will continue to be available in the future.

Subscription tax is calculated by reference to the net asset value of the relevant Fund or Share Class at the end of each quarter. Subscription tax will be paid from the Specific Additional Cost described in section 2.5 "Charges and Expenses". Any difference between the Specific Additional Costs collected and the tax payable will be paid by or credited to the Management Company.

Other taxes

No stamp duty or other tax will be payable in Luxembourg in connection with the issue of Shares of the Company. A fixed registration duty of Euro 75 will be levied upon amendments of the articles of incorporation of the Company.

The Company is exempt from net wealth tax.

Under current law in Luxembourg, no Luxembourg tax is payable on the realised capital appreciation of the assets of the Company.

The Company may be subject to withholding tax on dividends and interest and to tax on capital gains in the country of origin of its investments. As the Company itself is exempt from income tax, withholding tax levied at source, if any, is not refundable in Luxembourg. It is not certain whether the Company itself would be able to benefit from Luxembourg's double tax treaties network. Whether the Company may benefit from a double tax treaty concluded by Luxembourg must be analysed on a case-by-case basis. Indeed, as the Company is structured as an investment company (as opposed to a mere co-ownership of assets), certain double tax treaties signed by Luxembourg may directly apply to the Company.

The Company is considered in Luxembourg as a taxable person for value added tax ("VAT") purposes without any input VAT deduction right. A VAT exemption applies in Luxembourg for services qualifying as fund management services. Other services supplied to the Company could potentially trigger VAT. The Company is registered for VAT in Luxembourg and is required to self-assess Luxembourg VAT on taxable services purchased from abroad.

No VAT liability arises in principle in Luxembourg in respect of any payments by the Company to its Shareholders, to the extent that such payments are linked to their subscription to the Shares and do not constitute the consideration received for any taxable services supplied.

Withholding tax

Under current Luxembourg tax law, there is no withholding tax on any distribution, redemption or payment made by the Company to its Shareholders under the Shares. There is also no withholding tax on the distribution of liquidation proceeds to the Shareholders.

EU Mandatory Disclosure Regime ('MDR')

The EU Council Directive 2018/822 amending EU Council Directive 2011/16 in relation to the disclosure of cross-border tax arrangements, known as DAC6 or MDR, came into force on 25 June 2018. Generally, Member States were obliged to transpose the Directive into local law by 31 December 2019. Luxembourg transposed the Directive into national law with the law of 25 March 2020, as amended or supplemented from time to time (the "**DAC6 Law**").

MDR applies to cross-border tax arrangements, which meet one or more specified characteristics (hallmarks), and which concern either more than one EU country or an EU country and a non-EU country. It mandates a reporting obligation for these tax arrangements if in scope, regardless of whether the arrangement is in compliance with local tax law.

Intermediaries (i.e. any person that designs, markets, organises or makes available for implementation or manages the implementation of a reportable cross-border arrangement or any person that knows, or could reasonably be expected to know that they have undertaken to provide aid, assistance or advice with respect to designing, marketing, organising, making available for implementation or managing the implementation of a reportable cross-border arrangement), if any, or the relevant taxpayer will be required to file information with their national tax authority within thirty (30) days of the first of the following dates:

- on the day after the reportable cross-border arrangement is made available for implementation; or
- on the day after the reportable cross-border arrangement is ready for implementation; or
- when the first step in the implementation of the reportable cross-border arrangement has been made; or
- where an intermediary is involved, the date on which the intermediary provided aid, assistance or advice.

The Luxembourg law of 24 July 2020 implemented EU Council 2020/876 of 24 June 2020 amending DAC6 which provides for an extension of the starting date of the filing and exchange of information. Reports for reportable cross-border arrangements with the first step implemented between 25 June 2018 and 30 June 2020 should be filed by 28 February 2021. In addition, cross-border arrangements implemented during the period between 1 July 2020 and 31 December 2020 should be filed by 30 January 2021.

The Management Company is a potential intermediary for the purposes of MDR and hence may be obliged to report cross-border arrangements that meet one or more hallmarks. These hallmarks may be generic or specific. As regards to the generic and certain specific hallmarks, these may only be taken into account to the extent that they meet so called "main benefit test", i.e., if obtaining a tax advantage constitutes the main benefit or one of the main benefits a person is expected to derive from an arrangement. Shareholders, as taxpayers, may have a secondary liability to report in-scope arrangements and hence should consult their own tax advisors for more information.

Foreign Account Tax Compliance ("FATCA")

Being established in Luxembourg and subject to the supervision of the CSSF in accordance with the 2010 Law, the Company will be treated as a Foreign Financial Institution (within the meaning of FATCA) for FATCA purposes.

On 28 March 2014, Luxembourg has entered into a Model 1 Intergovernmental Agreement with the U.S. ("Luxembourg IGA") that was transposed in national law by the Luxembourg law dated 24 June 2015 (the "FATCA Law"), as amended by the Law of 18 June 2020. This means that the Company must comply with the requirements of the Luxembourg IGA. This includes the obligation for the Company to regularly assess the status of its investors. To this end, the Company will need to obtain and verify information on all of its investors. Upon request of the Company, each investor shall agree to provide certain information (the FATCA Information as defined previously), including, in case of a Non-Financial Foreign Entity (within the meaning of the FATCA Law), the direct or indirect owners above a certain threshold of ownership of such investors, along with the required supporting documentation. Similarly, each investor shall agree to actively provide to the Company within thirty days any information like for instance a new mailing address or a new residency address that would affect its status.

The Company will be required, in certain circumstances, to disclose the names, addresses, date and place of birth, jurisdiction(s) of tax residence and taxpayer identification number(s) (if issued by the jurisdictions) of the investors as well as information like account balances, income and gross proceeds (non-exhaustive list) to the Luxembourg tax authorities. The Luxembourg tax authorities will in turn share the information with the U.S. authorities. A failure for the Company to obtain such information from each investor and to transmit it to the Luxembourg tax authorities may trigger the 30% withholding tax to be imposed on payments of U.S. source income and on proceeds from the sale of property or other assets that could give rise to U.S. source interest and dividends.

Any investor that fails to comply with the Company's documentation requests may be charged with any taxes imposed on the Company attributable to such investor's failure to provide the information and the Company may, in its sole discretion, redeem the Shares of such investor.

Investors who invest through intermediaries are reminded to check if and how their intermediaries will comply with this U.S. withholding tax and reporting regime. Investors should consult a U.S. tax advisor or otherwise seek professional advice regarding the above requirements.

Common Reporting Standard

Capitalised terms used in this section should have the meaning as set forth in the CRS Law, unless provided otherwise herein.

Under the terms of the CRS Law, the Company is likely to be treated as a Luxembourg Reporting Financial Institution.

As such, as from 30 June 2017 and without prejudice to other applicable data protection provisions as set out in the Company documentation, the Company is required to annually report to the Luxembourg tax authorities personal and financial information related, inter alia, to the identification of, holdings by and payments made to (i) certain investors as per the CRS Law (the "Reportable Persons") and (ii) Controlling Persons (within the meaning of CRS) of certain non-financial entities ("NFEs") which are themselves Reportable Persons. This information, as exhaustively set out in Annex I of the CRS Law (the CRS Information as defined previously), will include personal data related to the Reportable Persons such as names, addresses, date and place of birth, jurisdiction(s) of tax residence and taxpayer identification number(s) if issued by the jurisdiction(s).

The Company's ability to satisfy its reporting obligations under the CRS Law will depend on each investor providing the Company with the CRS Information, along with the required supporting documentary evidence. In this context, the investors are hereby informed that, as data controller, the Company will process the CRS Information for the purposes as set out in the CRS Law.

The investors undertake to inform their Controlling Persons, if applicable, of the processing of their CRS Information by the Company.

Additionally, the Company is responsible for the processing of personal data and each investor and Controlling Person has a right to access the data communicated to the Luxembourg tax authorities and to correct such data (if necessary). Any data obtained by the Company are to be processed in accordance with the applicable Data Protection Laws.

The investors are further informed that the CRS Information related to Reportable Persons within the meaning of the CRS Law will be disclosed to the Luxembourg tax authorities annually for the purposes set out in the CRS Law. In particular, Reportable Persons are informed that certain operations performed by them will be reported to them through the issuance of statements, and that part of this information will serve as a basis for the annual disclosure to the Luxembourg tax authorities.

Similarly, the investors undertake to inform the Company within thirty (30) days of receipt of these statements should any included personal data be not accurate. The investors further undertake to immediately inform the Company of, and provide the Company with all supporting documentary evidence of any changes related to the Information after occurrence of such changes within thirty (30) days.

Any investor that fails to comply with the Company's CRS Information or documentation requests may be held liable for penalties imposed on the Company and attributable to such investor's failure to provide the CRS Information or subject to disclosure of the CRS Information by the Company to the Luxembourg tax authorities.

Taxation of Shareholders in Luxembourg

It is expected that Shareholders in the Company will be resident for tax purposes in many different countries. Consequently, except as set-out below, no attempt is made in this Prospectus to summarise the taxation consequences for each investor subscribing, converting, holding or redeeming or otherwise acquiring or disposing of Shares of the Company. These consequences will vary in accordance with the law and practice currently in force in a Shareholder's country of citizenship, residence, domicile or incorporation and with his personal circumstances.

Shareholders should consult their own professional advisors on the possible tax or other consequences of buying, holding, transferring or selling the Company's Shares under the laws of their countries of citizenship.

Under current legislation Shareholders are generally not subject to any capital gains or income tax in Luxembourg, except Shareholders who are resident in Luxembourg for tax purposes or non-resident Shareholders who have a permanent establishment or a permanent representative in Luxembourg to which or whom the Shares are attributable.

Luxembourg tax residency

A Shareholder will not become resident, nor be deemed to be resident, in Luxembourg by reason only of the holding and/or disposing of Shares or the execution, performance, delivery and/or enforcement of its rights and obligations thereunder.

Luxembourg resident individuals

Dividends and other payments derived from the Shares by a resident individual Shareholder, who acts in the course of the management of either his/her private wealth or his/her professional/business activity, are subject to income tax at the ordinary progressive rates.

Capital gains realized upon sale, disposal or redemption of the Shares by a resident individual Shareholder, who acts in the course of the management of his/her private wealth, are not subject to income tax, unless said capital gains qualify either as speculative gains or as gains on a substantial participation. Capital gains are deemed to be speculative and are thus subject to income tax at ordinary rates if the Shares are disposed of within 6 months after their acquisition or if their disposal precedes their acquisition. A participation is considered to be a substantial participation where a resident individual Shareholder holds or has held, either alone or together with his spouse/partner and/or his minor children, either directly or indirectly at any time within the 5 years preceding the disposal, more than 10% of the share capital of the company whose shares are being disposed of. A Shareholder is also deemed to alienate a substantial participation if he acquired free of charge, within the 5 years preceding the transfer, a participation that was constituting a substantial participation in the hands of the alienator (or the alienators in case of successive transfers free of charge within the same 5-year period).

Capital gains realized on a substantial participation more than 6 months after the acquisition thereof are taxed according to the half-global rate method (*i.e.* the average rate applicable to the total income is calculated according to progressive income tax rates and half of the average rate is applied to the capital gains realized on the substantial participation). A disposal may include a sale, an exchange, a contribution or any other kind of alienation of the Shares.

Capital gains realized on the disposal of the Shares by a resident individual Shareholder, who acts in the course of the management of his/her professional/business activity, are subject to income tax at ordinary rates. Taxable gains are determined as being the difference between the price for which the Shares have been disposed of and the lower of their cost or book value.

Luxembourg resident companies

A Luxembourg resident company (*société de capitaux*) must include any profits derived, as well as any gain realized on the sale, disposal or redemption of Shares, in its taxable profits for Luxembourg income tax assessment purposes. Taxable gains are determined as being the difference between the sale, repurchase or redemption price and the lower of the cost or book value of the Shares sold or redeemed.

Luxembourg residents benefiting from a special tax regime

Shareholders who are Luxembourg resident companies benefiting from a special tax regime, such as (i) undertakings for collective investment subject to the 2010 Law, (ii) specialized investment funds subject to the amended law of 13 February 2007, (iii) family wealth management companies governed by the amended law of 11 May 2007 and (iv) reserved alternative investment funds governed by the amended law of 23 July 2016 and treated as specialized investment funds for Luxembourg tax purposes, are income tax exempt entities in Luxembourg, and profits derived from the Shares are thus not subject to Luxembourg income tax.

Non-Luxembourg residents

A non-resident Shareholder, who has neither a permanent establishment nor a permanent representative in Luxembourg to which or whom the Shares are attributable, is not liable to any Luxembourg income tax on income received and capital gains realized upon the sale, disposal or redemption of the Shares.

A non-resident Shareholder who has a permanent establishment or a permanent representative in Luxembourg to which or whom the Shares are attributable, must include any income received, as well as any gain realized on the sale, disposal or redemption of Shares, in its taxable income for Luxembourg tax assessment purposes. The same inclusion applies to individuals, acting in the course of the management of a professional or business undertaking, who have a permanent establishment or a permanent representative in Luxembourg, to which or whom the Shares are attributable. Taxable gains are determined as being the difference between the sale, repurchase or redemption price and the lower of the cost or book value of the Shares sold or redeemed.

Shareholders resident in or citizens of certain countries which have anti-offshore fund legislation may have a current liability to tax on the undistributed income and gains of the Company. The Company and each of the Company's agents shall have no liability in respect of the individual tax affairs of Shareholders.

Net wealth tax

A Luxembourg resident, as well as a non-resident Shareholder who has a permanent establishment or a permanent representative in Luxembourg to which or whom the Shares are attributable, are subject to Luxembourg net wealth tax on such Shares, except if the Shareholder is (i) a resident or non-resident individual taxpayer, (ii) an undertaking for collective investment subject to the 2010 Law, (iii) a securitization company governed by the amended law of 22 March 2004 on securitization, (iv) a company governed by the amended law of 15 June 2004 on venture capital vehicles, (v) a specialized investment fund governed by the amended law of 13 February 2007, (vi) a family wealth management company governed by the amended law of 11 May 2007, (vii) a professional pension institution

governed by the amended law of 13 July 2005 or (viii) a reserved alternative investment fund governed by the amended law of 23 July 2016.

However, (i) a securitization company governed by the amended law of 22 March 2004 on securitization, (ii) a company governed by the amended law of 15 June 2004 on venture capital vehicles, (iii) a professional pension institution governed by the amended law of 13 July 2005 and (iv) an opaque reserved alternative investment fund governed by the amended law of 23 July 2016 and treated as a venture capital vehicle for Luxembourg tax purposes, remain subject to minimum net wealth tax.

Other taxes

Under Luxembourg tax law, where an individual Shareholder is a resident of Luxembourg for tax purposes at the time of his/her death, the Shares are included in his or her taxable basis for inheritance tax purposes. On the contrary, no inheritance tax is levied on the transfer of the Shares upon death of a Shareholder in cases where the deceased was not a resident of Luxembourg for inheritance purposes at the time of his death.

Gift tax may be due on a gift or donation of the Shares, if the gift is recorded in a Luxembourg notarial deed or otherwise registered in Luxembourg.

The information set forth above is based on present law and administrative practice and may be subject to modification.

Belgian Taxation

The Company is subject to an annual tax on Funds registered with the Belgian Financial Services and Markets Authority (the "FSMA"). The annual tax is due on the total net asset value of the Shares held in Belgium as at 31 December of the preceding year. Shares are considered held in Belgium if they are acquired through the intervention of a Belgian financial intermediary, unless to the extent the Belgian financial intermediary provides evidence that the Shares have been placed with non-residents of Belgium. The tax, which is currently payable at a rate of 0.0925% per annum will be paid by the Management Company. Note that a reduced rate of 0.01% can be relied upon for one or more compartments or Share Classes which only institutional or professional investors acting for their own account can invest in and of which the securities can only be acquired by these investors.

Taxation of the Funds as a result from investments in the PRC

By investing in China A-Shares and debt securities, the Funds may be subject to taxes imposed by the PRC.

a. Corporate Income Tax ("CIT")

Under general PRC tax law, if the Fund is considered a PRC tax resident, it will be subject to PRC CIT at 25% on its worldwide taxable income. If the Fund is considered a non-PRC resident but has a permanent establishment ("PE") in the PRC, the profits attributable to that PE would be subject to PRC CIT at 25%. If the Fund is a non-PRC resident without PE in the PRC, the income derived by it from the investment in China A-Shares and debt securities would in general be subject to 10% PRC CIT withholding in the PRC, unless exempt or reduced under specific tax circulars or relevant tax treaties or arrangements.

It is the intention of the Manager that the affairs of the Fund will be conducted in a manner such that it is not a PRC tax resident enterprise and will not have a PE in the PRC for PRC CIT purposes, although this cannot be guaranteed.

i. Shanghai Stock Connect

In respect of trading through Shanghai Stock Connect, the Ministry of Finance ("MoF"), the State Administration of Taxation ("SAT") and the China Securities Regulatory Commission ("CSRC") have jointly promulgated the Circular on Taxation Policy of the Pilot Programme for the Mutual Stock Market Access between Shanghai and Hong Kong Stock Markets (Caishui [2014] No.81) ("Circular 81") dated 31 October 2014 to clarify the relevant income tax liabilities.

Pursuant to Circular 81, taking effect on 17 November 2014, in respect of trading China A-Shares through Shanghai Stock Connect:

- income tax shall be exempt on a temporary basis on the gains earned by Hong Kong market investors (including corporate and individual investors) from the transfer of China A-Shares listed on SSE; and
- Hong Kong market investors are required to pay tax on dividends and bonuses of China A-Shares at a standard rate of 10%, which will be withheld and paid to the relevant PRC tax authority by the respective listed companies (before HKSCC is able to provide details such as investor identities and holding periods to CSDCC, the policy of differentiated rates of taxation based on holding periods will temporarily not be implemented). Where there is an applicable tax treaty/arrangement prescribing for a lower tax rate, the investor may apply for refund from the competent tax authority.

ii. Shenzhen Stock Connect

In respect of equity interest investments such as China A-Shares, MoF, SAT and CSRC have jointly promulgated the *Circular on Taxation Policy of the Pilot Programme for the Mutual Stock Market Access between Shenzhen and Hong Kong Stock Markets (Caishui [2016] No. 127)* ("Circular 127") dated 5 November 2016 to clarify the relevant income tax liabilities. The tax policies provided under Circular 127, which take effect on 5 December 2016, for investments through the Shenzhen Stock Connect generally mirror those applying to investments through the Shanghai Stock Connect. Accordingly, in respect of trading through Shenzhen Stock Connect:

- the income tax shall be exempt on a temporary basis on the gains earned by Hong Kong market investors (including corporate and individual investors) from the transfer of China A-Shares listed on the SZSE; and
- Hong Kong market investors are required to pay tax on dividends and bonuses of China A-Shares at a standard rate of 10%, which will be withheld and paid to the relevant PRC tax authority by the respective listed companies. Where there is any applicable tax treaty/arrangement prescribing for a lower tax rate, the investor may apply for refund from the competent tax authority.

iii. QFI

MoF, SAT and CSRC have jointly promulgated the *Circular on the Issues of Temporary Corporate Income Tax Exemption for Capital Gains Derived from the Transfer of PRC Shares and Equity Interests (Caishui (2014) No. 79)* ("Circular 79"), taking effect on 17 November 2014 in respect of capital gains derived by Qualified Foreign Institutional Investors ("QFIIs") and Renminbi Qualified Foreign Institutional Investors ("RQFIIs") (currently collectively referred to as QFIs). Circular 79 provides that QFIIs and RQFIIs shall be temporarily exempted from PRC CIT on capital gains derived from the trading of A-Shares and other PRC equity interest investments. However, QFIIs / RQFIIs shall be subject to PRC CIT on capital gains realised before 17 November 2014 in accordance with applicable PRC tax laws.

In respect of dividend and interest income, under the *Circular on Issues relating to Withholding Tax of Dividends and Interests Paid by a Resident Enterprise to a Qualified Foreign Institutional Investor (Guoshuihan (2009) No.47)* dated 23 January 2009 (the "**QFII Withholding Tax Notice**"), the PRC tax authorities have confirmed that QFIIs are subject to 10% PRC CIT withholding on dividends and interest derived from China (subject to any reduction or exemption under an applicable domestic tax law or tax treaty/arrangement). While this tax circular addresses only QFIIs, the circular has been interpreted to serve as guidance for the withholding tax treatment applicable to RQFIIs given the similarity in the policies underlying the two regimes and the latest combination of QFII and RQFII regimes into the QFI regime, although this remains subject to future confirmation from the PRC tax authorities.

iv. CIBM and Bond Connect

Under prevailing principles of the CIT Law and relevant regulations, non-PRC tax resident enterprises deriving China source interest and capital gains are subject to PRC CIT withholding of 10%, subject to any reduction or exemption under an applicable CIT Law, regulation (such as the CIT exemption for interest derived from qualified government bonds issued by MoF or local governments) or double taxation treaty/arrangement. Specifically with respect to investment on CIBM, on 22 November 2018, MoF and SAT issued the *Notice of Taxation Policy of the Company Income Tax and Value-Added Tax for Overseas Institutions' Investment in the Domestic Bond Market (Caishui [2018] No.108)* ("Circular 108"), which stipulates that foreign institutional investors are

exempt from PRC CIT in respect of bond interest income received from 7 November 2018 to 6 November 2021 from investments in the China bond markets. Therefore, interests generated from bond investment on CIBM by foreign investors (including investment through various regimes of QFI, direct CIBM access and Bond Connect) will be exempted from PRC CIT.

In terms of capital gains generated from bond investment on CIBM, according to PBOC's Operational Procedures for Overseas Institutional Investors to Enter China's Inter-bank Bond Market (2017) ("PBOC Operational Procedures for Direct CIBM Access") which directly applies to investment through the direct CIBM access, capital gains derived from trading bonds through the direct CIBM access is exempt from PRC CIT temporarily until new tax policies are released. However, the PBOC Operational Procedures for Direct CIBM Access is a guidance issued by PBOC and uncertainty exists since such position has not been formally confirmed by MoF or SAT. In addition, the PRC tax authorities have temporarily suspended tax on capital gains derived by foreign investors from the trading of equity investments through the QFI regime, as well as through the Shanghai and Shenzhen Stock Connects, reflecting a policy to encourage investments in the capital markets. Lacking any formal guidance, it is currently unclear whether similar preferential tax treatment or policies will apply to capital gains derived by foreign investors from bond investments under Bond Connect. In the event China does not provide for a specific tax exemption or sufficient clarification, capital gains realised by non-PRC tax resident investors from the trading of bonds through Bond Connect (although in practice the treatment under direct CIBM access may be referred to) may be treated as income sourced from China and subject to the 10% PRC CIT withholding, subject to any reduction or exemption under an applicable CIT Law, regulation or double taxation treaty/arrangement.

b. Value Added Tax ("VAT")

Circular 81 has provided that, in respect of trading through Shanghai Stock Connect, the business tax ("BT") shall be exempt on a temporary basis on the gains earned by Hong Kong market investors (including corporate and individual investors) from the sale and purchase of China A-Shares listed on the SSE.

With effect from 1 May 2016, all industries formerly subject to BT have transitioned to VAT pursuant to the *Circular on Overall Replacement of Business Tax by Value-Added Tax on A Pilot Basis (Caishui [2016] No. 36)* jointly issued by China's Ministry of Finance and State Administration of Taxation on 24 March 2016 ("Circular 36"). Circular 36 provides for a 6% VAT rate for financial services (including trading financial instruments), replacing the 5% tax rate formerly applied under the BT regime. However, the provisional exemption from BT provided under Circular 81 has been extended to VAT under Circular 36 for trading of securities under Shanghai Stock Connect by Hong Kong market investors (including corporate and individual investors). Likewise, Hong Kong market investors (including corporate and individual investors) trading securities through Shenzhen Stock Connect are provisionally exempt from VAT under Circular 127.

In respect of gains derived by QFIs, the PRC tax authorities had granted QFIIs exemption from BT since 1 December 2005. With the transition of BT to VAT, this exemption was extended to VAT under Circular 36. For RQFIIs, exemption from VAT on trading gains was subsequently confirmed in the *Supplemental Circular on Value Added Tax Policies on Interbank Transactions of Financial Institutions (Caishui (2016) No. 70)*, taking effect on 1 May 2016 ("Circular 70").

Under Circular 36, the general rule is that gains realised from the trading of China marketable securities are subject to VAT at 6% (plus local surcharges). However, under Circular 70, gains of approved foreign investors from the trading of bonds under CIBM are exempt from VAT. In addition, according to the PBOC Operational Procedures for Direct CIBM Access, capital gains obtained therefrom during the period of pilot transition from BT to VAT are exempt from VAT. It is expected that the principles of Circular 70, the PBOC Operational Procedures for Direct CIBM Access and related tax regulations may also extend to exempt VAT on gains from the trading of bonds through Bond Connect. However, this can only be confirmed with certainty when formal tax guidance for Bond Connect is issued. In respect of VAT on interest, as a general rule under Circular 36, China source interest from bonds derived by foreign investors in CIBM is subject to VAT at 6% (plus local surcharges), which has been exempted by Circular 108 until 6 November 2021. Given that the PRC tax authorities are still addressing issues in respect of the transition from BT to VAT on financial transactions, there remain significant uncertainties on the application of the new VAT regulations as well as the enforcement practice of the PRC tax authorities.

c. Stamp duty

Stamp duty under PRC laws generally applies to the execution and receipt of all taxable documents listed in the PRC's Provisional Rules on Stamp Duty.

Under Circular 81 and Circular 127, Hong Kong market investors trading through Shanghai and Shenzhen Stock Connects are required to pay stamp duty arising from the sale and purchase of China A-Shares and the transfer of China A-Shares by way of succession and gift in accordance with the prevailing PRC taxation regulations (currently, 0.1% on the transferor). This PRC stamp duty treatment also applies to trading of eligible securities by QFIs.

2.12 Pooling

For the purpose of effective management, and subject to the provisions of the Articles of Incorporation of the Company and to applicable laws and regulations, the Directors may invest and manage all or any part of the portfolio of assets established for two or more Funds (for the purposes hereof "Participating Funds") on a pooled basis. Any such asset pool shall be formed by transferring to it cash or other assets (subject to such assets being appropriate with respect to the investment policy of the pool concerned) from each of the Participating Funds. Thereafter, the Directors may from time to time make further transfers to each asset pool. Assets may also be transferred back to a Participating Fund up to the amount of the participation of the Class concerned.

The Share of a Participating Fund in an asset pool shall be measured by reference to notional units of equal value in the asset pool. On formation of an asset pool, the Directors shall, in their discretion, determine the initial value of notional units (which shall be expressed in such currency as the Directors consider appropriate) and shall allocate to each Participating Fund units having an aggregate value equal to the amount of cash (or to the value of other assets) contributed. Thereafter, the value of the notional unit shall be determined by dividing the net asset value of the asset pool by the number of notional units subsisting.

When additional cash or assets are contributed to or withdrawn from an asset pool, the allocation of units of the Participating Fund concerned will be increased or reduced, as the case may be, by a number of units determined by dividing the amount of cash or the value of assets contributed or withdrawn by the current value of a unit. Where a contribution is made in cash, it will be treated for the purpose of this calculation as reduced by an amount which the Directors consider appropriate to reflect fiscal charges and dealing and purchase costs which may be incurred in investing the cash concerned; in the case of cash withdrawal, a corresponding addition will be made to reflect costs which may be incurred in realising securities or other assets of the asset pool.

Dividends, interest and other distributions of an income nature received in respect of the assets in an asset pool will be immediately credited to the Participating Funds in proportion to their respective participation in the asset pool at the time of receipt. Upon the dissolution of the Company, the assets in an asset pool will be allocated to the Participating Funds in proportion to their respective participation in the asset pool.

Within any pooling arrangement, the Depositary shall ensure that at all times it is able to identify the assets which are owned by each Participating Fund.

Section 3

3.1 General Information

The Shares

The Shares of any Fund, within a given Class, are freely transferable (with the exception that Shares may not be transferred to a Prohibited Person). Shareholders may convert all or part of their Shares of one Fund into other Class of Shares of that Fund or into the same Class of Shares, or other Classes of Shares of other Funds, provided that the Shareholder meets the eligibility criteria for the Class of Shares into which they are converting, as detailed in Section 2.1. Upon issue, the Shares are entitled to participate equally in the profits and dividends of the Fund attributable to the relevant Class of Shares in which they have been issued as well as in the liquidation proceeds of such Fund.

The Shares do not carry any preferential or pre-emptive rights and each Share, irrespective of the Class to which it belongs or its net asset value, is entitled to one vote at all general meetings of Shareholders. The Shares are issued without par value and must be fully paid.

Luxembourg Stock Exchange Listing

Classes of Shares may be listed on the Luxembourg Stock Exchange as determined by the Management Company from time to time.

ISIN Codes

ISIN Codes for the Company's Share classes are available on www.morganstanleyinvestmentfunds.com.

FundSettle and Clearstream

The following Classes of Shares of the Company's Funds may be traded via FundSettle or via Clearstream:

	Settlement venue		
Share Class Indicator	Fundsettle	Clearstream	
A	Y	Y	
В	Y	N	
С	Y	N	
F	Y	Y	
Ι	Y	Y	
N	N	N	
S	N	N	
Z	Y	Y	

The Company

The Company has been incorporated under the laws of the Grand Duchy of Luxembourg as a "société d'investissement à capital variable" (SICAV) on 21 November 1988 for an unlimited period. The capital may not, at any time, be less than the equivalent in US Dollars of 1,250,000.00 Euro.

The Articles of Incorporation have been deposited with the Luxembourg Trade and Companies' Register and have been published in the *Recueil des Sociétés et Associations* (the "*Mémorial*") of 11 January 1989. The Company has been registered with the Luxembourg Trade and Companies' Register under number B 29192.

The Articles of Incorporation have been lastly amended by an extraordinary shareholders' meeting held on 16 December 2015. The amendments thereto were published in the *Mémorial* on 13 January 2016.

The Directors shall maintain for each Fund a separate pool of assets. As between Shareholders, each pool of assets shall be invested for the exclusive benefit of the relevant Fund.

The assets of the Company shall be segregated on a Fund by Fund basis with third party creditors only having recourse to the assets of the relevant Fund.

The Management Company

Pursuant to a Management Company Services agreement effective as from 1 January 2019 and which was amended and restated on 31 December 2020 (the "Management Company Services Agreement") MSIM Fund Management (Ireland) Limited, of The Observatory, 7-11 Sir John Rogerson's Quay, Dublin 2, Ireland, has been appointed as Management Company of the Company with responsibility for providing collective portfolio management services to the Company and each of the Funds, subject to the overall supervision and control of the Company. For its services, the Management Company receives fees, payable monthly, as set forth herein under the Section 2.5 "Charges and Expenses".

MSIM Fund Management (Ireland) Limited is an indirect wholly owned subsidiary of Morgan Stanley. MSIM Fund Management (Ireland) Limited was incorporated as a company limited by shares under the laws of Ireland on 5 December 2017.

The Management Company Services Agreement has been entered into for an unlimited period of time and may be terminated at any time by either party upon three (3) months' prior written notice or unilaterally with immediate effect by either party if, *inter alia*, the other party commits a material breach that it fails to remedy within thirty (30) days or if required by laws, regulations or any competent regulator or if the other party becomes insolvent or similar circumstances, fraud or bad faith on the part of the Management Company or if the interests of Shareholders so require.

The Management Company may delegate any of its responsibilities to any other party subject to approval by the Company but the Management Company's liability to the Company for the performance of collective portfolio management services shall not be affected by such delegation. In particular, the Management Company has delegated the investment management function for some of the Funds and has fully delegated the central administration and transfer agency functions as set out in the sections below. Subject to the compliance with applicable laws, the Management Company may, at its own expenses and without affecting its liability to the Company, select and rely upon affiliated Morgan Stanley group entities for investment advice, research and associated expertise of such selected entities with respect to the assets of a Fund provided that, unless each such entity is expressly disclosed in this Prospectus subject to the prior approval by the CSSF, no such entity shall have the power to undertake discretionary portfolio management on behalf of a Fund and, subject to any authorised delegation, the Management Company shall at all times ensure that it remains in charge of the effective portfolio management of the Fund.

The Management Company has a remuneration policy in place which seeks to ensure that the interests of the Company and the Shareholders are aligned. Such remuneration policy imposes remuneration rules on staff and senior management within the Management Company whose activities have an impact on the risk profile of the Company. The Management Company shall seek to ensure that such remuneration policies and practices will be consistent with sound and effective risk management and with UCITS Regulation. The Management Company shall also seek to ensure that such remuneration policies and practices shall not encourage risk taking which is inconsistent with the risk profile and constitutional documents of the Company.

The Management Company shall seek to ensure that the remuneration policy will, at all times, be consistent with the business strategy, objectives, values and interests of the Company and the Shareholders and that the remuneration policy will include measures that seek to ensure that all relevant conflicts of interest can be managed appropriately at all times.

In particular, the remuneration policy also complies with the following principles in a way and to the extent that is appropriate to the size, internal organisation and the nature, scope and complexity of the activities of the Management Company:

- (i) the assessment of performance is set in a multi-year framework appropriate to the holding period recommended to the Shareholders of the Company in order to ensure that the assessment process is based on the longer-term performance of the Company and its investment risks and that the actual payment of performance-based components of remuneration is spread over the same period; and
- (ii) fixed and variable components of total remuneration are appropriately balanced and the fixed component represents a sufficiently high proportion of the total remuneration to allow the operation of a fully flexible policy on variable remuneration components, including the possibility to pay no variable remuneration component.

Details regarding the remuneration to the Management Company and the Management Company's up-to-date remuneration policy, including, but not limited to, a description of how remuneration and benefits are calculated, the identity of persons responsible for awarding the remuneration and benefits, including the composition of the remuneration committee where such a committee exists, may be obtained free of charge during normal office hours at the registered office of the Company and is available on the following website http://www.morganstanleyinvestmentfunds.com.

The Directors of the Company

The Directors of the Company are responsible for the overall control and supervision of the performance of the tasks performed by the Management Company.

Directors of the Company who are not executive directors or employees of the Investment Advisers or any affiliate will be entitled to receive remuneration from the Company as disclosed in the Annual Report. Other than this, the Company does not pay remuneration to any individual. The Company does not pay Directors variable remuneration.

Each of the Directors of the Company has also been appointed to serve as director on the board of directors of one or more other collective investment schemes or management companies managed or operated by the Investment Adviser(s) or an affiliate.

The Investment Advisers

The Directors of the Company are responsible for determining the investment policy of the different Funds.

In determining the investment policies of the Funds, the Directors of the Company will be assisted by the Management Company or one or several investment advisers (the "Investment Adviser(s)") for the day to day responsibility of providing discretionary investment management and investment advisory services.

Under an Investment Advisory Agreement, Morgan Stanley Investment Management Limited, of 25 Cabot Square, Canary Wharf, London E14 4QA, United Kingdom, has been appointed as Investment Adviser with responsibility for providing discretionary asset management and investment advisory services to the Management Company, such services to be provided in relation to some of the Funds. The list of the Funds in relation to which advice is provided by the Investment Adviser and its delegate(s), if any, is available from www.morganstanleyinvestmentfunds.com and is included in the Company's Annual and Semi-Annual Reports. For its services, the Investment Adviser receives an annual fee, payable monthly, the details of which are set forth herein under the Section 2.5 "Charges and Expenses".

Morgan Stanley Investment Management Limited is a 100% indirect subsidiary of Morgan Stanley. Morgan Stanley Investment Management Limited was incorporated in 1986 under the laws of the United Kingdom. Its board of directors is currently composed of Terri Duhon, Fiona Kelly, Richard Lockwood, Zoë Parish, and Ruairi O'Healai.

The agreements between the Management Company and the Investment Advisers provide that they are to remain in force for an unlimited period and may be terminated at any time by either party to the agreement upon three months' prior written notice or immediately by either party if the Management Company Services Agreement is terminated.

The Investment Advisers have been appointed to provide discretionary investment management and investment advisory services to the Management Company and, subject to the Company's overall control and supervision, to provide advice in connection with the day-to-day management in respect of the relevant Funds.

Subject to an express delegation given by the Management Company, the Investment Advisers, pursuant to the agreement mentioned above, may furthermore have discretion, on a day-to-day basis and subject to the overall control of the Management Company, to purchase and sell securities, as appropriate, and otherwise to manage the portfolios of the different Funds for the account of the Company in relation to specific transactions.

During the term of any such specific delegation, the Investment Adviser will be authorised to act on behalf of the Management Company and to select agents, brokers and dealers through whom to execute transactions and provide the Management Company with reports in relation to such termination as the Management Company may require.

The Investment Adviser may delegate any of its responsibilities to any other party subject to approval by the Management Company and the CSSF but the Investment Adviser shall remain responsible for the proper performance by such party of those responsibilities.

Subject to the compliance with applicable laws and the prior consent of the Management Company, the Investment Adviser may, at its own expense and without affecting its liability to the Company, select and rely upon affiliated Morgan Stanley group entities and any of the Sub-Advisers for non-discretionary investment advice, research and associated expertise of such selected entities with respect to the assets of a Fund provided that, unless each such entity is expressly disclosed in this Prospectus subject to the prior approval by the CSSF, no such entity shall have the power to undertake discretionary portfolio management on behalf of a Fund and, subject to any authorised delegation, the Investment Adviser shall at all times ensure that it remains in charge of the effective portfolio management of the Fund.

A list of the Funds in relation to which discretionary and non-discretionary investment advice is provided by the Investment Adviser and its delegate(s), if any, is available from www.morganstanleyinvestmentfunds.com and is included in the Company's Annual and Semi-Annual Reports.

The Sub-Advisers

Pursuant to Sub-Advisory Agreements, Morgan Stanley Investment Management Inc, a company registered in the United States of America, has been appointed as Sub-Adviser of the Investment Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Morgan Stanley Investment Management Company in Singapore has been appointed as Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Morgan Stanley Asia Limited in Hong Kong has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Morgan Stanley Saudi Arabia in Saudi Arabia has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Morgan Stanley Investment Management (Australia) Pty Limited in Australia has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Calvert Research and Management in the United States of America has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Eaton Vance Advisers International Ltd. in the United Kingdom has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Eaton Vance Management in the United States of America has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Sub-Advisory Agreement, Parametric Portfolio Associates, LLC in the United States of America has been appointed by the Investment Adviser as a Sub-Adviser in relation to such Funds as agreed from time to time.

Pursuant to a Delegation Agreement, Mitsubishi UFJ Asset Management (UK) Limited in the United Kingdom has been appointed as Sub-Adviser in relation to Japanese Equity Fund and Japanese Small Cap Equity Fund.

Pursuant to the terms of the relevant agreement mentioned above, the Sub-Advisers may have discretion, on a day-to-day basis and subject to the overall control of the Investment Adviser to purchase and sell securities as agent for the Investment Adviser and otherwise to manage the portfolios of the relevant Funds for the account of the Company in relation to specific transactions.

The Sub-Advisers may delegate any of their responsibilities to delegates when expressly authorized by the relevant agreement, subject to the Adviser and CSSF's approval. In such case, the Investment Adviser shall remain responsible for the proper performance by such party of those responsibilities. In particular, Mitsubishi UFJ Kokusai Asset Management Co. Ltd. has been appointed by Mitsubishi UFJ Asset Management (UK) Ltd. pursuant to the Master Investment Management Delegation Agreement to provide discretionary investment management and/or investment advisory services to the Japanese Equity Fund and the Japanese Small Cap Equity Fund. The delegation of the discretionary investment management services by Mitsubishi UFJ Asset Management (UK) Ltd. to Mitsubishi UFJ Kokusai Asset Management Co., Ltd. with effect as from 1 July 2014.

Subject to the compliance with applicable laws and the prior consent of the Management Company and/or the Investment Adviser, each Sub- Adviser may, at its own expense and without affecting its liability to the Company and/or the Investment Adviser, select and rely upon affiliated Morgan Stanley group entities and any of the Sub-Advisers for non-discretionary investment advice, research and associated expertise of such selected entities with respect to the assets of a Fund provided that, unless each such entity is expressly disclosed in this Prospectus subject to the prior approval by the CSSF, no such entity shall have the power to undertake discretionary portfolio management on behalf of a Fund and, subject to any authorised delegation, the relevant Sub-Adviser shall at all times ensure that it remains in charge of the effective portfolio management of the assets of the relevant Fund entrusted to it for management.

The fees of each Sub-Adviser will be paid by the Investment Adviser.

A list of the Funds in relation to which discretionary and non-discretionary investment advice is provided by the Sub-Advisers and their delegate(s), if any, is available from www.morganstanleyinvestmentfunds.com and is included in the Company's Annual and Semi-Annual Reports.

Payment for investment research

Any third party research received in connection with investment management and investment advisory services that the Management Company or the Investment Adviser provides to the Funds (other than research that qualifies as a minor non-monetary benefit) will be paid for by the Management Company or the Investment Adviser out of its own resources.

Where possible, the Sub-Advisers will also pay for any third party equity research received by them in connection with the sub-investment advisory services that they provide to the relevant Fund(s) (other than research that qualifies as a minor non-monetary benefit) out of their own resources. However, to the extent the Sub-Advisers receive third party research other than equity research, or where research is received in a market that does not support the separation of execution commissions from research payments, they will implement systems and controls to ensure that the receipt of such research does not influence order routing and best execution decisions or give rise to conflicts of interest that risk detriment to the relevant Fund(s) and their investors.

The Depositary

Pursuant to a depositary agreement (the "**Depositary Agreement**"), J.P. Morgan SE, acting through its Luxembourg Branch has been appointed as the depositary (the "**Depositary**") to provide depositary, custodial, settlement and certain other associated services to the Company. For its services, the Depositary receives an annual fee, payable monthly as set forth herein under the Section 2.5 "Charges and Expenses".

The Depositary shall assume its functions and responsibilities in accordance with the UCITS Regulation as further described in the Depositary Agreement. In particular, the Depositary will be responsible for the safekeeping and ownership verification of the assets of the Company, cash flow monitoring and oversight in accordance with the UCITS Regulation.

Collateral posted in favour of a Fund under a title transfer arrangement should be held by the Depositary or one of its correspondents or sub-custodians. Collateral posted in favour of a Fund under a security interest arrangement (e.g., a pledge) can be held by a third party custodian which is subject to prudential supervision, and which is unrelated to the provider of the collateral.

In carrying out its role as depositary, the Depositary shall act independently from the Company and the Management Company and solely in the interest of the Company and its investors.

The Depositary will further, in accordance with the UCITS Regulation:

- a) ensure that the sale, issue, repurchase, redemption and cancellation of Shares effected by the Company or on its behalf are carried out in accordance with the applicable law and the Articles of Incorporation;
- b) ensure that the value per Share is calculated in accordance with the applicable law and the Articles of Incorporation;
- c) carry out, or where applicable, cause any sub-custodian or other custodial delegate to carry out the Instructions of the Company or the Management Company unless they conflict with the applicable law or the Articles of Incorporation;
- d) ensure that in transactions involving the assets of the Company, the consideration is remitted to it within the usual time limits; and
- e) ensure that the income of the Company is applied in accordance with the Articles of Incorporation and the applicable law.

The Depositary is liable to the Company and/or its Shareholders or its investors for the loss of any financial instrument held in custody by the Depositary or any of its sub-custodian or other custodial delegate. The Depositary shall however, not be liable if it can prove that the loss has arisen as a result of an external event beyond its reasonable control, the consequences of which would have been unavoidable despite all reasonable efforts to the contrary. The Depositary is also liable to the Company for all other losses suffered by them as a result of the Depositary's failure to exercise reasonable care and diligence or properly fulfil its duties in accordance with the UCITS Regulation.

The Depositary may entrust all or part of the assets of the Company that it holds in custody to such third party as may be determined by the Depositary from time to time ("sub-custodian"). Except as provided in the UCITS Regulation, the Depositary's liability shall not be affected by the fact that it has entrusted all or part of the assets in its care to a sub-custodian (see also above). Subject to the terms of the Depositary Agreement, entrusting the custody of assets to the operator of a securities settlement system is not considered to be a delegation of custody functions.

When selecting and appointing a sub-custodian or other delegate, the Depositary shall exercise all due skill, care and diligence as required by the UCITS Regulation to ensure that it entrusts the Company's assets only to a delegate who may provide an adequate standard of protection.

Where the law of a third country requires that certain financial instruments be held in custody by a local entity and there are no local entities that satisfy the delegation requirement (i.e. the effective prudential Regulation) under the 2010 Law, the Depositary may, but shall be under no obligation to, delegate to a local entity to the extent required by the law of such jurisdiction and as long as no other local entity meeting such requirements exists, provided however that (i) the investors, prior to their investment in the Company, have been duly informed of the fact that such a delegation is required, of the circumstances justifying the delegation and of the risks involved in such a delegation and (ii) instructions to delegate to the relevant local entity have been given by or for the Company.

The current list of sub-custodians used by the Depositary and sub-delegates that may arise from any delegation is attached in Appendix J, and the latest version of such list may be obtained by investors from the Company upon request.

The Depositary Agreement may be terminated by any party on 90 days' notice in writing. Subject to the UCITS Regulation, the Depositary Agreement may also be terminated by the Depositary on 30 days' notice in writing under the conditions and circumstances set forth in the Depositary Agreement.

The Depositary is not allowed to carry out activities with regard to the Company that may create conflicts of interest between the Company, the Shareholders and the Depositary itself, unless the Depositary has properly identified any such potential conflicts of interest, has functionally and hierarchically separated the performance of its depositary tasks from its other potentially conflicting tasks, and the potential conflicts of interest are properly identified, managed, monitored and disclosed to the Shareholders. Please refer to the section 2.9 Conflicts of Interests for further information on any such conflicts of interest.

Up-to-date information on the identity of the Depositary, a description of its duties and of conflicts of interests that may arise, a description of any delegated functions and any thereto related conflicts of interest may be obtained by the investors at the registered office of the Company upon request.

The Administrator and Paying Agent

Under an administration agreement, J.P. Morgan SE, acting through its Luxembourg Branch has been appointed as Administrator of the Company and the Funds, to administer the computation of the Net Asset Value per Share of the different Funds, and to perform other general administrative functions. For its services, the Administrator receives an annual fee, payable monthly, which is part of the Administration Charge as set forth herein under the Section 2.5 "Charges and Expenses".

As part of the implementation of the J.P. Morgan legal entity strategy in Europe, J.P. Morgan Bank Luxembourg S.A., the previous depositary, administrator and paying agent of the Company and the Funds, merged into J.P. Morgan AG on 22 January 2022 which at the same time changed its legal form from a German Stock Corporation (Aktiengesellschaft) to a European Company ("Societas Europaea"), being J.P. Morgan SE.

As from 22 January 2022, J.P. Morgan SE is a European Company ("Societas Europaea") organized under the laws of Germany and registered with the commercial register of the local court of Frankfurt; it is a credit institution subject to direct prudential supervision by the European Central Bank, the German Federal Financial Supervisory Authority (Bundesanstalt für Finanzdienstleistungsaufsicht, BaFin) and Deutsche Bundesbank, the German Central Bank. J.P. Morgan SE, Luxembourg Branch is authorized by the CSSF to act as depositary and fund administrator.

The Domiciliary Agent

Pursuant to a domiciliary agent agreement, the Company has appointed MSIM Fund Management (Ireland) Limited, Luxembourg branch as its Domiciliary Agent to provide the Company's registered address, to store its corporate documents and to perform other related administrative functions.

The Registrar and Transfer Agent

Pursuant to a registrar and transfer agency agreement, RBC Investor Services Bank S.A. in Luxembourg has been appointed as Registrar and Transfer Agent to administer the issue, conversion and redemption of Shares, the maintenance of records and other related administrative functions.

Data Protection

A detailed data protection notice is included as Appendix I to this prospectus. Shareholders and prospective Shareholders should read the information contained in Appendix I to understand how the Company, the Management Company, their affiliates and anyone acting on their behalf will process personal data.

Dissolution

The Company has been established for an unlimited period of time. However, the Company may be dissolved and liquidated at any time by a resolution of the general meeting of Shareholders.

In the event of dissolution, the liquidator(s) appointed by the Shareholders of the Company in accordance with the Supervisory Authority will realise the assets of the Company in the best interests of the Shareholders, and the Depositary, upon instruction given by the liquidator(s), will distribute the net proceeds of liquidation (after deducting all liquidation expenses) among the Shareholders of each Class of Shares in proportion to their respective rights. As provided for by Luxembourg law, at the close of liquidation, the proceeds of liquidation corresponding to Shares not surrendered for repayment will be kept in safe custody at the "Caisse de Consignation" until the statute of limitation has lapsed. If an event requiring liquidation arises, issue, redemption, exchange or conversion of the Shares is void.

In the event that for any reason the value of the total net assets in any Fund or the value of the net assets of any Class of Shares within a Fund has decreased to, or has not reached, an amount determined by the Directors to be the minimum level for such Fund to be operated in an economically efficient manner, which shall be not less than the amount provided for under "Compulsory Redemption" heretofore, or in case of substantial modification in the political, economic or monetary situation or as a matter of economic rationalization, the Directors may decide to compulsorily redeem all the Shares of the relevant Classes issued in such Fund at the Net Asset Value per Share (taking into account actual realisation prices of investments and realisation expenses as well as the liquidation costs, as the case may be), calculated at the valuation point at which such decision shall take effect. The Company shall serve a notice to the holders of the relevant Classes of Shares in writing prior to the effective date for the compulsory redemption, which will indicate the reasons for, and the procedure of, the redemption operations.

In addition, any feeder fund shall be liquidated and their Shares compulsory redeemed pursuant to the procedure set forth above in the event of the liquidation, division or merger of the relevant master fund, except to the extent permitted, and in compliance with the conditions set out under the 2010 Law and the CSSF Regulation 10-05.

In addition, the general meeting of Shareholders of the Classes of Shares issued in any Fund may, upon proposal from the Directors, redeem all the Shares of the relevant Classes issued in such Fund and refund to the Shareholders the net asset value of their Shares (taking into account actual realisation prices of investments and realisation expenses as well as the liquidation costs, as the case may be) calculated at the valuation point at which such decision shall take effect.

There shall be no quorum requirements for such general meeting of Shareholders which shall decide by resolution taken by simple majority of those present or represented.

Assets which may not be distributed to their beneficiaries upon the implementation of the redemption will be deposited with the "Caisse de Consignation" on behalf of the persons entitled thereto. The assets may be refunded upon duly justified request made to the "Caisse de Consignation". In case the assets have not been refunded within a period of thirty years as from the date of the consignment, they are forfeited by the Luxembourg State. Not later than six months before the expiry of this period of thirty years, the "Caisse de Consignation" shall notify by registered letter the beneficiaries whose domicile is known according to the documents in its possession of the forfeiture they incur. In the absence of a known domicile or in the absence of a claim of the rightful beneficiaries within two months of the sending of the aforementioned registered letter, the indications which allow the beneficiary to appear are published immediately in the Mémorial. The "Caisse de Consignation" is entitled to retain an annual fee at a rate between 0.5% and 3% of the estimated value of the assets (i.e. this annual fee has been set to 1% since a Grand Ducal Regulation dated 4 February 2000).

All redeemed Shares shall be cancelled.

Amalgamation

In the event that for any reason the value of assets in any Fund has decreased to, or has not reached, an amount determined by the Directors to be the minimum level for such Fund to be operated in an economically efficient manner or in case of a substantial modification in the political, economic or monetary situation, or as a matter of economic rationalisation the Directors may decide to proceed with a merger (within the meaning of the 2010 Law) of the assets of the Company or of any Fund with those of (i) another existing Fund within the Company or another Fund within such other Luxembourg or foreign UCITS (the "New Fund"), or of (ii) another Luxembourg or foreign UCITS (the "New UCITS") and to designate the Shares of the Company or the Fund concerned, as Shares of the New UCITS or of the New Fund, as applicable. The Directors are competent to decide on, or approve, the effective date of the merger. Such a merger shall be subject to the conditions and procedures imposed by the 2010 Law, in particular concerning the merger project to be established by the boards of directors and the information to be provided to the Shareholders.

The Directors may decide to proceed with the absorption by the Company or one or several Funds of (i) one or several sub-funds of another Luxembourg or a foreign UCI, irrespective of their form, or (ii) any Luxembourg or

foreign UCI constituted under a non-corporate form. The exchange ratio between the relevant shares of the Company and the shares or units of the absorbed UCI or of the relevant sub-fund thereof will be calculated on the basis of the relevant net asset value per share or unit as of the effective date of the absorption. Such a merger does not require the prior consent of the Shareholders except where the Company ceases to exist as a result of the merger; in such case, the general meeting of shareholders of the Company must decide on the merger and its effective date. Such general meeting will decide by resolution taken subject to a quorum and majority requirements referred to in Article 30 of the Articles of Incorporation.

Notwithstanding the powers conferred on the Directors by the preceding paragraphs, the Shareholders of the Company or any Fund may also decide on any of the mergers or absorptions described above and on their effective date thereof. The convening notice to the general meeting of shareholders will indicate the reasons for and the process of the proposed merger or absorption. Such decision can be adopted by a simple majority of the votes validly cast with no quorum requirement.

In addition to the above, the Company may also absorb another Luxembourg or foreign UCI incorporated under a corporate form in compliance with the Law of 10 August 1915 on commercial companies, as amended and any other applicable laws and regulations.

In the event that for any reason the value of the net assets of any Class of Shares within a Fund has decreased to, or has not reached, an amount determined by the Directors to be the minimum level for such Class of Shares, to be operated in an economically efficient manner in case of a substantial modification in the political, economic or monetary situation or as a matter of economic rationalization, the Directors may decide to amend the rights attached to any Class of Shares so as to include them in any other existing Class of Shares and re-designate the Shares of the Class or Classes concerned as Shares of another Class. Such decision will be subject to the right of the relevant Shareholders to request, without any charges, the redemption of their shares or, where possible, the conversion of those Shares into Shares of other Classes within the same Fund or into Shares of same or other Classes within another Fund in compliance with Section 2.4 "Conversion of Shares" above.

The assets which may not or are unable to be distributed to such Shareholders for whatever reasons will be deposited with the "Caisse de Consignation" on behalf of the persons entitled thereto.

General Meetings

The Annual General Meeting of Shareholders of the Company is held at the registered office of the Company on the second Tuesday of the month of May at 10:30 a.m.

Shareholders of any Class or Fund may hold, at any time, general meetings to decide on any matters which relate exclusively to such Fund or to such Class.

Notices of all general meetings are sent by mail to all registered Shareholders at their registered address at least eight days prior to the meeting. Such notice will indicate the time and place of the meeting, the conditions of admission thereto, will contain the agenda and refer to the requirements of Luxembourg law with regard to the necessary quorum and majorities at the meeting. To the extent required by law, further notices will be published in the *Mémorial* and in one Luxembourg newspaper.

Annual and Semi-Annual Reports

Audited reports to the Shareholders in respect of the preceding financial year of the Company, and the consolidated accounts of the Company, are made available at the registered office of the Company, of the Registrar and Transfer Agent and of the Management Company and shall be available at least eight days before the Annual General Meeting. In addition, unaudited semi-annual consolidated reports are also made available at such registered offices within two months after 30 June. The annual and semi-annual reports are also available on the Company's website (www.morganstanleyinvestmentfunds.com). The Company's financial year ends on 31 December. The Company's reference currency is USD.

The notice for the annual general meeting will contain an offer to provide to Shareholders upon request and free of charge a copy of the complete version of such annual and semi-annual reports.

In compliance with any applicable laws, Shareholders and third parties may, on request, receive additional information in relation to securities held by the Funds.

Documents Available for Inspection

Copies of the following documents may be inspected free of charge during usual business hours on any week day (Saturday and public holidays excepted) at the registered office of the Company: European Bank and Business Centre, 6B route de Trèves, L-2633 Senningerberg, Grand Duchy of Luxembourg.

- a) the Articles of Incorporation of the Company;
- b) the Articles of Incorporation of the Investment Adviser(s);
- c) the material contracts referred to above; and
- d) the financial reports of the Company.

The documents under (a) and (b) may be delivered to interested investors at their request.

Additional Information

Additional information such as details of Complaints Handling, Proxy Voting, Best Execution and Conflicts of Interest policies are available from MSIM Fund Management (Ireland) Limited, Luxembourg Branch, European Bank and Business Centre, 6B route de Tréves, L-2633 Senningerberg, Luxembourg. E-mail: cslux@morganstanley.com

z man <u>estan e morganistame j te s</u>

Probate

Upon the death of a Shareholder, the Directors reserve the right to require the provision of appropriate legal documentation to evidence the rights of the Shareholder's legal successor.

Morgan Stanley Investment Funds

Société d'Investissement à Capital Variable R.C.S. Luxembourg B 29192.

Directors of the Company

Henry Kelly, independent non-executive director and Chairman, Luxembourg Carine Feipel, independent non-executive director, Luxembourg Diane Hosie, non-executive director, UK William Jones, independent non-executive director, Luxembourg Andrew Mack, non-executive director, UK Zoë Parish, executive director, Morgan Stanley Investment Management, UK Susanne Van Dootingh, independent non-executive director, Belgium Arthur Lev, non-executive director, United States of America

Henry Kelly (Luxembourg) is a non-executive director and Chairman of the Company. Mr. Kelly is an independent board member of several investment funds and investment management companies domiciled in Luxembourg and internationally. He is the Managing Director of the Luxembourg-based consultancy firm that he founded in 1999, KellyConsult S.à r.l., which provides advisory services to the investment fund sector. He is a member and was the founding Chairman of the Fund Governance Forum for Association of the Luxembourg Fund Industry ("ALFI"). He is a member of the Management Committee of the *Institut Luxembourgeois des Administrateurs* (Luxembourg Institute of Directors) ("ILA") and is a founding member of the ILA Investment Funds Committee. He is a regular speaker at conferences on investment funds and corporate governance matters.

From 1993 – 1999 he was a Managing Director of Flemings Luxembourg (now JP Morgan Asset Management) following 5 years' experience in the capital markets division of BNP Paribas based in Paris. Prior to these posts, he acquired seven years' experience with Price Waterhouse in Paris, Frankfurt and New York. He has a master's degree in Modern Languages from Cambridge University, is a Member of the Institute of Chartered Accountants and holds the INSEAD Certificate in Corporate Governance. He obtained the qualification of ILA certified director in 2013.

Carine Feipel (Luxembourg) is a non-executive director of the Company. Ms. Feipel is a Luxembourg attorney and certified independent director. She is the current Chair of ILA.

Ms. Feipel obtained the INSEAD IDP Certificate on Corporate Governance in 2014 and was admitted by ILA as a Certified Director in the same year. Ms. Feipel serves the Board of Directors of a Luxembourg based bank, four insurance companies, both active in the life insurance and the non-life insurance fields. Ms. Feipel is further a member of the Board of Directors of various Luxembourg companies active in the financial and investment funds sectors. In 2014, Ms. Feipel was elected to the Board of Directors of ILA and became a member of the Management Committee of such association. She was appointed as Chair of such professional association in June 2019.

Since January 2014, Ms. Feipel practices as an independent lawyer, notably advising insurance and reinsurance companies as well as other financial institutions. Ms. Feipel has 20 years of experience as a lawyer with the Luxembourg law firm Arendt & Medernach where she headed the insurance law practice and was co-head of the employment law practice. Ms. Feipel was also a member of the Board of such firm and headed its New York office from 2007 to 2012. Since 2010, Ms. Feipel has been a member of various committees within the *Haut Comité pour la Place Financière*, a Government lead think tank coordinating promotion and innovation initiatives for the Luxembourg financial sector.

Diane Hosie (United Kingdom) is a non-executive director of the Company and formally International Head of Morgan Stanley Investment Managements client services and distribution oversight teams. Ms Hosie joined Morgan Stanley Investment Management in 1997 as a Senior Associate within Investment Management Operations and was named Managing Director in 2014. She has over 30 years' experience in Asset Management. Prior to Morgan Stanley Investment Management Ms Hosie spend nine years with Nomura Capital Management.

Ms. Hosie is an Executive Director on a number of Luxembourg fund boards including Morgan Stanley Investment Funds, Morgan Stanley Liquidity Funds Board, and Morgan Stanley Asset Management S.A. She is also a Non-Executive Director of MSIM Fund Management (Ireland) Ltd.

William Jones (Luxembourg) is a non-executive director of the Company. Mr. Jones is the founder of Beaumont Governance which provides directorship and corporate governance services from Luxembourg. Mr. Jones has over 32 years' experience in the funds industry and has held senior positions with Commodities Corporation, Goldman Sachs Asset Management International and Bank of Bermuda/ HSBC. Mr. Jones acts as director of investment funds and related entities sponsored by major US and international banks and asset managers. He also acts as advisor and director of non-financial start-up and early stage companies. He completed the first cohort of the INSEAD International Directors Program ("IIDP") in 2011, was a member of the IIDP Advisory Board (2011-2017) and served as its first President (2011-2015). Mr. Jones was accredited in 2011 as a certified director by IIDP and the ILA. He served on the Board of Directors of ILA (2016-2020) and its Fund Governance Committee (2010-2020) and cochaired the Alternative Investment Task Force of ILA's Fund Governance Committee which issued the "ILA Guide for Board Members in the Context of AIF and AIFM" in July 2014. He sits on multiple committees of ALFI. He was co-vice chair of the Alternative Investment Management Association's ("AIMA") working group which issued "AIMA's Fund Director Guide" in April 2015. He served on the Board of Governors of the International School of Luxembourg from 2011 to 2015 and was its Chairman during 2014-15. Mr. Jones has law and business degrees (1988) from Columbia University and a bachelor degree (1984) from Princeton University.

Andrew Mack (United Kingdom) is a non-executive director of the Company and formerly CEO of Morgan Stanley Investment Management EMEA. Mr. Mack joined Morgan Stanley in 1996 and has over 30 years of investment experience. Mr. Mack joined Morgan Stanley as a portfolio manager in the asset management business, where he launched and co-managed a global equity arbitrage fund. He subsequently headed the global market risk oversight team for all of Morgan Stanley before taking over as market risk manager of the European equities business. Mr. Mack was appointed as European head of multi asset class prime brokerage in 2004 and took over prime brokerage sales for Europe in 2006 before running the European listed derivative business. Mr. Mack rejoined Morgan Stanley Investment Management (MSIM) in 2008 as global chief risk officer, taking over as head of MSIM EMEA at the beginning of 2009. Mr. Mack was an employee of Morgan Stanley until 30 June 2010. Mr. Mack provided services as a consultant and senior adviser to Morgan Stanley between 1 July 2010 and 31 December 2013. Mr. Mack assumed a non-Executive Director role of Morgan Stanley Investment Management (ACD) Limited which was the Management Company from 1st April 2014 until 1st January 2019. Mr. Mack has been involved in taking and managing risk for most of his investment career, his previous experience includes portfolio management, trading and risk management positions at Cargill, Bankers Trust and Black River Asset Management, a Minneapolis based hedge fund.

Zoë Parish (United Kingdom) is a Managing Director of Morgan Stanley and the Head of International Product and Board Governance for Morgan Stanley Investment Management. Before returning to Morgan Stanley in 2019, Zoë spent 4 years at Coutts and Co and was most recently Head of Europe and Americas. Prior to that she was Head of Delivery responsible for developing and implementation of strategies for Coutts products for England and Wales. Prior to this she was business lead for several strategic projects for the firm including a web based investment platform. From 1993 to 2014, Zoë held leadership and product roles in Morgan Stanley's Private Wealth Management business in EMEA. In her latest role with the firm as Managing Director, she was Head of Product and also headed up the Alternatives platform, the Private Investment Club and Strategic Lead Management for EMEA. Zoë received a B. A. (Hons) and an LLB (Hons) both from the University of London. She also sits on the Board of Morgan Stanley Asset Management S.A and is the Chairman of the Board of Trustees for the University of London Boat Club.

Susanne van Dootingh (Belgium) is a non-executive director of the Company. Ms. van Dootingh is an independent non-executive director of various Luxembourg domiciled investment funds and management companies. Prior to becoming an independent non-executive director in 2017, Ms. van Dootingh worked at State Street Global Advisors from 2002 to 2017 with her final position being Senior Managing Director, Head of European Governance and Regulatory Strategy, EMEA. During her tenure at SSGA she held various other senior positions, amongst others in Global Product Development and Fixed Income Strategy, and was the chairwoman of the SSGA flagship Luxembourg Sicav and Management Company and various other boards for a decade. Before joining SSGA in 2002 Ms. van Dootingh gained experience in global fixed income portfolio management and strategy at Fortis Investment Management, Barclays Global Investors and ABN AMRO Asset Management. She graduated from the Vrije Universiteit Amsterdam with a Master's degree in Financial Sector Management.

Arthur Lev (United States of America) is a non-executive director of the Company. Arthur Lev is the former Chairman of Morgan Stanley Investment Management (MSIM), where he spent nearly twenty years in a variety of

business, risk and legal roles, most recently managing MSIM's Long Only and Alternatives businesses. Arthur also spent four years at FrontPoint Partners, a hedge fund firm that was acquired by Morgan Stanley. Arthur is currently an independent board member (and chair of the Audit and Risk Committee) of Russell Investments, a global money manager based in Seattle, WA, and Next Capital, a digital investment adviser and financial technology firm based in Chicago, IL. Arthur was previously an independent director at Techstars, a startup accelerator platform, based in Boulder, CO. Arthur is also an adviser to fintech startups focused on customized portfolio solutions and cryptocurrency trading. Arthur is a graduate of the University of Southern California (AB, 1983) and Harvard Law School (JD, 1986).

Directors of the Management Company

Eimear Cowhey, independent non-executive director, Ireland

Michael Hodson, independent non-executive director, Ireland.

Diane Hosie, non-executive director, United Kingdom

Elaine Keenan, Managing Director, Morgan Stanley Investment Management, CEO and COO of MSIM Fund Management (Ireland) Limited

Liam Miley, independent non-executive director and Chairman, Ireland

Ruairi O'Healai, Managing Director, Morgan Stanley Investment Management, COO Morgan Stanley Investment Management EMEA

Management Company

MSIM Fund Management (Ireland) Limited The Observatory, 7-11 Sir John Rogerson's Quay Dublin 2 D02VC42 Ireland

Investment Adviser

Morgan Stanley Investment Management Limited
25 Cabot Square
Canary Wharf
London E14 4QA
United Kingdom

Sub-Advisers

Morgan Stanley Investment Management Inc.
522 Fifth Avenue
New York
NY 10036
United States of America

Morgan Stanley Investment Management Company 23 Church Street 16-01 Capital Square Singapore, 049481

Mitsubishi UFJ Asset Management (UK) Ltd. 24 Lombard Street London EC3V 9AJ United Kingdom Morgan Stanley Asia Limited Level 46, International Commerce Centre 1 Austin Road West Kowloon Hong Kong

Morgan Stanley Saudi Arabia Al Rashid Tower, Floor 10 Al Ma'ather Street PO Box 66633 Riyadh 11586 Saudi Arabia

Morgan Stanley Investment Management (Australia) Pty
Limited
Level 61, Governor Phillip Tower
1 Farrer Place
Sydney NSW 2000
Australia

Calvert Research and Management 1825 Connecticut Avenue NW, Suite 400 Washington DC 20009 United States of America

Eaton Vance Management
2 International Place, Suite 1400
Boston
MA 02110
United States of America

Eaton Vance Advisers International Ltd.
125 Old Broad Street
London EC2N 1AR
United Kingdom

Parametric Portfolio Associates, LLC 800 Fifth Avenue, Suite 2800 Seattle WA 98104 United States of America

Delegated Adviser

Mitsubishi UFJ Kokusai Asset Management Co., Ltd. 1-12-1 Yurakucho, Chiyoda-ku Tokyo 100-0006 Japan

Administrator and Paying Agent of the Company

Domiciliary Agent of the Company

J.P. Morgan SE, Luxembourg Branch European Bank and Business Centre 6 route de Trèves L-2633 Senningerberg Grand Duchy of Luxembourg MSIM Fund Management (Ireland) Limited Luxembourg
Branch
European Bank and Business Centre
6B route de Trèves
L-2633 Senningerberg
Grand Duchy of Luxembourg

Registrar and Transfer Agent of the Company

Depositary of the Company

RBC Investor Services Bank S.A. 14, Rue Porte de France L-4360 Esch-sur-Alzette Grand Duchy of Luxembourg J.P. Morgan SE, Luxembourg Branch
 European Bank and Business Centre
 6 route de Trèves
 L-2633 Senningerberg
 Grand Duchy of Luxembourg

Auditor of the Company

Ernst & Young S.A. 35E, avenue J.F. Kennedy L-2082 Luxembourg Grand Duchy of Luxembourg

Legal Adviser of the Company as to Luxembourg law

Arendt & Medernach S.A. 41A, avenue J.F. Kennedy L-2082 Luxembourg Grand Duchy of Luxembourg

Appendix A – Investment Powers and Restrictions

Investment and Borrowing Powers

1. The Company's Articles permit it to invest in transferable securities and other liquid financial assets, to the full extent permitted by Luxembourg law. The Articles have the effect that, subject to the law, it is at the Directors' discretion to determine any restrictions on investment or on borrowing or on the pledging of the Company's assets.

Investment and Borrowing Restrictions

The following restrictions apply to all investments by the Company except investments by the Company in any wholly-owned subsidiary.

- 2. The following restrictions of Luxembourg law and (where relevant) of the Directors currently apply to the Company:
 - 2.1. The investments of each Fund shall consist of:
 - (a) transferable securities and money market instruments admitted to or dealt in on a regulated market within the meaning of Directive 2004/39/EC of the European Parliament and of the Council of 21 April 2004 on markets in financial instruments in Member States of the EU ("Regulated Market");
 - (b) transferable securities and money market instruments dealt in on other regulated markets that are operating regularly, are recognised and are open to the public ("Other Regulated Market") in Member States of the EU;
 - (c) transferable securities and money market instruments admitted to official listings on stock exchanges in any other country in Europe, Asia, Oceania, the American and African continents,
 - (d) Transferable securities and money market instruments dealt in on any Other Regulated Markets in Europe, Asia, Oceania, the American and African continents;
 - (e) recently issued transferable securities and money market instruments provided that the terms of the issue include an undertaking that application will be made for admission to the official listing on one of the stock exchanges as specified in a) and c) or Other Regulated Markets as specified in b) and d) and that such admission is secured within a year of issue;
 - (f) units of UCITS and/or other undertakings for collective investment ("UCIs") within the meaning of Article 1(2), first and second indents of the UCITS Directive, including shares/units of a master fund qualifying as UCITS (as defined below), whether they are situated in a Member State or not, provided that:
 - such other UCIs are authorized under laws which provide that they are subject to supervision considered by the CSSF to be equivalent to that laid down in EU law, and that cooperation between authorities is sufficiently ensured;
 - the level of protection for unitholders in the other UCIs is equivalent to that provided for unitholders in a UCITS, and in particular that the rules on asset segregation, borrowing, lending, and uncovered sales of transferable securities and money market instruments are equivalent to the requirements of the UCITS Directive;
 - the business of the other UCIs is reported in half-yearly and annual reports to enable an assessment to be made of the assets and liabilities, income and operations over the reporting period;
 - no more than 10% of the UCITS' or the other UCIs' assets (or of the assets of any subfund thereof, provided that the principle of segregation of liabilities of the different compartments is ensured in relation to third parties), whose acquisition is contemplated, can, according to their constitutional documents, be invested in aggregate in units of other UCITS or other UCIs. This restriction does not apply where a fund qualified as a feeder fund is investing in shares/units of a master fund qualifying as a UCITS;

For the purposes of this subparagraph (f), each sub-fund of a UCI with several sub-funds within the meaning of Article 181 of the 2010 Law must be considered as a separate issuer, provided that each sub-fund may be held severally liable for its own debts and obligations.

- (g) deposits with credit institutions which are repayable on demand or have the right to be withdrawn, and maturing in no more than 12 months, provided that the credit institution has its registered office in an EU Member State or, if the registered office of the credit institution is situated in a non-Member State, provided that it is subject to prudential rules considered by the CSSF as equivalent to those laid down in EU law;
- (h) financial derivative instruments, including equivalent cash-settled instruments, dealt in on a Regulated Market or an Other Regulated Market; and/or financial derivative instruments dealt in over-the-counter ('OTC derivatives'), provided that:
 - the underlying consists of instruments described in sub-paragraphs (a) to (g) above, financial indices, interest rates, foreign exchange rates or currencies, in which the Company may invest according to its investment objectives;
 - the counterparties to OTC derivative transactions are institutions subject to prudential supervision, and belonging to the categories approved by the CSSF and;
 - the OTC derivatives are subject to reliable and verifiable valuation on a daily basis and can be sold, liquidated or closed by an offsetting transaction at any time at their fair value at the Company's initiative;
- (i) money market instruments other than those dealt in on a Regulated Market or an Other Regulated Market, which fall under Article 1 of the 2010 Law, if the issue or issuer of such instruments is itself regulated for the purpose of protecting investors and savings, and provided that they are:
 - issued or guaranteed by a central, regional or local authority or central bank of an EU Member State, the European Central Bank, the EU or the European Investment Bank, a non-Member State or, in the case of a Federal State, by one of the members making up the federation, or by a public international body to which one or more Member States belong or;
 - issued by an undertaking any securities of which are dealt in on Regulated Markets or Other Regulated Market referred to in subparagraphs (a),(b) or (c) above, or;
 - issued or guaranteed by an establishment subject to prudential supervision, in accordance with criteria defined by EU law, or by an establishment which is subject to and complies with prudential rules considered by the CSSF to be at least as stringent as those laid down by EU law or;
 - issued by other bodies belonging to the categories approved by the CSSF provided that investments in such instruments are subject to investor protection equivalent to that laid down in the first, the second or the third indent and provided that the issuer is a company whose capital and reserves amount to at least EUR 10 million and which presents and publishes its annual accounts in accordance with Directive 78/660/EEC (1), is an entity which, within a group of companies which includes one or several listed companies, is dedicated to the financing of the group or is an entity which is dedicated to the financing of securitisation vehicles which benefit from a banking liquidity line.

2.2. Furthermore, each Fund may:

Invest no more than 10% of its net assets in securities and money market instruments other than those referred to in sub-paragraph 2.1.

2.3. Each Fund may acquire the units of UCITS and/or other UCIs referred to in paragraph 2.1 (f), provided that, unless (i) stated to the contrary in the investment policy of the relevant Fund and/or (ii) the Fund's denomination includes the term 'fund of funds', the aggregate investment in UCITS or other UCIs does not exceed 10% of the net assets of each Fund.

In the case of Funds not subject to the 10% restriction above, such Funds may acquire units of UCITS and/or UCIs provided that no more than 20% of its assets are invested in the units of a single UCITS or other UCI. Investments made by such Funds in units of UCIs other than UCITS may not in aggregate exceed 30% of the assets of the Fund.

When each Fund has acquired shares of UCITS and/or other UCIs, the assets of the respective UCITS or other UCIs do not have to be combined for the purposes of the limits laid down in paragraph 2.6.

When the Fund invests in the units of other UCITS and/or other UCIs that are managed, directly or by delegation, by the same investment manager or by any other company with which the investment manager is linked by common management or control, or by a substantial direct or indirect holding ("a substantial direct or indirect holding" is defined as more than 10% of the capital or voting rights), no subscription, redemption and management fees may be charged on the target fund level to the Company on its investment in the units of such other UCITS and/or UCIs.

This restriction neither applies to Funds which are feeder funds. A UCITS or a sub-fund thereof is qualified as feeder fund provided that it invests at least 85% of its assets in another UCITS or sub-fund thereof ("master fund") provided such master fund is neither a feeder fund nor hold units/shares of a feeder fund within the meaning of the 2010 Law. To be qualified as feeder fund a Fund shall, in addition to investing 85% in the master fund, not invest more than 15% of its assets in one or more of the following:

- ancillary liquid assets in accordance with Article 41 (1) a) and b) of the 2010 Law;
- financial derivative instruments, which may be used only for hedging purposes, in accordance with Article 41 (1) g) and Article 42 (2) and (3) of the 2010 Law;
- movable and immovable property which is essential for the direct pursuit of the Company's business.

Should a Fund qualify as feeder fund, a description of all remuneration and reimbursement of costs payable by the feeder fund by virtue of its investments in shares/units of the master fund, as well as the aggregate charges of both the feeder fund and the master fund, shall be disclosed in Section 2.5 "Charges and Expenses". The Company shall disclose in its Annual Report a statement on the aggregate charges of both the feeder fund and the master fund.

- 2.4. In addition, a Fund may subscribe, acquire and/or hold Shares of one or more Funds (the "Target Sub-Fund(s)"), without it being subject to the requirements of the Law of 10 August 1915 on commercial companies, as amended, with respect to the subscription, acquisition and/or the holding by a company of its own shares provided that:
 - the Target Sub-Fund does not, in turn, invest in the Fund invested in such Target Sub-Fund; and
 - no more than 10% of the net assets of the Target Sub-Fund the acquisition of which is contemplated may, be invested in aggregate in units/shares of other UCIs; and
 - voting rights, if any, attaching to the relevant Shares of the Target Sub-Fund(s) are suspended for as long as they are held by the Sub-Fund concerned and without prejudice to the appropriate processing in the accounts and the periodic reports; and
 - in any event, for as long as these Shares of the Target Sub-Fund(s) are held by the Fund, their value will not be taken into consideration for the calculation of the net assets of the Fund for the purposes of verifying the minimum threshold of the net assets of the Fund as imposed by law; and
 - there is no duplication of management/subscription or repurchase fees between those at the level of the Fund having invested in the Target Sub-Fund and such Target Sub-Fund.
- 2.5. A Fund may hold ancillary liquid assets (*i.e.*, bank deposits at sight, such as cash held in current accounts with a bank accessible at any time) up to 20% of its net assets, in order to cover current or exceptional payments, or for the time necessary to reinvest in eligible assets provided under article 41(1) of the 2010 Law or for a period of time strictly necessary in case of unfavourable market conditions. The above mentioned 20% limit shall only be temporarily breached for a period of time strictly necessary when, because of exceptionally unfavourable market conditions, circumstances so require and where such breach is justified having regard to the interests of the investors, for instance in highly serious circumstances such as the September 11 attacks or the bankruptcy of Lehman Brothers in 2008.
- 2.6. A Fund may not invest in any one issuer in excess of the limits set out below:
 - (a) not more than 10% of a Fund's net assets may be invested in transferable securities or money market instruments issued by the same entity;

- (b) not more than 20% of a Fund's net assets may be invested in deposits made with the same entity;
- (c) by way of exception, the 10% limit stated in the first paragraph of this section may be increased to:
 - a maximum of 35% if the transferable securities or money market instruments are issued or guaranteed by an EU Member State, by its local authorities, by a non-Member State or by public international bodies to which one or more Member States belong;
 - a maximum of 25% in the case of certain bonds which fall under the definition of covered bonds in point (1) of Article 3 of Directive (EU) 2019/2162 of the European Parliament and of the Council and for certain bonds that were issued before 8 July 2022 by a credit institution which has its registered office in an EU Member State and is subject by law to special public supervision designed to protect bond holders. In particular, sums deriving from the issue of these bonds issued before 8 July 2022 must be invested in conformity with the law in assets which, during the whole period of validity of the bonds, are capable of covering claims attaching to the bonds and which, in the event of failure of the issuer, would be used on a priority basis for the reimbursement of the principal and payment of the accrued interest. When a Fund invests more than 5% of its net assets in the bonds referred to in this paragraph and issued by one issuer, the total value of these investments may not exceed 80% of the value of the net assets of such Fund.
- (d) The total value of the transferable securities or money market instruments held by a Fund in the issuing bodies in each of which it invests more than 5% of its net assets must not then exceed 40% of the value of its net assets. This limitation does not apply to deposits and OTC derivative transactions made with financial institutions subject to prudential supervision. The transferable securities and money market instruments referred to in the two indents set out in 2.6 (c) above shall not be taken into account for the purpose of applying the limit of 40% referred to in this paragraph.

Notwithstanding the individual limits laid down in sub-paragraphs 2.6 (a) to (d) above, a Fund may not combine

- investments in transferable securities or money market instruments issued by a single entity, and/or
- deposits made with a single entity, and/or
- exposures arising from OTC derivative transactions undertaken with a single entity,

in excess of 20% of its net assets.

When a transferable security or money market instrument embeds a derivative, the latter must be taken into account when complying with the requirements of the above mentioned restrictions.

The limits provided for in sub-paragraphs 2.6 (a) to (d) above may not be combined, and thus investments in transferable securities or money market instruments issued by the same entity or in deposits or derivative instruments made with this entity carried out in accordance with paragraphs 2.6 (a) to (d) shall under no circumstances exceed in total 35% of the net assets of the Fund.

Companies which are included in the same group for the purposes of consolidated accounts, as defined in accordance with Directive 83/349/ EEC or in accordance with recognized international accounting rules, are regarded as a single entity for the purpose of calculating the investment limits mentioned in sub-paragraphs 2.6. (a) to (d) above.

The Fund may not invest cumulatively more than 20% of its net assets in transferable securities or money market instruments of the same group subject to restrictions 2.6 (a) and the three indents under 2.6 (d) above.

Without prejudice to the limits laid down in paragraph 2.8 below, the limit of 10% laid down in sub-paragraph 2.6 (a) above is raised to a maximum of 20% for investment in equity and/or debt securities issued by the same body when the aim of the investment policy of a Fund is to replicate the composition of a certain equity or debt securities index which is recognised by the CSSF, on the following basis:

- the composition of the index is sufficiently diversified,
- the index represents an adequate benchmark for the market to which it refers,
- it is published in an appropriate manner.

This limit is 35% where that proves to be justified by exceptional market conditions in particular in Regulated Markets where certain transferable securities or money market instruments are highly dominant. The investment up to this limit is only permitted for a single issuer.

By way of derogation, each Fund is authorised to invest up to 100% of its net assets in different transferable securities and money market instruments issued or guaranteed by an EU Member State, its local authorities, by another member state of the OECD or of the Group of twenty (G20), by the Republic of Singapore or Hong-Kong or by public international bodies of which one or more EU Member States are members, provided that (i) such securities are part of at least six different issues and (ii) securities from any one issue do not account for more than 30% of the net assets of such Fund.

- 2.7. The Company may not invest in shares with voting rights enabling it to exercise significant influence over the management of the issuing body.
- 2.8. The Company may not:
 - a. acquire more than 10% of the shares with non-voting rights of one and the same issuer:
 - b. acquire more than 10% of the debt securities of one and the same issuer;
 - c. acquire more than 25% of the units of one and the same undertaking for collective investment;
 - d. acquire more than 10% of the money market instruments of any single issuer.

The limits stipulated in sub-paragraphs 2.8. (b) (c) and (d) above may be disregarded at the time of acquisition if, at that time, the gross amount of debt securities or of the money market instruments, or the net amount of securities in issue cannot be calculated.

- 2.9. The limits stipulated in paragraphs 2.7. and 2.8. above do not apply to:
 - a. transferable securities and money market instruments issued or guaranteed by an EU
 Member State or its local authorities;
 - transferable securities and money market instruments issued or guaranteed by a non-EU Member State;
 - c. transferable securities and money market instruments issued by public international institutions to which one or more EU Member States are members;
 - d. transferable securities held by a Fund in the capital of a company incorporated in a non-Member State investing its assets mainly in the securities of issuing bodies having their registered offices in that State, where under the legislation of that State such a holding represents the only way in which such Fund can invest in the securities of issuing bodies of that State. This derogation, however, shall apply only if in its investment policy the company from the non-Member State complies with the limits laid down in Articles 43, 46 and 48 (1) and (2) of the 2010 Law. Where the limits set in Articles 43 and 46 of the 2010 Law are exceeded, Article 49 shall apply mutatis mutandis;
 - e. transferable securities held by the Company in the capital of subsidiary companies carrying on only the business of management, advice or marketing in the country where the subsidiary is located, in regard to the repurchase of units at unitholders' request exclusively on its or their behalf.
- 2.10. The Company may always, in the interest of the shareholders, exercise the subscription rights attached to securities, which forms part of its assets.

When the maximum percentages stated in paragraphs 2.2 through 2.8 above are exceeded for reasons beyond the control of the Company, or as a result of the exercise of subscription rights, the Company must adopt, as a priority objective, sales transactions to remedy the situation, taking due account of the interests of its shareholders.

- 2.11. A Fund may borrow to the extent of 10% of its total net assets (valued at market value) provided these borrowings are made on a temporary basis. However, the Company may acquire for the account of a Fund foreign currency by way of back-to-back loan. Collateral arrangements with respect to the writing of options or the purchase or sale of forward or future contracts are not deemed to constitute "borrowings" for the purpose of this restriction.
- 2.12. The Company may not grant credit facilities nor act as guarantor on behalf of third parties, provided that for the purpose of this restriction (i) the acquisition of transferable securities, money market instruments or other financial investments referred to in sub-paragraphs 2.1 (f), (h) and (i) above, in fully or partly paid form and (ii) the permitted lending of portfolio securities shall be deemed not to constitute the making of a loan.
- 2.13. The Company undertakes not to carry out uncovered sales transactions of transferable securities, money market instruments or other financial instruments referred to in sub-paragraphs 2.1 (f), (h) and (i) above; provided that this restriction shall not prevent the Company from making deposits or carrying out accounts in connection with financial derivatives instruments, permitted within the limits referred to below.
- 2.14. The Company's assets may not include precious metals or certificates representing them or commodities.
- 2.15. The Company may not purchase or sell real estate or any option, right or interest therein, provided that the Company may invest in securities secured by real estate or interests therein or issued by companies which invest in real estate or interests therein.
- 2.16. The Company will in addition comply with such further restrictions as may be required by the regulatory authorities in any country in which the Shares are marketed.
- 2.17. The Company shall not issue warrants or other rights to subscribe for Shares in the Company to its Shareholders.
- 2.18. For the Cash Equivalents, a Fund may invest in them pursuant to the applicable investment restrictions, in order to (i) achieve its investment goals, and/or for (ii) treasury purposes, and/or (iii) in case of unfavourable market conditions.

The Company shall take the risks that it deems reasonable to reach the assigned objective set for each Fund; however, it cannot guarantee that it shall reach its goals given stock exchange fluctuations and other risks inherent in investments in transferable securities.

3 Derivatives and efficient portfolio management techniques

- 3.1. The Funds are authorised to use derivatives either for hedging or efficient portfolio management purposes including duration management or as part of their investment strategies as described in the Funds' investment objectives.
- 3.2. The Company must employ a risk-management process which enables it to monitor and measure at any time the risk of the positions and their contribution to the overall risk profile of the portfolio; it must employ a process for accurate and independent assessment of the value of OTC derivative instruments. It must communicate to the CSSF regularly and in accordance with the detailed rules defined by the latter, the types of derivative instruments, the underlying risks, the quantitative limits and the methods which are chosen in order to estimate the risks associated with transactions in derivative instruments.
- 3.3. The Company will ensure that the global exposure relating to derivatives shall not exceed the total net value of a Fund. The exposure is calculated taking into account the current value of the

underlying assets, the counterparty risk, future market movements and the time available to liquidate the positions.

The Funds may invest, as part of their investment policy and within the limits laid down in paragraph 2.6 (a) to (d) above in financial derivative instruments provided that the exposure to the underlying assets does not exceed in aggregate the investment limits laid down in paragraph 2.6. When a Fund invests in index-based financial derivative instruments, these investments do not have to be combined to the limits laid down in paragraph 2.6.

When a transferable security or money market instrument embeds a derivative, the latter must be taken into account when complying with the requirements of the above-mentioned restrictions.

3.4 On an ancillary basis, the Company can also enter into one or several TRS to gain exposure to reference assets, which may be invested according to the investment policy of the relevant Fund. In particular, TRS may be used to gain exposures where a direct investible instrument is not available, or to implement the stated investment policy of the relevant Fund in a more efficient manner.

A TRS is an agreement in which one party (total return payer) transfers the total economic performance of a reference obligation, which may for example be a share, bond or index, to the other party (total return receiver). The total return receiver must in turn pay the total return payer any reduction in the value of the reference obligation and possibly certain other cash flows. Total economic performance includes income from interest and fees, gains or losses from market movements, and credit losses. A Fund may use a TRS to gain exposure to an asset (or other reference obligation), which it does not wish to buy and hold itself, or otherwise to make a profit or avoid a loss. TRS entered into by a Fund may be in the form of funded and/or unfunded swaps. An unfunded swap means a swap where no upfront payment is made by the total return receiver at inception. A funded swap means a swap where the total return receiver pays an upfront amount in return for the total return of the reference obligation.

Where a Fund enters into a TRS or invests in other derivatives with similar characteristics (within the meaning of, and under the conditions set out in, applicable laws, regulations and CSSF circulars issued from time to time, in particular, but not limited to, Regulation (EU) 2015/2365):

- the assets held by the Fund should comply with the investment limits set out in paragraphs 2.3, 2.6, 2.7, 2.8 and 2.9 above; and
- the underlying exposures of such derivatives must be taken into account to calculate the investment limits laid down in paragraph 2.6 above.
- none of the counterparties will have discretion over the composition or management of the portfolio of the Fund or the underlying assets of the financial derivative instruments.

In addition, the Company may only enter into TRS with regulated financial institutions which have their registered office in one of the OECD countries, and which are specialised in such types of transactions and have a minimum credit rating of investment grade quality.

3.5 The expected and maximum proportion of the total assets which may be subject to TRS is summarized, for each relevant Fund, in the table below. In certain circumstances this proportion may be higher.

Relevant Funds	Expected level (in % of total assets)	Maximum level (in % of total assets)	
Calvert Sustainable Global Green Bond Fund	0	25	
China A-shares Fund	0	15	
China Equity Fund	0	15	
Emerging Leaders Equity Fund	0	15	
Emerging Markets Corporate Debt Fund	0	25	
Emerging Markets Debt Fund	0	25	
Emerging Markets Debt Opportunities Fund	0	15	
Emerging Markets Domestic Debt Fund	0	25	
Emerging Markets Fixed Income Opportunities Fund	0	25	
Emerging Markets Local Income Fund	2	15	
Europe Opportunity Fund	0	25	
European High Yield Bond Fund	0	25	
Frontier Markets Local Debt Fund	0	10	
Global Balanced Defensive Fund	0	50	
Global Balanced Fund	0	50	
Global Balanced Income Fund	0	50	
Global Balanced Risk Control Fund of Funds	0	50	
Global Bond Fund	0	25	
Global Brands Equity Income Fund	190	250	
Global Credit Fund	0	25	
Global Fixed Income Opportunities Fund	0	25	
Global High Yield Bond Fund	0	25	
Global Insight Fund	0	15	
Global Macro Fund	10	500	
Global Multi-Asset Opportunities Fund	25	170	
Indian Equity Fund	0	15	
Latin American Equity Fund	0	15	
Multi-Asset Risk Control Fund	0	50	

NextGen Emerging Markets Fund	0	15
Short Duration US Government Income Fund	0	100
Sustainable Asia Equity Fund	0	15
Sustainable Emerging Markets Equity Fund	0	15
Sustainable Global Fixed Income Opportunities Fund	0	25
US Advantage Fund	0	15
US Dollar Corporate Bond Fund	0	25
US Dollar High Yield Bond Fund	0	25
US Dollar Short Duration Bond Fund	0	25
US Dollar Short Duration High Yield Bond Fund	0	25
US Growth Fund	0	15

- 3.6 Each Fund may incur costs and fees in connection with TRS or other financial derivative instruments with similar characteristics, upon entering into TRS and/or any increase or decrease of their notional amount. The amount of these fees may be fixed or variable. Information on costs and fees incurred by each Fund in this respect, as well as the identity of the recipients and any affiliation they may have with the Depositary, the Investment Adviser or the Management Company, if applicable, may be available in the Annual Report. All revenues arising from TRS, net of direct and indirect operational costs and fees, will be returned to the relevant Fund.
- 3.7. The Annual Reports will contain, in respect of each Fund that has entered into financial derivative instruments over the relevant reporting period, details of:
 - the underlying exposure obtained through financial derivative instruments;
 - the identity of the counterparty(ies) to these financial derivative instruments;
 - the costs and fees in connection with TRS or other financial derivative instruments with similar characteristics;
 - the type and amount of collateral received to reduce counterparty risk exposure.
- 3.8. The Funds are authorised to employ techniques and instruments relating to transferable securities or money market instruments subject to the following conditions:
 - a) they are economically appropriate in that they are realised in a cost-effective way;
 - b) they are entered into for one or more of the following specific aims:
 - (i) reduction of risk;
 - (ii) reduction of cost;
 - (iii) generation of additional capital or income for the relevant Fund with a level of risk which is consistent with its risk profile and applicable risk diversification rules;
 - c) their risks are adequately captured by the Company's risk management process.
- 3.9. The efficient portfolio management techniques ("**EPM Techniques**") that may be employed by the Funds in accordance with paragraph 3.8 above include securities lending and borrowing

transactions, repurchase agreements and reverse repurchase agreements, and buy-sell back transactions (together "SFTs") (within the meaning of, and under the conditions set out in, applicable laws, regulations and CSSF circulars issued from time to time, in particular, but not limited to, Regulation (EU) 2015/2365).

Securities lending and borrowing transactions consist in transactions whereby a lender transfers the ownership of an asset (securities or instruments) to a third party, a borrower, subject to a commitment that the borrower will pay a fee to the lender for the use of the loaned asset and will agree to return equivalent securities or instruments on a future date or when requested to do so by the lender, such transaction being considered as securities lending for the party transferring the securities or instruments and being considered as securities borrowing for the counterparty to which they are transferred. Even though the parties are called lender and borrower, actual ownership of the assets is transferred.

Repurchase agreements consist of transactions governed by an agreement whereby a party sells securities or instruments to a counterparty, subject to a commitment to repurchase them, or substituted securities or instruments of the same description, from the counterparty at a specified price with a financing cost on a future date specified, or to be specified, by the transferor. Such transactions are commonly referred to as repurchase agreements for the party selling the securities or instruments, and reverse repurchase agreements for the counterparty buying them. A reverse repurchase agreement transaction is a forward transaction at the maturity of which the seller (counterparty) has the obligation to repurchase the assets sold and the relevant Fund has the obligation to return the assets received under the transaction. None of the Funds has entered into repurchase agreements and reverse repurchase agreements.

The Funds may engage in securities lending transactions and repurchase and reverse repurchase agreements (collectively, "securities financing transactions"), provided that they are in the best interests of investors to do so and the associated risks have been properly mitigated and addressed. The counterparties to securities financing transactions should be financial institutions which are subject to ongoing prudential regulation and supervision.

Buy-sell back transactions consist of transactions, not being governed by a repurchase agreement or a reverse repurchase agreement as described above, whereby a party buys or sells securities or instruments to a counterparty, agreeing, respectively, to sell to or buy back from that counterparty securities or instruments of the same description at a specified price on a future date. Such transactions are commonly referred to as buy-sell back transactions for the party buying the securities or instruments, and sell-buy back transactions for the counterparty selling them. **None of the Funds has entered into buy-sell back transactions.**

- 3.10. The use of EPM Techniques by the Funds is subject to the following conditions:
 - a) The Company may enter into EPM Techniques only through a standardised system organised by a recognised clearing institution or through a regulated financial institutions which have their registered office in one of the OECD countries, and which are specialised in such types of transactions and have a minimum credit rating of investment grade quality.
 - Authorised counterparties or securities lending agents to SFT must be specialised in the relevant types of transactions and either credit institutions with a registered office in an EU Member State or an investment firm authorised under the MiFID directive or an equivalent set of rules, and subject to prudential supervision, with an investment grade credit rating.
 - b) When entering into a securities lending agreement, the Company should ensure that it is able at any time to recall any security that has been lent out or terminate the securities lending agreement.
 - c) When entering into a reverse repurchase agreement, the Company should ensure that it is able at any time to recall the full amount of cash or to terminate the reverse repurchase agreement on either an accrued basis or a mark-to-market basis. When the cash is recallable at any time on a mark-to-market basis, the mark-to-market value of the reverse repurchase agreement should be used for the calculation of the net asset value of the relevant Fund.

- d) When entering into a repurchase agreement, the Company should ensure that it is able at any time to recall any securities subject to the repurchase agreement or to terminate the repurchase agreement into which it has entered.
- e) Fixed-term repurchase and reverse repurchase agreements that do not exceed seven days should be considered as arrangements on terms that allow the assets to be recalled at any time by the Company.
- f) The Company's Annual Report will include the following information:
 - i. the exposure obtained through EPM Techniques;
 - ii. the identity of the counterparty(ies) to these EPM Techniques;
 - iii. the type and amount of collateral received by the Company to reduce counterparty exposure; and
 - iv. the revenues arising from EPM Techniques for the entire reporting period together with the direct and indirect operational costs and fees incurred.
- 3.11. The Funds listed in the table below may enter into securities lending transactions opportunistically and on a temporary basis. The Funds may use securities lending for the purpose of efficient portfolio management to generate additional capital or income through the transaction itself or through the reinvestment of the cash collateral. The expected and maximum proportion of the total assets which may be subject to securities lending and borrowing transactions is summarized, for each relevant Fund, in the table below. In certain circumstances this proportion may be higher.

Relevant Funds	Expected level (in % of total assets)	Maximum level (in % of total assets)	
American Resilience Fund	3	33	
Asia Opportunity Fund	0	33	
Asian Property Fund	8	33	
Calvert Sustainable Climate Transition Fund	12	33	
Calvert Sustainable Global Green Bond Fund	0	33	
China A-shares Fund	0	33	
China Equity Fund	1	33	
Counterpoint Global Fund	9	33	
Developing Opportunity Fund	0	33	
Emerging Leaders Equity Fund	0	33	
Emerging Markets Corporate Debt Fund	8	33	
Emerging Markets Debt Fund	13	33	
Emerging Markets Debt Opportunities Fund	0	15	
Emerging Markets Domestic Debt Fund	0	33	
Emerging Markets Fixed Income Opportunities Fund	0	33	
Emerging Markets Local Income Fund	5	15	
Euro Bond Fund	12	33	
Euro Corporate Bond Fund	8	33	
Euro Corporate Bond – Duration Hedged Fund	7	33	
Euro Strategic Bond Fund	4	33	
European Fixed Income Opportunities Fund	6	33	
European High Yield Bond Fund	15	33	
European Property Fund	7	33	
Floating Rate ABS Fund	3	33	
Frontier Markets Local Debt Fund	0	15	
Global Bond Fund	2	33	
Global Convertible Bond Fund 9 33		33	

Global Credit Fund	2	33
Global Credit Opportunities Fund	3	33
Global Endurance Fund	0	33
Global Fixed Income Opportunities Fund	2	33
Global Focus Property Fund	0	33
Global High Yield Bond Fund	0	33
Global Infrastructure Fund	3	33
Global Macro Fund	0	33
Global Opportunity Fund	1	33
Global Permanence Fund	0	33
Global Property Fund	4	33
Indian Equity Fund	0	33
Latin American Equity Fund	1	33
NextGen Emerging Markets Fund	0	33
Saudi Equity Fund	0	33
Short Duration US Government Income Fund	0	25
Short Maturity Euro Bond Fund	6	33
Short Maturity Euro Corporate Bond Fund	3	33
Sustainable Asia Equity Fund	1	33
Sustainable Emerging Markets Equity Fund	1	33
Sustainable Euro Corporate Bond Fund	6	33
Sustainable Euro Strategic Bond Fund	5	33
Sustainable Global Fixed Income Opportunities Fund	1	33
US Advantage Fund	1	33
US Dollar High Yield Bond Fund	0	33
US Focus Property Fund	0	33
US Insight Fund	4	33
US Growth Fund	2	33
US Property Fund	1	33
Vitality Fund	1	33

- 3.12. None of the Funds utilise repurchase agreements, reverse repurchase agreements or buy-sell-back transactions.
- 3.13. Each Fund may incur costs and fees in connection with SFTs. In particular, a Fund may pay fees to agents and other intermediaries, which may be affiliated with the Depositary, the Investment Adviser or the Management Company, in consideration for the functions and risks they assume. The amount of these fees may be fixed or variable. Information on direct and indirect operational costs and fees incurred by each Fund in this respect, as well as the identity of the entities to which such costs and fees are paid and any affiliation they may have with the Depositary, the Investment Adviser or the Management Company, if applicable, may be available in the Annual Report. All revenues arising from SFTs, net of direct and indirect operational costs and fees, will be returned to the Fund.
- 3.14. The counterparty risk arising from OTC derivative instruments and EPM Techniques may not exceed 10% of the assets of a Fund when the counterparty is a credit institution domiciled in the EU or in a country where the CSSF considers that supervisory regulations are equivalent to those prevailing in the EU. This limit is set at 5% in any other case.
- 3.15. For the purpose of the restriction set out in paragraph 3.14., above, the counterparty risk of a Fund towards a counterparty under OTC derivative instruments or EPM Techniques is reduced by the amount of collateral posted in favour of the Fund. Collateral received by the Funds must comply at all times with the eligibility requirements set out in the Collateral Policy (Appendix B).
- 3.16. The collateral eligibility requirements set out in the Collateral Policy (Appendix B) stem from the ESMA Guidelines 2014/937on ETFs and other UCITS issues (the "ESMA Guidelines 2014/937") that apply to Luxembourg UCITS in accordance with CSSF Circular 14/592.

Appendix B – Collateral Policy

1. General

Funds are allowed to enter into OTC financial derivative transactions and to use EPM Techniques subject to the restrictions set out in Appendix A - Investment Powers and Restrictions, section 3 – "Derivatives and efficient portfolio management techniques". In particular, the counterparty risk arising from OTC derivative instruments and EPM Techniques may not exceed 10% of the assets of a Fund when the counterparty is a credit institution domiciled in the EU or in a country where the CSSF considers that supervisory regulations are equivalent to those prevailing in the EU. This limit is set at 5% in any other case.

The counterparty risk of a Fund vis-à-vis a counterparty will be equal to the positive mark-to-market value of all OTC derivative and EPM Techniques transactions with that counterparty, provided that:

- if there are legally enforceable netting arrangements in place, the risk exposure arising from OTC derivative and EPM Techniques transactions with the same counterparty may be netted; and
- if collateral posted in favour of the Fund and such collateral complies at all times with the criteria set out in section 2. below, the counterparty risk of a Fund towards a counterparty under OTC derivative or EPM Techniques transactions is reduced by the amount of such collateral.

The purpose of this Appendix is to set the collateral policy that will be followed by all Funds.

2. Eligible collateral

2.1 General principles

Collateral received by a Fund may be used to reduce its counterparty risk exposure with a counterparty if it complies at all times with the criteria laid down in the ESMA Guidelines 2014/937. By way of derogation to the principle of collateral diversification laid down under 43 (e) of the ESMA Guidelines 2014/937, each Fund may have an exposure for up to 100% of its net assets in securities issued or guaranteed by a Member State, its local authorities, a member State of the OECD or by a public international body of which one or more Member States are members, provided that the Fund holds securities of at least six different issues and that the securities from any one issue do not account for more than 30% of the net assets of the Fund.

2.2 For the purpose of paragraph 2.1. above, all assets received by a Fund in the context of EPM Techniques should be considered as collateral.

2.3 Eligible assets

Collateral received by a Fund will only be taken into account for reducing its counterparty risk exposure with a counterparty if it consists of assets which are part of the following list:

- (a) liquid assets. Liquid assets include not only cash and short term bank certificates, but also money market instruments such as defined within the UCITS Directive. A letter of credit or a guarantee at first-demand given by a first class credit institution not affiliated to the counterparty are considered as equivalent to liquid assets;
- (b) bonds issued or guaranteed by a Member State of the OECD or by their local public authorities or by supranational institutions and undertakings with EU, regional or world-wide scope;
- shares or units issued by money market UCIs calculating a daily net asset value and being assigned a rating of AAA or its equivalent;
- (d) shares or units issued by UCITS investing mainly in bonds/shares mentioned in (e) and (f) below;
- (e) bonds issued or guaranteed by first class issuers offering an adequate liquidity;
- (f) shares admitted to or dealt in on a Regulated Market of a Member State of the EU or on a stock exchange of a Member State of the OECD, on the condition that these shares are included in a main index.

The above general collateral eligibility requirements are without prejudice to the more specific requirements which may apply to a Fund under Section 1.2 "Investment Objectives and Policies" of the Prospectus.

3. Reinvestment of collateral

3.1 Non-cash collateral

Non-cash collateral received by a Fund may not be sold, re-invested or pledged.

3.2 Cash collateral

Cash collateral received by a Fund can only be:

- (a) placed on deposit with credit institutions which either have their registered office in an EU Member State or are subject to prudential rules considered by the CSSF as equivalent to those laid down in EU law;
- (b) invested in high-quality government bonds;
- (c) used for the purpose of reverse repo transactions provided the transactions are with credit institutions subject to prudential supervision and the Company is able to recall at any time the full amount of cash on accrued basis;
- (d) invested in Short-Term Money Market Funds.

Re-invested cash collateral should be diversified in accordance with the diversification requirements applicable to non-cash collateral as set out in the ESMA Guidelines 2014/937.

4. Safekeeping of collateral

Collateral posted in favour of a Fund under a title transfer arrangement should be held by the Depositary or one of its correspondents or sub-custodians. Collateral posted in favour of a Fund under a security interest arrangement (e.g., a pledge) can be held by a third party custodian which is subject to prudential supervision, and which is unrelated to the provider of the collateral.

5. Level and valuation of collateral

The Company will determine the required level of collateral for OTC financial derivatives transactions and efficient portfolio management techniques by reference to the applicable counterparty risk limits set out in this prospectus and taking into account the nature and characteristics of transactions, the creditworthiness and identity of counterparties and prevailing market conditions.

Collateral will be valued, on a daily basis, using available market prices and taking into account appropriate discounts which will be determined by the Company for each asset class based on its haircut policy.

6. Haircut policy

The Company has a haircut policy relating to the classes of assets received as collateral. The Company typically receives cash, high-quality government and non-government bonds as collateral. The Company will typically apply haircuts ranging from 0.5-10% for government bonds and from 5-15% for non-government bonds. No haircut will generally be applied to cash collateral. Haircuts are assessed based on collateral credit quality, price volatility and tenor, and the Company may vary the haircut outside the above ranges if it consider it to be appropriate based on these factors.

Appendix C - Additional Information for U.K. Investors

Morgan Stanley Investment Funds Société d'Investissement à Capital Variable Luxembourg

General

This Supplement should be read in conjunction with the Company's Prospectus, of which it forms part. References to the "Prospectus" are to be taken as references to that document as supplemented or amended hereby.

The Company is a recognised collective investment scheme for the purposes of Section 264 of the Financial Services and Markets Act 2000 ("FSMA") of the United Kingdom. The Prospectus is distributed in the United Kingdom by or on behalf of the Company and is approved by Morgan Stanley Investment Management Limited, whose registered office is at 25 Cabot Square, Canary Wharf, London, E14 4QA. Morgan Stanley Investment Management Limited is regulated by the Financial Conduct Authority ("FCA"), and is an "authorised person" for the purposes of Section 21 of the FSMA. The registered office of the Company is at European Bank and Business Centre, 6B route de Trèves, L-2633 Senningerberg, Grand Duchy of Luxembourg.

Morgan Stanley Investment Management Limited is acting for the Company in relation to the Prospectus and matters relating thereto and it or any of its associates may have an interest or position in Shares of the Company. It is not acting for, or advising or treating as its customer any other person (unless other arrangements apply between Morgan Stanley Investment Management Limited and such person) in relation to investment in the Company.

Important

A United Kingdom investor who enters into an investment agreement with the Company to acquire Shares in response to the Prospectus will not have the right to cancel the agreement under the cancellation rules made by the Financial Conduct Authority in the United Kingdom. The agreement will be binding upon acceptance of the order by the Company.

The Company does not carry on any regulated activity from a permanent place of business in the United Kingdom and United Kingdom investors are advised that most of the protections afforded by the United Kingdom regulatory system will not apply to an investment in the Company. Shareholders in the Company may not be protected by the Financial Services Compensation Scheme established in the United Kingdom.

Morgan Stanley Investment Management Limited shall provide the following facilities to United Kingdom investors at 25 Cabot Square, Canary Wharf, London, E14 4QA:

- (i) arrangement for the payment of dividends in respect of shares in certificated form, if any, and for the payment of redemption or repurchase proceeds;
- (ii) during normal business hours on any week day (Saturday and public holidays excepted), the inspection, free of charge and the supply of copies, if required, of the following in English:
 - the Prospectus;
 - the Articles of Incorporation of the Company;
 - the key investor information document;
 - the most recently published audited annual reports and the unaudited semi-annual reports;
 - notices which have been given or sent to investors;
 - and the Net Asset Value per Share of each Class of Shares on each Dealing Day.
- (iii) the receipt of complaints for forwarding to the Company.

Potential investors should note that the investments of the Company are subject to normal market fluctuations and other risks inherent in investing in shares and other securities, in addition to the additional risks associated with investment in certain of the Funds, as described under Sections 1.2 "Investment Objectives and Policy" and 1.5 "Risk Factors".

The value of investments and the income from them, and therefore the value of, and income from, the Shares of each Class can go down as well as up and an investor may not get back the amount he invests. Changes in exchange rates between currencies may also cause the value of the investment to diminish or increase.

Shareholders may view the Net Asset Value per Share on the Company's web-site (www.morganstanleyinvestmentfunds.com).

UK Taxation

The following information is based on the law in force in the United Kingdom ("UK") and HM Revenue & Customs published practice understood to be applicable on the date of this Supplement, is subject to changes therein and is not exhaustive. Unless expressly stated otherwise, it is intended to offer guidance only to UK resident, and (in the case of individuals) domiciled investors holding Shares as an investment as the absolute beneficial owners thereof on the UK taxation of the Company and such investors, but does not constitute legal or tax advice. The following summary is not a guarantee to any investor of the taxation results of investing in the Company and may not apply to certain categories of investors.

The rates, bases and levels of, and any relief from, taxation can change. Tax treatment depends on the individual circumstances of the investor, and investors and prospective investors should inform themselves of, and where appropriate take advice on, the tax consequences applicable to the subscription, purchase, holding, disposal and redemption of Shares and the receipt of distributions (whether or not on redemption) in respect of such Shares in the country of their citizenship, residence or domicile and any other country in which they are liable to taxation.

The Company

It is the intention of the Directors to conduct the affairs of the Company so that it does not become resident in the UK for taxation purposes. Accordingly, and provided that the Company does not carry on a trade in the UK (whether or not through a permanent establishment situated therein), the Company will not be liable to UK income tax or corporation tax on its income or gains earned on or derived from the Company's investments save for tax on certain income deriving from a UK source, for example, interest with a UK source (UK tax on this interest potentially being levied by withholding at source).

UK resident Shareholders

Individual Shareholders that are resident in the UK for taxation purposes may, subject to their personal circumstances, be liable to UK income tax in respect of any dividends paid by the Company or any other income distributions made by the Company (whether or not reinvested and including undistributed reported income under the "reporting fund" regime). Individual investors should also refer to the Section on Anti-avoidance provisions below. Investors who are within the charge to UK corporation tax in respect of Shares in the Company will generally be exempt from corporation tax on dividends and other income distributions, unless the bond fund rules (described below) or other anti-avoidance provisions apply.

UK individual Shareholders are not charged to UK income tax in respect of the first £2,000 of dividend income received in a tax year, which is taxed at a nil rate.

A UK resident investor's liability to UK income tax or UK corporation tax in respect of income distributions made by the Company may be adjusted for a number of reasons, in particular as a result of equalisation arrangements if such investor subscribes for Shares otherwise than at the beginning of a period over which distributions are calculated.

Investors should note that special rules apply to certain categories of taxpayer, including insurance companies, pension schemes, investment trusts, authorised unit trusts, open-ended investment companies, charities and certain governmental bodies, among others.

Taxation of investors in Bond Funds

A fund is considered a bond fund for UK taxation purposes if the market value of its "qualifying investments" (as detailed below) at any time exceeds 60% of the market value of all its investments. Given the current structure and investment objectives of certain Funds, these rules are likely to be relevant to certain classes of Shares in certain Funds.

"Qualifying investments" are: (a) money placed at interest (other than cash awaiting investment); (b) securities (other than shares in a company); (c) shares in a building society; (d) qualifying holdings in a unit trust scheme, an offshore fund or an open-ended investment company (this can be interpreted as a holding in a unit trust, offshore fund or OEIC which itself fails, at that time or at any other time in the relevant accounting period, the non-qualifying investment test (as set out above) in respect of its holdings of investments; (e) alternative finance arrangements; (f) derivative contracts in respect of currency or any of the matters listed in (a) to (e) above; (g)

contracts for differences relating to interest rates, creditworthiness or currency; and (h) derivative contracts not within (f) or (g) where there is a hedging relationship between the derivative contract and an asset within (a) to (d) above.

An individual resident in the UK for taxation purposes who holds Shares of a bond fund will be taxed on dividends from that bond fund as if that payment was a payment of interest. These rules may apply to a dividend received by UK resident investor from a bond fund notwithstanding that it may have divested of its holding at the date it receives the dividend.

A company resident in the UK for taxation purposes which is within the charge to UK corporation tax, and which holds Shares of a bond fund will be taxed on a fair value basis for each year of its investment as if its interest in the relevant Shares was a right under a creditor loan relationship. These rules will apply to a UK resident corporate investor if the 60% limit is exceeded at any time during that investor's accounting period, even if it was not holding Shares of that class at that time.

"Reporting fund" status

On the basis that the Company provides arrangements for the separate pooling of the contributions of investors to the Company and the profits or income out of which payments are made to investors in the Company, the Company is an umbrella fund for United Kingdom tax purposes. In addition, all of the Funds within the Company consist of different classes of Shares. The United Kingdom offshore funds rules therefore apply in relation to each separate class of Shares within each Fund, as if each class of Shares in each Fund formed a separate offshore fund for United Kingdom tax purposes.

The UK operates a specific regime for the taxation of UK resident investors in offshore funds came into force. Under this regime, funds may be certified as "reporting funds" and UK resident investors in such funds should retain capital gains treatment on disposal of their holdings (subject to the rules outlined above for corporate investors in bond funds). The Company intends to maintain "reporting fund" status for the share classes highlighted as such on www.morganstanleyinvestmentfunds.com. Details of the reportable income for such share classes will be made available on the above website within 6 months of the Company's year-end or the closure or merger of a relevant fund or share class. Any excess of reportable income over amounts paid out as distributions must be declared by UK investors on their own tax returns and will be taxed as dividends or interest, depending on whether the Fund is a bond fund or not. Shareholders without access to the internet may arrange to receive details of reportable income by alternative means by contacting Morgan Stanley Investment Management Limited. In all cases, there can, however, be no guarantee that "reporting fund" certification will be obtained or maintained.

If approval as a "reporting fund" is granted, "reporting fund" status will apply to Shares of the relevant class for each period of account of the Company provided the Company continues to comply with the applicable rules in respect of Shares of that class and does not elect for Shares of that class to become a "non-reporting fund".

For so long as "reporting fund" status is maintained, any profit on a disposal of Shares of the relevant class (for example, by way of transfer or redemption including switching between classes of Shares) by a UK resident investor should fall to be taxed as a capital gain (subject to the rules outlined above for corporate investors in bond funds).

If "reporting fund" status is not maintained for any accounting period (or is not initially sought or obtained) in respect of a class of Shares, any gain arising on a disposal of Shares of that class (for example, by way of transfer or redemption including switching between classes of Shares) will constitute income for all purposes of United Kingdom taxation.

Anti-avoidance provisions

The attention of companies resident in the UK for taxation purposes is drawn to the "controlled foreign companies" provisions contained in Part 9A Taxation (International and Other Provisions) Act 2010. These provisions affect UK resident companies which are deemed to be interested, either alone or together with certain connected or associated persons, in at least 25% of the profits (not including capital gains) of a non-UK resident company (such as the Company) which (i) is controlled by companies or other persons who are resident in the UK for taxation purposes and (ii) is subject to a low level of taxation in the territory in which it is resident.

The attention of individuals resident in the UK for taxation purposes is drawn to the provisions of Chapter 2 of Part 13 of the Income Tax Act 2007. These provisions are aimed at preventing the avoidance of income tax by individuals through transactions resulting in the transfer of assets or income to persons or companies resident or

domiciled abroad, and may render them liable to income tax in respect of the undistributed income (if any) of the Company.

The attention of persons resident in the UK is drawn to the fact that the provisions of section 13 of the Taxation of Chargeable Gains Act 1992 could be material to any such person who, either alone or together with certain "connected persons", holds more than 25% of the Shares in the Company if, at the same time, the Company is controlled in such a manner as to render it a company that would, were it to be resident in the UK, be a "close" company for UK taxation purposes. These provisions could, if applicable, result in such a person being treated for the purposes of UK taxation as if a proportionate part of any gain accruing to the Company (such as on a disposal of any of its investments) had accrued to that person at the time when the chargeable gain accrued to the Company.

Appendix D - Additional Information for Irish Investors

Facilities Agent

The Company and J.P. Morgan SE, Luxembourg Branch (the "Depositary") have appointed J.P. Morgan Administration Services (Ireland) Limited as facilities agent of the Company in Ireland for the tasks listed a) and b) below in Article 92 of the UCITS Directive, as amended by Directive 2019/1160, and it has agreed to provide such facilities at its offices at International Financial Services Centre, Dublin 1, Ireland. They can be contacted by email to Dublin.TAOperations@jpmorgan.com.

According to Article 92 of the UCITS Directive, as amended by Directive 2019/1160, the facilities tasks are as follows:

- a. Process subscription, repurchase and redemption orders and make other payments to Shareholders relating to the Shares of the Company, in accordance with the conditions set out in the Prospectus and the Key Information Documents:
- b. Provide investors with information on how orders referred to in point a. can be made and how repurchase and redemption proceeds are paid.
- c. Facilitate the handling of information and access to procedures and arrangements referred to in Article 15 the UCITS Directive relating to the investors' exercise of their rights arising from their investment in the Company in the Member State where the Company is marketed;
- d. Make the information and documents required pursuant to Chapter IX of the UCITS Directive available to investors under the conditions laid down in Article 94 of the UCITS Directive, for the purposes of inspection and obtaining copies thereof;
- e. Provide investors with information relevant to the tasks that the Facilities Agent performs in a durable medium;
- f. Act as a contact point for communicating with the competent authorities. Morgan Stanley Investment Funds, will provide the tasks listed under points c) to e) above in Article 92 of the UCITS Directive, as amended by Directive 2019/1160, at its offices 6B, Route de Trèves, L-2633 Senningerberg, Email: cslux@morganstanley.com. The information stated can be found on: https://morganstanleyinvestmentfunds-website.publifund.com/en_IE

Arendt Regulatory & Consulting S.A. 41A Avenue J.F Kennedy, L-2082 Luxembourg, will act as the contact point for communicating with the competent authorities, according to point f) above of Article 92 of the UCITS Directive, as amended by Directive 2019/1160. They can be contacted by email at DistributionARC@arendt.com.

Applications for Shares and redemptions as well as for conversions may also be made to the Transfer Agent in Luxembourg at the address below:

RBC Investor Services Bank S.A. 14, Rue Porte de France L-4360 Esch-sur-Alzette Grand Duchy of Luxembourg Tel (352) 25 47 01 9511, Fax (352) 2460 9902

Publications

The Net Asset Value per Share of each fund of each Class of Shares in respect of each Dealing Day is available at the registered office of the Company and at the office of Morgan Stanley Investment Management Limited.

Shareholders may view the Net Asset Value per Share on the Company's website (www.morganstanleyinvestmentfunds.com).

Irish Taxation

The following information is based on the law in force in Ireland on the date of this Prospectus. This summary deals only with Shares held as capital assets by Irish resident Shareholders and does not address special classes of Shareholders such as dealers in securities or persons that may be exempt from tax such as Irish pension funds and charities. This summary is not exhaustive and Shareholders are advised to consult their own tax advisors with respect to the taxation consequences of the ownership or disposition of Shares.

The Company

It is the intention of the Directors to conduct the affairs of the Company so that it is neither resident in Ireland for tax purposes nor carrying on a trade in Ireland through a branch or agency. Accordingly, the Company will not be subject to Irish corporation tax.

Irish Investors

(a) Taxation generally

Shares in the Company are likely to constitute a "material interest" in an offshore fund for the purposes of Chapter 4 Part 27 of the Taxes Consolidation Act, 1997.

(b) Reporting of acquisition

An Irish resident or ordinarily resident person acquiring Shares in the Company is required to disclose details of the acquisition in his annual tax return. Where an intermediary in the course of carrying on a business in Ireland acquires Shares in the Company it must report details of the acquisition to the Irish Revenue Commissioners.

(c) Income and capital gains

An Irish resident corporate Shareholder will be liable to corporation tax at 25% on income distributions received from the Company and on gains on disposal of shares in the Company, except where the corporate shareholder holds shares as part of its trading activities in which case the rate of corporation tax applicable will be that applicable to trading income. It should be noted that no indexation allowance is available in respect of the gain.

Where an Irish resident or ordinarily resident person who is not a company holds Shares in the Company and receives an income distribution from the Company, that Shareholder will be liable to Irish tax at 41% on the amount of such distribution.

Where an Irish resident or ordinarily resident person who is not a company disposes of a Share a liability to Irish tax at 41% will arise on the amount of the gain. No indexation allowance is available and the death of a Shareholder would constitute a deemed disposal of a Share.

There is a deemed disposal and reacquisition at market value for the purposes of Irish tax of Shares held by an Irish resident or ordinarily resident investor on a rolling 8 year basis where the Shares are acquired on or after 1 January 2001. This deemed disposal takes place at market value so that Irish resident or ordinarily resident shareholders will be subject to tax at the rate of 41% for individuals or 25% for a corporate shareholder on the increase in value of their Shares at 8 year intervals commencing on 8th anniversary of the date of acquisition of the Shares.

To the extent that any tax arises on such a deemed disposal such tax will be taken into account to ensure that any tax payable on the subsequent encashment, redemption, cancellation or transfer of the relevant Shares does not exceed the tax that would have arisen had the deemed disposal not occurred.

Anti-avoidance provision

There is an anti-avoidance provision imposing higher rates of tax on Irish resident investors in "personal portfolio investment undertakings" ("PPIU"). A PPIU is a fund in which the investor, or a person connected with the investor, has a right under the terms of the fund or any other agreement, to influence the selection of the assets of the fund. If a fund is treated as a PPIU the Irish resident investor can suffer tax at rates of up to 60% (80% where details of the payments/disposal are not correctly included in the shareholder's annual tax return) on amounts received from the fund, or on the rolling 8 year deemed disposal.

Specific exemptions apply where the ability to select the property invested in has been clearly identified in the offshore fund's marketing and promotional literature and the investment is widely marketed to the public.

Withholding obligation on Facilities agents

If any dividend is paid through an Encashment Agent established in Ireland, such an agent would be obliged to deduct tax from such dividend and account for this to the Irish Revenue Commissioners. The recipient of the dividend would be entitled to claim a credit for the sum deducted by the Facilities Agent against his tax liability for the relevant year.

Stamp duty

Transfers for cash or Shares in the Company will not be subject to Irish stamp duty.

Gift and inheritance tax

A gift or inheritance of Shares in the Company received from a person who is resident or ordinarily resident in Ireland or received by such a person will be within the charge to Irish capital acquisitions tax. Capital acquisitions tax is charged at a rate of 33% above a tax-free threshold which is determined by the amount of the benefit and of previous benefits within the charge to capital acquisitions tax, and the relationship between the person treated as disposing of such shares and the successor or donee. Tax chargeable on the gain arising on a deemed disposal by an individual on their death shall be treated as an amount paid in respect of capital gains tax for the purposes of crediting such amount paid against gift on inheritance tax arising on such death.

Transfers between funds

The Directors have been advised that in the Republic of Ireland the exchange of Shares in one sub-fund for shares in another sub-fund of an umbrella scheme should not constitute a disposal of such Shares and will not give rise to a charge to tax.

Appendix E - Additional Information for Chilean Investors

All of the information provided by the Company for the purpose of registering the Shares on the "Registro de Valores Extranjeros" ("Chilean Foreign Securities Registry" (or the "FSR"), maintained by the "Comisión para el Mercado Financiero" of Chile ("Financial Markets Commission" or the "CMF") is available to the public at the offices of (i) Santander Investment S.A. Corredores de Bolsa, (ii) Larrain Vial S.A. Corredora de Bolsa, (iii) Cruz del Sur Corredora de Bolsa S.A., (iv) Credicorp Capital S.A. Corredores de Bolsa, (v) Valores Security S.A. Corredores de Bolsa, (vi) Corredores de Bolsa Sura S.A., (vii) BICE Inversiones Corredores de Bolsa SA and (viii) Seguros de Vida SURA SA at the addresses below.

Santander Investment S.A. Corredores de Bolsa

Bandera 140, Piso 14

Santiago Chile

Telephone: (56-2) 23663400

Fax: (56-2) 26962097

Larraín Vial S.A. Corredora de Bolsa.

El Bosque Norte 0177 Piso 3

Las Condes Santiago Chile

Telephone: (56-2) 2339-8617

Fax: (56-2) 2332-0131

Valores Security S.A. Corredores de Bolsa

Apoquindo 3150, 7th Floor

Las Condes Santiago Chile

Telephone: (56-2) 2584-4601 Facsimile: (56-2) 2584-4015

BICE Inversiones Corredores de Bolsa SA

Teatinos 280, Piso 5

Santiago Chile

Telephone: (56-2) 2692-1850

Cruz del Sur Corredora de Bolsa S.A.

Magdalena 121 Las Condes Santiago Chile

Telephone: (56-2) 24618810

Credicorp Capital S.A. Corredores de Bolsa

Av. Apoquindo 3721 Piso 9

Las Condes Santiago Chile

Telephone: (56-2) 2450-1600

Corredores de Bolsa Sura S.A.

Apoquindo 4820 Floor 16 Las Condes Santiago Chile

Telephone: (56-2) 2915-2021

Seguros de Vida SURA SA

Apoquindo 4820 Las Condes Santiago Chile

Telephone: (56-2) 2915-2018

Taxation

In accordance with article 11 of the Chilean Income Tax Law (Decree-Law 824 of 1974, as modified by Law 21.210 which regulates the taxation of offerings to the public of foreign securities in Chile (the "Income Tax Law")), shares of the Company which are acquired on the Chilean market are not considered to be situated in Chile, as they are securities issued by an entity incorporated outside the country and are registered in the FSR, in accordance with the Title XXIV of the Securities Law. Consequently, as stated in article 10 and 11 of the Income Tax Law, any income arising from the Shares is not considered to be income from a Chilean source. Therefore, Investors who are not domiciled in or resident in Chile will not be subject to tax in Chile in respect of their holding of Shares. Investors who are domiciled or resident in Chile will be subject to tax in Chile in respect of the Shares in accordance with the Income Tax Law.

Private placement sales restriction

CERTAIN SECURITIES INCLUDED IN THIS DOCUMENT ARE NOT REGISTERED IN THE FOREIGN SECURITIES REGISTER HELD BY THE FINANCIAL MARKETS COMMISSION (COMISIÓN PARA EL MERCADO FINANCIERO OR "CMF"). ANY OFFER REGARDING SUCH SECURITIES WILL BE CONDUCTED SUBJECT TO GENERAL RULE N°336 OF THE CMF, BEGINNING AT THE DATE OF THIS PROSPECTUS.

THESE SECURITIES ARE NOT SUBJECT TO THE SUPERVISION OF THE CMF.

BECAUSE THESE SECURITIES ARE NOT REGISTERED, THERE IS NO OBLIGATION TO THE ISSUER TO PROVIDE PUBLIC INFORMATION ON THEM, IN CHILE.

THESE SECURITIES CANNOT BE SUBJECT OF PUBLIC OFFERING AS LONG AS THEY ARE NOT REGISTERED IN THE FOREIGNS SECURITIES REGISTER.

Appendix F – Additional Information for Taiwanese Investors

Any Fund registered for sale with the Taiwan Financial Supervisory Commission (the "FSC") will be subject to the following restrictions:

The Funds offered and sold in Taiwan will be subject to the following restrictions:

- (a) unless otherwise approved or announced for exemption by the FSC, the total value of such Fund's non offset short position in derivatives for hedging purposes will not exceed the total market value of the relevant securities held by that Fund and the risk exposure of such Fund's non offset position in derivatives for purposes of increase of investment efficiency will not exceed 40% of the Net Asset Value of that Fund;
- (b) a Fund may not invest in gold, direct commodities or direct real estate;
- (c) a Fund is not permitted to invest in Mainland China securities other than securities listed on exchanges in Mainland China and bonds traded in the Chinese Interbank Bond Market and the percentage of the Fund's holdings in such securities listed on exchanges in Mainland China and bonds traded in the Chinese Interbank Bond Market may not, at any time, exceed certain percentage limits as stipulated by the FSC from time to time;
- (d) the total investment in such Funds by domestic investors in Taiwan may not, at any time, exceed certain percentage limits as stipulated by the FSC from time to time; and
- (e) the securities market of Taiwan may not constitute a major investment region in the portfolio of any Fund. The investment amount of each Fund in the securities market of Taiwan shall be subject to a percentage stipulated by the FSC from time to time.

Appendix G – Additional Information for Peruvian Investors

The Shares have not been registered before the Superintendencia del Mercado de Valores (the "SMV") and are being placed by means of a private offer subject to the laws of the Grand Duchy of Luxembourg. SMV has not reviewed the information provided to the investor. This Prospectus is only for the exclusive use of institutional investors in Peru and is not for public distribution.

Taxation

In accordance with article 6 and 7 of the Peruvian Income Tax Law, Peruvian resident individuals are subject to income tax for their Peruvian and foreign source income. Consequently: (i) dividends and other payments derived from the Shares, and (ii) capital gains realized upon sale, disposal or redemption of the Shares by a Peruvian resident individual will qualify as foreign source income and, thus, subject to income tax. Foreign source income is added to labour income obtained in each fiscal year and subject to taxation at the ordinary progressive rate of 8%, 14%, 17%, 20% and 30%. However, in the event of a sale of the Shares through the Lima Stock Exchange (Bolsa de Valores de Lima – BVL), the resulting capital gain would be subject to a reduced 6.25% rate or tax exempt³³.

For Peruvian tax purposes, the capital gain is calculated by deducting the corresponding cost basis from the transfer value.

Any deposit made in a Peruvian bank account is subject to a Financial Transactions Tax (FTT)³⁴ with a rate of 0.005%, applicable to the amount of such deposit. However, FTT may be deducted as an ordinary expense in order to calculate the labour income tax.

The income tax on foreign source income is calculated on an annual basis and must be reported to the Peruvian Tax Authority by filing the Annual Income Tax Return, during the chronogram approved by SUNAT at the end of the fiscal year.

In accordance with article 51 of the Peruvian Income Tax Law, capital losses obtained by a Peruvian resident individual may be offset against other foreign source income, except the ones that are obtained in jurisdictions with low or no taxation (Tax Havens) which are not deductible in order to determine the net foreign source income. However, as from January 1, 2019, all jurisdictions which are members of the OECD, such as Luxembourg, are not being considered as Tax Havens for Peruvian tax purposes. Hence, foreign source losses cannot be offset against Peruvian source income or carried forward.

Peruvian resident individuals who hold Shares indirectly, through a foreign controlled entity, should consult their counsels and pose their queries with respect to the applicable tax matters regarding the International Tax Transparency Regime regulated under the Peruvian Income Tax Law.

By the other hand, investors who are not domiciled in or residents in Peru will not be subject to tax in Peru.

³³ To apply the Income Tax exemption, the transaction of the shares must comply with the following requirements: (a) the sale must be traded through the Lima Stock Exchange; (b) the traded security must have market presence (*presencia bursátil*), which is a liquidity threshold established in the Peruvian tax regulations (the securities that fulfil the market presence requirement are published by the BVL, View: https://www.bvl.com.pe/mercpresencia.html); and, (c) trade less than 10% of equity.

³⁴ The FTT is generally imposed on debits and credits in Peruvian bank accounts.

Appendix H – Additional Information for Asian-Pacific and Middle Eastern Investors

This Appendix provides you with important information about the Company and its Funds in respect of those jurisdictions set out below. This Appendix refers to and should, where appropriate (taking into account the important information below), be read in conjunction with the contents of the most up-to-date Prospectus and/or, as applicable, any jurisdiction specific supplement and/or prospectus. This Appendix by itself does not constitute a solicitation, recommendation or offer to buy or sell any shares in any Funds and any description of the Company, the Funds or shares are not intended to be complete and are qualified in their entirety by the Prospectus, subscription documents and all and any other constituent documents of the Company and/or the applicable Fund. If you are in any doubt about the contents of this Appendix and/or anything referred to in this Appendix then you should consult your financial adviser, bank, stockbroker or other relevant financial intermediary qualified to provide such information and advice.

Australia

The Prospectus is not a prospectus or product disclosure statement under the Corporations Act 2001 (Cth) (Corporations Act) and does not constitute a recommendation to acquire, an invitation to apply for, an offer to apply for or buy, an offer to arrange the issue or sale of, or an offer for issue or sale of, any securities in Australia, except as set out below. The Company has not authorised nor taken any action to prepare or lodge with the Australian Securities & Investments Commission an Australian law compliant prospectus or product disclosure statement.

The Prospectus may not be issued or distributed in Australia and the shares in the Funds may not be offered, issued, sold or distributed in Australia by the fund manager, or any other person, under the Prospectus, other than by way of or pursuant to an offer or invitation that does not need disclosure to investors under Part 6D.2 or Part 7 of the Corporations Act, whether by reason of the investor being a 'wholesale client' (as defined in section 761G of the Corporations Act and applicable regulations) or otherwise.

The Prospectus does not constitute or involve a recommendation to acquire, an offer or invitation for issue or sale, an offer or invitation to arrange the issue or sale, or an issue or sale, of shares to a 'retail client' (as defined in Section 761G of the Corporations Act and applicable regulations) in Australia.

Brunei

The Company does not have a capital market service licence under the Securities Market Order 2013 to provide investment advice to clients, or to undertake any investment business, in Brunei. By requesting or accepting financial services or financial product information from the Company, you agree, represent and warrant that you are engaging our services or in receipt of information wholly outside the jurisdiction of Brunei and, as applicable, subject to the foreign laws of the contract governing the terms of our engagement.

The Prospectus does not constitute an offer or an advertisement within the meaning of the Securities Market Order 2013 and shall not be distributed or circulated to any person in Brunei, except to a specific and selected class of investors only who are an accredited investor, an expert investor or an institutional investor as defined in the Securities Market Order, 2013 at their request so that they may consider an investment and subscription in the Funds. This Prospectus is not issued to the public or section of the public in Brunei.

The Autoriti Monetari Brunei Darussalam (the "Authority") is not responsible for reviewing or verifying any prospectus or other documents in connection with this collective investment scheme. The Authority has not approved the Prospectus or any other associated documents nor taken any steps to verify the information set out in the Prospectus and has no responsibility for it.

People's Republic of China

Neither the Prospectus nor this Document constitute a public offer of the Funds, whether by sale or subscription, in the People's Republic of China (excluding Hong Kong Special Administrative Region, Macau Special Administrative

Region and Taiwan) (the "PRC"). Shares in the Funds are not being offered or sold directly or indirectly in the PRC to or for the benefit of, legal or natural persons in the PRC.

Further, no legal or natural persons in the PRC may directly or indirectly purchase any shares in the Funds or any beneficial interest therein without obtaining all prior PRC governmental approvals that are required, whether statutorily or otherwise. Potential investors who are PRC nationals are responsible for obtaining the required approvals from all relevant government authorities in the PRC before purchasing the shares.

Neither the Prospectus nor this Document constitute any securities or investment advice in the PRC, if provided to citizens of the PRC in the PRC, or nationals in the PRC with permanent residence in the PRC, or to any corporation, partnership, or other entity incorporated or established in the PRC.

Persons who come into possession of the Prospectus are required by the issuer and its representatives to observe these restrictions.

Hong Kong

WARNING FOR HONG KONG INVESTORS

The contents of the attached fund offering document relating to the Unauthorized Funds (as defined below) have not been reviewed by any regulatory authority in Hong Kong.

You are advised to exercise caution in relation to the offer of Unauthorized Funds. If you are in any doubt about any of the contents of this document, you should obtain independent professional advice.

The contents of this document relating to Unauthorized Funds have not been delivered for registration to the Registrar of Companies in Hong Kong and have not been reviewed by any regulatory authority in Hong Kong. Accordingly: (i) the shares of the Unauthorised Funds may not be offered or sold in Hong Kong by means of any document other than to persons who are "professional investors" within the meaning of the Securities and Futures Ordinance (Cap. 571) of Hong Kong and the Securities and Futures (Professional Investor) Rules made thereunder or in other circumstances which do not result in the document being a "prospectus" within the meaning of the Companies Ordinance (Cap. 32) of Hong Kong or which do not constitute an offer to the public within the meaning of the Companies Ordinance; and (ii) no person may issue any invitation, advertisement or other document relating to the shares of the Unauthorised Funds whether in Hong Kong or elsewhere, which is directed at, or the contents of which are likely to be accessed or read by, the public in Hong Kong (except if permitted to do so under the securities laws of Hong Kong) other than with respect to the shares of the Unauthorised Funds which are or are intended to be disposed of only to persons outside Hong Kong or only to "professional investors" within the meaning of the Securities and Futures Ordinance and the Securities and Futures (Professional Investor) Rules made thereunder.

Unauthorized Funds:

The funds set out below have not been authorized by the Securities and Futures Commission of Hong Kong pursuant to section 104 of the Securities and Futures Ordinance:

Morgan Stanley Investment Funds American Resilience Fund

Morgan Stanley Investment Funds Calvert Sustainable Climate Aligned Fund³⁵

Morgan Stanley Investment Funds Calvert Sustainable Climate Transition Fund³⁵

Morgan Stanley Investment Funds Calvert Sustainable Developed Europe Equity Select Fund³⁵

Morgan Stanley Investment Funds Calvert Sustainable Developed Markets Equity Select Fund³⁵

³⁵ The Unauthorized Fund is not available for subscription at the date of this Prospectus. The Unauthorized Fund may be launched at the Directors' discretion, at which time, confirmation of the launch of the Unauthorized Fund will be made available at the registered office of the Company.

```
Morgan Stanley Investment Funds Calvert Sustainable Diversity, Equity and Inclusion Fund<sup>35</sup>
```

Morgan Stanley Investment Funds Calvert Sustainable Emerging Markets Equity Select Fund 35

Morgan Stanley Investment Funds Calvert Sustainable Global Equity Engagement Fund³⁵

Morgan Stanley Investment Funds Calvert Sustainable Global Green Bond Fund³⁵

Morgan Stanley Investment Funds Calvert Sustainable US Equity Select Fund³⁵

Morgan Stanley Investment Funds China A-shares Fund

Morgan Stanley Investment Funds China Equity Fund

Morgan Stanley Investment Funds Counterpoint Global Fund

Morgan Stanley Investment Funds Developing Opportunity Fund

Morgan Stanley Investment Funds Emerging Leaders Equity Fund

Morgan Stanley Investment Funds Emerging Markets Corporate Debt Fund

Morgan Stanley Investment Funds Emerging Markets Debt Opportunities Fund³⁵

Morgan Stanley Investment Funds Emerging Markets Fixed Income Opportunities Fund

Morgan Stanley Investment Funds Emerging Markets Local Income Fund³⁵

Morgan Stanley Investment Funds Euro Bond Fund

Morgan Stanley Investment Funds Euro Corporate Bond Fund

Morgan Stanley Investment Funds Euro Corporate Bond – Duration Hedged Fund

Morgan Stanley Investment Funds Euro Strategic Bond Fund

Morgan Stanley Investment Funds Morgan Stanley Investment Funds Europe Opportunity Fund

Morgan Stanley Investment Funds European Fixed Income Opportunities Fund

Morgan Stanley Investment Funds European High Yield Bond Fund

Morgan Stanley Investment Funds European Property Fund

Morgan Stanley Investment Funds Floating Rate ABS Fund

Morgan Stanley Investment Funds Frontier Markets Local Debt Fund³⁵

Morgan Stanley Investment Funds Global Asset Backed Securities Fund

Morgan Stanley Investment Funds Global Balanced Defensive Fund

Morgan Stanley Investment Funds Global Balanced Fund

Morgan Stanley Investment Funds Global Balanced Income Fund

Morgan Stanley Investment Funds Global Balanced Risk Control Fund of Funds

Morgan Stanley Investment Funds Global Balanced Sustainable Fund

Morgan Stanley Investment Funds Global Brands Equity Income Fund

Morgan Stanley Investment Funds Global Core Equity Fund

Morgan Stanley Investment Funds Global Credit Fund

Morgan Stanley Investment Funds Global Credit Opportunities Fund

Morgan Stanley Investment Funds Global Endurance Fund

Morgan Stanley Investment Funds Global Fixed Income Opportunities Fund

Morgan Stanley Investment Funds Global Focus Property Fund

Morgan Stanley Investment Funds Global High Yield Bond Fund

Morgan Stanley Investment Funds Global Insight Fund

Morgan Stanley Investment Funds Global Macro Fund³⁵

Morgan Stanley Investment Funds Global Multi-Asset Opportunities Fund

Morgan Stanley Investment Funds Global Permanence Fund

Morgan Stanley Investment Funds Global Property Fund

Morgan Stanley Investment Funds Global Quality Fund

Morgan Stanley Investment Funds Global Sustain Fund

Morgan Stanley Investment Funds Indian Equity Fund

Morgan Stanley Investment Funds International Equity (ex US) Fund

Morgan Stanley Investment Funds Japanese Equity Fund

Morgan Stanley Investment Funds Japanese Small Cap Equity Fund³⁵

Morgan Stanley Investment Funds Multi-Asset Risk Control Fund

Morgan Stanley Investment Funds Parametric Emerging Markets Fund³⁵

Morgan Stanley Investment Funds Parametric Global Defensive Equity Fund³⁵

Morgan Stanley Investment Funds Saudi Equity Fund

Morgan Stanley Investment Funds Short Duration US Government Income Fund³⁵

Morgan Stanley Investment Funds Short Maturity Euro Bond Fund

Morgan Stanley Investment Funds Short Maturity Euro Corporate Bond Fund

```
Morgan Stanley Investment Funds Sustainable Asia Equity Fund
```

Morgan Stanley Investment Funds Sustainable Euro Strategic Bond Fund

Morgan Stanley Investment Funds Sustainable European High Yield Bond Fund³⁵

Morgan Stanley Investment Funds Sustainable Global Fixed Income Opportunities Fund³⁵

Morgan Stanley Investment Funds Sustainable Global High Yield Bond Fund³⁵

Morgan Stanley Investment Funds Sustainable US High Yield Bond Fund

Morgan Stanley Investment Funds US Core Equity Fund

Morgan Stanley Investment Funds US Dollar Corporate Bond Fund

Morgan Stanley Investment Funds US Dollar High Yield Bond Fund

Morgan Stanley Investment Funds US Dollar Short Duration Bond Fund

Morgan Stanley Investment Funds US Dollar Short Duration High Yield Bond Fund

Morgan Stanley Investment Funds US Focus Property Fund

Morgan Stanley Investment Funds US Insight Fund

Morgan Stanley Investment Funds US Permanence Fund

Morgan Stanley Investment Funds US Value Fund³⁵

Morgan Stanley Investment Funds Vitality Fund³⁵

India

The shares to which the Prospectus relates are not being offered to the Indian public for sale or subscription. The shares are not registered and/or approved by the securities and exchange board of India, the reserve bank of India or any other governmental/ regulatory authority in India. The Prospectus is not and should not be deemed to be a 'prospectus' as defined under the provisions of the companies act, 2013 (18 of 2013) and the same shall not be filed with any regulatory authority in India.

Morgan Stanley does not guarantee or promise to return all or any portion of the money invested towards the shares by an investor and an investment in the shares is subject to applicable risks associated with an investment in the shares and shall not constitute a deposit within the meaning of the banning of unregulated deposits schemes act, 2019.

Pursuant to the foreign exchange management act, 1999 and the regulations issued there under, any investor resident in India may be required to obtain prior special permission of the reserve bank of India before making investments outside of India, including any investment in the Funds.

Morgan Stanley has neither obtained any approval from the reserve bank of India, the securities and exchange board of India or any other regulatory authority in India nor does it intend to do so.

Indonesia

The offering contained in the Prospectus does not constitute a public offering in Indonesia under Capital Markets Law No. 8 Year 1995. The Prospectus may not be distributed to the general public in Indonesia and the securities may not be offered or sold, directly or indirectly, in Indonesia or to Indonesian citizens wherever they are domiciled, or to Indonesian residents, in a manner which constitutes a public offering under the laws and regulations of Indonesia.

Malaysia

No action has been, or will be, taken to comply with Malaysian laws for making available, offering for subscription or purchase, or issuing any invitation to subscribe for or purchase or sale of the shares of the Funds in Malaysia or to persons in Malaysia as the shares are not intended by the issuer to be made available, or made the subject of any offer or invitation to subscribe or purchase, by persons in Malaysia. Neither the Prospectus nor any document or other material in connection with the shares should be distributed, caused to be distributed or circulated to persons in Malaysia. No person should make available or make any invitation or offer or invitation to sell or purchase the shares to persons in Malaysia.

Morgan Stanley Investment Funds Sustainable Euro Corporate Bond Fund

New Zealand

The Prospectus is not a product disclosure statement for the purposes of the Financial Markets Conduct Act 2013 (the FMCA) and does not contain all the information typically included in such offering documentation.

An offer of shares in the Funds does not constitute a "regulated offer" for the purposes of the FMCA and, accordingly, there is neither a product disclosure statement nor a register entry available in respect of the offer. Shares in the Funds may only be offered in New Zealand in accordance with the FMCA and the Financial Markets Conduct Regulations 2014.

Accordingly, the Funds have not been, and may not be, offered, issued or sold to any person in New Zealand other than:

- (1) to persons who are "wholesale investors" within the meaning of clause 3(2)(a), (c) or (d) of schedule 1 of the FMCA, being persons who fall within one or more of the following categories of "wholesale investor":
 - a) A person that is an "investment business" within the meaning of clause 37 of schedule 1 of the FMCA;
 - b) A person that is "large" within the meaning of clause 39 of schedule 1 of the FMCA; or
- c) A person that is a "government agency" within the meaning of clause 40 of schedule 1 of the FMCA. (2) in other circumstances where there is no contravention of the FMCA.

Philippines

Under Republic Act No. 8799, known as the Securities Regulation Code of the Philippines (the "Code"), and its implementing rules, securities, such as the Funds, are not permitted to be sold or offered for sale or distribution within the Philippines unless such securities are approved for registration by the Securities and Exchange Commission of the Philippines ("SEC") or are otherwise exempt securities or sold pursuant to an exempt transaction.

To the extent that the Code and its implementing rules apply to the offer and sale of the Funds (but without prejudice to the last paragraph of this section), the offer and sale are or will be made to "qualified buyers", as such term is defined Section 10.1(1) of the Code, its implementing rules and other applicable regulations of the SEC. Hence, such offer and sale of Funds qualify as a transaction exempt from the securities registration requirement pursuant to Section 10.1(1) of the Code. A confirmation of exemption from the SEC that the offer and sale of the Funds in the Philippines qualify as an exempt transaction under the Code will not be obtained.

THE SECURITIES BEING OFFERED OR SOLD HEREIN HAVE NOT BEEN REGISTERED WITH THE SECURITIES AND EXCHANGE COMMISSION UNDER THE SECURITIES REGULATION CODE OF THE PHILIPPINES. ANY OFFER OR SALE THEREOF IS SUBJECT TO REGISTRATION REQUIREMENTS UNDER THE CODE UNLESS SUCH OFFER OR SALE QUALIFIES AS AN EXEMPT TRANSACTION.

By purchase of the Funds, the investor will be deemed to acknowledge that the issue of, offer for subscription or purchase of, or invitation to subscribe for or purchase of such Fund was made outside the Philippines.

Singapore

The Funds are not authorised under section 286 of the Securities and Futures Act, Chapter 289 of Singapore (the "SFA") or recognised under section 287 of the SFA by the Monetary Authority of Singapore but some of the Funds may be notified as restricted schemes under section 305 of the SFA and marketed to prospective investors in Singapore, subject to certain conditions. Prospective investors resident in Singapore should refer to the separate "Information Memorandum for Investors in Singapore" (the "Information Memorandum") together with the Prospectus, which are available through your financial intermediary. The Information Memorandum contains a current list of the Funds that are notified as restricted schemes under section 305 of the SFA.

The remainder of the Funds which are not listed in the Information Memorandum have not been notified as restricted schemes under section 305 of the SFA, are not registered for marketing purposes in Singapore and accordingly, the Prospectus and any other document or material in connection with the offer or sale, or invitation for subscription or

purchase, of the shares of such Funds may not be circulated or distributed, nor may such shares be offered or sold, or be made the subject of an invitation for subscription or purchase, whether directly or indirectly, to persons in Singapore.

Taiwan

Some of the Funds are registered for sale to the public in Taiwan. Prospective investors resident in Taiwan should refer to the "Prospectus for Taiwanese Investors" available through Taiwanese licensed financial institutions.

There are other Funds that have been approved by CSSF but which are not currently registered for sale to the public in Taiwan. In respect of these Funds, Taiwanese residents should refer to the disclaimer below:

The shares may be made available:

- (i) to Taiwan resident investors outside Taiwan for purchase by such investors outside Taiwan;
- (ii) to the offshore banking units (as defined in the R.O.C. statute for offshore banking operations) of Taiwan banks, the offshore securities units (as defined in the R.O.C. statute for offshore banking operations) of Taiwan securities firms or the offshore insurance units (as defined in the R.O.C. statute for offshore banking operations) of Taiwan insurance companies purchasing the shares either for their proprietary account or for the accounts of their non-Taiwan clients; or
- (iii) to all Taiwan resident investors through licensed financial institutions after registration of the relevant Fund pursuant to the R.O.C. Rules Governing Offshore Funds.

Each subscriber or purchaser of the shares must seek professional advice as to whether he/she/it is qualified to subscribe to or purchase the shares. Purchasers/subscribers may be restricted or prohibited from re-selling the shares.

Thailand

The Funds have not been granted permission by the Securities and Exchange Commission (SEC) of Thailand to be publicly offered in Thailand. No interests in the Funds may be advertised or offered for sale to the public in Thailand or marketed through any means of communication to the general public in Thailand.

All the Funds' materials have been prepared strictly for informational purposes only and the contents of these documents have not been reviewed by the SEC. The contents contained in these materials should not be construed as a public offer of the Funds in Thailand.

The Prospectus and this Document are distributed on a confidential basis to the person to whom it is addressed. The Prospectus and this Document may not be reproduced in any form or transmitted to any person other than the person to whom it is addressed. Transmission of these documents to the general public in Thailand shall not constitute solicitation in Thailand by Morgan Stanley or any other Morgan Stanley entity or any of its or their representatives or agents to invest in the Funds.

United Arab Emirates

FOR UNITED ARAB EMIRATES (EXCLUDING DUBAI INTERNATIONAL FINANCIAL CENTRE AND ABU DHABI GLOBAL MARKET) RESIDENTS ONLY

The Prospectus does not constitute, and is not intended to constitute, a public offer of securities in the United Arab Emirates ("UAE") and accordingly should not be construed as such. Unless neither the provisions of the SCA Board of Directors' Chairman Decision No. 9/R.M of 2016 (concerning the regulations on mutual funds) nor the SCA Board of Directors' Chairman Decision No. 3/R.M of 2017 (concerning the marketing, distribution, advertisement, publication or provision of any data, information or advertising materials relating to a Financial Product in any way

or by any means) do not apply, Shares in the Funds are only being offered to a limited number of exempt investors in the UAE who fall under one of the following categories of Qualified Investors:

- (1) an investor which is able to manage its investments on its own (unless such person does not wish to be classified as a Qualified Investor), namely:
 - (a) the federal government, local governments and governmental entities, institutions and authorities, or companies wholly-owned by any such entities;
 - (b) foreign governments, their respective entities, institutions and authorities or companies wholly owned by any such entities;
 - (c) international entities and organisations;
 - (d) entities licensed by the Securities and Commodities Authority (the "SCA") or a counterpart authority (i.e. a regulatory authority that is an ordinary or associate member of the International Organisation of Securities Commissions) or
 - (e) any legal person that meets, as at the date of its most recent financial statements, at least two of the following conditions:
 - (i) it has a total assets or balance sheet of AED 75 million;
 - (ii) it has a net annual turnover of AED 150 million;
 - (iii) it has total equity or paid-up capital of AED 7 million; or
- (2) any natural person licensed by the SCA or a counterpart authority to carry out any of the functions related to financial activities or services, (each a "Qualified Investor").

Shares in the Funds have not been approved by or licensed or registered with the UAE Central Bank, the SCA, the Dubai Financial Services Authority, the Financial Services Regulatory Authority or any other relevant licensing authorities or governmental agencies in the UAE (the "Authorities"). The Authorities assume no liability for any investment that the named addressee makes as a Qualified Investor. The Prospectus is for the use of the named addressee only and should not be given or shown to any other person (other than employees, agents or consultants in connection with the addressee's consideration thereof).

FOR UNITED ARAB EMIRATES (EXCLUDING DUBAI INTERNATIONAL FINANCIAL CENTRE AND ABU DHABI GLOBAL MARKET) RESIDENTS ONLY

The Prospectus does not constitute, and is not intended to constitute, a public offer of securities in the United Arab Emirates ("UAE") and accordingly should not be construed as such. Unless neither the provisions of the SCA Board of Directors' Chairman Decision No. 9/R.M of 2016 (concerning the regulations on mutual funds) nor the SCA Board of Directors' Chairman Decision No. 3/R.M of 2017 (concerning the marketing, distribution, advertisement, publication or provision of any data, information or advertising materials relating to a Financial Product in any way or by any means) do not apply, Shares in the Funds are only being offered to a limited number of investors in the UAE who (a) are willing and able to conduct an independent investigation of the risks involved in an investment in such shares, and (b) upon their specific request. Shares in the Funds have not been approved by or licensed or registered with the UAE Central Bank, the Securities and Commodities Authority or any other relevant licensing authorities or governmental agencies in the UAE.

The Prospectus is for the use of the named addressee only, who has specifically requested it without a promotion effected by Morgan Stanley, any other Morgan Stanley entity, their promoters or the distributors of their shares, and should not be given or shown to any other person (other than employees, agents or consultants in connection with the addressee's consideration thereof). No transaction will be concluded in the UAE and any enquiries regarding the shares should be made to your financial intermediary.

Appendix I – Data Protection Notification

- a) In accordance with the data protection law applicable to the Grand-Duchy of Luxembourg including the Law of 2 August 2002 on the protection of persons with regard to the processing of personal data (as may be amended from time to time), the General Data Protection Regulation (EU/2016/679) and all applicable laws (together, the "Data Protection Laws"), the Company, acting as "data controller" within the meaning of the Data Protection Laws, hereby informs prospective investors and holders of shares in the Company (together, "Shareholders", and each a "Shareholder") that personal data provided by each Shareholder to the Company ("Personal Data", as defined in paragraph (b), below) may be collected, recorded, stored, adapted, transferred or otherwise processed, by electronic means or otherwise, for the following purposes (each a "Processing Purpose"):
 - to enable and process the subscription and redemption of Shares in the Company by investors, including (without limitation) the facilitation and processing of payments by and to the Company (including the payment of subscription monies and redemption proceeds, the payment of fees by and to Shareholders and the payment of distributions on Shares), and generally to enable and give effect to the participation of investors in the Company;
 - 2) to enable an account to be maintained of all payments referenced in sub-paragraph (1), above;
 - 3) to enable the maintaining of a register of Shareholders in accordance with applicable laws;
 - 4) to carry out or to facilitate the carrying out with respect to Shareholders of credit, money laundering, due diligence and conflict checks, for the purposes of fraud, money-laundering, financial crime prevention and tax identification laws (including FATCA and CRS and applicable anti-money laundering laws), and generally to enable the Company to comply with its legal obligations arising in connection therewith;
 - 5) to enable the Company to perform controls in respect of late trading and market timing practices;
 - 6) to facilitate the provision to the Company of services by the service providers referenced in this Prospectus, including (without limitation) the authorisation or confirmation of billing transactions and payments by and to the Company;
 - 7) to facilitate the operational support and development necessary to the Company's investment objectives and strategies with respect to its Sub-Funds, including (without limitation) the Company's risk management processes, and the evaluation of services provided to the Company by third-party service providers;
 - 8) in relation to any litigation, disputes or contentious matter in which the Company is involved;
 - 9) to comply with legal and regulatory obligations, (including any legal or regulatory guidance, codes or opinions), applicable to the Company anywhere in the world:
 - 10) to comply with legal and regulatory requests made to the Company anywhere in the world;
 - 11) to facilitate reporting, including (without limitation) transaction reporting to, and audits by, national and international regulatory, enforcement or exchange bodies, and tax authorities (including the Luxembourg Tax Authority) and the compliance by the Company with court orders associated therewith;
 - 12) for the Monitoring Purposes defined and specified in Section (e) below; and
 - 13) for direct marketing purposes specified in Section (g) below.

The Company may not collect Personal Data without a valid legal ground. Accordingly, the Company will only process and use Personal Data:

- a. if necessary to enter into, to execute or to carry out a contract with each Shareholder for the services or products they required by the Shareholder (as described in Processing Purposes 1 to 3 inclusive, above);
- b. if necessary for the Company's legitimate interests, provided in each case that such interests are not overridden by the privacy interests of impacted individuals. The Company's legitimate interests are described in the Processing Purposes 1 to 12 inclusive, above;
- c. to exercise and defend the Company's legal rights anywhere in the world as described in Processing Purpose 8 above; and
- d. if necessary to comply with legal obligations, (including any legal or regulatory guidance, codes or opinions), applicable to the Company anywhere in the world as described in Processing Purposes 4, 9 and 10 above.
- b) "Personal Data" includes data that is personal to a Shareholder (whether a Shareholder is a natural or a legal person) and which the Company obtains directly from a Shareholder and/or indirectly from a data processor such as personal details (including, at a minimum, a Shareholder's name, legal organization, country of residence, address and contact details) and financial account information. Some of this information will be publicly accessible.

Under certain conditions set out under the Data Protection Laws, a Shareholder shall have the right to:

- (i) access to his/her/its Personal Data;
- (ii) to correct or amend his/her/its Personal Data when such Personal Data is inaccurate or incomplete;
- (iii) to object to the processing of his/her/its Personal Data;
- (iv) to refuse at his/her/its own discretion to provide his/her/its Personal Data to the Company; (v) to request the erasure of his/her/its Personal Data; and
- (vi) to request the portability of his/her/its Personal Data in accordance with the Data Protection Laws.

Shareholders should note in particular that a refusal to provide Personal Data to the Company may result in the Company being required to reject his/her/its application for Shares in the Company.

Shareholders may exercise these rights by contacting the Company at dataprotectionoffice@morganstanley.com. In addition to exercising these rights, Shareholders have also a right to lodge a complaint in connection with matters concerning the processing and protection of Personal Data with the Company at dataprotectionoffice@morganstanley.com, without prejudice to their ability to submit a complaint to the National Commission for Data Protection in Luxembourg (the "CNPD").

c) For any Processing Purpose, the Company will delegate the processing of Personal Data, in accordance with the Data Protection Laws, to other parties, including the Management Company, the Administrator and Paying Agent, the Domiciliary Agent, the Registrar and Transfer Agent, and the Depositary, as well as other parties such as settlement agents, foreign banks or exchange or clearing houses, credit reference, fraud prevention and other similar agencies, and other financial institutions, together with parties to which the Company and/or the Management Company may assign or novate Personal Data (each a "Data Processor", and together the "Data Processors").

A Data Processor may, subject to the approval of the Company, sub-delegate the processing of Personal Data (and, pursuant to such sub-delegation, the transfer thereof) to its parent company or organization, affiliates, branch offices or third party agents (together, the "Delegates").

Data Processors and Delegates may be located in countries outside of the EEA (which may include Malaysia, India, United States of America and Hong Kong), where data protection laws may not provide an adequate level of protection. In such cases the Data Processor, with the supervision of the Company, will ensure (i)

that it has put in place appropriate data transfer mechanisms with the Company, and (ii) if applicable, that the Delegate has put in place appropriate data transfer mechanisms, in each case such as European Commission Standard Contractual Clauses. Shareholders can obtain a copy of the relevant data transfer mechanism that the Company has put in place by contacting the Company at dataprotectionoffice@morganstanley.com.

The Company will disclose Personal Data to the Luxembourg tax authority, which in turn, acting as data controller, may disclose that Personal Data to foreign tax authorities.

- d) In compliance with the Data Protection Laws, the Company will retain Personal Data in an identifiable form in accordance with the Company's information management policy which establishes general standards and procedures regarding the retention, handling and disposal of Personal Data. Personal Data shall not be retained for longer than is necessary with regard to the Processing Purposes, subject to any limitation periods imposed by law. Upon request, the Company will provide a Shareholder with more information on the exact retention periods applying to its Personal Data. The retention period may be extended in the sole discretion of the Company if the Company is required to preserve Personal Data in connection with litigation, regulatory investigations and legal proceedings.
- e) To the extent permitted by the Data Protection Laws, the Company and the Management Company (acting as "data controller" within the meaning of the Data Protection Laws) will access, review, disclose, intercept, monitor and record (together, "Monitoring") (i) verbal and electronic messaging and communications (for example, and without limitation, telephone, sms, instant message, email, Bloomberg and any other electronic or recordable communications) with a Shareholder or Shareholders' agent (together, "Communications"), and (ii) a Shareholder's use of technology owned, provided or made accessible by the Company and the Management Company, including (without limitation) systems that facilitate Communications with Shareholders, information processing, transmission, storage and access, including remote access (together, "Systems").

The Company and the Management Company will subject Communications and Systems to Monitoring only for the following purposes (together, "Monitoring Purposes"):

- 1) to establish the existence of facts (e.g., keeping records of transactions);
- 2) to ascertain compliance with regulatory or self-regulatory practices or procedures which are applicable to the Company and/or the Management Company;
- 3) to ascertain or demonstrate standards which are achieved or ought to be achieved by persons using Systems, including compliance with any terms of use associated use of Systems;
- 4) to prevent, detect or investigate crime, money laundering, fraud, financial crime and/or other breaches of applicable law;
- 5) to comply with applicable laws and regulations, any material contract and any applicable policies and procedures;
- 6) to safeguard against the loss, theft, unauthorised and unlawful collection, use, disclosure, destruction or other processing or misuse of confidential and proprietary information;
- 7) to prevent, detect or investigate unauthorised use of Systems and/or data (e.g., Monitoring to ensure compliance with the policies and procedures of the Company and/or the Management Company, including without limitation those relating to information security and cyber security);
- 8) to ensure the effective operation of Systems (including telephones, email and internet) systems;
- 9) for support and administration purposes;
- 10) to assist with investigations, complaints, regulatory requests, litigation, arbitration, mediation or requests from individuals; and

in particular, in the course of the operational support and development of the business of the Company and/or the Management Company, such as to evaluate the quality of customer service, efficiency, cost and risk management purposes.

Monitoring will be conducted by the Company and/or the Management Company using various methods, including: (i) the use of "intelligent" automated monitoring tools; (ii) IT filtering tools which randomly review Systems; (iii) random monitoring of Systems, e.g. by authorised supervisors randomly joining ongoing telephone calls on sales and trading floors; (iv) specific monitoring of Systems, e.g. in relation to investigations, regulatory requests, subject access requests, litigation, arbitration or mediation; (v) data tracking, aggregation and analysis tools that collect data from various sources to extrapolate linkages and/or detect behavioural patterns, interactions or preferences for analysis (including predictive analysis); and/or (vi) using other similar Monitoring technology that may become available from time to time.

The Company and/or the Management Company also use cookies and similar technologies to collect information about Shareholders as part of and/or in connection with services provided by them or in connection with any System owned or provided by them. By accessing or using services or a System, a Shareholder signifies his/her/its understanding that the Company and/or the Management Company will use such cookies and similar technologies as detailed in the Company's privacy policy, and that if the Shareholder chooses to reject such cookies, some or all parts of the services or the relevant System may not function properly or may not be accessible. To find out more about how the Company and/or the Management Company uses cookies and similar technologies, how the Company and/or the Management Company processes the information obtained through cookies and how a Shareholder may reject cookies, see the Company's privacy policy at www.morganstanley.com/privacy_pledge.

- f) Any documentation or records relating to the Monitoring of Systems shall be prima facie evidence of any orders or communications that have been subjected to Monitoring, and Shareholders agree that such records shall be admissible as such in any legal proceedings. Furthermore, Shareholders confirm that they will not use, file, or cite as a reason for objecting to the admission of such records as evidence in any legal proceedings either that the records are not originals, or are not in writing or are documents produced by a computer. The Company and/or the Management Company will retain such records in accordance with its operational procedures which may change from time to time in its absolute discretion, however, such records shall not be held by the Company for longer than is necessary with regard to the Monitoring Purposes, subject to any limitation periods imposed by law. Shareholders are hereby informed that this record keeping should not be deemed to be a substitute for their own keeping of adequate records in accordance with any applicable rules or regulations to which they are subject.
- If there are any products or services that the Company and/or the Management Company believes may be of particular interest to a Shareholder, whether provided or sponsored by the Company and/or the Management Company or their respective affiliates, or by third party investment services providers (for example, a fund manager or insurance service provider not affiliated with the Company, the Management Company or their respective affiliates), the Company, the Management Company or their respective affiliates will contact that Shareholder by means which may include mail, email, sms and telephone), including outside standard working hours. When required by the Data Protection Laws, a Shareholder's prior consent will be requested before its Personal Data is used to make or facilitate direct marketing of this nature. If a Shareholder does not wish the Company, the Management Company or their respective affiliates to use its Personal Data in this way, or does not wish to provide Personal Data for such direct marketing purposes, the Shareholder may notify the Company, the Management Company or their respective affiliates at any time in accordance with section (b) above or as directed in any marketing materials that may be received by Shareholders. In this respect, each Shareholder has a right to object to the use of his/her/its Personal Data for marketing purposes. This objection must be made in writing by letter addressed to the Company, the Management Company or their respective affiliates at European Bank and Business Centre, 6B route de Trèves, L-2633 Senningerberg, Grand Duchy of Luxembourg, or at cslux@morganstanley.com.
- h) Before providing the Company and/or the Management Company with access to, or permitting any access to, or permitting the processing of, Personal Data which contains any data regarding an individual in

connection with this Prospectus, a Shareholder should ensure that: (i) the individual understands that the Shareholder will be providing their Personal Data to the Company, the Management Company or their respective affiliates; (ii) the individual has been provided with the information set out herein regarding the collection, use, processing, disclosure and overseas transfer of Personal Data, the use of Personal Data for direct marketing purposes, and the possibility of monitoring or recording of their or their agent's communications by the Company, the Management Company or their respective affiliates (in each case if permitted by the Data Protection Laws); (iii) if required, the individual has provided their consent to the processing by the Company, the Management Company or their respective affiliates of their Personal Data or that another legal basis to process Personal Data is satisfied; and (iv) the individual is aware of their data protection rights and how to exercise these.

Appendix J – List of sub-custodians used by the Depositary and sub-delegates that may arise from any delegation of the Depositary's duties

This list is accurate as at the date of the Prospectus. The latest version of such list may be obtained by investors from the Company upon request.

CASH CORRESPONDENT BANK MARKET SUBCUSTODIAN ARGENTINA HSBC Bank Argentina S.A. HSBC Bank Argentina S.A. Bouchard 557, 18th Floor **Buenos Aires** Buenos Aires C1106ABJ **ARGENTINA AUSTRALIA** JPMorgan Chase Bank N.A.** Australia and New Zealand Banking Group Level 31, 101 Collins Street Melbourne 3000 Melbourne **AUSTRALIA** JPMorgan Chase Bank N.A., Sydney Branch (for clients utilizing J.P. Morgan's domestic AUD solution))** Sydney **AUSTRIA** UniCredit Bank Austria AG J.P. Morgan AG ** Julius Tandler Platz - 3, Frankfurt Vienna A-1090 **AUSTRIA BAHRAIN HSBC** Bank Middle East Limited HSBC Bank Middle East Limited Road No 2832 Al Seef Al Seef 428 **BAHRAIN BANGLADESH** Standard Chartered Bank Standard Chartered Bank Portlink Tower, Level-6, 67 Gulshan Avenue, Dhaka Gulshan Dhaka 1212 **BANGLADESH BELGIUM** J.P. Morgan SE, Luxembourg Branch (for clients J.P. Morgan AG** contracting with this entity and JPMorgan Frankfurt am Main Chase Bank, N.A.)** European Bank & Business Centre, 6, route de Senningerberg L-2633 LUXEMBOURG BNP Paribas Securities Services S.C.A. (for clients contracting with J.P. Morgan (Suisse) SA and for all Belgian Bonds settling in the National Bank of Belgium (NBB) and Physical Securities held by clients) Central Plaza Building, Rue de Loxum, 25, 7th Floor Brussels 1000 **BELGIUM** J.P. Morgan Bank (Ireland) PLC (for clients contracting with this entity)** 200 Capital Dock, 79 Sir John Rogerson's Quay Dublin D02 RK57 **IRELAND**

BERMUDA

HSBC Bank Bermuda Limited

37 Front Street

Hamilton HM 11 BERMUDA **HSBC** Bank Bermuda Limited

Hamilton

BOTSWANA

Standard Chartered Bank Botswana Limited 5th Floor, Standard House, P.O. Box 496, Queens Road, The Mall Gaborone BOTSWANA Standard Chartered Bank Botswana Limited Gaborone

BRAZIL

J.P. Morgan S.A. DTVM** Av. Brigadeiro Faria Lima, 3729, Floor 06 Sao Paulo SP 04538 905 BRAZIL J.P. Morgan S.A. DTVM** Sao Paulo

BULGARIA

Citibank Europe plc Serdika Offices, 10th Floor, 48 Sitnyakovo Blvd Sofia 1505 BULGARIA ING Bank N.V. Sofia

CANADA

CIBC Mellon Trust Company (Note: Clients please refer to your issued settlement instructions) 1 York Street, Suite 900 Toronto Ontario M5J 0B6 CANADA Royal Bank of Canada Toronto

Royal Bank of Canada (Note: Clients please refer to your issued settlement instructions) 155 Wellington Street West Toronto M5V 3L3 CANADA

CHILE

Banco Santander Chile Bandera 140, Piso 4 Santiago CHILE Banco Santander Chile Santiago

CHINA A-SHARE

JPMorgan Chase Bank (China) Company Limited (Note: Clients please refer to your issued settlement instructions)*** 41st floor, Park Place, No. 1601, West Nanjing Road, Jingan District SHANGHAI

The People's Republic of China
HSBC Bank (China) Company Limited (Note:
Clients please refer to your issued settlement
instructions)
33/F, HSBC Building, Shanghai IFC, 8 Century Avenue,
Pudong
Shanghai 200120
THE PEOPLE'S REPUBLIC OF CHINA

JPMorgan Chase Bank (China) Company Limited (Note: Clients please refer to your issued settlement instructions)** Shanghai

HSBC Bank (China) Company Limited (Note: Clients please refer to your issued settlement instructions)
Shanghai

CHINA B-SHARE

HSBC Bank (China) Company Limited 33/F, HSBC Building, Shanghai IFC, 8 Century Avenue, Pudong Shanghai 200120 THE PEOPLE'S REPUBLIC OF CHINA

JPMorgan Chase Bank, N.A. ** New York

JPMorgan Chase Bank, N.A.** Hong Kong **CHINA CONNECT**

JPMorgan Chase Bank, N.A.**
48th Floor, One Island East, 18 Westlands
Road, Quarry Bay
Hong Kong Island
HONG KONG

JPMorgan Chase Bank, N.A.** Hong Kong

COLOMBIA

Cititrust Colombia S.A. Carrera 9 A #99-02, 3rd Floor Bogota COLOMBIA Cititrust Colombia S.A. Bogota

COSTA RICA

Banco BCT S.A. 150 Metros Norte de la Catedral Metropolitana, Edificio BCT San Jose COSTA RICA Banco BCT S.A. San Jose

CROATIA

Privredna banka Zagreb d.d. Radnicka cesta 50 Zagreb 10000 CROATIA Zagrebacka banka d.d. Zagreb

CYPRUS

HSBC France Athens Branch 109-111, Messogion Ave. Athens 11526 GREECE J.P. Morgan AG ** Frankfurt am Main

CZECH REPUBLIC

UniCredit Bank Czech Republic and Slovakia, a.s. BB Centrum - FILADELFIE, Zeletavska 1525-1, Prague 1 Prague 140 92 CZECH REPUBLIC eskoslovenská obchodní banka a.s. Prague

DENMARK

Nordea Bank Abp Christiansbro, Strandgade 3, P.O. Box 850 Copenhagen DK-0900 DENMARK Nordea Bank Abp Copenhagen

EGYPT

Citibank N.A., Egypt Boomerang Building, Plot 46, Zone J, 1st district, 5th Settlement, New Cairo 11511 EGYPT Citibank N.A., Egypt New Cairo

ESTONIA

Swedbank AS Liivalaia 8 Tallinn 15040 ESTONIA J.P. Morgan AG** Frankfurt am Main

FINLAND

Nordea Bank Abp Satamaradankatu 5 Helsinki FIN-00020 Nordea FINLAND J.P. Morgan AG** Frankfurt am Main

FRANCE

BNP Paribas Securities Services S.C.A. (for clients contracting with J.P. Morgan (Suisse) SA and for Physical Securities and Ordre de Mouvement (ODMs) held by clients) 3, Rue d'Antin Paris 75002 FRANCE J.P. Morgan AG** Frankfurt am Main

J.P. Morgan SE, Luxembourg Branch (for clients contracting with this entity and JPMorgan Chase Bank, N.A.)**
European Bank & Business Centre, 6, route de Treves
Senningerberg L-2633
LUXEMBOURG

J.P. Morgan Bank (Ireland) PLC (for clients contracting with this entity)**
200 Capital Dock, 79 Sir John Rogerson's Quay Dublin D02 RK57
IRELAND

GERMANY

J.P. Morgan AG (for domestic German custody clients only)**
Taunustor 1 (TaunusTurm)
Frankfurt am Main 60310
GERMANY

Deutsche Bank AG Alfred-Herrhausen-Allee 16-24 Eschborn D-65760 GERMANY J.P. Morgan AG** Frankfurt am Main

GHANA

Standard Chartered Bank Ghana Limited Accra High Street, P.O. Box 768 Accra GHANA Standard Chartered Bank Ghana Limited Accra

GREECE

HSBC France Athens Branch 109-111, Messogion Ave. Athens 11526 GREECE J.P. Morgan AG** Frankfurt am Main

HONG KONG

JPMorgan Chase Bank, N.A.**
48th Floor, One Island East, 18 Westlands
Road, Quarry Bay
Hong Kong Island
HONG KONG

JPMorgan Chase Bank, N.A.** Hong Kong

HUNGARY

Deutsche Bank AG Hold utca 27 Budapest H-1054 HUNGARY ING Bank N.V. Budapest

ICELAND

Islandsbanki hf. Kirkjusandur 2 Reykjavik IS-155 ICELAND Islandsbanki hf. Reykjavik INDIA

JPMorgan Chase Bank, N.A.**
6th Floor, Paradigm B Wing, Mindspace, Malad (West)
Mumbai 400 064
INDIA

JPMorgan Chase Bank, N.A. ** Mumbai

INDONESIA

PT Bank HSBC Indonesia WTC 3 Building - 8th floor Jl. Jenderal Sudirman Kav. 29-31 Jakarta 12920 INDONESIA PT Bank HSBC Indonesia Jakarta

IRELAND

JPMorgan Chase Bank, N.A.** 25 Bank Street Canary Wharf London E14 5JP UNITED KINGDOM J.P. Morgan AG** Frankfurt am Main

ISRAEL

Bank Leumi le-Israel B.M. 35, Yehuda Halevi Street Tel Aviv 65136 ISRAEL Bank Leumi le-Israel B.M. Tel Aviv

ITALY

J.P. Morgan Bank (Ireland) PLC (for clients contracting with this entity. Clients contracting with J.P. Morgan SE, Luxembourg Branch please refer to your issued settlement instructions)** 200 Capital Dock, 79 Sir John Rogerson's Quay Dublin D02 RK57 IRELAND

J.P. Morgan AG** Frankfurt am Main

J.P. Morgan Bank (Ireland) PLC (for clients contracting with this entity. Clients contracting with J.P. Morgan SE, Luxembourg Branch please refer to your issued settlement instructions)** 200 Capital Dock, 79 Sir John Rogerson's Quay Dublin D02 RK57 IRELAND

JAPAN

Mizuho Bank Ltd. (Note: Clients please refer to your issued settlement instructions) 2-15-1, Konan, Minato-ku Tokyo 108-6009 JAPAN JPMorgan Chase Bank, N.A** Tokyo

MUFG Bank, Ltd. (Note: Clients please refer to your issued settlement instructions) 1-3-2 Nihombashi Hongoku-cho, Chuo-ku Tokyo 103-0021 JAPAN

JORDAN

Standard Chartered Bank Shmeissani Branch, Al-Thaqafa Street, Building #2 P.O. Box 926190 Amman JORDAN

Standard Chartered Bank Amman KAZAKHSTAN

JSC Citibank Kazakhstan Park Palace, Building A, Floor 2, 41 Kazybek Bi Almaty 050010 KAZAKHSTAN

Almaty

KENYA

Standard Chartered Bank Kenya Limited Chiromo, 48 Westlands Road Nairobi 00100 KENYA Standard Chartered Bank Kenya Limited

Nairobi

KUWAIT

HSBC Bank Middle East Limited Kuwait City, Sharq Area Safat 13017 KUWAIT **HSBC** Bank Middle East Limited

JSC Citibank Kazakhstan

Safat

LATVIA

Swedbank AS Balast dambis 1a Riga LV-1048 LATVIA J.P. Morgan AG**
Frankfurt am Main

LITHUANIA

AB SEB Bankas 12 Gedimino pr. Vilnius LT 2600 LITHUANIA J.P. Morgan AG** Frankfurt am Main

LUXEMBOURG

BNP Paribas Securities Services S.C.A. 60 Avenue John F. Kennedy Luxembourg L-1855 LUXEMBOURG J.P. Morgan AG** Frankfurt am Main

MALAWI

Standard Bank Limited 1st Floor Kaomba House, Cnr Glyn Jones Road & Victoria Avenue Blantyre MALAWI Standard Bank Limited

Blantyre

MALAYSIA

HSBC Bank Malaysia Berhad 2 Leboh Ampang, 12th Floor, South Tower Kuala Lumpur 50100 MALAYSIA HSBC Bank Malaysia Berhad

Kuala Lumpur

MAURITIUS

The Hongkong and Shanghai Banking Corporation Limited HSBC Centre, 18 Cybercity Ebene MAURITIUS The Hongkong and Shanghai Banking

Corporation Limited

Ebene

MEXICO

Banco Nacional de Mexico S.A. Act. Roberto Medellin No. 800 3er Piso Norte Colonia Santa Fe Mexico, D.F. 1210 MEXICO

Banco Santander (Mexico) S.A. Ciudad de México, C.P.

MOROCCO

Société Générale Marocaine de Banques 55 Boulevard Abdelmoumen Casablanca 20100 MOROCCO Attijariwafa Bank S.A. Casablanca

NAMIBIA

Standard Bank Namibia Limited 2nd Floor, Town Square Building, Corner of Werner List and Post Street Mall, P.O. Box 3327 Windhoek NAMIBIA The Standard Bank of South Africa Limited Johannesburg

NETHERLANDS

J.P. Morgan SE, Luxembourg Branch (for clients contracting with this entity and JPMorgan Chase Bank, N.A.)
European Bank & Business Centre, 6, route de Treves
Senningerberg L-2633
LUXEMBOURG

BNP Paribas Securities Services S.C.A. (for clients contracting with J.P. Morgan (Suisse) SA) Herengracht 595 Amsterdam 1017 CE NETHERLANDS

J.P. Morgan Bank (Ireland) PLC (for clients contracting with this entity)**
200 Capital Dock, 79 Sir John Rogerson's Quay Dublin D02 RK57
IRELAND

J.P. Morgan AG** Frankfurt am Main

NEW ZEALAND

JPMorgan Chase Bank, N.A.** Level 13, 2 Hunter Street Wellington 6011 NEW ZEALAND JPMorgan Chase Bank, N.A. New Zealand Branch (for clients utilizing J.P. Morgan's domestic NZD solution)** Wellington

Westpac Banking Corporation Wellington

NIGERIA

Stanbic IBTC Bank Plc Plot 1712, Idejo Street Victoria Island Lagos NIGERIA Stanbic IBTC Bank Plc Lagos

NORWAY

Nordea Bank Abp Essendropsgate 7, P.O. Box 1166 Oslo NO-0107 NORWAY Nordea Bank Abp Oslo

OMAN

HSBC Bank Oman S.A.O.G. 2nd Floor Al Khuwair P.O. Box 1727 Seeb PC 111 OMAN HSBC Bank Oman S.A.O.G. Seeb

PAKISTAN

Standard Chartered Bank (Pakistan) Limited P.O. Box 4896, Ismail Ibrahim Chundrigar Road Karachi 74000 PAKISTAN Standard Chartered Bank (Pakistan) Limited Karachi

PERU

Citibank del Perú S.A. Canaval y Moreryra 480 Piso 3, San Isidro San Isidro, L-27 L-27 Lima, PERU Banco de Crédito del Perú Lima 012

PHILIPPINES

The Hongkong and Shanghai Banking Corporation Limited 7/F HSBC Centre, 3058 Fifth Avenue West, Bonifacio Global City Taguig City 1634 PHILIPPINES The Hongkong and Shanghai Banking Corporation Limited Taguig City

POLAND

Bank Handlowy w. Warszawie S.A. ul. Senatorska 16 Warsaw 00-923 POLAND mBank S.A. Warsaw

PORTUGAL

BNP Paribas Securities Services S.C.A. Avenida D.João II, Lote 1.18.01, Bloco B, 7º andar Lisbon 1998-028 PORTUGAL J.P. Morgan AG** Frankfurt am Main

QATAR

HSBC Bank Middle East Limited 2nd Floor, Ali Bin Ali Tower, Building 150, Airport Road Doha QATAR The Commercial Bank (P.Q.S.C.) Doha

ROMANIA

Citibank Europe plc 145 Calea Victoriei, 1st District Bucharest 10072 ROMANIA ING Bank N.V. Bucharest

RUSSIA

Commercial Bank "J.P. Morgan Bank International" (Limited Liability Company)** 10, Butyrsky Val, White Square Business Centre, Floor 12 Moscow 125047 RUSSIA Public Joint-Stock Company Rosbank Moscow JPMorgan Chase Bank, N.A.** New York

SAUDI ARABIA

J.P. Morgan Saudi Arabia Company (Note: Clients please refer to your issued settlement instructions)** Al Faisaliah Tower, Level 8, P.O. Box 51907 Riyadh 11553 SAUDI ARABIA

HSBC Saudi Arabia (Note: Clients please refer to your issued settlement instructions) 2/F HSBC Building, 7267 Olaya Street North, Al Murooj Riyadh 12283-2255 SAUDI ARABIA The Saudi British Bank Riyadh

SERBIA

Unicredit Bank Srbija a.d. Rajiceva 27-29 Belgrade 11000 SERBIA Unicredit Bank Srbija a.d. Belgrade

SINGAPORE

DBS Bank Ltd 10 Toh Guan Road, DBS Asia Gateway, Level 04-11 (4B) 608838 SINGAPORE Oversea-Chinese Banking Corporation Singapore

SLOVAK REPUBLIC

UniCredit Bank Czech Republic and Slovakia, a.s. Sancova 1/A Bratislava SK-813 33 SLOVAK REPUBLIC J.P. Morgan AG** Frankfurt am Main

SLOVENIA

UniCredit Banka Slovenija d.d. Smartinska 140 Ljubljana SI-1000 SLOVENIA J.P. Morgan AG** Frankfurt am Main

SOUTH AFRICA

FirstRand Bank Limited
1 Mezzanine Floor, 3 First Place, Bank City Cnr Simmonds and Jeppe Streets
Johannesburg 2001
SOUTH AFRICA

The Standard Bank of South Africa Limited Johannesburg

SOUTH KOREA

Kookmin Bank Co. Ltd. (Note: Clients please refer to your issued settlement instructions) 84, Namdaemun-ro, Jung-gu Seoul 100-845 SOUTH KOREA

Standard Chartered Bank Korea Limited (Note: Clients please refer to your issued settlement instructions) 47 Jongro, Jongro-Gu Seoul 3160

SOUTH KOREA

Kookmin Bank Co. Ltd. (Note: Clients please refer to your issued settlement instructions) Seoul

Standard Chartered Bank Korea Limited (Note: Clients please refer to your issued settlement instructions) Seoul **SPAIN**

Santander Securities Services, S.A. Parque Empresarial La Finca, Pozuelo de Alarcón Madrid 28223 SPAIN J.P. Morgan AG**
Frankfurt am Main

SRI LANKA

The Hongkong and Shanghai Banking Corporation Limited 24 Sir Baron Jayatillaka Mawatha Colombo 1 SRI LANKA The Hongkong and Shanghai Banking Corporation Limited Colombo

SWEDEN

Nordea Bank Abp Hamngatan 10 Stockholm SE-105 71 SWEDEN Svenska Handelsbanken Stockholm

SWITZERLAND

UBS Switzerland AG 45 Bahnhofstrasse Zurich 8021 SWITZERLAND UBS Switzerland AG Zurich

TAIWAN

JPMorgan Chase Bank, N.A.** 8th Floor, Cathay Xin Yi Trading Building, No. 108, Section 5, Xin Yi Road Taipei 11047 TAIWAN PMorgan Chase Bank, N.A.** Taipei

TANZANIA

Stanbic Bank Tanzania Limited Stanbic Centre, Corner Kinondoni and A.H. Mwinyi Roads, P.O. Box 72648 Dar es Salaam TANZANIA Stanbic Bank Tanzania Limited Dar es Salaam

THAILAND

Standard Chartered Bank (Thai) Public Company Limited 14th Floor, Zone B, Sathorn Nakorn Tower, 90 North Sathorn Road Bangrak, Silom, Bangrak Bangkok 10500 THAILAND Standard Chartered Bank (Thai) Public Company Limited Bangkok

TUNISIA

Banque Internationale Arabe de Tunisie S.A. 70-72 Avenue Habib Bourguiba, P.O. Box 520 Tunis 1000 TUNISIA Banque Internationale Arabe de Tunisie S.A. Tunis

TURKEY

Citibank A.S. Inkilap Mah., Yilmaz Plaza, O. Faik Atakan Caddesi No. 3, Umraniye Istanbul 34768 TURKEY JPMorgan Chase Bank, N.A. Istanbul Branch**
Istanbul

UGANDA

Standard Chartered Bank Uganda Limited 5 Speke Road, PO Box 7111 Kampala **UGANDA**

Standard Chartered Bank Uganda Limited Kampala

UKRAINE

JSC Citibank 16-G Dilova Street Kiev 03150 UKRAINE

JPMorgan Chase Bank, N.A.**

New York

JSC Citibank Kiev

UNITED ARAB EMIRATES

HSBC Bank Middle East Limited Emaar Square, Level 4, Building No. 5, P.O. Box 502601 Dubai **UNITED ARAB EMIRATES**

JPMorgan Chase Bank, N.A.**

The National Bank of Abu Dhabi

Dubai

UNITED KINGDOM

JPMorgan Chase Bank, N.A.** 4 New York Plaza New York 10004 **UNITED STATES**

JPMorgan Chase Bank, N.A.** New York

Deutsche Bank AG Depository and Clearing Centre 10 Bishops Square London E1 6EG UNITED KINGDOM

UNITED STATES

JPMorgan Chase Bank, N.A.** 4 New York Plaza New York 10004 **UNITED STATES**

JPMorgan Chase Bank, N.A.** New York

URUGUAY

Banco Itaú Uruguay S.A. Zabala 1463 Montevideo 11000 URUGUAY

Banco Itaú Uruguay S.A. Montevideo

VIETNAM

HSBC Bank (Vietnam) Ltd. 106 Nguyen Van Troi Street, Phu Nhuan District Ho Chi Minh City VIETNAM

HSBC Bank (Vietnam) Ltd. Ho Chi Minh City

WAEMU - BENIN, BURKINA FASO, GUINEA-BISSAU, IVORY COAST, MALI, NIGER, SENEGAL, TOGO

Standard Chartered Bank Côte d'Ivoire SA 23 Boulevard de la Republique 1 Abidjan 01 B.P. 1141 **IVORY COAST**

Standard Chartered Bank Côte d'Ivoire SA Abidjan

ZAMBIA

Standard Chartered Bank Zambia Plc Standard Chartered House, Cairo Road P.O. Box 32238 Lusaka 10101 ZAMBIA

Standard Chartered Bank Zambia Plc

ZIMBABWE

Stanbic Bank Zimbabwe Limited Stanbic Centre, 3rd Floor, 59 Samora Machel Avenue Harare ZIMBABWE Stanbic Bank Zimbabwe Limited Harare

This document is for information only and its contents are subject to change. This document is intended neither to influence your investment decisions nor to amend or supplement any agreement governing your relations with J.P. Morgan. Neither this document nor any of its contents may be disclosed to any third party or used for any other purpose without the proper written consent of J.P. Morgan has gathered the information from a source it considers reliable, however, it cannot be responsible for inaccuracies, incomplete information or updating of the information furnished hereby.

Appendix K – Facilities Agents and Services

According to Article 92 of the UCITS Directive, the facilities tasks are as follows:

- a. process subscription, repurchase and redemption orders and make other payments to Shareholders relating to the Shares of the Company, in accordance with the conditions set out in the Prospectus and the key information documents;
- b. provide investors with information on how orders referred to in point a. can be made and how repurchase and redemption proceeds are paid;
- facilitate the handling of information and access to procedures and arrangements referred to in Article 15 of the UCITS Directive relating to the investors' exercise of their rights arising from their investment in the Company in the Member State where the Company is marketed;
- d. make the information and documents required pursuant to Chapter IX of the UCITS Directive available to investors under the conditions laid down in Article 94 of the UCITS Directive, for the purposes of inspection and obtaining copies thereof;
- e. provide investors with information relevant to the tasks that the facilities agent performs in a durable medium;
- f. act as a contact point for communicating with the competent authorities.

The Company has appointed the facilities agents for the tasks listed a) and b) above in Article 92 of the UCITS Directive, and it has agreed to provide such facilities at its offices set out below.

Country	Facilities for EU investors
Austria	UniCredit Bank Austria AG
Austria	
	Rothschildplatz 1
	A-1020 Vienna
	Austria
	Email: stephan.hans@unicreditgroup.at
Belgium	RBC Investor Services Belgium S.A
	11, Place Rogier
	1210 Brussels
	Belgium
	Email: RBCIS BE FundCorporateServices@rbc.com
Czech Republic	Unicredit Bank Czech Republic and Slovakia, a.s.
	Prague 4 – Michle, Zeletavska 1525/1
	Postal code 14092
	Czech Republic
	ICO:64948242
	Email: Tomas.bartek@unicreditgroup.cz & Jaroslava.slatinska@unicreditgroup.cz
Denmark	Nordea Danmark
	Standgade 3
	0900 Copenhagen C
	Denmark
	Email: <u>issuerservice.dk@nordea.com</u>
France	Caceis Bank SA
	1/3 Place Valhubert

Country	Facilities for EU investors
	75013 Paris
	France
	Email: BK-OPCVM-IVS@caceis.com
Germany	Morgan Stanley Bank AG
	Grosse Gallusstrasse 18
	D-60312 Frankfurt am Main
	Germany
	Email: msim-germany@morganstanley.com
Greece	Alpha Bank
	40, Stadiou Street
	102 52 Athens
	Greece
	Email: ForeignCustodyGroup2@alpha.gr
Ireland	J.P. Morgan Administration Services (Ireland) Limited
	International Financial Services Centre
	Dublin 1 Ireland
	Email: <u>Dublin.TAOperations@jpmorgan.com</u>
	Email. <u>Dubini. l'Aoperations@jpinorgan.com</u>
Italy	All Funds Bank S.A (Italian branch)
•	Via Bocchetto, 6
	20121 Milano
	Italy
	Email: fundgroupsitaly@allfunds.com; milan ta customer services@allfunds.it
	Allianz Bank Financial Advisors SpA
	Piazza Tre Torri
	320145 Milano
	Italy
	Email: correspondentbank@allianzbank.it
	Caceis Bank SA (Italian Branch)
	Piazza Cavour, 2
	20121 Milano
	Italy
	Email: milan.lpa.customer@caceis.com
	ICCREA Banca S.p.A.
	Via Lucrezia Romana 41/47
	00178 Roma
	Italy
	Email: banca.corrispo@iccrea.bcc.it
	Monte Dei Paschi di Siena SpA
	Via Grossi 3
	46100 Mantova
	Italy Empile approximation has a commission and anti-Commission in the commission of the commission o
	Email: operation.bancacorrispondente@mps.it

Country	Facilities for EU investors	
	Sociéte Générale Securities Services S.p.A.	
	Via Benigno Crespi, 19/A - MAC 2	
	20159 Milano	
	Italy	
	Email: SGSS.TAOICESTERI@socgen.com	
	State Street Bank International GmbH (Italian Branch)	
	Via Ferrante Aporti 10	
	20125 Milano	
	Italy	
	Email: <u>LPA_SSBITALY_TA@statestreet.com</u>	
	Banca Sella Holding SPA	
	Piazza Gaudenzio Sella 1	
	13900 Biella BI	
	Italy	
	Email: fondiesteri@sella.it	
Luxembourg	J.P. Morgan SE, Luxembourg Branch (effective 22 January 2022)	
	European Bank and Business Centre, 6 route de Treves	
	L-2633 Senningerberg	
	Grand Duchy of Luxembourg	
	Email: <u>luxpayingagency@jpmorgan.com</u>	
Portugal	BEST – Banco Electronico De Servico Total S.A.	
	Praça Marquês de Pombal, 3A, 3	
	1250-096 Lisbon	
	Portugal	
	Email: bestdi.oferta@bancobest.pt	
Spain	Paying Agents (refer to Annual Report and Accounts). Every distributor registered in	
	Spain acts as a paying agent for the company and a list of them are available on	
	www.cnmv.es	
Sweden	MFEX Mutual Funds Exchange AB	
	Box 5378, Grev Turegatan 19	
	SE-10249 Stockholm	
	Sweden	
	Email: fundinformation@mfex.com	

The Company will provide the tasks listed under points c) to e) above in Article 92 of the UCITS Directive, at its registered office 6B, Route de Trèves, L-2633 Senningerberg, Email: cslux@morganstanley.com.

The countries that do not have a facilities agent appointed are listed below.

Estonia

Finland

Iceland

Latvia

Lithuania

Norway

Romania

Slovakia

The Netherlands

RBC Investor Services Bank S.A., 14, Rue Porte de France, L-4360 Esch-sur-Alzette, Grand Duchy of Luxembourg, Tel (352) 2605 9782, Fax (352) 2460 9902. Email csmorganstanley@rbc.com, will provide the tasks listed under point a) above.

For tasks listed under points b) to e) above, please contact Morgan Stanley Client Services at CSLux@morganstanley.com.

Arendt Regulatory & Consulting S.A. 41A Avenue J.F Kennedy, L-2082 Luxembourg, Grand Duchy of Luxembourg will act as the contact point for communicating with the competent authorities, according to point f) above of Article 92 of the UCITS Directive. Email at DistributionARC@arendt.com.

This appendix is available in the official language of your local jurisdiction at: www.morganstanleyinvestmentfunds.com. The Luxembourg webpage provides this appendix in additional languages of other EU regions.